# UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

#### FORM 8-K

## CURRENT REPORT Pursuant to Section 13 or 15(d) of the Securities Exchange Act of 1934

Date of Report (Date of earliest event reported): November 7, 2012

#### CLEAN ENERGY FUELS CORP.

(Exact Name of Registrant as Specified in Charter)

**Delaware**(State or Other Jurisdiction of Incorporation)

**001-33480** (Commission File Number)

**33-0968580** (IRS Employer Identification No.)

3020 Old Ranch Parkway, Suite 400 Seal Beach, California

90740

(Address of Principal Executive Offices) Zip Code

(562) 493-2804

(Registrant's telephone number, including area code)

Check the appropriate box below if the Form 8-K filing is intended to simultaneously satisfy the filing obligation of the registrant under any of the following provisions (see General Instruction A.2. below):

- Written communications pursuant to Rule 425 under the Securities Act (17 CFR 230.425)
- Soliciting material pursuant to Rule 14a-12 under the Exchange Act (17 CFR 240.14a-12)
- o Pre-commencement communications pursuant to Rule 14d-2(b) under the Exchange Act (17 CFR 240.14d-2(b))
- o Pre-commencement communications pursuant to Rule 13e-4(c) under the Exchange Act (17 CFR 240.13e-4(c))

#### Item 1.01 Entry into a Material Definitive Agreement.

On November 7, 2012, Clean Energy Fuels Corp. ("Clean Energy"), through two wholly owned subsidiaries (the "Borrowers"), entered into a Credit Agreement ("Credit Agreement") with General Electric Capital Corporation ("GE"). Pursuant to the Credit Agreement, GE agreed to loan to the Borrowers up to an aggregate of \$200 million to finance the development, construction and operation of two liquefied natural gas ("LNG") production facilities (individually a "Project" and together the "Projects"), each with an expected production capacity of approximately 250,000 LNG gallons per day. Clean Energy expects to sell the LNG produced by the Projects through America's Natural Gas Highway®, a nationwide network of LNG truck fueling stations, which Clean Energy is building along major transportation corridors in the United States.

The Borrowers' ability to obtain loans under the Credit Agreement (collectively, "Loans" and, with respect to each Project "Tranche A Loans" and "Tranche B Loans") for the Projects is subject to the satisfaction of certain conditions, including each of the (i) acquisition of title to, or leasehold interests in, the sites upon which the Projects will be constructed, (ii) receipt of all governmental approvals necessary in connection with the design, development, ownership, construction, installation, operation and maintenance of the Projects, (iii) commitment of all utility services necessary for the construction and operation of the Projects, and (iv) execution of an engineering, procurement and construction ("EPC") Contract for each Project by Clean Energy and GE Oil & Gas, Inc.

The Credit Agreement further provides that (i) if initial Loans are not made prior to December 31, 2014, the Credit Agreement will automatically terminate, (ii) each Project must be completed by the earlier of (a) the date thirty months after the funding of the initial Loans with respect to such Project and (b) December 31, 2016 (with respect to each Project, the "Date Certain"), (iii) the then existing Loans with respect to each Project must be converted into term loans with eight year amortization schedules ("Term Loans") on or before the Date Certain with respect to such Project (the date of such conversion with respect to each Project, the "Conversion Date"), provided that if such Loans are not converted into Term Loans by the applicable Date Certain, such Loans must be repaid by the applicable Date Certain, (iv) each Term Loan will be due and payable on the eighth anniversary of the Conversion Date with respect to such Term Loan, and (v) at any time prior to the applicable Conversion Date, the Loans may be prepaid in whole, and at any time after the applicable Conversion Date, the Loans may be prepaid in whole or in part. Clean Energy expects the Loans to bear interest at an annual rate equal to the then-current LIBOR rate plus 7.00%, provided that for purposes of the Credit Agreement the then-current LIBOR rate will always be at least 1.00%. The Credit Agreement includes various customary covenants, including debt service coverage ratios, and also provides for customary events of default which, if such events occur, would permit or require the Loans to become or to be declared due and payable.

The Loans are secured by (i) a first priority security interest in all of the Borrowers' assets, including the Projects, and (ii) a pledge of the Borrowers' outstanding ownership interests. In addition, Clean Energy has executed a guaranty in favor of GE ("Guaranty"), pursuant to which Clean Energy has

guaranteed all of the Borrowers' obligations under the Credit Agreement, including repayment of all Loans.

Clean Energy and GE also entered an equity contribution agreement (the "EC Agreement") pursuant to which Clean Energy agreed to pay at least 25% of the budgeted cost of the Projects and all additional costs that exceed such expected budgeted costs, in each case, in the form of equity contributions to the Borrowers ("Equity Contributions"). Clean Energy expects the aggregate total amount of Equity Contributions to be approximately \$67 million. The EC Agreement also requires Clean Energy to provide, concurrent with GE's extension of the initial Loans under the Credit Agreement, letter(s) of credit in an amount equal to Clean Energy's then-current unfunded Equity Contributions.

Concurrently with the execution of the Credit Agreement, and in consideration of the lending terms for the Loans, lender fees forgone by GE, and the significant strategic benefits to Clean Energy of the transactions contemplated by the Credit Agreement, and as an upfront commitment fee payment, Clean Energy issued to GE a warrant (the "Warrant") to purchase up to an aggregate of five million shares of Clean Energy's common stock, par value \$0.0001 per share ("Shares"), at a price per share of \$0.01. The Shares subject to the Warrant are exercisable pursuant to the following schedule: (i) 500,000 Shares are immediately exercisable, (ii) an additional 1,250,000 shares will become exercisable at the time that the first Tranche A Loan is made under the Credit Agreement, (iv) an additional 1,250,000 shares will become exercisable at the time that the first Tranche B Loan is made under the Credit Agreement, (iv) an additional 1,000,000 shares will become exercisable at the time that Tranche A Loans in aggregate principal amount of at least \$15 million have been made under the Credit Agreement, and (v) the

2

remaining 1,000,000 Shares will become exercisable at the time that Tranche B Loans in aggregate principal amount of at least \$15 million have been made under the Credit Agreement; provided, however, that if no Loans are made as contemplated by (ii) through (v) above pursuant to the Credit Agreement, an additional 500,000 Shares will be exercisable. The Warrant terminates on November 7, 2022.

In connection with the Warrant, on November 7, 2012, Clean Energy and GE entered into a warrant agreement (the "Warrant Agreement") and a registration rights agreement (the "Registration Rights Agreement"). Pursuant to the Registration Rights Agreement Clean Energy agreed, subject to the terms and conditions of that agreement, to (i) file with the Securities and Exchange Commission one or more registration statements relating to the resale of Shares issuable upon exercise of the Warrant, and (ii) at the request of the Warrant holder, and if a registration statement with respect to the Shares is not then effective, participate in one or more underwritten offerings of Shares issuable upon exercise of the Warrant. If Clean Energy does not meet certain of its obligations under the Registration Rights Agreement with respect to the registration of Shares, it will be required to pay certain liquidated damages.

The issuance of the Warrant and the Shares issuable upon exercise of the Warrant has not been registered under the Securities Act of 1933, as amended (the "Securities Act"). The Warrant and the Shares issuable upon exercise of the Warrant were and will be sold in reliance upon exemptions from registration under Section 4(2) of the Securities Act. The Warrant and the Shares issuable upon exercise of the Warrant may not be offered or sold in the United States absent registration under or exemption from the Securities Act and any applicable state securities laws. GE represented that it is an accredited investor as defined in the rules and regulations under the Securities Act and that it was acquiring the Warrant and the Shares issuable upon exercise of the Warrant for investment only and not with a view towards, or for resale in connection with, the public sale or distribution thereof.

The foregoing description of the Credit Agreement, Guaranty, EC Agreement, Warrant, Warrant Agreement and Registration Rights Agreement is a summary of the respective terms of such agreements. Clean Energy's press release, issued on November 13, 2012, announcing the Credit Agreement and the other transactions contemplated thereby is attached hereto as Exhibit 99.1 and is incorporated herein by reference.

#### Item 2.03 Creation of a Direct Financial Obligation or an Obligation under an Off-Balance Sheet. Arrangement.

The information set forth in Item 1.01 of this Current Report on Form 8-K is incorporated by reference into this Item 2.03.

#### Item 3.02 Unregistered Sales of Equity Securities.

The information set forth in Item 1.01 of this Current Report on Form 8-K is incorporated by reference into this Item 3.02.

3

#### Item 9.01 Financial Statements and Exhibits.

herewith)

(d) Exhibits.

Exhibit	Description
4.8	Warrant dated November 7, 2012 (filed herewith).
10.70	Credit Agreement among Clean Energy Tranche A LNG Plant, LLC and Clean Energy Tranche B LNG Plant, LLC, as Borrowers, the Various Financial Institutions from Time to Time Party thereto, as Lenders, and General Electric Capital Corporation, as Administrative Agent and Collateral Agent, dated as of November 7, 2012 (filed herewith).
10.71	Guaranty by Clean Energy Fuels Corp. in favor of General Electric Capital Corporation, dated as of November 7, 2012 (to be filed with Clean Energy's annual report on Form 10-K for the fiscal year ending December 31, 2012).
10.72	Equity Contribution Agreement by and among Clean Energy Tranche A LNG Plant, LLC and Clean Energy Tranche B LNG Plant, LLC, as Borrowers, Clean Energy Fuels Corp., as Equity Investor, and General Electric Capital Corporation, as Administrative Agent and Collateral Agent, dated as of November 7, 2012 (filed herewith).
10.73	Warrant Agreement dated as of November 7, 2012 between Clean Energy Fuels Corp. and GE Energy Financial Services, Inc. (filed

10.74 Registration Rights Agreement dated as of November 7, 2012 by and between Clean Energy Fuels Corp. and GE Energy Financial Services, Inc. (filed herewith)

99.1 Press release issued by Clean Energy Fuels Corp., dated November 13, 2012. (filed herewith)

1

#### **SIGNATURES**

Pursuant to the requirements of the Securities Exchange Act of 1934, as amended, the registrant has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

Date: November 13, 2012 Clean Energy Fuels Corp.

By: /s/ Richard R. Wheeler

Name: Richard R. Wheeler Title: Chief Financial Officer

5

THIS WARRANT AND THE SECURITIES ISSUABLE UPON THE EXERCISE HEREOF HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933, AS AMENDED, OR REGISTERED OR QUALIFIED UNDER THE SECURITIES LAWS OF ANY STATE OR OTHER JURISDICTION AND MAY NOT BE SOLD, PLEDGED, TRANSFERRED OR OTHERWISE ASSIGNED IN VIOLATION OF SUCH ACT AND LAWS OR THE PROVISIONS OF THIS WARRANT.

#### WARRANT

#### TO PURCHASE SHARES OF CLEAN ENERGY FUELS CORP.

THIS CERTIFIES THAT, pursuant to the Warrant Agreement dated November 7, 2012 (the "Warrant Agreement") by and between GE Energy Financial Services, Inc. (the "Holder") and Clean Energy Fuels Corp. (the "Issuer"), the Holder, or its successor or permitted assign, is entitled, at any time during the Exercise Period (as defined herein), to purchase from the Issuer, an aggregate of up to 5,000,000 shares of Common Stock, par value \$0.0001 per share, of the Issuer (the "Shares") (subject to adjustment and limitation as provided herein), in whole or in part, at a purchase price of \$0.01 per Share (the "Exercise Price"), all on the terms and conditions and pursuant to the provisions hereinafter set forth. This Warrant is issued in consideration of the execution of the Credit Agreement.

**DEFINITIONS.** All capitalized terms defined in the Warrant Agreement and not otherwise defined herein shall have the meanings ascribed to such terms in the Warrant Agreement.

#### 2. EXERCISE OF WARRANT.

- 2.1 <u>General</u>. At any time and from time to time after the Closing Date and until 5:00 p.m., New York time, on the tenth (10<sup>th</sup>) anniversary of the Closing Date or such earlier date as provided in <u>Section 4.2</u> below (the "*Exercise Period*"), the Holder may exercise this Warrant, on any Business Day, for all or any part of such number of Shares, at the stated Exercise Price, equal to the sum of:
  - (a) 10% of the aggregate number of Shares purchasable hereunder as an upfront commitment fee, *plus* (b) or (c)

(b)

- (i) after the first Tranche A Loan (as defined in the Credit Agreement) is made, 25% of the aggregate number of Shares purchasable hereunder,
- (ii) after the first Tranche B Loan (as defined in the Credit Agreement) is made, 25% of the aggregate number of Shares purchasable hereunder.
- (iii) after Tranche A Loans in aggregate principal amount of at least \$15,000,000 have been made, 20% of the aggregate number of Shares purchasable hereunder, and

1

- (iv) after Tranche B Loans in aggregate principal amount of at least \$15,000,000 have been made, 20% of the aggregate number of Shares purchasable hereunder, *or* (in the alternative)
- (c) if no Borrowings (as defined in the Credit Agreement) are made prior to December 31, 2014, 10% of the aggregate number of Shares purchasable hereunder.

Upon the occurrence of any of the events described in the foregoing clauses (b) and (c), the Issuer shall deliver written notice thereof to Holder.

Cash Exercise. The Holder may exercise this Warrant, in whole or in part and at any time and from time to time during the Exercise Period, by delivering to the Issuer at the Issuer's principal offices at the address set forth in the Warrant Agreement: (i) a written notice of the Holder's election to exercise this Warrant specifying the number of Shares to be purchased, (ii) payment of the Warrant Price and (iii) this Warrant. "Warrant Price" shall mean an amount equal to (i) the number of Shares being purchased upon any exercise of the Warrant pursuant to this Section 2, multiplied by (ii) the Exercise Price as adjusted pursuant to the terms of the Warrant as of the date of such exercise. Such notice shall be substantially in the form of the subscription form appearing at the end of this Warrant as Annex A, duly executed by the Holder or its agent or attorney. Upon receipt thereof, the Issuer shall, as promptly as practicable, and in any event within five (5) Business Days thereafter, execute or cause to be executed and deliver or cause to be delivered to the Holder a certificate or certificates reflecting the Holder's ownership of the aggregate number of Shares issuable upon such exercise, together with cash in lieu of any fraction of a Share, as hereinafter provided in Section 2.5. The Share certificate or certificates so delivered shall be in such denomination or denominations as the Holder shall request in the notice and shall be registered in the name of the Holder or, subject to any restrictions on Transfer, such other name as shall be designated in the notice. This Warrant shall be deemed to have been exercised and such Warrant Shares shall be deemed to have been issued, and the Holder or any other Person so designated to be named therein shall be deemed to have become a holder of record of such Shares for all purposes, as of the date of delivery of the Warrant Shares by the Issuer. If this Warrant has been exercised in part, the Issuer shall, at the time of delivery of the certificate or certificates representing Warrant Shares, deliver to the Holder a new Warrant evidencing the rights of the Holder to purchase the unpurchased Shares called for by this Warrant, which new Warrant shall in all other respects be identical with this Warrant, or, at the request of the Holder, appropriate notation may be made on this Warrant and the same returned to the Holder. Payment of the Warrant Price and all taxes required to be paid by the Holder, if any, pursuant to Section 2.4, shall be paid by the Holder prior to delivery of the Warrant Shares by the Issuer and shall be made at the option of the Holder by certified or official bank check or by wire transfer of immediately available funds.

#### 2.3 <u>Cashless Exercise</u>.

(a) In lieu of the payment of the Warrant Price, the Holder shall have the right (but not the obligation), to require the Issuer to convert this Warrant, in whole or in part, into Shares as provided for in this Section 2.3 (the "Conversion Right"). Upon exercise of the Conversion Right, the Issuer shall

2

obtained by dividing (x) the value of this Warrant (or portion thereof as to which the Conversion Right is being exercised if the Conversion Right is being exercised in part) at the time the Conversion Right is exercised (determined by subtracting the aggregate Warrant Price of the Warrant Shares as to which the Conversion Right is being exercised in effect immediately prior to the exercise of the Conversion Right from the aggregate Current Market Price of the Warrant Shares as to which the Conversion Right is being exercised immediately prior to the exercise of the Conversion Right) by (y) the Current Market Price of one (1) Warrant Share immediately prior to the exercise of the Conversion Right.

- (b) The Conversion Rights provided under this Section 2.3 may be exercised in whole or in part and at any time and from time to time during the Exercise Period. In order to exercise the Conversion Right, the Holder shall surrender to the Issuer, at its offices, this Warrant, and the Notice of Conversion in the form attached hereto as Annex B duly executed. The presentation and surrender shall be deemed a waiver of the Holder's obligation to pay all or any portion of the aggregate purchase price payable for the Warrant Shares as to which such Conversion Right is being exercised. This Warrant (or so much thereof as shall have been surrendered for conversion) shall be deemed to have been converted immediately prior to the close of business on the day of surrender of such Warrant for conversion in accordance with the foregoing provisions. If this Warrant has been exercised in part, the Issuer shall, at the time of delivery of the acknowledgement reflecting the ownership of the Warrant Shares, deliver to the Holder a new Warrant evidencing the rights of the Holder to purchase the unpurchased Shares called for by this Warrant, which new Warrant shall in all other respects be identical with this Warrant, or, at the request of the Holder, appropriate notation may be made on this Warrant and the same returned to the Holder.
- 2.4 Payment of Taxes. When the Warrant Price is paid to the Issuer, all such Warrant Shares shall be validly issued, fully paid and nonassessable and without any preemptive rights. The Issuer shall pay all expenses in connection with, and all taxes and other governmental charges that may be imposed with respect to, the issue or delivery thereof, unless such tax or charge is imposed by law upon the Holder, in which case, the Holder shall pay such taxes or charges. The Issuer shall not be required to pay any tax or other charge imposed in connection with any Transfer involved in the issue or delivery of any certificate for Shares issuable upon exercise of this Warrant in any name other than that of the Holder, and in such case, the Issuer shall not be required to register such Shares in any name other than the Holder until such tax or other charge has been paid or it has been established to the reasonable satisfaction of the Issuer that no such tax or other charge is due.
- 2.5 <u>Fractional Shares</u>. The Issuer shall not be required to issue a fractional Share upon the exercise of this Warrant. As to any fraction of a Share which the Holder would otherwise be entitled to purchase upon such exercise, the Issuer shall pay a cash adjustment in respect of such final fraction (calculated on an aggregate basis for all Warrants exercised) in an amount equal to the same fraction of the Current Market Price per Share of a Warrant Share on the date of exercise.
- 2.6 <u>Limitation on Conversion</u>. Notwithstanding anything herein to the contrary, the Issuer shall not be obligated to affect any conversion of all or any portion of this Warrant, and the Holder shall not have the right to convert all or any portion of this Warrant, to the extent that,

3

after giving effect to an attempted conversion, the Holder (together with its Affiliates and its permitted assignees and any other Person whose beneficial ownership of Shares would be aggregated with the Holder and its permitted assignees for purposes of Section 13(d) of the Exchange Act and the applicable rules and regulations of the Securities and Exchange Commission, including any "group" of which Holder and its permitted assignees is a member) would have acquired a number of Shares as a result of one or more conversions or otherwise in excess of 19.99% of the number of Shares outstanding immediately prior to the Closing Date. The Holder and each permitted assignee shall supply all information necessary to ensure compliance with this Section 2.6, and the Issuer shall be entitled to rely on representations made to it by the Holder and its permitted assignees regarding the ownership limitation set forth in this Section 2.6.

#### 3. TRANSFER, DIVISION AND COMBINATION.

- 3.1 Transfer. Subject to compliance with all applicable securities laws, this Warrant may be Transferred to an Affiliate of the Holder at any time and from time to time, in whole or, in the case of a partial Transfer, in a Transfer of not less than 1,000,000 Shares subject to this Warrant. Such Transfer shall be registered on the books of the Issuer to be maintained for such purpose, upon surrender of this Warrant at the principal office of the Issuer at the address set forth in the Warrant Agreement, together with a written assignment of this Warrant duly executed by the Holder or its agent or attorney. Upon such surrender, the Issuer shall execute and deliver a new Warrant or Warrants in the name of the transferee or transferees and in the denomination specified in such instrument of assignment, and shall issue to the assignor a new Warrant evidencing the portion of this Warrant not so assigned, and this Warrant shall promptly be cancelled. Such Warrant, if properly assigned in compliance with any restrictions on Transfer and properly registered on the books of the Issuer, may be exercised by a new Holder for the purchase of Shares without having a new Warrant issued.
- 3.2 <u>Maintenance of Books</u>. The Issuer agrees to maintain, at its aforesaid office or agency, a ledger recording the Holder(s) of this Warrant and any subsequent Transfer of this Warrant in compliance with <u>Section 3.1</u>.
- **4. ADJUSTMENTS.** The number of Shares for which this Warrant is exercisable shall be subject to adjustment from time to time as set forth in this <u>Section 4</u>. The Issuer shall give the Holder notice of any event described below, which requires an adjustment pursuant to this <u>Section 4</u> at the time of such event.
  - 4.1 <u>Distributions, Subdivisions and Combinations</u>. If, at any time, the Issuer:
- (a) takes a record of holders of its Shares for the purpose of entitling them to receive a distribution payable in, or other distribution of, Additional Shares,
  - (b) subdivides its outstanding Shares into a larger number of Shares, or

then the number of Shares for which this Warrant is exercisable immediately after the occurrence of any such event shall be adjusted to equal the number of Shares that a record holder of the same number of Shares for which this Warrant is exercisable immediately prior to the occurrence of such event would own or be entitled to receive after the happening of such event. If the Issuer subdivides its outstanding Shares into a larger number of Shares, the Exercise Price shall be proportionately decreased. If the Issuer combines its outstanding Shares into a lesser number of Shares, the Exercise Price shall be proportionately increased.

- 4.2 Reorganization, Consolidation, Merger and Other Changes. In case of any capital reorganization or change in the Common Stock (other than as a result of a subdivision, combination, or stock dividend provided for in Section 4.1), or a spinoff, consolidation or merger of the Issuer with or into another entity, or the sale of all or substantially all of its assets to another entity shall be effected in such a way that holders of Common Stock shall be entitled to receive stock, securities or assets with respect to or in exchange for Common Stock, then, as a condition of such reorganization, change, spinoff, consolidation, merger or sale (each a "Change"), lawful provision shall be made, and duly executed documents evidencing the same from the Issuer or its successor shall be delivered to the Holder, so that the Holder shall have the right at any time prior to the Expiration Date to purchase, at a total price equal to that payable upon the exercise of this Warrant, the kind and amount of shares of stock and other securities and property receivable in connection with such Change by a holder of the same number of Shares as were purchasable by the Holder immediately prior to such Change. In any such case appropriate provisions shall be made with respect to the rights and interest of the Holder so that the provisions hereof shall thereafter be applicable with respect to any shares of stock or other securities and property deliverable upon exercise hereof, and appropriate adjustments shall be made to the purchase price per share payable hereunder, provided the aggregate purchase price shall remain the same; provided, however, that notwithstanding the foregoing, if as a result of such Change, holders of the Issuer's Common Stock shall receive consideration other than solely in shares of stock or other securities in exchange for their Common Stock, the Issuer may, at its option, fulfill its obligations hereunder by causing a notice to be mailed to Holder at least 10 days prior to the record date for the Change, which notice shall (i) describe the nature of the proposed Change and any material terms applicable thereto, (ii) indicate the record date applicable to the Change and (iii) set forth the first and last date on which the Holder may exercise this Warrant. The Holder shall have the opportunity to exercise this Warrant in full before the applicable record date for the Change, and thereby receive consideration with respect to any such Change, on the same basis as other previously outstanding shares of Common Stock of the Issuer. If the notice specified above is provided to the Holder in accordance with this Section 4.2, this Warrant to the extent not exercised before the consummation of the Change shall be cancelled and become null and void on the effective date of the Change.
  - 4.3 <u>Adjustment upon Dilutive Issuance</u>.
- (a) If the Issuer at any time, and from time to time, during the Exercise Period issues or sells any Shares, other than Exempted Securities, for a price per share less than a price equal to eighty percent (80%) of the Market Price on the day of such issue or sale (the foregoing a "*Dilutive Issuance*"), then immediately after such Dilutive Issuance the number of Shares then

5

purchasable hereunder shall be increased by an amount determined in accordance with the following formula:

$$S2 = [((A - B) * 0.93 * C) * (S1 ÷ D)] ÷ A$$

For purposes of the foregoing formula, the following definitions shall apply:

"S2" means the number of additional Shares purchasable hereunder immediately after the Dilutive Issuance;

"S1" means the number of Shares purchasable hereunder immediately prior to the Dilutive Issuance;

"A" means the Market Price on the day of the Dilutive Issuance;

"B" means the per share price for which Shares were issued in the Dilutive Issuance;

"C" means the number of Shares issued pursuant to the Dilutive Issuance; and

- "D" means the number of Shares deemed outstanding on a fully diluted basis immediately prior to the Dilutive Issuance (treating for this purpose as outstanding all Shares issuable upon exercise of Options outstanding immediately prior to such Dilutive Issuance or upon conversion or exchange of Convertible Securities outstanding (assuming exercise of any outstanding Options therefor) immediately prior to such Dilutive Issuance).
- (b) Notwithstanding any other provision of this Warrant, (i) no adjustment in the number of Shares purchasable hereunder shall be made pursuant to this Section 4.3 if the Market Price on the date of the Dilutive Issuance is greater than or equal to the Market Price on the Closing Date, and (ii) no adjustment shall be made in respect of any Shares issued under this Warrant prior to the date of the Dilutive Issuance.
  - (c) For purposes of this <u>Section 4.3</u>, the following terms shall have the following meanings:
- (i) "Exempted Securities" means (1) Shares issued to (or issuable upon exercise of Options issued to) employees or directors of, or consultants or advisors to, the Issuer or any of its subsidiaries pursuant to a plan, agreement or arrangement approved by the Board of Directors of the Issuer; (2) Shares issued upon the exercise or exchange of or conversion of any Options or Convertible Securities outstanding on the Closing Date and (3) Shares issued pursuant to (or issuable upon the exercise or exchange or conversion of any Options or Convertible Securities issued pursuant to) mergers, consolidations, acquisitions or any other transactions with Persons with whom the Issuer has business relationships, provided such issuances are for other than primarily equity financing purposes.

- (ii) "Market Price" shall mean, in respect of any Shares on any date herein specified, the last reported closing price on the Trading Market on which the Shares are listed.
- 4.4 <u>Timing of Issuance of Additional Shares Upon Adjustments</u>. In any case in which the provisions of this <u>Section 4</u> shall require that an adjustment shall become effective immediately after a record date for an event, the Issuer, after such record date and before the occurrence of such event, may defer until the occurrence of such event issuing to the Holder the Additional Shares or other property issuable or deliverable upon exercise by reason of the adjustment required by such event over and above the Shares issuable or deliverable upon such exercise before giving effect to such adjustment; provided, however, that the Issuer shall, upon request of the Holder, deliver to the Holder a due bill or other appropriate instrument evidencing the Holder's right to receive such Additional Shares or other property upon the occurrence of the event requiring such adjustment.
- 4.5 <u>When Adjustment Not Required.</u> If the Issuer takes a record of holders of its Shares for the purpose of entitling them to receive a distribution or subscription or purchase rights and, thereafter and before the distribution to holders thereof, legally abandons its plan to pay or deliver such distribution, subscription or purchase rights, then thereafter no adjustment shall be required by reason of the taking of such record and any such adjustment previously made in respect thereof shall be rescinded and annulled.
- 4.6 Notice of Adjustments. Whenever the number of Shares for which this Warrant is exercisable, or whenever the price at which such Shares may be purchased upon exercise of the Warrants, is adjusted pursuant to this Section 4, the Issuer shall prepare a certificate to be executed by its chief financial officer, if any, or its principal financial officer(s) in case there is no chief financial officer, setting forth, in reasonable detail, the event requiring the adjustment and the method by which such adjustment was calculated, specifying the number of Shares for which this Warrant is exercisable and describing the number and kind of any other Shares or Other Property for which this Warrant is exercisable, and any change in the purchase price or prices thereof, after giving effect to such adjustment or change. The Issuer shall promptly cause a signed copy of such certificate to be delivered to the Holder. The Issuer shall keep at its office or agency copies of all such certificates and cause the same to be available for inspection at said office during normal business hours by the Holder.
- **5. NO IMPAIRMENT.** The Issuer shall not by any action, including, without limitation, through any amendment to its articles of incorporation, through any reorganization, transfer of assets, consolidation, merger, dissolution, issue or sale of securities or any other voluntary action, avoid or seek to avoid the observance or performance of any of the terms of this Warrant, but shall at all times in good faith assist in carrying out all such actions as may be reasonably necessary or appropriate to protect the rights of the Holder against impairment. Without limiting the generality of the foregoing, the Issuer shall (a) take all such action as may be reasonably necessary or appropriate in order that the Issuer may validly and legally issue fully-paid and nonassessable Shares upon the exercise of this Warrant, and (b) use its reasonable best efforts to obtain all such authorizations, exemptions or consents from any public regulatory body having jurisdiction thereof as may be necessary to enable the Issuer to perform its obligations under this Warrant.

7

- **6. RESERVATION AND AUTHORIZATION OF SHARES.** From and after the Closing Date, the Issuer shall at all times reserve and keep available for issue upon the exercise of Warrants such number of its authorized but unissued Shares as shall be sufficient to permit the exercise in full of all outstanding Warrants. All Shares, when issued upon exercise of this Warrant and payment therefor in accordance with the terms of this Warrant, shall be duly and validly issued and fully paid and nonassessable, and not subject to preemptive rights.
- **7. TAKING OF RECORD; SHARES AND WARRANT TRANSFER BOOKS.** In the case of all distributions by the Issuer to holders of its Shares with respect to which any provision of Section 4 refers to the taking of a record of such holders, the Issuer shall in each such case take such a record as of the close of business on a Business Day. The Issuer shall not at any time, except upon dissolution, liquidation or winding up of the Issuer, close its transfer books so as to result in preventing or delaying the exercise or transfer of any Warrant.
- **8. RESTRICTIVE LEGEND.** This Warrant and any Warrant issued upon transfer or partial exercise of this Warrant shall be imprinted with the following legend, in addition to any legend required under applicable state securities laws:

"THIS WARRANT AND THE SECURITIES ISSUABLE UPON THE EXERCISE HEREOF HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933, AS AMENDED, OR REGISTERED OR QUALIFIED UNDER THE SECURITIES LAWS OF ANY STATE OR OTHER JURISDICTION AND MAY NOT BE SOLD, PLEDGED, TRANSFERRED OR OTHERWISE ASSIGNED IN VIOLATION OF SUCH ACT AND LAWS OR THE PROVISIONS OF THIS WARRANT AND THE WARRANT AGREEMENT."

Each Share certificate representing Warrant Shares shall bear the following legend:

"THE SECURITIES REPRESENTED BY THIS CERTIFICATE HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933, AS AMENDED, OR REGISTERED OR QUALIFIED UNDER THE SECURITIES LAWS OF ANY STATE OR OTHER JURISDICTION AND MAY NOT BE SOLD, PLEDGED, TRANSFERRED OR OTHERWISE ASSIGNED IN VIOLATION OF SUCH ACT AND LAWS OR THE PROVISIONS OF THIS WARRANT AND THE WARRANT AGREEMENT."

Upon request of the holder of a Share certificate, the Issuer shall issue to that holder a new certificate free of the foregoing legend, if, with such request, such holder provides the Issuer with an opinion of counsel (including in-house counsel) reasonably acceptable to the Issuer to the effect that the securities evidenced by such certificate may be sold without restriction under Rule 144 (or any other rule permitting resales of securities without restriction) promulgated under the Securities Act.

**9. SUPPLYING INFORMATION.** The Issuer shall cooperate with each Holder of a Warrant and each holder of Warrant Shares in supplying such information as may be reasonably necessary for such holder to complete and file any information reporting forms presently or hereafter required by the SEC as a condition to the availability of an exemption from the Securities Act for the sale or transfer of any Warrant or Warrant Shares.

- **10. LOSS OR MUTILATION.** Upon receipt by the Issuer of evidence reasonably satisfactory to it of the ownership of and the loss, theft, destruction or mutilation of any Warrant (which evidence shall be, in the case of an institutional investor, notice from such institutional investor of such ownership and such loss, theft, destruction or mutilation), and
- (a) in the case of loss, theft or destruction, of indemnity reasonably satisfactory to it (provided that if the Holder of such Warrant has a minimum net worth of at least \$100,000,000, such Holder's own unsecured agreement of indemnity shall be deemed to be satisfactory), or
  - (b) in the case of mutilation, upon surrender and cancellation thereof,

the Issuer at its own expense shall execute and deliver, in lieu thereof, a new Warrant, dated the date of the original Warrant.

- 11. OFFICE OF THE ISSUER. As long as any Warrant remains outstanding, the Issuer shall maintain an office or agency (which may be the principal executive offices of the Issuer) where the Warrants may be presented for exercise, registration of transfer, division or combination as provided in this Warrant.
- **12. LIMITATION OF LIABILITY.** No provision hereof, in the absence of affirmative action by holder to purchase Shares, and no enumeration of the rights or privileges of holder contained herein, shall give rise to any liability of holder for the purchase price of any Share or as a holder of Shares of the Issuer, whether such liability is asserted by the Issuer or by creditors of Company.

#### 13. MISCELLANEOUS.

- 13.1 <u>Nonwaiver and Expenses</u>. If either party fails to comply with any provision of this Warrant, it shall pay to the other party such amounts as shall be sufficient to cover any costs and expenses including, but not limited to, reasonable attorneys' fees, including those of appellate proceedings, incurred by the other party in enforcing any of its rights, powers, or remedies hereunder. No course of dealing or any delay or failure to exercise any right hereunder on the part of a party shall operate as a waiver of such right or otherwise prejudice its rights, powers, or remedies.
- 13.2 <u>Non-Survival</u>. The parties hereby agree that all the provisions of this Warrant shall terminate and be of no further force or effect on the exercise in full of this Warrant.
- 13.3 <u>Miscellaneous Provisions in Warrant Agreement</u>. The provisions set forth in Section 6 of the Warrant Agreement shall apply to this Warrant, *mutatis mutandis*, and are hereby incorporated by reference herein.

[Signature page follows.]

9

IN WITNESS WHEREOF, each party hereto has caused this Warrant to be duly executed by its authorized officer and dated effective as of the Closing Date.

#### CLEAN ENERGY FUELS CORP.

By: /s/ Andrew J. Littlefair

Name: Andrew J. Littlefair

Title: President & Chief Executive Officer

#### GE ENERGY FINANCIAL SERVICES, INC.

By: /s/ Tyson Yates
Name: Tyson Yates

Title: Managing Director

SIGNATURE PAGE TO WARRANT

### ANNEX A SUBSCRIPTION FORM

(To be executed only upon exercise of the attached Warrant)

The undersigned registered Holder of this Warrant irrevocably exercises this Warrant for purchase of Shares of Clean Energy Fuels Corp., Delaware corporation, [and herewith makes payment therefore][pursuant to the cashless exercise provisions set forth in Section 2.3 of the Warrant, with the calculation for such cashless exercise attached to this Subscription Form], all at the price and on the terms and conditions specified in this Warrant and requests that certificates for the Shares hereby purchased (and any securities or property issuable upon such exercise) to be issued in the name of the undersigned and delivered to the undersigned as follows:

Name Address

If the certificates representing the Shares being purchased pursuant hereto are to be registered in a name or names other than the name of the holder of this Warrant, all transfer taxes payable upon such transfer shall be paid by the undersigned at the time of delivering the notice of exercise and such request.

Solely with respect to the Warrant Shares being purchased pursuant to this Subscription Form, the representations and warranties of the Holder contained in Section 15.1 of the Warrant are repeated at and as of the time of delivery hereof and are true and correct in all material respects at and as of the time of delivery hereof, except to the extent such representations and warranties are expressly limited to an earlier date or the Issuer has expressly consented in writing to the contrary.

The undersigned acknowledges that each certificate for Warrant Shares issued upon exercise of the Warrant shall bear a legend to the effect that such Warrant Shares may not be transferred except upon compliance with the provisions of the Securities Act and applicable state securities laws, and each certificate for Warrant Shares transferred shall bear such a legend unless, in the opinion of counsel for the Issuer, such legend is not required.

If the number of Shares shall not be all the Warrant Shares purchasable under this Warrant, a new Warrant of like tenor is to be issued in the name of and delivered to the undersigned for the remaining balance of the Shares purchasable thereunder.

and delivered	to the undersigned for the remaining balance of the Shares purch	asable thereund	er.	
(Name of Registered Owner)		(Street Address)		
(Signature of	Registered Owner)	(City)	(State)	(Zip Code)
NOTICE:	The signature on this subscription must correspond with the without alteration or enlargement or any change whatsoever.	names as writte	n upon the face of the attach	ed Warrant in every particular,
	Anne	ex A-1		
exercise provi executed a cer in lieu of any	NOTICE OF (  (To be executed only upon con undersigned registered owner irrevocably elects to surrender this	Warrant for the derlying this W he aggregate nu	number of Shares as shall be farrant, and requests that umber of Shares issuable upo	execute or cause to be on such exercise, together with cash
	<u>Name</u>		Address	
Warrant Share certificate for Solel contained in S	undersigned acknowledges that each certificate for Warrant Shares may not be transferred except upon compliance with the provise Warrant Shares transferred shall also bear such a legend unless, in the last of the Warrant Shares being received pursuant to the Section 15.1 of the Warrant are repeated at and as of the time of dury hereof, except to the extent such representations and warranties the contrary.	tions of the Sect in the opinion of his Notice of Co elivery hereof a	urities Act and applicable sta f counsel for the Issuer, such conversion, the representation and are true and correct in all	ate securities laws, and each a legend is not required. as and warranties of the Holder material respects at and as of the
	e number of Shares shall not be all the Warrant Shares issuable ur ne undersigned for the remaining balance of the Shares issuable th		t, a new Warrant of like tend	or is to be issued in the name of and
(Name of Reg	gistered Owner)	(Street Addre	ss)	
(Signature of	Registered Owner)	(City)	(State)	(Zip Code)

Annex B-1

without alteration or enlargement or any change whatsoever.

The signature on this subscription must correspond with the names as written upon the face of the within Warrant in every particular,

NOTICE:

#### **CREDIT AGREEMENT**

#### among

CLEAN ENERGY TRANCHE A LNG PLANT, LLC, and CLEAN ENERGY TRANCHE B LNG PLANT, LLC, as Borrowers,

# THE VARIOUS FINANCIAL INSTITUTIONS FROM TIME TO TIME PARTY HERETO, as Lenders,

and

### GENERAL ELECTRIC CAPITAL CORPORATION, as Administrative Agent and Collateral Agent

Dated as of November 7, 2012

Micro-LNG Liquefaction Project

#### TABLE OF CONTENTS

SECTION 1.	DEFINITIONS AND RULES OF INTERPRETATION	-
1.1	Defined Terms	-
1.2	Rules of Interpretation	35
1.3	Accounting Principles	36
SECTION 2.	AMOUNTS AND TERMS OF CREDIT FACILITY	37
2.1	Tranche A Construction Loan Facility	33
2.2	Tranche B Construction Loan Facility	37
2.3	Tranche A Term Loan Facility	38
2.4	Tranche B Term Loan Facility	38
2.5	Notice of Borrowing	38
2.6	Pro Rata Borrowings; Availability	39
2.7	Minimum Amount and Maximum Number of Borrowings, etc.	39
2.8	Disbursement of Funds	39
2.9	Evidence of Obligations and Notes	40
2.10	Conversions	43
2.11	Interest	42
2.12	Interest Periods	42
2.13	Termination and Reduction of Commitments	43
2.14	Taxes	44
2.15	Illegality	48
2.16	Increased Costs and Reduction of Return	48
2.17	Funding Losses	49
2.18	Inability to Determine Rates	50
2.19	Survival	50
2.20	Replacement of Lenders	50
2.21	Defaulting Lenders	5.3
2.22	Interest During Construction	52
SECTION 3.	CONDITIONS PRECEDENT	53
3.1	Conditions to Closing	53
3.2	Conditions to Initial Funding	52
3.3	Initial and Subsequent Construction Loans	62
3.4	Conversion Dates	65
SECTION 4.	REPRESENTATIONS, WARRANTIES AND AGREEMENTS	67

4.1	Organization	67
4.2	Authority and Consents	68
4.3	Capitalization; Indebtedness; Investments	69
4.4	Financial Condition	69
4.5	Litigation; Labor Disputes	69
	i	
4.6	Covernmental Approvals	70
4.6 4.7	Governmental Approvals	71
4.7	Margin Stock, Etc. ERISA Compliance	71 71
4.0	Taxes	71
4.9	Investment Company Act; Public Utility Holding Company Act	71 71
4.10	Title; Security Documents	72
4.12	Environmental Matters	72
4.13	Subsidiaries	73
4.14	Intellectual Property	73
4.15	Insurance	74
4.16	Project Documents	74
4.17	No Default	75
4.18	Compliance with Laws	75
4.19	Disclosure	75
4.20	Utilities, Etc.	76
4.21	Transactions with Affiliates	76
4.22	Project Completion Date; Project Costs	76
4.23	Single-Purpose Entity	76
4.24	Anti-Terrorism Laws	77
SECTION 5.	AFFIRMATIVE COVENANTS	77
5.1	Financial Statements and Other Information	77
5.2	Other Notices	80
5.3	Maintenance of Existence; Conduct of Business	82
5.4	Compliance with Laws	82
5.5	Payment of Taxes, Etc.	82
5.6	Accounting and Financial Management	83
5.7	Inspection	83
5.8	Governmental Approvals	84
5.9	Insurance	84
5.10	Events of Loss and Project Document Claims	88
5.11	Application of Loss Proceeds	89
5.12	Required Hedging Agreements	91
5.13	Chief Place of Business; etc.	91
5.14	Use of Proceeds	92
5.15	Project Construction; Maintenance	92
5.16	Performance of Project Documents	93
5.17	Operating Plan and Budget	93
5.18	Environmental Compliance	94
5.19	Security Documents	95
5.20	Required Debt Service Reserve Amounts	95
5.21	Payment of Project Costs with Project Revenues	96
5.22	Anti-Terrorism Laws	96
5.23	Further Assurances	96
5.24	Separateness	96
5.25	Collateral Account Agreement	96
	ii	
SECTION 6.	NEGATIVE COVENANTS	96
6.1	Limitation on Liens	96
6.2	Indebtedness	98
6.3	Leases	98
6.4	Investments; Subsidiaries	98
6.5	Distributions	98
6.6	Transactions with Affiliates	99
6.7	Merger; Sales and Purchases of Assets	99
6.8	Amendment of Transaction Documents; Additional Project Documents; Change Orders; etc.	100
6.9	Completion; Acceptance Tests	101
6.10	ERISA	101

101

6.11

Sale and Leaseback Transactions

6.12	Certain Agreements	102
6.13	Transfers of Equity Interests	102
6.14	Bank Accounts	102
6.15	Fiscal Year and Fiscal Quarter; Accounting Changes	102
SECTION 7.	FINANCIAL COVENANTS	102
7.1	Debt Service Coverage Ratios	102
SECTION 8.	PAYMENT PROVISIONS; FEES	102
8.1	Repayment of Principal	102
8.2	Voluntary Prepayments	103
8.3	Mandatory Prepayments	104
8.4	Method and Place of Payment	105
8.5	Net Payments	106
8.6	Fees	106
8.7	Application of Payments; Sharing	106
0.7	Application of Payments, Sharing	100
SECTION 9.	EVENTS OF DEFAULT AND REMEDIES	108
9.1	Events of Default	108
9.2	Acceleration	112
9.3	Other Remedies	113
SECTION 10.	THE AGENTS	114
10.1	Appointment.	114
10.2	Delegation of Duties	115
10.3	Liability of the Agents	115
10.4	Reliance by the Agents	115
10.4	Notice of Default	
		116
10.6 10.7	Credit Decision Indemnification of Agents	116 117
10.7	indeminification of rigents	117
	iii	
10.8	Agents in Individual Canacities	117
10.8	Agents in Individual Capacities	117
10.8 10.9	Agents in Individual Capacities Successor Agents	117 117
10.9 SECTION 11.	Successor Agents MISCELLANEOUS	117 118
10.9 SECTION 11. 11.1	Successor Agents  MISCELLANEOUS  Costs and Expenses	117 118 118
10.9 SECTION 11. 11.1 11.2	Successor Agents  MISCELLANEOUS  Costs and Expenses Indemnity	117 118 118 118
10.9 SECTION 11. 11.1 11.2 11.3	Successor Agents  MISCELLANEOUS  Costs and Expenses Indemnity Notices	117 118 118 118 118 121
10.9 SECTION 11. 11.1 11.2 11.3 11.4	Successor Agents  MISCELLANEOUS  Costs and Expenses Indemnity Notices Benefit of Agreement	117 118 118 118 121 122
10.9 SECTION 11. 11.1 11.2 11.3 11.4 11.5	Successor Agents  MISCELLANEOUS  Costs and Expenses Indemnity Notices Benefit of Agreement No Waiver; Remedies Cumulative	117 118 118 118 119 119 119 119 120 120
10.9 SECTION 11. 11.1 11.2 11.3 11.4 11.5 11.6	Successor Agents  MISCELLANEOUS  Costs and Expenses Indemnity Notices Benefit of Agreement No Waiver; Remedies Cumulative No Third Party Beneficiaries	117 118 118 118 119 119 120 122 122
10.9 SECTION 11. 11.1 11.2 11.3 11.4 11.5 11.6 11.7	Successor Agents  MISCELLANEOUS  Costs and Expenses Indemnity Notices Benefit of Agreement No Waiver; Remedies Cumulative No Third Party Beneficiaries Reinstatement	117 118 118 118 119 121 122 122 122 123
10.9 SECTION 11. 11.1 11.2 11.3 11.4 11.5 11.6 11.7 11.8	Successor Agents  MISCELLANEOUS  Costs and Expenses Indemnity Notices Benefit of Agreement No Waiver; Remedies Cumulative No Third Party Beneficiaries Reinstatement No Immunity	117 118 118 118 119 121 122 122 122 123 123
10.9  SECTION 11.  11.1 11.2 11.3 11.4 11.5 11.6 11.7 11.8 11.9	Successor Agents  MISCELLANEOUS  Costs and Expenses Indemnity Notices Benefit of Agreement No Waiver; Remedies Cumulative No Third Party Beneficiaries Reinstatement No Immunity Counterparts	117 118 118 118 119 121 122 122 122 123 123 123
10.9  SECTION 11.  11.1 11.2 11.3 11.4 11.5 11.6 11.7 11.8 11.9 11.10	Successor Agents  MISCELLANEOUS  Costs and Expenses Indemnity Notices Benefit of Agreement No Waiver; Remedies Cumulative No Third Party Beneficiaries Reinstatement No Immunity Counterparts Amendment or Waiver	117 118 118 118 119 121 122 122 122 123 123 123 123 123
10.9  SECTION 11.  11.1 11.2 11.3 11.4 11.5 11.6 11.7 11.8 11.9 11.10 11.11	Successor Agents  MISCELLANEOUS  Costs and Expenses Indemnity Notices Benefit of Agreement No Waiver; Remedies Cumulative No Third Party Beneficiaries Reinstatement No Immunity Counterparts Amendment or Waiver Assignments, Participations, etc.	117  118  118  118  119  121  122  122  122
10.9  SECTION 11.  11.1 11.2 11.3 11.4 11.5 11.6 11.7 11.8 11.9 11.10 11.11 11.12	Successor Agents  MISCELLANEOUS  Costs and Expenses Indemnity Notices Benefit of Agreement No Waiver; Remedies Cumulative No Third Party Beneficiaries Reinstatement No Immunity Counterparts Amendment or Waiver Assignments, Participations, etc. Survival	117  118  118  118  119  121  122  122  122
10.9  SECTION 11.  11.1 11.2 11.3 11.4 11.5 11.6 11.7 11.8 11.9 11.10 11.11 11.12 11.13	Successor Agents  MISCELLANEOUS  Costs and Expenses Indemnity Notices Benefit of Agreement No Waiver; Remedies Cumulative No Third Party Beneficiaries Reinstatement No Immunity Counterparts Amendment or Waiver Assignments, Participations, etc. Survival WAIVER OF JURY TRIAL	117 118 118 118 118 119 121 122 122 122 123 123 123 123 123 124 127
10.9  SECTION 11.  11.1 11.2 11.3 11.4 11.5 11.6 11.7 11.8 11.9 11.10 11.11 11.12 11.13 11.14	Successor Agents  MISCELLANEOUS  Costs and Expenses Indemnity Notices Benefit of Agreement No Waiver; Remedies Cumulative No Third Party Beneficiaries Reinstatement No Immunity Counterparts Amendment or Waiver Assignments, Participations, etc. Survival WAIVER OF JURY TRIAL Right of Set-off	117  118  118  118  118  121  122  122
10.9  SECTION 11.  11.1 11.2 11.3 11.4 11.5 11.6 11.7 11.8 11.9 11.10 11.11 11.12 11.13 11.14 11.15	Successor Agents  MISCELLANEOUS  Costs and Expenses Indemnity Notices Benefit of Agreement No Waiver; Remedies Cumulative No Third Party Beneficiaries Reinstatement No Immunity Counterparts Amendment or Waiver Assignments, Participations, etc. Survival WAIVER OF JURY TRIAL Right of Set-off Severability	117  118  118  118  118  121  122  122
10.9  SECTION 11.  11.1 11.2 11.3 11.4 11.5 11.6 11.7 11.8 11.9 11.10 11.11 11.12 11.13 11.14 11.15 11.16	Successor Agents  MISCELLANEOUS  Costs and Expenses Indemnity Notices Benefit of Agreement No Waiver; Remedies Cumulative No Third Party Beneficiaries Reinstatement No Immunity Counterparts Amendment or Waiver Assignments, Participations, etc. Survival WAIVER OF JURY TRIAL Right of Set-off Severability Domicile of Loans	117  118  118  118  118  121  122  122
10.9  SECTION 11.  11.1 11.2 11.3 11.4 11.5 11.6 11.7 11.8 11.9 11.10 11.11 11.12 11.13 11.14 11.15 11.16 11.17	Successor Agents  MISCELLANEOUS  Costs and Expenses Indemnity Notices Benefit of Agreement No Waiver; Remedies Cumulative No Third Party Beneficiaries Reinstatement No Immunity Counterparts Amendment or Waiver Assignments, Participations, etc. Survival WAIVER OF JURY TRIAL Right of Set-off Severability Domicile of Loans [Reserved]	117  118  118  118  118  119  121  122  122
10.9  SECTION 11.  11.1 11.2 11.3 11.4 11.5 11.6 11.7 11.8 11.9 11.10 11.11 11.12 11.13 11.14 11.15 11.16 11.17 11.18	Successor Agents  MISCELLANEOUS  Costs and Expenses Indemnity Notices Benefit of Agreement No Waiver; Remedies Cumulative No Third Party Beneficiaries Reinstatement No Immunity Counterparts Amendment or Waiver Assignments, Participations, etc. Survival WAIVER OF JURY TRIAL Right of Set-off Severability Domicile of Loans [Reserved] Governing Law; Submission to Jurisdiction	117  118  118  118  118  119  121  122  122
10.9  SECTION 11.  11.1 11.2 11.3 11.4 11.5 11.6 11.7 11.8 11.9 11.10 11.11 11.12 11.13 11.14 11.15 11.16 11.17 11.18 11.19	Successor Agents  MISCELLANEOUS  Costs and Expenses Indemnity Notices Benefit of Agreement No Waiver; Remedies Cumulative No Third Party Beneficiaries Reinstatement No Immunity Counterparts Amendment or Waiver Assignments, Participations, etc. Survival WAIVER OF JURY TRIAL Right of Set-off Severability Domicile of Loans [Reserved] Governing Law; Submission to Jurisdiction Patriot Act Notification	117 118 118 118 118 118 119 121 122 122 122 123 123 123 123 123 124 127 127 128 128 128 128 128 128 129
10.9  SECTION 11.  11.1 11.2 11.3 11.4 11.5 11.6 11.7 11.8 11.9 11.10 11.11 11.12 11.13 11.14 11.15 11.16 11.17 11.18 11.19 11.20	Successor Agents  MISCELLANEOUS  Costs and Expenses Indemnity Notices Benefit of Agreement No Waiver; Remedies Cumulative No Third Party Beneficiaries Reinstatement No Immunity Counterparts Amendment or Waiver Assignments, Participations, etc. Survival WAIVER OF JURY TRIAL Right of Set-off Severability Domicile of Loans [Reserved] Governing Law; Submission to Jurisdiction Patriot Act Notification Complete Agreement	117 118 118 118 118 118 119 121 122 122 122 123 123 123 123 123 124 127 127 127 128 128 128 128 128 129 129
10.9  SECTION 11.  11.1 11.2 11.3 11.4 11.5 11.6 11.7 11.8 11.9 11.10 11.11 11.12 11.13 11.14 11.15 11.16 11.17 11.18 11.19 11.20 11.21	Successor Agents  MISCELLANEOUS  Costs and Expenses Indemnity Notices Benefit of Agreement No Waiver; Remedies Cumulative No Third Party Beneficiaries Reinstatement No Immunity Counterparts Amendment or Waiver Assignments, Participations, etc. Survival WAIVER OF JURY TRIAL Right of Set-off Severability Domicile of Loans [Reserved] Governing Law; Submission to Jurisdiction Patriot Act Notification Complete Agreement No Strict Construction	117 118 118 118 118 118 119 121 122 122 122 123 123 123 123 123 124 127 127 128 128 128 128 128 129 129
10.9  SECTION 11.  11.1 11.2 11.3 11.4 11.5 11.6 11.7 11.8 11.9 11.10 11.11 11.12 11.13 11.14 11.15 11.16 11.17 11.18 11.19 11.20 11.21 11.21	Successor Agents  MISCELLANEOUS  Costs and Expenses Indemnity Notices Benefit of Agreement No Waiver; Remedies Cumulative No Third Party Beneficiaries Reinstatement No Immunity Counterparts Amendment or Waiver Assignments, Participations, etc. Survival WAIVER OF JURY TRIAL Right of Set-off Severability Domicile of Loans [Reserved] Governing Law; Submission to Jurisdiction Patriot Act Notification Complete Agreement No Strict Construction Release of Collateral	117 118 118 118 118 118 119 121 122 122 122 122 123 123 123 123 123
10.9  SECTION 11.  11.1 11.2 11.3 11.4 11.5 11.6 11.7 11.8 11.9 11.10 11.11 11.12 11.13 11.14 11.15 11.16 11.17 11.18 11.19 11.20 11.21 11.22 11.23	Successor Agents  MISCELLANEOUS  Costs and Expenses Indemnity Notices Benefit of Agreement No Waiver; Remedies Cumulative No Third Party Beneficiaries Reinstatement No Immunity Counterparts Amendment or Waiver Assignments, Participations, etc. Survival WAIVER OF JURY TRIAL Right of Set-off Severability Domicile of Loans [Reserved] Governing Law; Submission to Jurisdiction Patriot Act Notification Complete Agreement No Strict Construction Release of Collateral Confidentiality	117 118 118 118 118 119 121 122 122 122 123 123 123 123 123 123
10.9  SECTION 11.  11.1 11.2 11.3 11.4 11.5 11.6 11.7 11.8 11.9 11.10 11.11 11.12 11.13 11.14 11.15 11.16 11.17 11.18 11.19 11.20 11.21 11.22 11.23 11.24	Successor Agents  MISCELLANEOUS  Costs and Expenses Indemnity Notices Benefit of Agreement No Waiver; Remedies Cumulative No Third Party Beneficiaries Reinstatement No Immunity Counterparts Amendment or Waiver Assignments, Participations, etc. Survival WAIVER OF JURY TRIAL Right of Set-off Severability Domicile of Loans [Reserved] Governing Law; Submission to Jurisdiction Patriot Act Notification Complete Agreement No Strict Construction Release of Collateral Confidentiality Joint and Several Liability of the Borrowers	117 118 118 118 118 118 119 121 122 122 122 122 123 123 123 123 124 127 127 127 128 128 128 128 129 129 129 129 129 129 129 129
10.9  SECTION 11.  11.1 11.2 11.3 11.4 11.5 11.6 11.7 11.8 11.9 11.10 11.11 11.12 11.13 11.14 11.15 11.16 11.17 11.18 11.19 11.20 11.21 11.22 11.23	Successor Agents  MISCELLANEOUS  Costs and Expenses Indemnity Notices Benefit of Agreement No Waiver; Remedies Cumulative No Third Party Beneficiaries Reinstatement No Immunity Counterparts Amendment or Waiver Assignments, Participations, etc. Survival WAIVER OF JURY TRIAL Right of Set-off Severability Domicile of Loans [Reserved] Governing Law; Submission to Jurisdiction Patriot Act Notification Complete Agreement No Strict Construction Release of Collateral Confidentiality	117 118 118 118 118 119 121 122 122 122 123 123 123 123 123 123
10.9  SECTION 11.  11.1 11.2 11.3 11.4 11.5 11.6 11.7 11.8 11.9 11.10 11.11 11.12 11.13 11.14 11.15 11.16 11.17 11.18 11.19 11.20 11.21 11.22 11.23 11.24	Successor Agents  MISCELLANEOUS  Costs and Expenses Indemnity Notices Benefit of Agreement No Waiver; Remedies Cumulative No Third Party Beneficiaries Reinstatement No Immunity Counterparts Amendment or Waiver Assignments, Participations, etc. Survival WAIVER OF JURY TRIAL Right of Set-off Severability Domicile of Loans [Reserved] Governing Law; Submission to Jurisdiction Patriot Act Notification Complete Agreement No Strict Construction Release of Collateral Confidentiality Joint and Several Liability of the Borrowers No Fiduciary Duty	117 118 118 118 118 118 119 121 122 122 122 122 123 123 123 123 124 127 127 127 128 128 128 128 129 129 129 129 129 129 129 129

Insurance Provisions

Appendix A

**SCHEDULES**:

Schedule 4.3 Capitalization; Indebtedness
Schedule 4.6 Necessary Governmental Approvals
Schedule 4.11 Storage Locations
Schedule 4.21 Affiliate Transactions

iv

#### **EXHIBITS**:

Exhibit A Form of Notice of Borrowing
Exhibit B-1 Form of Construction Note
Exhibit B-2 Form of Term Note

Exhibit C-1 Form of Construction Requisition

Exhibit C-2 Form of Project Engineer Disbursement Certificate
Exhibit D-1 Form of Borrower Completion Certificate
Exhibit D-2 Form of Project Engineer Completion Certificate

Exhibit E Form of Consent Agreement

Exhibit F Form of Assignment and Acceptance

Exhibit G Form of Notice of Conversion and Continuation

Exhibit H Form of Solvency Certificate

Exhibit I-1 Form of Tax Certificate (Lenders That Are Not Partnerships)
Exhibit I-2 Form of Tax Certificate (Participants That Are Not Partnerships)
Exhibit I-3 Form of Tax Certificate (Participants That Are Partnerships)
Exhibit I-4 Form of Tax Certificate (Lenders That Are Partnerships)

Exhibit J Form of Collateral Account Agreement

#### ANNEXES:

Annex I Construction Loan Commitments
Annex II Applicable Lending Offices
Annex III Term Loan Amortization

V

This **CREDIT AGREEMENT** (as amended, restated, supplemented or otherwise modified from time to time, this "**Agreement**"), dated as of November 7, 2012, is by and among (i) Clean Energy Tranche A LNG Plant, LLC, a limited liability company organized under the laws of the State of Delaware (the "**Tranche A Borrower**"), (ii) Clean Energy Tranche B LNG Plant, LLC, a limited liability company organized under the laws of the State of Delaware (the "**Tranche B Borrower**" and, together with the Tranche A Borrower, collectively, the "**Borrowers**"), (iii) the financial institutions from time to time party hereto as Lenders and (iv) General Electric Capital Corporation, as Administrative Agent and Collateral Agent. Capitalized terms used herein shall have the meanings specified in Section 1.1.

#### WITNESSETH:

WHEREAS, the Borrowers have been organized as Delaware limited liability companies to undertake the development, construction and ownership of two (2) micro-LNG facilities, each with a capacity of approximately 250,000 LNG Gallons per day;

WHEREAS, to finance the development, construction, ownership and operation of the Projects and certain other costs and expenditures associated with the development of the Projects and the financing contemplated herein, the Borrowers have requested that the Lenders provide the credit facilities described herein; and

WHEREAS, the Lenders are willing to provide the credit facilities described herein upon the terms and conditions herein set forth;

NOW, THEREFORE, in consideration of the premises and mutual agreements hereinafter contained, the parties hereto agree as follows:

#### SECTION 1. DEFINITIONS AND RULES OF INTERPRETATION

- 1.1 <u>Defined Terms</u>. Except as otherwise expressly provided herein, capitalized terms used in this Agreement and its Schedules, Appendices, Annexes, Exhibits and Recitals shall have the following meanings:
- "<u>Acceptable Swap Counterparty</u>" shall mean a bank or other financial institution whose long-term senior unsecured debt is rated at least A- by Standard & Poor's and A3 by Moody's or that is otherwise acceptable to the Required Lenders.
- "Acceptance Tests" shall mean the equipment and systems performance testing required to determine that Facility A or Facility B, as applicable, (a) proves all warranties and performance guarantees set forth in the applicable Construction Contracts and (b) will be consistently able to satisfy the operational requirements of the applicable LNG Purchase Agreement. The nature, procedures and testing periods of such Acceptance Tests shall be mutually agreed by the applicable Borrower and the Administrative Agent (in consultation with the Project Engineer).
  - "Additional Project Document" shall mean, with respect to a Project, a Material Additional Project Document or an Immaterial Project Document.

- "Adjusted Eurodollar Rate" shall mean, with respect to any Eurodollar Loans for any Interest Period, (a) an interest rate *per annum* (rounded upward, if necessary, to the nearest 1/100th of 1%) determined by the Administrative Agent to be equal to the Eurodollar Rate for such Eurodollar Loans in effect for such Interest Period divided by (b) 1 *minus* the Statutory Reserves (if any) for such Eurodollar Loans for such Interest Period; *provided* that the Adjusted Eurodollar Rate shall not, at any time, be less than 1.00%.
- "<u>Administrative Agent</u>" shall mean General Electric Capital Corporation, acting in its capacity as administrative agent for the Lenders pursuant hereto, and shall include any successor Administrative Agent appointed pursuant to <u>Section 10.9</u> hereof.
- "<u>Affected Property</u>" shall mean, with respect to any Event of Loss, the Property lost, destroyed, damaged, condemned (including through a Taking) or otherwise taken as a result of such Event of Loss.
- "Affiliate" shall mean, in relation to any Person, any other Person that directly or indirectly, Controls, is Controlled by, or is under common Control with such Person.
  - "Affiliate Project Document" shall mean a Project Document entered into with an Affiliate Project Participant.
  - "Affiliate Project Participant" shall mean a Project Participant that is an Affiliate of a Borrower.
- "Agent-Related Persons" shall mean each Agent, and any successor Agent appointed pursuant to <u>Section 10.9</u> hereof, together with their respective Affiliates and its and their officers, directors, employees, representatives, attorneys and agents.
  - "Agents" shall mean, collectively, the Administrative Agent, the Collateral Agent and the Depositary Agent.
  - "**Agreement**" shall have the meaning provided in the Preamble hereto.
  - "Anti-Terrorism Law" shall have the meaning provided in Section 4.24 hereof.
- "Applicable Lending Office" shall mean, for each Lender and for each Type of Loan, the "Lending Office" of such Lender (or of an Affiliate thereof) designated for such Type of Loan in Annex II hereto or such other office of such Lender (or an Affiliate thereof) as such Lender may from time to time specify to the Administrative Agent and the Borrowers by written notice in accordance with the terms hereof as the office by which its Loans of such Type are to be made and maintained.
  - "Applicable Margin" shall mean (i) in respect of Eurodollar Loans, 7.00% and (ii) in respect of Base Rate Loans, 6.00%.
  - "Approved Gas Supplier" shall mean a supplier of natural gas reasonably acceptable to the Required Lenders.

2

- "Approved Power Supplier" shall mean a supplier of power reasonably acceptable to the Required Lenders.
- "Approved Fund" shall mean any (i) investment company, fund, trust, securitization vehicle or conduit that is (or will be) engaged in making, purchasing, holding or otherwise investing in commercial loans and similar extensions of credit in the ordinary course of its business or (ii) any Person (other than a natural person) that temporarily warehouses loans for any Lender or any entity described in the preceding clause (i) and that, with respect to each of the preceding clauses (i) and (ii), is administered or managed by (A) a Lender, (B) an Affiliate of a Lender or (C) a Person (other than a natural person) or an Affiliate of a Person (other than a natural person) that administers or manages a Lender.
  - "Assignee" shall have the meaning provided in Section 11.11(a) hereof.
  - "Assignment and Acceptance" shall have the meaning provided in Section 11.11(a) hereof.
- "Assumed Interest Rate" shall mean, with respect to any Obligation bearing interest at a floating rate of interest on any date of determination, the rate of interest on such Obligation as in effect on such date of determination; *provided* that in the event that the applicable Borrower has entered into an Interest Rate Hedging Agreement to protect itself against the risk of interest rate fluctuations, such Interest Rate Hedging Agreement shall, to the extent applicable, be taken into account in determining the rate of interest in effect on such date of determination.
  - "Attorney Costs" shall mean all fees and disbursements of any law firm or other legal counsel.
- "Authorized Officer" shall mean (i) with respect to any Person that is a corporation or a limited liability company, the Chairman, Chief Executive Officer, Chief Financial Officer, President, Treasurer or Secretary of such Person or (ii) with respect to any Person that is a partnership, the Chief Executive Officer, Chief Financial Officer, President, Treasurer or Secretary of such Person or of a general partner or managing partner of such Person or such general partner or managing partner, in each case whose name appears on a certificate of incumbency of such Person delivered in accordance with this Agreement, as such certificate may be amended from time to time.
  - "Bank Parties" shall have the meaning provided in Section 11.24 hereof.
  - "Bankruptcy Code" shall mean Title 11 of the United States Code entitled "Bankruptcy" as now and hereafter in effect, or any successor statute.
- "Bankruptcy Law" shall mean the Bankruptcy Code and any other Law of any jurisdiction relating to bankruptcy, insolvency, liquidation, reorganization, moratorium, winding-up or composition or readjustment of debts or any similar Law.
- "Base Case Projections" shall mean, with respect to a Project, a projection of operating results for such Project over a period ending no sooner than eight (8) years beyond the Facility A Date Certain or the Facility B Date Certain, as applicable to such Project, showing the

Borrower's reasonable good faith estimates, as of the applicable Initial Funding Date, of revenue, operating expenses, debt service and minimum and annual (and, for the period covered thereby, average) Debt Service Coverage Ratios, Leverage Ratios and sources and uses of revenues over the forecast period.

"Base Rate" for any day, shall mean the rate per annum equal to the highest of (a) the Federal Funds Rate for such day *plus* 0.50%, (b) the Prime Rate for such day and (c) the then current one-month Adjusted Eurodollar Rate *plus* 1%. Any changes in the Base Rate due to a change in the Prime Rate, the Federal Funds Rate or the Adjusted Eurodollar Rate shall be effective on the effective date of such change in the Prime Rate, the Federal Funds Rate or the Adjusted Eurodollar Rate.

"Base Rate Loans" shall mean Loans that bear interest based upon the Base Rate.

"Borrowers" shall have the meaning provided in the Preamble hereto.

"Borrower Completion Certificate" shall mean, with respect to a Borrower, a certificate, substantially in the form of Exhibit D-1 hereto (such form being subject to review by the Project Engineer subsequent to the appointment thereof) or otherwise in form and substance acceptable to the Administrative Agent, dated the applicable Conversion Date, duly completed and signed by an Authorized Officer of such Borrower.

"Borrowing" shall mean the borrowing of Loans of one Type from the Lenders on a given date (or the conversion of a Loan or Loans of a Lender or Lenders from one Type to another Type on a given date).

"Business Day." shall mean (i) for all purposes other than as covered by clause (ii) below, any day except Saturday, Sunday and any day which shall be in New York City a legal holiday or a day on which banking institutions are authorized, required or permitted by law or other government action to close and (ii) with respect to all notices and determinations in connection with, and payments of principal and interest on, Eurodollar Loans, any day which is a Business Day described in clause (i) above and which is also a day for trading by and between banks in the London interbank eurodollar market.

"Capital Adequacy Regulation" means any guideline, request or directive of any central bank or other Governmental Authority, or any other Law, whether or not having the force of law, in each case, regarding capital adequacy of any bank or of any corporation controlling a bank.

"<u>Capital Lease Obligations</u>" shall mean, for any Person, the obligations of such Person to pay rent or other amounts under a lease of (or other agreement conveying the right to use) Property which obligations are required to be classified and accounted for as a capital lease on a balance sheet of such Person under GAAP and, for purposes hereof, the amount of such obligations shall be the capitalized amount thereof, determined in accordance with GAAP.

"<u>Capital Stock</u>" shall mean, with respect to any Person, any and all shares, interests, participations and/or rights in or other equivalents (however designated, whether voting or nonvoting, ordinary or preferred) in the equity or capital of such Person, now or hereafter outstanding, and any and all rights, warrants or options exchangeable for or convertible into any thereof.

4

"Cash Flow Available for Debt Service" shall mean, with respect to a Borrower, for any period, the excess (if any) of (i) Project Revenues that are deposited into the applicable Project Revenues Collection Account during such period *over* (b) Operation and Maintenance Expenses paid from the applicable Operating Account during such period.

"Change of Control" shall mean (i) Holdings shall cease to own, directly, all of the outstanding Capital Stock of either Borrower, (ii) the Sponsor shall cease to own, directly, at least 66 2/3% of the outstanding Capital Stock of Holdings, (iii) if the Sponsor is not the LNG Purchaser, the Sponsor shall cease to own, directly or indirectly all of the outstanding Capital Stock of the LNG Purchaser, (iv) the Sponsor shall cease to Control (a) either Borrower, (b) Holdings or (c) if the Sponsor is not the LNG Purchaser, the LNG Purchaser, (v) the acquisition of ownership, directly or indirectly, beneficially or of record, by any Person or group (within the meaning of the Securities Exchange Act of 1934 and the rules of the Securities and Exchange Commission thereunder as in effect on the date hereof), other than an employee benefit or stock ownership plan of the Sponsor, of Capital Stock representing more than 50% of the aggregate ordinary voting power represented by the issued and outstanding Capital Stock of the Sponsor or (vi) occupation of a majority of the seats on the board of directors of the Sponsor by Persons who were neither (x) nominated by the board of directors of the Sponsor nor (y) appointed by directors so nominated.

"Change in Law" means (a) the adoption of any law, rule, regulation or treaty (including any rules or regulations issued under or implementing any existing law) after the date of this Agreement, (b) any change in any law, rule, regulation or treaty or in the interpretation or application thereof by any Governmental Authority after the date of this Agreement or (c) compliance by any Lender (or, for purposes of Section 2.16(b), by any lending office of such Lender or by such Lender's holding company, if any) with any request, guideline or directive (whether or not having the force of law) of any Governmental Authority made or issued after the date of this Agreement; provided that notwithstanding anything herein to the contrary: (i) the Dodd-Frank Wall Street Reform and Consumer Protection Act and all requests, rules, guidelines or directives thereunder, issued in connection therewith or in implementation thereof; and (ii) all requests, rules, guidelines and directives promulgated pursuant to Basel III by the Bank for International Settlements, the Basel Committee on Banking Supervision (or any successor or similar authority) or the United States or foreign regulatory authorities, shall in each case be deemed to be a "Change in Law", regardless of the date enacted, adopted, issued or implemented.

"Change Order" shall mean any change order, amendment, supplement, work order, amendment and restatement, waiver or forbearance made with respect to any Construction Agreement.

"Charter Documents" shall mean, with respect to any Person, (i) the articles of incorporation or other similar organizational document of such Person, (ii) the by-laws, operating agreement or other similar document of such Person, (iii) any certificate of designation or instrument relating to the rights of preferred shareholders or other holders of Capital Stock of such Person, and (iv) any shareholder rights agreement or other similar agreement among the holders of Capital Stock of such Person.

- "Checking Accounts" shall mean the deposit accounts permitted to be maintained by a Borrower in accordance with Section 6.14.
- "Closing Date" shall mean the date upon which the conditions precedent set forth in Section 3.1 hereof have been satisfied (or waived by the appropriate Secured Parties).
  - "Closing Date Project Documents" shall mean each LNG Purchase Agreement.
  - "Closing Date Project Participants" shall mean the LNG Purchaser.
  - "Code" shall mean the Internal Revenue Code of 1986.
- "Collateral" shall mean all Property that, in accordance with the terms of the Security Documents, is intended to be subject to any Lien in favor of the Collateral Agent for the benefit of the Secured Parties.
- "Collateral Account Agreement" shall mean a Collateral Account Agreement, substantially in the form of Exhibit J hereto, to be executed among the Borrowers, the Administrative Agent, the Collateral Agent and the Depositary Agent.
  - "Collateral Accounts" shall have the meaning provided in the Collateral Account Agreement.
- "Commitments" shall mean with respect to any Lender, such Lender's Construction Loan Commitments and/or Term Loan Commitments, as the context shall require.
- "Commodity Hedging Agreement" shall mean any agreement (including each confirmation entered into pursuant to any master agreement) providing for swaps, caps, collars, puts, calls, floors, futures, options, spots, forwards, each with respect to or involving the hedge of any energy, generation capacity or fuel, or any other energy related commodity or service, price or price indices for any such commodities or services or any other similar derivative agreements, and any other similar agreements.
  - "Communications" shall have the meaning provided in Section 11.3(a) hereof.
- "Connection Income Taxes" means Other Connection Taxes that are imposed on or measured by net income (however denominated) or that are franchise Taxes or branch profits Taxes.
- "Consent Agreement" shall mean each Acknowledgment and Consent Agreement entered into between a Project Participant or a Specified Site RE Right Grantor and the Collateral Agent and acknowledged by the Borrower, in each case substantially in the form of Exhibit E hereto or otherwise in form and substance reasonably acceptable to the Required Lenders.

6

- "Construction Agreements" shall mean, collectively, (i) the GE EPC Contracts and (ii) any other agreement to which any Finance Party or any Affiliate of any Finance Party is now or hereafter becomes a party providing for the design, development, construction, testing or start-up of the Projects.
- "Construction Budget" shall mean, in respect of a Project, the budget for such Project delivered to each Lender on or before the Closing Date and prepared and certified as such by an Authorized Officer of the applicable Borrower of all Project Costs theretofore incurred and thereafter expected to be incurred by such Borrower in respect of such Project on or prior to the applicable Conversion Date.
- "Construction Loan Commitments" shall mean the Tranche A Construction Loan Commitments and/or the Tranche B Construction Loan Commitments, as the context shall require.
  - "Construction Loans" shall mean Tranche A Construction Loans and/or Tranche B Construction Loans, as the context shall require.
- "Construction Requisition" shall mean, in respect of a Project, a certificate, substantially in the form of Exhibit C-1 hereto (such form being subject to review by the Project Engineer subsequent to the appointment thereof) or otherwise in form and substance acceptable to the Administrative Agent, executed and delivered by an Authorized Officer of the applicable Borrower to the Administrative Agent, including all attachments referred to therein (i) pursuant to Section 3.3(a) hereof in connection with each Disbursement of Construction Loans and (ii) pursuant to Section 5.21 hereof in connection with each application of Project Revenues to the payment of Project Costs.
  - "Contingency" shall mean the aggregate amount specified in the "Contingency" line item in the applicable Construction Budget.
- "Contingent Obligation" shall mean, as to any Person, any obligation of such Person guaranteeing or intending to guarantee any Indebtedness, leases, dividends or other obligations ("primary obligations") of any other Person (the "primary obligor") in any manner, whether directly or indirectly, including any obligation of such Person, whether or not contingent, (a) to purchase any such primary obligation or any property constituting direct or indirect security therefor, (b) to advance or supply funds (i) for the purchase or payment of any such primary obligation or (ii) to maintain working capital or equity capital of the primary obligor or otherwise to maintain the net worth or solvency of the primary obligor, (c) to purchase Property, securities or services primarily for the purpose of assuring the owner of any such primary obligation of the ability of the primary obligor to make payment of such primary obligation, or (d) otherwise to assure or hold harmless the owner of such primary obligation against loss in respect thereof. The amount of any Contingent Obligation shall be deemed to be an amount equal to the stated or determinable amount of the primary obligation in respect of which such Contingent Obligation is made or, if not stated or determinable, the maximum reasonably anticipated liability in respect thereof (assuming such Person is required to perform thereunder) as determined by such Person in good faith.

- "Contractors" shall mean, collectively, the GE Contractor and each other party to a Construction Agreement (other than a Borrower).
- "Control" shall mean, in relation to any Person, the ability to direct or cause the direction of the management or policies of another Person, whether by means of the ownership of stock or other securities with the right to vote, by contract or otherwise, and "Controlled" and "Controlling" shall have meanings correlative thereto.
  - "Conversion Date" shall mean the Facility A Conversion Date and/or the Facility B Conversion Date, as the context shall require.
- "Cost" shall mean, in respect of any Change Order, all costs incurred or to be incurred by the applicable Borrower in respect thereof, any Tranche A Debt Service, Tranche B Debt Service or other costs attributable to a delay in the milestones or completion deadlines set forth in the Construction Agreements, or any other cost incurred by the applicable Borrower directly or indirectly as a result of such Change Order.
  - "  $\underline{\textbf{Debt Service Coverage Statement}} \text{ shall mean each statement delivered pursuant to } \underline{\textbf{Section 5.1(d)(i)(A)}} \text{ or } \underline{\textbf{Section 5.1(d)(ii)(A)}} \text{ hereof.}$
  - "Debt Service and Related Payments Account" shall have the meaning provided in the Collateral Account Agreement.
- "<u>Debt Service Reserve Account</u>" shall mean the Tranche A Debt Service Reserve Account and/or the Tranche B Debt Service Reserve Account, as the context may require.
  - "Default" shall mean any event or circumstance which with notice or lapse of time or both would become an Event of Default.
  - "Default Rate" shall have the meaning provided in Section 2.11(c) hereof.
- "Defaulting Lender" shall mean, subject to Section 2.21(b), any Lender that (a) has refused or failed to (i) fund all or any portion of its Loans within three Business Days after the date such Loans were required to be funded hereunder unless such Lender notifies the Administrative Agent and the Borrowers in writing that such failure is the result of such Lender's determination that one or more conditions precedent to funding (each of which conditions precedent, together with any applicable default, shall be specifically identified in such writing) has not been satisfied, or (ii) pay to the Administrative Agent or any Lender any other amount required to be paid by it hereunder within three Business Days after the date when due, unless such payment is being contested in good faith by such Lender, (b) has notified the Borrowers or the Administrative Agent in writing that it does not intend to comply with its funding obligations hereunder, or has made a public statement to that effect (unless such writing or public statement relates to such Lender's obligation to fund a Loan hereunder and such position is based on such Lender's determination that a condition precedent to funding (which condition precedent, together with any applicable default, shall be specifically identified in such writing or public statement) cannot be satisfied), (c) has failed, within three Business Days after written request by the Administrative Agent or the Borrowers, to confirm in writing to the Administrative Agent and the Borrowers that it will comply with its prospective funding

8

obligations hereunder (provided that such Lender shall cease to be a Defaulting Lender pursuant to this clause (c) upon receipt of such written confirmation by the Administrative Agent and the Borrowers), or (d) has, or has a direct or indirect parent company that has, (i) become the subject of a proceeding under any bankruptcy, insolvency, reorganization, liquidation or similar law or proceeding, or (ii) had appointed for it a receiver, custodian, conservator, trustee, administrator, assignee for the benefit of creditors or similar Person charged with reorganization or liquidation of its business or assets, including the Federal Deposit Insurance Corporation or any other state or federal regulatory authority acting in such a capacity; *provided*, *however*, that a Lender shall not be a Defaulting Lender solely by virtue of the ownership or acquisition of any equity interest in that Lender or any direct or indirect parent company thereof by a Governmental Authority so long as such ownership interest does not result in or provide such Lender with immunity from the jurisdiction of courts within the United States or from the enforcement of judgments or writs of attachment on its assets or permit such Lender (or such Governmental Authority) to reject, repudiate, disavow or disaffirm any contracts or agreements made with such Lender. Any determination by the Administrative Agent that a Lender is a Defaulting Lender under any one or more of clauses (a) through (d) above shall be conclusive and binding absent manifest error, and such Lender shall be deemed to be a Defaulting Lender (subject to Section 2.21(b)) upon delivery of written notice of such determination to the Borrowers and each Lender.

"<u>Depositary Agent</u>" shall mean a commercial bank or trust company selected by the Administrative Agent (and reasonably satisfactory to the Borrowers), to act in the capacity of depositary agent in accordance with the Collateral Account Agreement, and shall include any successor Depositary Agent appointed pursuant to <u>Section 5.4</u> of the Collateral Account Agreement.

- "Disbursement" shall mean any disbursement of a Loan pursuant to this Agreement.
- "<u>Disbursement Date</u>" shall mean the date, including the Initial Funding Date, specified in a Construction Requisition as the date on which Disbursement of a Construction Loan is requested by a Borrower.
- "<u>Disposition</u>" shall mean any sale, transfer or other disposition by the Borrowers to any Person of any Property, and "<u>Dispose</u>" shall have the meaning correlative thereto.
  - "Distribution" shall have the meaning provided in Section 6.5 hereof.
  - "  $\underline{\textbf{Dollars}}" \text{ and the sign } \underline{\textbf{\$}}" \text{ shall each mean freely transferable, lawful money of the United States.}$
- "<u>Drawdown Schedule</u>" shall mean, in respect of a Project, the schedule of expected Disbursements of the Construction Loans to be made during each month prior to the Conversion Date of such Project prepared by the applicable Borrower and delivered to the Administrative Agent on the applicable Initial Funding Date pursuant to <u>Section 3.2(b)</u> hereof.
- "EBITDA" shall mean, in respect of a Borrower, for any period, such Borrower's net income for such period <u>plus</u>, without duplication and to the extent deducted in determining net income for such period, the sum of (a) total income tax expense, (b) interest expense, amortization or write-off of debt

and other fees and charges associated with Indebtedness, (c) depreciation and amortization expense, (d) any extraordinary expense or loss, and (e) any other non-cash charges, and minus, to the extent included in the Borrower's net income for such period, the sum of (a) interest income, (b) any extraordinary income or gains, and (c) any other non-cash income, all as determined in accordance with GAAP. For purposes of the foregoing, interest expense shall be determined after giving effect to any net payments made or received by a Borrower with respect to Required Hedging Agreements.

"Eligible Assignee" shall mean (i) a Lender (other than a Defaulting Lender), (ii) an Affiliate of a Lender (other than a Defaulting Lender), (iii) an Approved Fund, and (iv) any other Person (other than a natural person or a Defaulting Lender) approved by the Administrative Agent (such approval not to be unreasonably withheld); provided that, notwithstanding the foregoing, "Eligible Assignee" shall not include the Borrowers or any of the Borrowers' Affiliates or Subsidiaries.

"Enforcement Action" shall mean any action or proceeding against the Borrowers, Holdings, the Sponsor, the Project or all or any part of the Collateral taken for the purpose of (i) enforcing the rights of any Secured Party in respect of the Collateral or otherwise under the Security Documents, including the initiation of action in any court or before any administrative agency or governmental tribunal to enforce such rights, and any action to exercise any rights provided in Section 9.3 hereof and (ii) adjudicating or seeking a judgment on a claim.

"Environmental Claim" shall mean, with respect to any Person, (i) any notice, claim, administrative, regulatory or judicial or equitable action, suit, Lien, judgment or demand by any other Person or (ii) any other written communication by any Governmental Authority or third party, in either case alleging or asserting such Person's liability for investigatory costs, cleanup costs, consultants' fees, governmental response costs, damages to natural resources (including, without limitation, wetlands, wildlife, aquatic and terrestrial species and vegetation) or other Property, property damages, personal injuries, fines or penalties arising out of, based on or resulting from, in the case of either (i) or (ii), (x) the presence, or Release into the environment, of or exposure of any person or Property to any Hazardous Material at any location, whether or not owned by such Person or (y) circumstances forming the basis of any violation, or alleged violation, of any Environmental Law, or Governmental Approval issued under any Environmental Law.

"Environmental Laws" shall mean any and all Laws relating to the environment (including natural resources), occupational health and safety, remediation of contamination, restoration of environmental quality, or to Releases or threatened Releases of Hazardous Materials into the environment including ambient air, surface water, groundwater, or land, or otherwise relating to the manufacture, processing, distribution, use, treatment, storage, disposal, transport, or handling of Hazardous Materials.

"<u>Equity Contribution</u>" shall mean one or more cash contributions by the Sponsor to the Borrower, in Dollars, that would be treated as a cash equity contribution under GAAP, and not as Indebtedness of the Borrower. For the avoidance of doubt, no Project Revenue shall be credited to or otherwise be considered to be an Equity Contribution.

10

"Equity Contribution Agreement" shall mean that certain Equity Contribution Agreement, dated as of November 7, 2012, among the Sponsor, the Borrowers, the Administrative Agent and the Collateral Agent.

"ERISA" shall mean the Employee Retirement Income Security Act of 1974, as amended from time to time, and the regulations promulgated and rulings issued thereunder. Section references to ERISA are to ERISA, as in effect at the date of this Agreement and any subsequent provisions of ERISA, amendatory thereof, supplemental thereto or substituted therefor.

"ERISA Affiliate" shall mean each Person (as identified in Section 3(9) of ERISA) that together with a Borrower or a Subsidiary of a Borrower would be deemed to be a "single employer" within the meaning of Section 4001 of ERISA or Section 414(b), (c), (m), or (o) of the Code.

"Eurodollar Loans" shall mean Loans which bear interest based on the Adjusted Eurodollar Rate.

"Eurodollar Rate" shall mean, with respect to any Eurodollar Loans for any Interest Period, the rate per annum determined by the Administrative Agent to be the arithmetic mean of the offered rates for deposits in Dollars with a term comparable to such Interest Period that appears on the Reference Page (as defined below) at approximately 11:00 a.m., London, England time, on the Interest Determination Date; provided, however, that (i) if no comparable term for an Interest Period is available, the Eurodollar Rate shall be determined using the weighted average of the offered rates for the two terms most nearly corresponding to such Interest Period and (ii) if there shall at any time no longer exist a Reference Page, "Eurodollar Rate" shall mean, with respect to each day during each Interest Period pertaining to Eurodollar Loans, the rate per annum equal to the rate at which a major financial institution reasonably selected by the Administrative Agent is offered deposits in Dollars at approximately 11:00 a.m., London, England time on the Interest Determination Date in the London interbank market for delivery on the first day of such Interest Period for the number of days comprised therein and in an amount comparable to their respective portions of the amount of such Eurodollar Loan to be outstanding during such Interest Period. "Reference Page" shall mean the Reuters Screen LIBOR01 Page (or such other page as may replace such page on such service for the purpose of displaying the rates at which Dollar deposits are offered by leading banks in the London interbank deposit market).

"Event of Default" shall have the meaning provided in Section 9.1 hereof.

"Event of Loss" shall mean, with respect to any Property of the Borrowers, any loss of, destruction of, or damage to, or any condemnation (including a Taking) or other taking of, such Property.

"Excluded Taxes" shall mean any of the following Taxes imposed on or with respect to a Recipient or required to be withheld or deducted from a payment to a Recipient, (a) Taxes imposed on or measured by net income (however denominated), franchise Taxes, and branch profits Taxes, in each case, (i) imposed by the jurisdiction (or any political subdivision thereof)

under the laws of which such Recipient is organized, or in which its principal office or, in the case of any Lender, its applicable lending office is located or (ii) that are Other Connection Taxes, (b) in the case of a Lender, U.S. federal withholding Taxes imposed on amounts payable to or for the account of such Lender with respect to an applicable interest in a Loan or Commitment pursuant to a law in effect on the date on which (i) such Lender acquires such interest in the Loan or Commitment (other than pursuant to an assignment request by the Borrower under Section 2.20) or (ii) such Lender changes its lending office, except in each case to the extent that, pursuant to Section 2.14, amounts with respect to such Taxes were payable either to such Lender's assignor immediately before such Lender became a party hereto or to such Lender immediately before it changed its lending office, (c) Taxes attributable to such Recipient's failure to comply with Section 2.14(f) and (d) any U.S. federal withholding Taxes imposed under FATCA.

"Expropriation Event" shall mean (a) any condemnation, nationalization, seizure or expropriation by a Governmental Authority of all or a substantial portion of the Project or the Property of a Borrower or of its share capital, (b) any assumption by a Governmental Authority of control of a material portion of the Property or business operations of a Borrower or of its share capital, (c) any taking of any action by a Governmental Authority for the dissolution or disestablishment of a Borrower or (d) any taking of any action by a Governmental Authority that would prevent a Borrower from carrying on its business or operations or a substantial part thereof.

"Extraordinary Receipt" shall mean any cash in excess of \$250,000 received by or paid to or for the account of either Borrower not in the ordinary course of business, including, without limitation, purchase price adjustments, tax refunds, judgments and litigation settlements, pension plan reversions and indemnity payments.

"Facilities" shall mean the collective reference to Facility A and Facility B.

"<u>Facility A</u>" shall mean a micro-LNG facility with a capacity of approximately 250,000 LNG Gallons per day, to be located on Site A and owned by the Tranche A Borrower, all as more fully described in the Facility A Plans and Specifications, together with all other amenities and improvements related thereto owned by the Tranche A Borrower.

"Facility A Conversion Date" shall mean the date on which the conditions precedent set forth in Section 3.4 hereof are satisfied with respect to Facility A.

"Facility A Date Certain" shall mean the earlier of (a) the date thirty (30) months after the Initial Funding Date for Tranche A Construction Loans and (b) December 31, 2016.

"<u>Facility A Plans and Specifications</u>" shall mean all engineering plans and project specifications prepared in connection with the Projects and relating to Facility A, together with any approved amendments, changes or *addenda* thereto, in each case approved by the Tranche A Borrower, the Administrative Agent and the Project Engineer.

"Facility A Project Completion Date" shall mean the date upon which all of the following events shall have occurred with respect to Facility A:

12

- (a) the Commercial Operations Date (as defined in the applicable LNG Purchase Agreement) shall have occurred;
- (b) the total production capacity of Facility A is at least 250,000 LNG Gallons per day (taking into account reasonable measurement tolerances), as verified by the Project Engineer using data generally accepted in the natural gas liquefaction industry;
- (c) all portions of the Work (except for punch list items the total cost of which to complete shall not exceed \$500,000) shall have been completed in accordance with the Construction Agreements and in compliance with all applicable Laws and Governmental Approvals, and all clearing, landscaping, lighting and paving of Site A, and all ancillary construction, upgrades and improvements necessary for the operation of the Project as contemplated by the Transaction Documents, shall have been completed;
  - (d) the Acceptance Tests shall have been successfully completed to the satisfaction of the Project Engineer;
  - (e) the Tranche A Borrower shall have delivered the Borrower Completion Certificate to the Administrative Agent; and
- (f) the Administrative Agent shall have received an executed counterpart of the Project Engineer Completion Certificate with respect to Facility A.

"Facility A Project Costs" shall mean (i) all costs and expenses reasonably and necessarily incurred or to be incurred by the Tranche A Borrower to finance and complete Facility A and achieve the Facility A Project Completion Date (and complete all punch list items) in the manner contemplated by the Transaction Documents, including all reasonable and necessary costs and expenses incurred in connection with the negotiation and preparation of the Transaction Documents, all fees payable in respect of the Tranche A Construction Loans, and all other reasonable and necessary expenses required for the financing, development, design, construction, equipment procurement, installation, start-up and initial operation of Facility A, (ii) all Tranche A Interest Expense and Operation and Maintenance Expenses incurred during the construction, installation and start-up of Facility A, and (iii) the deposit into the Tranche A Debt Service Reserve Account contemplated by Section 3.4(m) hereof. "Facility A Project Costs" shall not include (a) payments of principal of any Indebtedness or (b) any payments of any kind to the Sponsor or any Affiliate thereof (other than payments pursuant to an Affiliate Project Document relating to Facility A).

"Facility A Project Schedule" shall mean the Project Schedule for Facility A.

"<u>Facility B</u>" shall mean a micro-LNG facility with a capacity of approximately 250,000 LNG Gallons per day, to be located on Site B and owned by the Tranche B Borrower, all as more fully described in the Facility B Plans and Specifications, together with all other amenities and improvements related thereto owned by the Tranche B Borrower.

"Facility B Date Certain" shall mean the earlier of (a) the date thirty (30) months after the Initial Funding Date for Tranche B Construction Loans and (b) December 31, 2016.

"<u>Facility B Plans and Specifications</u>" shall mean all engineering plans and project specifications prepared in connection with the Projects and relating to Facility B, together with any approved amendments, changes or *addenda* thereto, in each case approved by the Tranche B Borrower, the Administrative Agent and the Project Engineer.

"Facility B Project Completion Date" shall mean the date upon which all of the following events shall have occurred with respect to Facility B:

- (a) the Commercial Operations Date (as defined in the applicable LNG Purchase Agreement) shall have occurred;
- (b) the total production capacity of Facility B is at least 250,000 LNG Gallons per day (taking into account reasonable measurement tolerances), as verified by the Project Engineer using data generally accepted in the natural gas liquefaction industry;
- (c) all portions of the Work (except for punch list items the total cost of which to complete shall not exceed \$500,000) shall have been completed in accordance with the Construction Agreements and in compliance with all applicable Laws and Governmental Approvals, and all clearing, landscaping, lighting and paving of Site B, and all ancillary construction, upgrades and improvements necessary for the operation of the Project as contemplated by the Transaction Documents, shall have been completed;
  - (d) the Acceptance Tests shall have been successfully completed to the satisfaction of the Project Engineer;
  - (e) the Tranche B Borrower shall have delivered the Borrower Completion Certificate to the Administrative Agent; and
- (f) the Administrative Agent shall have received an executed counterpart of the Project Engineer Completion Certificate with respect to Facility B.

"Facility B Project Costs" shall mean (i) all costs and expenses reasonably and necessarily incurred or to be incurred by the Tranche B Borrower to finance and complete Facility B and achieve the Facility B Project Completion Date (and complete all punch list items) in the manner contemplated by the Transaction Documents, including all reasonable and necessary costs and expenses incurred in connection with the negotiation and preparation of the Transaction Documents, all fees payable in respect of the Tranche B Construction Loans, and all other reasonable and necessary expenses required for the financing, development, design, construction, equipment procurement, installation, start-up and initial operation of Facility B, (ii) all Tranche B Interest Expense and Operation and Maintenance Expenses incurred during the construction, installation and start-up of Facility B, and (iii) the deposit into the Tranche B Debt Service Reserve Account contemplated by Section 3.4(m) hereof. "Facility B Project Costs" shall not include (a) payments of principal of any Indebtedness or (b) any payments of any kind to the Sponsor or any Affiliate thereof (other than payments pursuant to an Affiliate Project Document relating to Facility B).

14

"Facility B Project Schedule" shall mean the Project Schedule for Facility B.

"FATCA" shall mean Sections 1471 through 1474 of the Code, as of the date of this Agreement (or any amended or successor version that is substantively comparable and not materially more onerous to comply with), any current or future regulations or official interpretations thereof and any agreements entered into pursuant to Section 1471(b)(1) of the Code.

"Federal Funds Rate" shall mean, for any day, the rate *per annum* (rounded upwards, if necessary, to the nearest 1/100 of 1%) equal to the weighted average of the rates on overnight Federal funds transactions with members of the Federal Reserve System arranged by Federal funds brokers on such day, as published by the Federal Reserve Bank of New York on the Business Day next succeeding such day; *provided* that (i) if the day for which such rate is to be determined is not a Business Day, the Federal Funds Rate for such day shall be such rate on such transactions on the next preceding Business Day as so published on the next succeeding Business Day and (ii) if such rate is not so published for any day, the Federal Funds Rate for such day shall be the average rate charged to the Administrative Agent (in its individual capacity) on such day on such transactions as determined by the Administrative Agent.

"Fees" shall mean all amounts payable pursuant to or referred to in Section 8.6 hereof.

"<u>Finance Documents</u>" shall mean, collectively, this Agreement, the Notes (if any), the Sponsor Guaranty, the Security Documents, the Equity Contribution Agreement and the Required Hedging Agreements entered into by any Secured Swap Counterparty.

"Finance Party" shall mean, collectively, the Borrowers, the Sponsor and Holdings.

"Foreign Lender" shall mean any Lender that is not a U.S. Person.

"GAAP" shall mean generally accepted accounting principles and practices as in effect from time to time in the United States.

"Gas Supply Agreements" shall mean, collectively, (i) a gas supply agreement for the supply of natural gas to Facility A, to be entered into after the Closing Date and before the Initial Funding Date for Tranche A Construction Loans, between an Approved Gas Supplier and the Tranche A Borrower and (ii) a gas supply agreement for the supply of natural gas to Facility B, to be entered into after the Closing Date and before the Initial Funding Date for Tranche B Construction Loans, between an Approved Gas Supplier and the Tranche B Borrower.

"GE Capital" shall mean General Electric Capital Corporation, a Delaware corporation.

"GE Contractor" shall mean GE Oil & Gas, Inc., a Delaware corporation.

"GE EPC Contracts" shall mean, collectively, (i) the EPC Contract for Facility A to be entered into by and among the GE Contractor and the Tranche A Borrower and (ii) the EPC Contract for Facility B to be entered into by and among the GE Contractor and the Tranche B Borrower.

15

"Governmental Approval" shall mean any authorization, consent, approval, license, ruling, permit, tariff, rate, certification, exemption, filing, variance, claim, order, judgment, decree, publication, notice to, declaration of or with, or registration by or with, any Governmental Authority.

"Governmental Authority" shall mean any government, governmental department, commission, board, bureau, agency, regulatory authority, instrumentality, judicial or administrative body, central bank, domestic or foreign, federal, state or local having jurisdiction over the matter or matters in question.

"Hazardous Material" shall mean any substance, material, or waste that is regulated as a hazardous waste, hazardous material, hazardous substance, hazardous chemical, toxic substance, contaminant or pollutant under any applicable Environmental Law or which could give rise to liability under any Environmental Law, including, but not limited to, any petroleum or petroleum product, friable asbestos, mercury, transformers or other equipment that contain dielectric fluid containing polychlorinated biphenyls (PCB's).

"Hedging Agreement" shall mean (i) any Interest Rate Hedging Agreement, (ii) any Commodity Hedging Agreement and (iii) any other agreement in respect of a basis swap, forward rate transaction, commodity swap, commodity option, equity or equity index swap, equity or equity index option, bond option, interest rate option, cap transaction, floor transaction, collar transaction, currency swap transaction, cross-currency rate swap transaction, currency option or any other similar transaction (including any option with respect to any of the foregoing transactions) or any combination of the foregoing transactions entered into by the Borrower.

"Hedging Claims" shall mean any settlement amounts or other breakage or termination fees in connection with any unwinding, breach or termination of any Required Hedging Agreement.

"Holdings" shall mean Clean Energy LNG Holdings, LLC, a Delaware limited liability company.

"Immaterial Additional Project Document" shall mean, with respect to a Project, any contract or agreement relating to the development, construction, testing, operation, maintenance, repair, insurance, financing or use of such Project entered into by the applicable Borrower with any other Person subsequent to the date of this Agreement (including any contract(s) or agreement(s) entered into in substitution for any Project Document that has been terminated in accordance with its terms or otherwise) other than a Material Additional Project Document.

"Immaterial Project Participants" shall mean each party (other than the Borrowers) to an Immaterial Additional Project Document.

"Indebtedness" of any Person shall mean (i) all indebtedness of such Person for borrowed money, (ii) the deferred purchase price of assets or services which in accordance with GAAP would be shown on the liability side of the balance sheet of such Person, (iii) the face amount of all letters of credit issued for the account of such Person and, without duplication, all drafts drawn thereunder, (iv) all Indebtedness of a second Person secured by any Lien on any Property owned by such first Person, whether or not such Indebtedness has been assumed, (v) all

16

Capital Lease Obligations of such Person, (vi) all obligations of such Person to pay a specified purchase price for goods or services whether or not delivered or accepted, *i.e.*, take-or-pay and similar obligations, (vii) all net obligations of such Person under Hedging Agreements, and (viii) all Contingent Obligations of such Person; *provided* that Indebtedness shall not include trade payables arising in the ordinary course of business so long as such trade payables are payable within ninety (90) days of the date the respective goods are delivered or the respective services are rendered and are not overdue.

"Indemnified Liabilities" shall have the meaning provided in Section 11.2(a) hereof.

"Indemnified Person" shall have the meaning provided in Section 11.2(a) hereof.

"<u>Indemnified Taxes</u>" shall mean (a) Taxes, other than Excluded Taxes, imposed on or with respect to any payment made by or on account of any obligation of any Finance Party under any Finance Document and (b) to the extent not otherwise described in (a), Other Taxes.

"<u>Initial Funding Date</u>" shall mean, in respect of a Tranche, the date upon which the conditions precedent set forth in <u>Section 3.2</u> hereof have been satisfied (or waived by the appropriate Secured Parties) with respect to such Tranche.

"Insurance Advisor" shall mean any Person from time to time appointed by the Required Lenders to act as Insurance Advisor for the purposes of this Agreement.

"Insurance Proceeds" shall mean all amounts payable to a Borrower or any Secured Party in respect of any insurance required to be maintained (or caused to be maintained) by such Borrower pursuant to Section 5.9 hereof (other than general liability insurance, delayed completion insurance and business interruption insurance), regardless of whether such payments are received from any insurer or from the Contractor or any other party pursuant to any Construction Agreement or otherwise.

"Interest Determination Date" shall mean, with respect to any Eurodollar Loan, the second Business Day prior to the commencement of any Interest Period relating to such Eurodollar Loan.

"Interest Period" shall have the meaning provided in Section 2.12 hereof.

"Interest Rate Hedging Agreement" shall mean any agreement in respect of any interest rate swap transaction, cap or collar agreement or similar transaction.

"Investment" in any Person shall mean, without duplication: (i) the acquisition (whether for cash, securities, other Property, services or otherwise) or holding of Capital Stock, bonds, notes, debentures, partnership or other ownership interests or other securities of such Person, or any agreement to make any such acquisition or to make any capital contribution to such Person; or (ii) the making of any deposit with (excluding deposits with financial institutions available for withdrawal on demand), or advance, loan or other extension of credit to, such Person.

"IRS" means the United States Internal Revenue Service.

17

"<u>Law</u>" shall mean, with respect to any Person (i) any statute, law, regulation, ordinance, rule, judgment, order, decree, permit, concession, grant, treaty, franchise, license, agreement or other governmental restriction or any interpretation or administration of any of the foregoing by any Governmental Authority (including Governmental Approvals) and (ii) any directive, guideline, policy, requirement or any similar form of decision of or determination by any Governmental Authority which is applicable to or binding on such Person or its Property, in each case, whether now or hereafter in effect (including, in each case, any Environmental Law).

"Lender Affiliate Project Document" shall mean any Project Document to which any Lender Affiliate Project Participant is a party.

"Lender Affiliate Project Participant" shall mean any Project Participant that is an Affiliate of General Electric Capital Corporation.

"Lenders" shall mean the lenders listed on Annex I hereto, and any other Person that shall have become a party hereto as a lender pursuant to an Assignment and Acceptance, other than such Person that ceases to be a party hereto pursuant to an Assignment and Acceptance or any other documentation, in each case, in accordance with Section 11.11.

"Leverage Ratio Statement" shall mean each statement delivered pursuant to Section 5.1(d)(i)(C) or Section 5.1(d)(ii)(C) hereof.

"Lien" shall mean, with respect to any Property of any Person, any mortgage, lien, deed of trust, hypothecation, fiduciary transfer of title, assignment by way of security, lien, pledge, charge, lease, sale and lease-back arrangement, easement, servitude, trust arrangement (including any statutory trust), or security interest or encumbrance of any kind in respect of such Property, or any preferential arrangement having the practical effect of constituting a security interest with respect to the payment of any obligation with, or from the proceeds of, any Property of any kind (and any Property acquired or held subject to the interest of a vendor or lessor under any conditional sale agreement, Capital Lease Obligation or other title retention agreement relating to such Property shall be deemed subject to a Lien).

"LNG" shall mean liquefied natural gas, natural gas that has been cooled at normal pressure to achieve a liquid state suitable for transport.

"LNG Gallon" shall have the meaning provided in the LNG Purchase Agreements.

"LNG Purchase Agreements" shall mean (i) that certain LNG Sale and Purchase Agreement, dated as of November 7, 2012, by and between the Tranche A Borrower and the LNG Purchaser and (ii) (i) that certain LNG Sale and Purchase Agreement, dated as of November 7, 2012, by and between the Tranche B Borrower and the LNG Purchaser.

"LNG Purchaser" shall mean Clean Energy, a California corporation.

"Loans" shall mean Construction Loans and/or Term Loans, as the context shall require.

"<u>Loss Proceeds</u>," shall mean, with respect to any Event of Loss, any Insurance Proceeds, condemnation awards or other compensation, awards, damages and other payments or relief

18

(including any compensation payable in connection with a Taking) with respect to such Event of Loss (excluding, in each case, the proceeds of general liability insurance, delayed completion insurance, and business interruption insurance).

"Loss Proceeds Account" shall have the meaning provided in the Collateral Account Agreement.

"Margin Stock" shall mean margin stock within the meaning of Regulation U and Regulation X.

"Material Additional Project Document" shall mean, with respect to a Project, any contract or agreement relating to the development, construction, testing, operation, maintenance, repair, insurance, financing or use of such Project entered into by the applicable Borrower with any other Person subsequent to the date of this Agreement (including any contract(s) or agreement(s) entered into in substitution for any Project Document that has been terminated in accordance with its terms or otherwise) other than any such contract or agreement or series of related contracts or agreements (other than any Gas Supply Agreement or any Power Purchase Agreement) which (i) provides for the payment by Borrowers of less than \$500,000 per annum, or the provision to Borrowers of less than \$500,000 per annum in value of goods or services, and (ii) is not reasonably necessary or critical to a Project or does not cover goods or services that cannot be replaced on commercially reasonable terms.

"Material Adverse Effect" shall mean a material adverse effect on (i) a Project, (ii) the business, liabilities (actual and contingent), operations, condition (financial or otherwise) or a material portion of the Property of a Borrower, Holdings or the Sponsor, (iii) the legality, validity or enforceability of any provision of any Transaction Document (other than any Immaterial Additional Project Document), (iv) the ability of a Borrower, any other Finance Party

or any other Project Participant (other than any Immaterial Project Participant) to timely perform any of its obligations under any of the Transaction Documents to which it is a party, (v) the rights and remedies of the Secured Parties under any of the Finance Documents (other than as a result of the gross negligence or willful misconduct of the Collateral Agent) or (vi) the Security Interests provided under the Security Documents (other than as a result of the gross negligence or willful misconduct of the Collateral Agent).

"<u>Modified Unutilized Tranche A Construction Loan Commitment Amount</u>" shall mean, for each Lender, at any time, the product of (i) the Unutilized Tranche A Construction Loan Commitment of such Lender at such time and (ii) Unutilized Commitment Coefficient.

"Modified Unutilized Tranche B Construction Loan Commitment Amount" shall mean, for each Lender, at any time, the product of (i) the Unutilized Tranche B Construction Loan Commitment of such Lender at such time and (ii) Unutilized Commitment Coefficient.

"Monthly Transfer Date" shall mean the last Business Day of each month commencing on the first such day occurring on or after (i) in respect of the Tranche A Borrower, the Facility A Conversion Date and (ii) in respect of the Tranche B Borrower, the Facility B Conversion Date.

"Moody's" shall mean Moody's Investors Services, Inc. or its successor.

19

"Mortgage" shall mean any mortgage or deed of trust (including a leasehold mortgage or leasehold deed of trust) with respect to a Site granted by a Borrower to the Collateral Agent for the benefit of the Secured Parties which shall be in form and substance reasonably satisfactory to the Administrative Agent.

"<u>Multiemployer Plan</u>" shall mean a multiemployer plan, as defined in Section 4001(a)(3) of ERISA, to which the Borrower or any ERISA Affiliate is making or accruing an obligation to make contributions or has within any of the preceding five plan years made or accrued an obligation to make contributions.

"Necessary Governmental Approval" shall have the meaning provided in Section 4.6(a) hereof.

"<u>Net Available Amount</u>" shall mean, with respect to any Event of Loss or any Project Document Claim, the aggregate amount of Loss Proceeds (in the case of an Event of Loss) or gross proceeds (in the case of a Project Document Claim) received by the applicable Borrower or any Secured Party in respect of such Event of Loss or Project Document Claim, as the case may be, net of reasonable expenses incurred in connection with the collection thereof.

"Net Disposition Proceeds" shall mean, with respect to any Disposition, the gross cash proceeds received from such Disposition (including any cash received by way of deferred payment pursuant to a promissory note, receivable or otherwise, but only as and when received), net of (i) the reasonable out-of-pocket costs of such Disposition, including fees, expenses and commissions with respect to legal, accounting, financial advisory, brokerage and other professional services provided in connection with such Disposition and (ii) the incremental taxes paid or payable by the applicable Borrower as a result of such Disposition.

"Net Issuance Proceeds" shall mean, with respect to any incurrence of Indebtedness or issuance of Capital Stock, the gross cash proceeds received from such incurrence or issuance, in each case net of brokers', advisors' and investment banking fees and other customary out-of-pocket underwriting discounts, commissions and other customary out-of-pocket cash costs, fees and expenses (including reasonable attorneys' and accountants' fees), in each case incurred in connection with such transaction.

"Notes" shall mean the collective reference to Tranche A Construction Notes, Tranche B Construction Notes, Tranche A Term Notes and Tranche B Term Notes.

"Notice of Borrowing" shall have the meaning provided in Section 2.5 hereof.

"Notice of Conversion or Continuation" shall have the meaning provided in Section 2.10 hereof.

"<u>Notice Office</u>" shall mean the office of the Administrative Agent located at 800 Long Ridge Road, Stamford, CT 06927, or such other office, telephone or facsimile number as the Administrative Agent may hereafter designate in writing as such to each of the other parties hereto.

20

"Obligations" shall mean, collectively, (i) all loans, advances, debts, liabilities, and obligations (including all interest that accrues after the commencement of any case or proceeding by or against a Borrower in bankruptcy, whether or not allowed in such case or proceeding), howsoever arising, owed by either Borrower under any Finance Document to the Administrative Agent or any other Secured Party of every kind and description (whether or not evidenced by any note or instrument and whether or not for the payment of money), direct or indirect, absolute or contingent, due or to become due, now existing or hereafter arising, including all interest, fees, charges, expenses, attorneys' fees and consultants' fees chargeable to such Borrower, (ii) any and all sums advanced by the Administrative Agent or any other Secured Party in order to preserve the Collateral or to preserve the Security Interests, (iii) in the event of any Enforcement Action, the reasonable expenses of retaking, holding, preparing for sale or lease, selling or otherwise disposing of or realizing on the Collateral, or of any exercise by the Administrative Agent and/or the Secured Parties of their rights under the Security Documents, together with attorneys' fees and court costs, and (iv) the obligations of either Borrower under any Required Hedging Agreement entered into with any Secured Swap Counterparty and any obligations of either Borrower to General Electric Capital Corporation or an Affiliate thereof in connection therewith. For the avoidance of doubt, "Obligations" shall not include any obligations in respect of the Warrant Agreement, the Warrant or the Registration Rights Agreement

"Operating Account" shall have the meaning provided in the Collateral Account Agreement.

"Operating Budget" shall have the meaning provided in Section 5.17(a) hereof.

"Operating Year" shall mean (i) with respect to Facility A, each fiscal year (or portion thereof) after the Facility A Conversion Date and (ii) with respect to Facility B, each fiscal year (or portion thereof) after the Facility B Conversion Date.

"Operation and Maintenance Expenses" shall mean, with respect to each portion of a Project that is in operation (regardless of whether the Facility A Project Completion Date or the Facility B Project Completion Date has occurred) for any period on or after the date such portion of such Project has begun operation, collectively, without duplication, all reasonable (i) expenses incurred by the applicable Borrower of administering and operating such portion of such Project and of maintaining it in accordance with Prudent Industry Practices, (ii) fuel, feedstock and other input procurement and transportation costs payable by the applicable Borrower, (iii) direct operating and maintenance costs of the applicable Facility payable by the applicable Borrower, (iv) insurance premiums payable by the applicable Borrower, (v) property, sales, value-added and excise taxes payable by the applicable Borrower (other than taxes imposed on or measured by income or receipts), (vi) costs and fees incurred by the applicable Borrower in connection with obtaining and maintaining in effect the Governmental Approvals required in connection with the applicable Project and (vii) legal, accounting and other professional fees incurred by the applicable Borrower in the ordinary course of business in connection with the applicable Project, in each case as set forth in an Operating Budget (or any amendment thereof) approved in accordance with Section 5.17; provided that (a) with effect from the Facility A Project Completion Date or the Facility B Project Completion Date, as applicable, "Operation and Maintenance Expenses" shall include all of the foregoing with respect to the entire Project and (b) "Operation and Maintenance Expenses" shall not include payments into the Tranche A Debt Service Reserve Account or Tranche B Debt Service Reserve Account, depreciation, or any items properly chargeable by GAAP to fixed capital accounts.

21

"Other Connection Taxes" shall mean, with respect to any Recipient, Taxes imposed as a result of a present or former connection between such Recipient and the jurisdiction imposing such Tax (other than connections arising from such Recipient having executed, delivered, become a party to, performed its obligations under, received payments under, received or perfected a security interest under, engaged in any other transaction pursuant to or enforced any Finance Document, or sold or assigned an interest in any Loan or Finance Document).

"Other Taxes" shall mean all present or future stamp, court or documentary, intangible, recording, filing or similar Taxes that arise from any payment made under, from the execution, delivery, performance, enforcement or registration of, from the receipt or perfection of a security interest under, or otherwise with respect to, any Finance Document.

"Participant" shall have the meaning assigned to such term in clause (f) of Section 11.11.

"Participant Register" has the meaning specified in clause (f) of Section 11.11.

"Patriot Act" shall mean the Uniting and Strengthening America by Providing Appropriate Tools Required to Intercept and Obstruct Terrorism Act of 2001, Public Law 107-56, signed into law October 26, 2001.

"Permitted Investments" shall mean, for any Person: (a) direct obligations of the United States, or of any agency thereof, or obligations guaranteed as to principal and interest by the United States, or of any agency thereof, in either case maturing not more than 180 days from the date of acquisition thereof by such Person; (b) Dollar time deposits in the London interbank market with, or certificates of deposit issued by, any bank or trust company licensed under the laws of the United States or any state thereof which has outstanding senior long-term unsecured indebtedness which is rated (on the date of acquisition thereof) A+ or Al or better by Standard & Poor's or Moody's, respectively, maturing not more than 180 days from the date of acquisition thereof by such Person; (c) commercial paper rated A-1 or P-1 or better by Standard & Poor's or Moody's, respectively, maturing not more than 180 days from the date of acquisition thereof by such Person; and (d) money market funds rated at least "AA" by Standard & Poor's or "Aa" or better by Moody's.

"Permitted Lien" shall mean any Lien permitted to be incurred by the Borrowers pursuant to Section 6.1 hereof.

"Person" shall mean any individual, corporation, limited liability company, company, voluntary association, partnership, joint venture, trust, or other enterprise or unincorporated organization or government (or any agency, instrumentality or political subdivision thereof).

"Plan" shall mean any pension plan as defined in Section 3(2) of ERISA other than a Multiemployer Plan.

22

"Plans and Specifications" shall mean the Facility A Plans and Specifications and/or the Facility B Plans and Specifications, as the context shall require.

"Platform" shall have the meaning provided in Section 11.3(d) hereof.

"Pledge Agreement" shall mean the Pledge and Guaranty Agreement, dated as of November 7, 2012, between Holdings and the Collateral Agent for the benefit of the Secured Parties.

"Power Purchase Agreements" means collectively, (i) a power purchase agreement for the supply of power to Facility A, to be entered into after the Closing Date and before the Initial Funding Date for Tranche A Construction Loans, between an Approved Power Supplier and the Tranche A Borrower and (ii) a power purchase agreement for the supply of power to Facility B, to be entered into after the Closing Date and before the Initial Funding Date for Tranche B Construction Loans, between an Approved Power Supplier and the Tranche B Borrower.

"Prime Rate" shall mean, at any time, a rate per annum equal to the rate last quoted by The Wall Street Journal as the "U.S. prime rate" or, if The Wall Street Journal ceases to quote such rate, the highest per annum interest rate published by the Federal Reserve Board in Federal Reserve Statistical Release H.15 (519) (Selected Interest Rates) as the "bank prime loan" rate or, if such rate is no longer quoted therein, any similar rate quoted therein (as determined by the Administrative Agent) or any similar release by the Federal Reserve Board (as determined by the Administrative Agent).

"Project Costs" shall mean Facility A Project Costs and/or Facility B Project Costs, as the context shall require.

"Projected Debt Service Coverage Statement" shall mean each statement delivered pursuant to Section 5.1(d)(i)(B) or Section 5.1(d)(ii)(B) hereof.

"Project Document Claim" shall mean any claim of the Borrowers against any Project Participant under any Project Document.

"Project Documents" shall mean each Construction Agreement, each LNG Purchase Agreement, each Gas Supply Agreement, each Power Purchase Agreement and each Additional Project Document.

"Project Engineer" shall mean any Person from time to time appointed by the Required Lenders to act as Project Engineer for the purposes of this Agreement.

"<u>Project Engineer Completion Certificate</u>" shall mean a certificate, substantially in the form of <u>Exhibit D-2</u> hereto (such form being subject to review by the Project Engineer subsequent to the appointment thereof) or otherwise in form and substance acceptable to the required Lenders, dated the applicable Conversion Date, duly completed and signed by the Project Engineer.

"Project Participants" shall mean the LNG Purchaser, each Contractor, each Approved Gas Supplier, each Approved Power Supplier, the Sponsor and each other party (other than the

23

Borrowers) to a Project Document, and each Replacement Project Participant. For the avoidance of doubt, after the termination of a Project Document by virtue of scheduled expiration in the ordinary course in accordance with its terms, "Project Participant" shall not include the party to such Project Document (except to the extent such Person is also party to another Project Document).

"Project Revenues" shall mean, for any period, without duplication, the aggregate of all revenues received by a Borrower in respect of a Project during such period from (i) payments made by the LNG Purchaser pursuant to the applicable LNG Purchase Agreement, (ii) interest accrued on, and other income derived from, the balance outstanding during such period in the applicable Collateral Accounts (including from Permitted Investments), and (iii) the proceeds of any business interruption insurance; provided that Project Revenues shall exclude, to the extent otherwise included, (1) net amounts receivable under any Hedging Agreements, (2) proceeds payable in respect of any insurance (other than business interruption insurance) or (3) warranty or indemnity payments or damages payable to the applicable Borrower under any Project Document.

"Project Revenues Collection Account" shall have the meaning provided in the Collateral Account Agreement.

"Project Schedule" has the meaning specified in clause (iii) of Section 3.2(n) hereof.

"<u>Projects</u>" shall mean, collectively, the Facilities, the Sites, and all easements, leasehold interests, licenses, permits, contract rights and other real and personal property interests now owned or hereafter acquired by either Borrower or in which either Borrower has any rights.

"Property" shall mean any property of any kind whatsoever, whether real, personal or mixed and whether tangible or intangible, and any right or interest therein.

"Prudent Industry Practices" means the professional practices, methods, equipment, specifications and safety and output standards and industry codes applicable to projects of the same type and capacity as the Projects, with respect to the design, installation, operation, maintenance and use of equipment and similar machinery, all of the above in compliance with applicable standards of safety, output, dependability, efficiency, environmental protection and economy, including recommended practice, of a good, safe, prudent and workman-like character. Prudent Industry Practices are not intended to be limited to the optimum or minimum practice or method to the exclusion of all others, but rather to be a spectrum of commercially reasonable and prudent practices and methods as practiced in the industry.

"Quarterly Dates" shall mean the last Business Day of each March, June, September and December.

"Quarterly Distribution Date" shall mean each Monthly Transfer Date occurring in the months of February, May, August and November.

"Recipient" shall mean (a) the Administrative Agent, and (b) any Lender.

"Register" shall have the meaning provided in Section 2.9(c) hereof.

24

"Registration Rights Agreement" shall mean that certain Registration Rights Agreement, dated as of November 7, 2012, between the Sponsor and GE Energy Financial Services, Inc.

"Regulation D" shall mean Regulation D of the Board of Governors of the Federal Reserve system (or any successor).

"Regulation U" shall mean Regulation U of the Board of Governors of the Federal Reserve system (or any successor).

"Regulation X" shall mean Regulation X of the Board of Governors of the Federal Reserve system (or any successor).

"Release" shall mean any depositing, spilling, leaking, pumping, pouring, placing, emitting, discarding, abandoning, emptying, discharging, migrating, injecting, escaping, leaching, dumping, or disposing into the environment (including the abandonment or discarding of barrels, containers, and other closed receptacles containing any Hazardous Material).

"Replaced Lender" shall have the meaning set forth in Section 2.20 hereof.

"Replacement Lender" shall have the meaning set forth in Section 2.20 hereof.

"Replacement Project Document" shall mean any contract or agreement in form and substance reasonably satisfactory to the Required Lenders entered into by either Borrower with any Replacement Project Participant subsequent to the date of this Agreement in substitution for any Project Document that has been terminated (other than by virtue of scheduled expiration in the ordinary course in accordance with its terms).

"Replacement Project Participant" shall mean, (a) in the case of any Project Participant other than an Immaterial Project Participant, with respect to such Project Participant, any Person reasonably satisfactory to the Required Lenders and having credit, or acceptable credit support, equal to or greater than that of the replaced Project Participant on the date that the applicable Project Document was entered into who, pursuant to a Replacement Project Document, assumes the obligations of the replaced Project Participant on terms and conditions no less favorable to the Borrower than those applicable to the replaced Project Participant pursuant to the applicable Project Document; and (b) in the case of any Immaterial Project Participant, any Person selected by the applicable Borrower.

"Required Debt Service Reserve Amount" shall mean, the Tranche A Required Debt Service Reserve Amount and/or the Tranche B Required Debt Service Reserve Amount, as the context may require.

"Required Hedging Agreement" shall have the meaning provided in Section 5.12 hereof.

"Required Lenders" shall mean, without duplication, (a) the Lenders holding at least 50.1% of the aggregate outstanding principal amount of the Loans or (b) if no Loans are outstanding, the Lenders holding at least 50.1% of the Construction Loan Commitments; provided that the Construction Loan Commitments and the principal amount of the Loans of the Lenders that are Defaulting Lenders (if any) shall be excluded from the determination of Required Lenders.

25

"Reserve Requirement" shall mean, at any time, the maximum rate at which reserves (including any marginal, special, supplemental, or emergency reserves) are required to be maintained under regulations issued from time to time by the Board of Governors of the Federal Reserve System (or any successor) by member banks of the Federal Reserve System against "Eurocurrency liabilities" (as such term is used in Regulation D). Without limiting the effect of the foregoing, the Reserve Requirement shall reflect any other reserves required to be maintained by such member banks with respect to (i) any category of liabilities which includes deposits by reference to which the Adjusted Eurodollar Rate is to be determined or (ii) any category of extensions of credit or other assets which include Eurodollar Loans. The Adjusted Eurodollar Rate shall be adjusted automatically on and as of the effective date of any change in the Reserve Requirement.

"Restoration Requisition" shall have the meaning provided in Section 4.8(e)(i) of the Collateral Account Agreement.

"Restoration Work" shall mean the design, engineering, construction and other work with respect to the Restoration of Affected Property.

"Restore" shall mean, with respect to any Affected Property, to rebuild, repair, restore or replace such Affected Property. The term "Restoration" shall have a correlative meaning.

"Scheduled Principal Payments" shall mean the scheduled amounts payable in respect of principal of the Loans pursuant to Section 8.1 hereof.

"Secured Parties" shall mean, collectively, the Administrative Agent, the Collateral Agent, the Lenders, and each Secured Swap Counterparty.

"Secured Swap Counterparty" shall mean any party to a Required Hedging Agreement (other than a Finance Party) that, at the time such party enters into such Required Hedging Agreement is (i) an Acceptable Swap Counterparty and (ii) (a) a Lender or an Affiliate thereof or (b) a Person with whom either Borrower has entered into a Required Hedging Agreement provided or arranged by General Electric Capital Corporation or an Affiliate of General Electric Capital Corporation, and any assignee thereof

"Security Agreement" shall mean that certain Security Agreement, dated as of November 7, 2012, made by the Borrowers in favor of the Collateral Agent for the benefit of the Secured Parties.

"Security Documents" shall mean, collectively, the Collateral Account Agreement, the Security Agreement, the Pledge Agreement, each Mortgage, each Consent Agreement, each other agreement that creates or purports to create a Lien in favor of the Collateral Agent for the benefit of the Secured Parties to secure the Obligations, and all UCC financing statements and other filings, recordings or registrations required by any Finance Document to be filed or made in respect of any such Finance Document.

26

"Security Interest" shall mean the Lien on the Collateral or any other collateral purported to be granted to the Collateral Agent under any Security Document for the benefit of one or more of the Secured Parties (or any trustee, sub-agent or other Person acting for or on behalf of the Collateral Agent).

"Sites" shall mean Site A and/or Site B, as the context shall require.

"<u>Site A</u>" shall mean the site upon which Facility A will be constructed, which site shall have been approved by the Required Lenders in their sole discretion, together with any fixtures and civil works constructed thereon, and any and all Site Real Estate Rights relating thereto.

"Site B" shall mean the site upon which Facility B will be constructed, which site shall have been approved by the Required Lenders in their sole discretion, together with any fixtures and civil works constructed thereon, and any and all Site Real Estate Rights relating thereto.

"<u>Site Real Estate Rights</u>" shall mean, in respect of a Site, all easements, real property leases, surface use agreements, and other real property rights, privileges, and interests required for the installation, ownership, operation, use or maintenance of a Facility upon such Site.

"Solvent" shall mean, when used with respect to any Person as of any date of determination, (a) the amount of the "present fair saleable value" of the assets of such Person will, as of such date, exceed the amount of all "liabilities of such Person, contingent or otherwise", as of such date, as such quoted terms are determined in accordance with applicable federal and state Laws governing determinations of the insolvency of debtors, (b) the present fair saleable value of the assets of such Person will, as of such date, be greater than the amount that will be required to pay the liability of such Person on its debts as such debts become absolute and matured, (c) such Person will not have, as of such date, an unreasonably small amount of capital with which to conduct its business, and (d) such Person will be able to pay its debts as they mature. For purposes of this definition, (i) "debt" means liability on a "claim", and (ii) "claim" means any (x) right to payment, whether or not such a right is reduced to judgment, liquidated, unliquidated, fixed, contingent, matured, unmatured, disputed, undisputed, legal, equitable, secured or unsecured or (y) right to an equitable remedy for breach of performance if such breach gives rise to a right to payment, whether or not such right to an equitable remedy is reduced to judgment, fixed, contingent, matured or unmatured, disputed, undisputed, secured or unsecured.

"Specified Site RE Right Grantor" shall mean each owner, grantor or lessor of any Specified Site Real Estate Rights.

"<u>Specified Site Real Estate Right</u>" shall mean each Site Real Estate Right that is not by its terms assignable by the applicable Borrower to the Collateral Agent without the consent of its owner, grantor or lessor in a manner reasonably satisfactory to the Required Lenders.

"Sponsor" shall mean Clean Energy Fuels Corp., a Delaware corporation.

"Sponsor Guaranty" shall mean that certain Guaranty, dated as of November 7, 2012, made by the Sponsor in favor of the Secured Parties.

27

"Standard & Poor's" shall mean Standard & Poor's Ratings Group, a division of The McGraw Hill Corporation, or its successor.

"Statutory Reserves" shall mean the aggregate of the maximum reserve percentages (including any marginal, special, emergency or supplemental reserves) expressed as a decimal established by the Board of Governors of the Federal Reserve System of the United States and any other banking authority, domestic or foreign, to which the Administrative Agent or any Lender (including any branch, Affiliate, or other fronting office making or holding a Loan) is subject for Eurocurrency Liabilities (as defined in Regulation D). Eurodollar Loans shall be deemed to constitute Eurocurrency Liabilities (as defined in Regulation D) and to be subject to such Reserve Requirements without benefit of or credit for proration, exemptions or offsets that may be available from time to time to any Lender under such Regulation D. Statutory Reserves shall be adjusted automatically on and as of the effective date of any change in any reserve percentage.

"Subsidiary" shall mean, for any Person, any corporation, partnership, limited liability company or other entity of which at least a majority of the securities or other ownership interests having by the terms thereof ordinary voting power to elect a majority of the board of directors or other persons performing similar functions of such corporation, partnership, limited liability company or other entity (irrespective of whether or not at the time securities or other ownership interests of any other class or classes of such corporation, partnership or other entity shall have or might have voting power by reason of the happening of any contingency) is at the time directly or indirectly owned or Controlled by such Person or one or more Subsidiaries of such Person or by such Person and one or more Subsidiaries of such Person.

"<u>Taking</u>" shall mean any circumstance or event, or series of circumstances or events (including an Expropriation Event), in consequence of which a Project or any portion thereof shall be condemned, nationalized, seized, compulsorily acquired or otherwise expropriated by any Governmental Authority under power of eminent domain or otherwise.

"Tax Distributions" shall mean all amounts distributed to the Sponsor pursuant to Section 4.6(b) of the Collateral Account Agreement.

"<u>Taxes</u>" shall mean all present or future taxes, levies, imposts, duties, deductions, withholdings (including backup withholding), assessments, fees or other charges imposed by any Governmental Authority, including any interest, additions to tax or penalties applicable thereto.

"Term Loan Commitments" shall mean the Tranche A Term Loan Commitments and/or the Tranche B Term Loan Commitments, as the context shall require.

"Term Loan Principal Payment" shall mean a Tranche A Term Loan Principal Payment and/or a Tranche B Term Loan Principal Payment, as the context shall require.

"<u>Term Loan Principal Payment Date</u>" shall mean the Tranche A Term Loan Principal Payment Date and/or the Tranche B Term Loan Principal Payment Date, as the context shall require.

20

"Term Loans" shall mean Tranche A Term Loans and/or Tranche B Term Loans, as the context shall require.

"<u>Title Insurance</u>" shall having the meaning given to such term in <u>3.2(g)(iii)</u> hereof.

"<u>Title Insurance Company</u>" shall mean any Person from time to time appointed by the Required Lenders to act as a Title Insurance Company for the purposes of this Agreement.

"Total Construction Commitment" shall mean \$200,000,000.

"Total Funded Construction Loans" shall mean, at any time, the aggregate principal amount of Construction Loans funded as of such time.

"Tranche" shall mean Tranche A and/or Tranche B, as the context shall require.

"Tranche A Borrower" shall have the meaning provided in the Preamble hereto.

"Tranche B Borrower" shall have the meaning provided in the Preamble hereto.

"Tranche A Consolidated Total Debt" shall mean, at any time, the sum of (i) all outstanding Indebtedness of the Tranche A Borrower at such time plus (ii) all future rental obligations of the Tranche A Borrower under leases (other than Capital Lease Obligations) minus (iii) the principal amount of Tranche B Loans then outstanding.

"Tranche A Construction Account" shall have the meaning provided in the Collateral Account Agreement.

"Tranche A Construction Loan" shall have the meaning provided in Section 2.1(a) hereof.

"<u>Tranche A Construction Loan Availability Period</u>" shall mean the period commencing on the Initial Funding Date for Tranche A Construction Loans, and ending on the earliest to occur of (i) the full utilization of the Tranche A Construction Loan Commitments, (ii) the Facility A Date Certain, (iii) the Facility A Conversion Date and (iv) the termination of the Tranche A Construction Loan Commitment pursuant to the provisions of this Agreement.

"Tranche A Construction Loan Commitment" shall mean, as to any Tranche A Lender, the applicable amount set forth opposite such Tranche A Lender's name in Annex I hereto. The aggregate Tranche A Construction Loan Commitments of all Tranche A Lenders is \$100,000,000, as the same may be reduced pursuant to Section 2.13 hereof.

"<u>Tranche A Construction Loan Maturity Date</u>" shall mean the earliest to occur of (i) the Facility A Conversion Date, (ii) the Facility A Date Certain and (iii) the date on which any Tranche A Loans become due pursuant to <u>Section 9.2</u> hereof.

"Tranche A Construction Notes" shall have the meaning provided in Section 2.9(b) hereof.

29

"Tranche A Debt Service" shall mean, for any period with respect to the Tranche A Borrower, the sum, without duplication, of (i) all amounts payable by the Tranche A Borrower during such period pursuant to the terms and conditions of the Finance Documents (including all amounts that are required to be prepaid and all amounts overdue from any prior period) in respect of principal on the Tranche A Term Loans during such period, (ii) all amounts payable in respect of Tranche A Interest Expense for such period, and (iii) all scheduled Fees payable in respect of the Tranche A Loans and the Tranche A Construction Loan Commitment during such period.

"Tranche A Debt Service Coverage Ratio" shall mean, as at any date with respect to the Tranche A Borrower, for the period of four consecutive fiscal quarters of the Tranche A Borrower then most recently ended (or, if fewer, the number of full fiscal quarters that have elapsed since the applicable Conversion Date) for which financial statements have been delivered pursuant to Section 5.1(a) (or any other period referred to in this Agreement, in each case, taken as one accounting period), the ratio of (i) Cash Flow Available for Debt Service of the Tranche A Borrower for such period to (ii) Tranche A Debt Service for such period (including Scheduled Principal Payments, but excluding mandatory prepayments in respect of the Tranche A Term Loans payable during such period pursuant to the Finance Documents). Any determination of the Tranche A Debt Service Coverage Ratio shall be made by the Tranche A Borrower using methodology and assumptions satisfactory to the Administrative Agent.

"Tranche A Debt Service Reserve Account" shall have the meaning provided in the Collateral Account Agreement.

"Tranche A Distribution Account" shall have the meaning provided in the Collateral Account Agreement.

"Tranche A Excess Cash Flow" shall have the meaning provided in the Collateral Account Agreement.

"<u>Tranche A Final Maturity Date</u>" shall mean the earliest to occur of (i) the eighth (8th) anniversary of the Facility A Conversion Date and (ii) the date on which any Tranche A Loans become due pursuant to <u>Section 9.2</u> hereof.

"Tranche A Interest Expense" shall mean, for any period in respect of the Tranche A Borrower, the sum of the following: (i) all interest on the Tranche A Term Loans accrued or capitalized during such period (whether or not actually paid during such period) pursuant to the Finance Documents <u>plus</u> (ii) the net amounts payable (or minus the net amounts receivable) by the Tranche A Borrower under the Required Hedging Agreements that are Interest Rate Hedging Agreements accrued during such period (whether or not actually paid or received during such period).

"Tranche A Leverage Ratio" shall mean, as at any date for the Tranche A Borrower, the ratio of (i) Tranche A Consolidated Total Debt to (ii) EBITDA of the Tranche A Borrower for the period of four consecutive fiscal quarters of the Tranche A Borrower then most recently ended for which financial statements have been delivered pursuant to Section 5.1(a); provided, that if less than four full fiscal quarters shall have elapsed since the Facility A Conversion Date,

30

this clause (ii) shall be calculated as the product of (A) EBITDA of the Tranche A Borrower for each full fiscal quarter that has elapsed since the Facility A Conversion Date and (B) a fraction, the numerator of which is four (4) and the denominator of which is the number of full fiscal quarters that have elapsed since the Facility A Conversion Date. Any determination of the Tranche A Leverage Ratio shall be made by the Tranche A Borrower using methodology and assumptions satisfactory to the Administrative Agent.

"Tranche A Obligations" shall mean all Obligations relating to, or arising out of, the Tranche A Construction Loans or the Tranche A Term Loans.

"Tranche A Projected Debt Service Coverage Ratio" shall mean, as at any date with respect to the Tranche A Borrower, for the period of four consecutive fiscal quarters of the Borrowers commencing on the day after the most recent Quarterly Date prior to such date, the ratio of (i) projected Cash Flow Available for Debt Service of the Tranche A Borrower for such period to (ii) projected Tranche A Debt Service for such period (including Scheduled Principal Payments, but excluding mandatory prepayments in respect of the Tranche A Term Loans payable during such period pursuant to the Finance Documents). Any determination of the Tranche A Projected Debt Service Coverage Ratio shall be made by the Tranche A Borrower using methodology and assumptions satisfactory to the Administrative Agent.

"<u>Tranche A Required Debt Service Reserve Amount</u>" shall mean, on any date, an amount equal to the Tranche A Debt Service payable over the six-month period following such date, calculated based on the Assumed Interest Rate (including Scheduled Principal Payments, but excluding mandatory prepayments in respect of the Tranche A Term Loans payable during such period pursuant to the Finance Documents).

"Tranche A Term Loan Commitment" shall mean, for each Tranche A Lender, the aggregate principal amount of Tranche A Construction Loans of such Tranche A Lender outstanding as of the Facility A Conversion Date (after giving effect to any Borrowing of Tranche A Construction Loans on such date in accordance with Section 2.1 hereof and any prepayment of Tranche A Construction Loans on such date in accordance with Section 8 hereof); provided that the aggregate Tranche A Term Loan Commitment of all Tranche A Lenders shall not in any event exceed \$100,000,000, as the same may be reduced pursuant to Section 2.13 hereof.

"Tranche A Term Loan Principal Payment" shall have the meaning provided in Section 8.1(b) hereof.

"Tranche A Term Loan Principal Payment Date" shall have the meaning provided in Section 8.1(b) hereof.

"Tranche A Term Loans" shall have the meaning provided in Section 2.3(a) hereof.

"Tranche A Term Notes" shall have the meaning provided in Section 2.9(b) hereof.

31

"Tranche B Consolidated Total Debt" shall mean, at any time, the sum of (i) all outstanding Indebtedness of the Tranche B Borrower at such time plus (ii) all future rental obligations of the Tranche B Borrower under leases (other than Capital Lease Obligations) minus (iii) the principal amount of Tranche A Loans then outstanding.

"Tranche B Construction Account" shall have the meaning provided in the Collateral Account Agreement.

"Tranche B Construction Loan" shall have the meaning provided in Section 2.2(a) hereof.

"Tranche B Construction Loan Availability Period" shall mean the period commencing on the Initial Funding Date for Tranche B Construction Loans, and ending on the earliest to occur of (i) the full utilization of the Tranche B Construction Loan Commitments, (ii) the Facility B Date Certain, (iii) the Facility B Conversion Date and (iv) the termination of the Tranche B Construction Loan Commitment pursuant to the provisions of this Agreement.

"Tranche B Construction Loan Commitment" shall mean, as to any Tranche B Lender, the applicable amount set forth opposite such Tranche B Lender's name in <u>Annex I</u> hereto. The aggregate Tranche B Construction Loan Commitments of all Tranche B Lenders is \$100,000,000, as the same may be reduced pursuant to **Section 2.13** hereof.

"<u>Tranche B Construction Loan Maturity Date</u>" shall mean the earliest to occur of (i) the Facility B Conversion Date, (ii) the Facility B Date Certain and (iii) the date on which any Tranche B Loans become due pursuant to <u>Section 9.2</u> hereof.

"Tranche B Construction Notes" shall have the meaning provided in Section 2.9(b) hereof.

"Tranche B Debt Service" shall mean, for any period with respect to the Tranche B Borrower, the sum, without duplication, of (i) all amounts payable by the Tranche B Borrower during such period pursuant to the terms and conditions of the Finance Documents (including all amounts that are required to be prepaid and all amounts overdue from any prior period) in respect of principal on the Tranche B Term Loans during such period, (ii) all amounts payable in respect of Tranche B Interest Expense for such period, and (iii) all scheduled Fees payable by in respect of the Tranche B Loans and the Tranche B Construction Loan Commitment during such period.

"Tranche B Debt Service Coverage Ratio" shall mean, as at any date with respect to the Tranche B Borrower, for the period of four consecutive fiscal quarters of the Tranche B Borrower then most recently ended (or, if fewer, the number of full fiscal quarters that have elapsed since the applicable Conversion Date) for which financial statements have been delivered pursuant to Section 5.1(a) (or any other period referred to in this Agreement, in each case, taken as one accounting period), the ratio of (i) Cash Flow Available for Debt Service of the Tranche B Borrower for such period to (ii) Tranche B Debt Service for such period (including Scheduled Principal Payments, but excluding mandatory prepayments in respect of the Tranche B Term Loans payable during such period pursuant to the Finance Documents). Any determination of the Tranche B Debt Service Coverage Ratio shall be made by the Tranche B Borrower using methodology and assumptions satisfactory to the Administrative Agent.

32

"Tranche B Debt Service Reserve Account" shall have the meaning provided in the Collateral Account Agreement.

"Tranche B Distribution Account" shall have the meaning provided in the Collateral Account Agreement.

"Tranche B Excess Cash Flow" shall have the meaning provided in the Collateral Account Agreement.

"<u>Tranche B Final Maturity Date</u>" shall mean the earliest to occur of (i) the eighth (8th) anniversary of the Facility B Conversion Date and (ii) the date on which any Tranche B Loans become due pursuant to <u>Section 9.2</u> hereof.

"Tranche B Interest Expense" shall mean, for any period in respect of the Tranche B Borrower, the sum of the following: (i) all interest on the Tranche B Term Loans accrued or capitalized during such period (whether or not actually paid during such period) pursuant to the Finance Documents <u>plus</u> (ii) the net amounts payable (or minus the net amounts receivable) by the Tranche B Borrower under the Required Hedging Agreements that are Interest Rate Hedging Agreements accrued during such period (whether or not actually paid or received during such period).

"Tranche B Leverage Ratio" shall mean, as at any date for the Tranche B Borrower, the ratio of (i) Tranche B Consolidated Total Debt to (ii) EBITDA of the Tranche B Borrower for the period of four consecutive fiscal quarters of the Tranche B Borrower then most recently ended for which financial statements have been delivered pursuant to Section 5.1(a); provided, that if less than four full fiscal quarters shall have elapsed since the Facility B Conversion Date, this clause (ii) shall be calculated as the product of (A) EBITDA of the Tranche B Borrower for each full fiscal quarter that has elapsed since the Facility B Conversion Date and (B) a fraction, the numerator of which is four (4) and the denominator of which is the number of full fiscal quarters that have elapsed since the Facility B Conversion Date. Any determination of the Tranche B Leverage Ratio shall be made by the Tranche B Borrower using methodology and assumptions satisfactory to the Administrative Agent.

"Tranche B Loans" means Tranche B Construction Loans and/or Tranche B Term Loans, as the context shall require.

"Tranche B Obligations" shall mean all Obligations relating to, or arising out of, the Tranche B Construction Loans or the Tranche B Term Loans.

"Tranche B Projected Debt Service Coverage Ratio" shall mean, as at any date with respect to the Tranche B Borrower, for the period of four consecutive fiscal quarters of the Borrowers commencing on the day after the most recent Quarterly Date prior to such date, the ratio of (i) projected Cash Flow Available for Debt Service of the Tranche B Borrower for such period to (ii) projected Tranche B Debt Service for such period (including Scheduled Principal Payments, but excluding mandatory prepayments in respect of the Tranche B Term Loans payable during such period pursuant to the Finance Documents). Any determination of the Tranche B Projected Debt Service Coverage Ratio shall be made by the Tranche B Borrower using methodology and assumptions satisfactory to the Administrative Agent.

33

"<u>Tranche B Required Debt Service Reserve Amount</u>" shall mean, on any date, an amount equal to the Tranche B Debt Service payable over the six-month period following such date, calculated based on the Assumed Interest Rate (including Scheduled Principal Payments, but excluding mandatory prepayments in respect of the Tranche B Term Loans payable during such period pursuant to the Finance Documents).

"Tranche B Term Loan Commitment" shall mean, for each Tranche B Lender, the aggregate principal amount of Tranche B Construction Loans of such Tranche B Lender outstanding as of the Facility B Conversion Date (after giving effect to any Borrowing of Tranche B Construction Loans on such date in accordance with Section 2.2 hereof and any prepayment of Tranche B Construction Loans on such date in accordance with Section 8 hereof); provided that the aggregate Tranche B Term Loan Commitment of all Tranche B Lenders shall not in any event exceed \$100,000,000, as the same may be reduced pursuant to Section 2.13 hereof.

"Tranche B Term Loan Principal Payment" shall have the meaning provided in Section 8.1(c) hereof.

"Tranche B Term Loan Principal Payment Date" shall have the meaning provided in Section 8.1(c) hereof.

"Tranche B Term Loans" shall have the meaning provided in Section 2.4(a) hereof.

"Tranche B Term Notes" shall have the meaning provided in Section 2.9(b) hereof.

"Transaction Documents" shall mean, collectively, the Warrant, the Warrant Agreement, the Registration Rights Agreement, the Project Documents and the Finance Documents.

"Transfer Date Certificate" shall mean a "Transfer Date Certificate" as defined in and delivered under the Collateral Account Agreement.

"<u>Type</u>" shall mean the type of Loan determined with regard to the interest option applicable thereto, *i.e.*, whether a Base Rate Loan or a Eurodollar Loan.

"UCC" means the Uniform Commercial Code as in effect from time to time in any relevant jurisdiction.

"United States" and "U.S." shall each mean the United States of America.

"<u>Unutilized Commitment Coefficient</u>" shall mean the lesser of (i) a fraction, the numerator of which shall be the excess of the Total Construction Commitment over the amount of Total Funded Construction Loans and the denominator of which shall be the sum of the Unutilized Tranche A Construction Loan Commitments and the Unutilized Tranche B Construction Loan Commitments and (ii) one (1).

34

"<u>Unutilized Tranche A Construction Loan Commitment</u>" shall mean, for each Lender, at any time, the greater of (i) the Tranche A Construction Loan Commitment of such Lender at such time <u>less</u> the aggregate principal amount at such time of all Tranche A Construction Loans made by such Lender and (ii) zero (0).

"<u>Unutilized Tranche B Construction Loan Commitment</u>" shall mean, for each Lender, at any time, the greater of, (i) the Tranche B Construction Loan Commitment of such Lender at such time <u>less</u> the aggregate principal amount at such time of all Tranche B Construction Loans made by such Lender and (ii) zero (0).

- "U.S. Person" shall mean any Person that is a "United States Person" as defined in Section 7701(a)(30) of the Code.
- "U.S. Tax Compliance Certificate" shall have the meaning assigned to such term in paragraph (f) of Section 2.14.
- "Warrant" shall mean that certain Warrant, dated as of November 7, 2012, issued by the Sponsor and GE Energy Financial Services, Inc.
- "<u>Warrant Agreement</u>" shall mean that certain Warrant Agreement, dated as of November 7, 2012, between the Sponsor and GE Energy Financial Services, Inc.
  - "Withholding Agent" shall mean each of the Borrowers and the Administrative Agent.
- "<u>Work</u>" shall mean all design, fabrication, construction, erection and other services to be performed, and all items to be procured, fabricated, constructed, manufactured, completed and warranted, by a Contractor pursuant to a Construction Agreement.
  - 1.2 <u>Rules of Interpretation</u>. In each Finance Document (other than any Required Hedging Agreement), unless otherwise indicated:
  - (1) each reference to, and the definition of, any document (including any Transaction Document) shall be deemed to refer to such document as it may be amended, amended and restated, supplemented, revised or otherwise modified from time to time in accordance with its terms and, to the extent applicable, the terms of this Agreement;
  - (2) each reference to a Law or Governmental Approval shall be deemed to refer to such Law or Governmental Approval as the same may be amended, supplemented or otherwise modified from time to time;
  - (3) any reference to a Person in any capacity includes a reference to its permitted successors and assigns in such capacity and, in the case of any Governmental Authority, any Person succeeding to any of its functions and capacities;

35

- (4) references to days shall refer to calendar days unless Business Days are specified; references to weeks, months or years shall be to calendar weeks, months or years, respectively, unless otherwise specified;
- (5) all references to a "Section," "Appendix," "Annex," "Schedule" or "Exhibit" of a Finance Document are to a Section of such Finance Document or to an Appendix, Annex, Schedule or Exhibit attached to such Finance Document;
- (6) the table of contents and Section headings and other captions therein are for the purpose of reference only and do not affect the interpretation of such Finance Document;
- (7) defined terms in the singular shall include the plural and vice versa, and the masculine, feminine or neuter gender shall include all genders;
- (8) the words "hereof', "herein" and "hereunder", and words of similar import, when used in any Finance Document, shall refer to such Finance Document as a whole and not to any particular provision of such Finance Document;
  - (9) the words "include," "includes" and "including" are deemed to be followed by the phrase "without limitation";
- (10) where the terms of any Finance Document require that the approval, opinion, consent or other input of any Secured Party be obtained, such requirement shall be deemed satisfied only where the requisite approval, opinion, consent or other input is given by or on behalf of the relevant Secured Party in writing; and
- (11) where the terms of any Finance Document require or permit any action to be taken by any Agent, such action shall be taken strictly in accordance with the applicable provisions of the relevant Finance Documents.
- 1.3 Accounting Principles. Except as otherwise provided in this Agreement, all computations and determinations as to financial matters, and all financial statements to be delivered under this Agreement shall be made or prepared in accordance with GAAP (including principles of consolidation where appropriate) applied on a consistent basis (except to the extent approved or required by the independent public accountants certifying such statements and disclosed therein); provided that, if the Borrower notifies the Administrative Agent that the Borrower requests an amendment to any provision hereof to eliminate the effect of any change occurring after the date hereof in GAAP or in the application thereof on the operation of such provision (or if the Administrative Agent notifies the Borrowers that the Required Lenders request an amendment to any provision hereof for such purpose), regardless of whether any such notice is given before or after such change in GAAP or in the application thereof, then such provision shall be interpreted on the basis of GAAP as in effect and applied immediately before such change shall have become effective until such notice shall have been withdrawn or such provision amended in accordance herewith. Notwithstanding the foregoing, no change in the accounting principles used in the preparation of any financial statement hereafter adopted by the Borrower shall be given effect if such change would affect a calculation that measures compliance with any provision of this Agreement or any other Finance Document unless the

36

Borrower, the Administrative Agent and the Required Lenders agree to modify such provisions to reflect such changes in GAAP and, unless such provisions are modified, all financial statements, officer's certificates and similar documents provided hereunder shall be provided together with a reconciliation between the calculations and amounts set forth therein before and after giving effect to such change in GAAP.

#### 2.1 <u>Tranche A Construction Loan Facility.</u>

- (a) Subject to and upon the terms and conditions set forth herein, each Tranche A Lender severally agrees to make, from time to time during the Tranche A Construction Loan Availability Period, Tranche A Loans (each a "<u>Tranche A Construction Loan</u>" and, collectively, the "<u>Tranche A Construction Loans</u>") to the Tranche A Borrower, which Tranche A Loans shall: (i) at the option of the Tranche A Borrower, be Base Rate Loans or Eurodollar Loans (*provided*, *however*, that all Tranche A Construction Loans comprising the same Borrowing shall at all times be of the same Type); (ii) be made and maintained in Dollars; (iii) not exceed for any Tranche A Lender, in aggregate principal amount at any time outstanding, the Tranche A Construction Loan Commitment of such Tranche A Lender; (iv) after giving effect to the funding thereof, and after giving effect to the funding of any Tranche B Construction Loans requested pursuant to <u>Section 2.2(a)</u>, do not cause the Total Funded Construction Loans to exceed to Total Construction Commitment; and (v) mature on the Tranche A Construction Loan Maturity Date.
- (b) The Tranche A Construction Loans are available only on the terms and conditions specified hereunder, and once repaid, in whole or in part, at maturity or by prepayment, may not be reborrowed in whole or in part.

#### 2.2 <u>Tranche B Construction Loan Facility.</u>

- (a) Subject to and upon the terms and conditions set forth herein, each Tranche B Lender severally agrees to make, from time to time during the Tranche B Construction Loan Availability Period, Tranche B Loans (each a "<u>Tranche B Construction Loan</u>" and, collectively, the "<u>Tranche B Construction Loans</u>") to the Tranche B Borrower, which Tranche B Loans shall: (i) at the option of the Tranche B Borrower, be Base Rate Loans or Eurodollar Loans (*provided*, *however*, that all Tranche B Construction Loans comprising the same Borrowing shall at all times be of the same Type); (ii) be made and maintained in Dollars; (iii) not exceed for any Tranche B Lender, in aggregate principal amount at any time outstanding, the Tranche B Construction Loan Commitment of such Tranche B Lender; (iv) after giving effect to the funding thereof, and after giving effect to the funding of any Tranche A Construction Loans requested pursuant to <u>Section 2.1(a)</u>, do not cause the Total Funded Construction Loans to exceed the Total Construction Commitment; and (v) mature on the Tranche B Construction Loan Maturity Date.
- (b) The Tranche B Construction Loans are available only on the terms and conditions specified hereunder, and once repaid, in whole or in part, at maturity or by prepayment, may not be reborrowed in whole or in part.

37

#### 2.3 <u>Tranche A Term Loan Facility</u>.

- (a) Subject to and upon the terms and conditions set forth herein, each Tranche A Lender agrees that on the Facility A Conversion Date all Tranche A Construction Loans of such Tranche A Lender outstanding on such date (after giving effect to any Borrowing of Tranche A Construction Loans on such date and any prepayment of Tranche A Construction Loans on such date in accordance herewith) shall automatically convert into term loans (each a "<u>Tranche A Term Loan</u>" and, collectively, the "<u>Tranche A Term Loans</u>") in an aggregate principal amount not exceeding such Tranche A Lender's Tranche A Term Loan Commitment in effect as of such date. Such Tranche A Term Loans shall: (i) at the option of the Borrower, be Base Rate Loans or Eurodollar Loans; (ii) be made and maintained in Dollars; (iii) not exceed for any Tranche A Lender, in aggregate principal amount at any time outstanding, the Tranche A Term Loan Commitment of such Tranche A Lender; and (iv) mature on the Tranche A Final Maturity Date.
- (b) Tranche A Construction Loans that are converted into Tranche A Term Loans shall not be deemed to be prepaid, repaid or discharged but shall be deemed to be continued as Tranche A Term Loans as provided hereby.
- (c) The Tranche A Term Loans are available only on the terms and conditions specified hereunder, and once repaid, in whole or in part, at maturity or by prepayment, may not be reborrowed in whole or in part.

#### 2.4 <u>Tranche B Term Loan Facility</u>.

- (a) Subject to and upon the terms and conditions set forth herein, each Tranche B Lender agrees that on the Facility B Conversion Date all Tranche B Construction Loans of such Tranche B Lender outstanding on such date (after giving effect to any Borrowing of Tranche B Construction Loans on such date and any prepayment of Tranche B Construction Loans on such date in accordance herewith) shall automatically convert into term loans (each a "Tranche B Term Loan" and, collectively, the "Tranche B Term Loans") in an aggregate principal amount not exceeding such Tranche B Lender's Tranche B Term Loan Commitment in effect as of such date. Such Tranche B Term Loans shall: (i) at the option of the Borrower, be Base Rate Loans or Eurodollar Loans; (ii) be made and maintained in Dollars; (iii) not exceed for any Tranche B Lender, in aggregate principal amount at any time outstanding, the Tranche B Term Loan Commitment of such Tranche B Lender; and (iv) mature on the Tranche B Final Maturity Date.
- (b) Tranche B Construction Loans that are converted into Tranche B Term Loans shall not be deemed to be prepaid, repaid or discharged but shall be deemed to be continued as Tranche B Term Loans as provided hereby.
- (c) The Tranche B Term Loans are available only on the terms and conditions specified hereunder, and once repaid, in whole or in part, at maturity or by prepayment, may not be reborrowed in whole or in part.
- 2.5 <u>Notice of Borrowing</u>. Whenever a Borrower desires to make a Borrowing pursuant to <u>Section 2.1</u> or <u>2.2</u> hereof, it shall give the Administrative Agent at its Notice Office

38

Borrowing (which shall be a Business Day) and (iii) whether the Construction Loans being made pursuant to such Borrowing are to be initially maintained as Base Rate Loans or Eurodollar Loans and, if Eurodollar Loans, the initial Interest Period to be applicable thereto. The Administrative Agent shall promptly give each applicable Lender notice of the proposed Borrowing, of such Lender's proportionate share thereof and of the other matters required by the immediately preceding sentence to be specified in the Notice of Borrowing.

2.6 <u>Pro Rata Borrowings; Availability</u>. Each Borrowing of Loans shall be incurred contemporaneously and ratably among the applicable Lenders based upon the amount of their respective Construction Loan Commitments of the applicable Tranche. It is agreed that no Lender shall be responsible for any default by any other Lender of its obligation to make a Loan hereunder and that each Lender shall be obligated to make the Loans provided to be made by it hereunder regardless of the failure of any other Lender to make a Loan hereunder.

#### 2.7 <u>Minimum Amount and Maximum Number of Borrowings, etc.</u>

- (a) Except in the case of Borrowings of Construction Loans incurred solely to pay any of the Obligations, the aggregate principal amount of each Borrowing of Construction Loans or a particular Tranche shall not be less than (i) \$1,000,000 or (ii) if less than \$1,000,000, the remaining aggregate Construction Loan Commitments of all Lenders in respect of such Tranche.
- (b) Except in the case of Borrowings of Construction Loans incurred solely to pay any of the Obligations, each Borrower shall be limited to a maximum of one (1) Borrowing of Construction Loans per month.
  - (c) At no time shall there be outstanding more than six (6) separate Interest Periods in respect of Eurodollar Loans of any Tranche.
- 2.8 <u>Disbursement of Funds</u>. Subject to the terms and conditions hereof, no later than 11:00 a.m. (New York City time) on the date specified in each Notice of Borrowing, each Lender will make available, through such Lender's Applicable Lending Office, its *pro rata* portion of the aggregate principal amount of the Loans requested to be made on such date, in Dollars and in immediately available funds at the payment office of the Administrative Agent (as notified to the Lenders from time to time by the Administrative Agent), and the Administrative Agent will deposit the aggregate of the amounts so made available by the Lenders into the Tranche A Construction Account (in the case of Tranche B Construction Loans). Unless the Administrative Agent shall have been notified by any Lender prior to the applicable date of the Borrowing that such Lender does not intend to make available to the Administrative Agent such Lender's

39

portion of the Borrowing on such date, the Administrative Agent may assume that such Lender has made such amount available to the Administrative Agent on such date, and the Administrative Agent may (but shall have no obligation to), in reliance upon such assumption, make available to the applicable Borrower a corresponding amount. If such corresponding amount is not in fact made available to the Administrative Agent by such Lender, the Administrative Agent shall be entitled to recover such corresponding amount from such Lender on demand. If such Lender does not pay such corresponding amount forthwith upon the Administrative Agent's demand therefor, the Administrative Agent shall promptly notify the applicable Borrower, and such Borrower shall immediately pay such corresponding amount to the Administrative Agent. The Administrative Agent shall also be entitled to recover on demand from such Lender or such Borrower, as the case may be, interest on such corresponding amount in respect of each day from the date such corresponding amount was made available by the Administrative Agent to such Borrower until the date such corresponding amount is recovered by the Administrative Agent, at a rate *per annum* equal to (i) if such amount is recovered from such Lender, the cost to the Administrative Agent of acquiring overnight federal funds at the then applicable rate and (ii) if such amount is recovered from a Borrower, the then applicable rate of interest as provided herein. Nothing in this Section 2.8 shall be deemed to relieve any Lender from its obligation to make a Loan hereunder or to prejudice any rights which a Borrower may have against any Lender as a result of any failure by such Lender to make Loans hereunder. Notwithstanding anything to the contrary herein, any amounts repaid by the Borrowers pursuant to this Section 2.8 shall be available for reborrowing as if such amounts had never been disbursed.

#### 2.9 <u>Evidence of Obligations and Notes.</u>

- (a) Each Lender will maintain in accordance with its usual practice an account or accounts evidencing the indebtedness of the Borrowers to such Lender as a result of the Loans of such Lender, including the amounts of principal, interest and other amounts payable and paid to such Lender from time to time under this Agreement and, subject to Section 2.9(b) hereof, the Notes. The entries made by each Lender pursuant to the foregoing sentence shall constitute prima facie evidence of the existence and amounts of the Loans and other Obligations therein recorded; provided, however, that the failure of any Lender to maintain such account or accounts, or any error therein, shall not in any manner affect the obligations of the Borrowers to repay or pay the Loans made by such Lender, accrued interest thereon and the other Obligations of the Borrowers to such Lender hereunder in accordance with the terms of this Agreement. Each Lender will advise each Borrower of the outstanding indebtedness hereunder to such Lender upon written request therefor. If any amounts paid by a Borrower pursuant to this Agreement and the Notes, if any, issued pursuant hereto are insufficient to repay or pay the principal amount of all outstanding Loans owed by such Borrower, all accrued and unpaid interest thereon and all other Obligations, such Borrower shall be responsible for the deficiency.
- (b) Each Borrower agrees that, upon request to the Administrative Agent by any Lender, such Borrower will execute and deliver to such Lender (i) in the case of Tranche A Construction Loans or Tranche B Construction Loans, a promissory note duly executed and delivered by each Borrower substantially in the form of Exhibit B-1 hereto with blanks appropriately completed in conformity herewith (each, a "Tranche A Construction Note") or a "Tranche B Construction Note", as applicable), and (ii) in the case of Tranche A Term Loans

40

or Tranche B Term Loans, a promissory note duly executed and delivered by each Borrower substantially in the form of <u>Exhibit B-2</u> hereto with blanks appropriately completed in conformity herewith (each, a "<u>Tranche A Term Note</u>" or a "<u>Tranche B Term Note</u>", as applicable).

(c) Each Borrower hereby designates the Administrative Agent to serve as such Borrower's agent, solely for the purpose of this clause, to maintain a register (the "**Register**") on which the Administrative Agent will record each Lender's Commitments, the Loans made by each Lender, the principal amount (and stated interest) with respect to each Loan, and each repayment in respect of the principal amount of the Loans, annexed to which the Administrative Agent shall retain a copy of each Assignment and Acceptance delivered to the Administrative Agent pursuant to **Section 11.11** hereof.

Failure to make any recordation, or any error in such recordation, shall not affect the Obligations. The entries in the Register shall be rebuttably presumptive evidence of the accuracy of the information so set forth, and the Borrowers, the Administrative Agent and the Lenders shall treat each Person in whose name a Loan is registered as the owner thereof for the purposes of all Finance Documents, notwithstanding notice or any provision herein to the contrary. Any assignment or transfer of a Commitment or the Loans made pursuant hereto shall be registered in the Register only upon delivery to the Administrative Agent of an Assignment and Acceptance that has been executed by the requisite parties pursuant to Section 11.11 and compliance with the other provisions set forth in Section 11.11. No assignment or transfer of a Lender's Commitments or Loans shall be effective unless such assignment or transfer shall have been recorded in the Register by the Administrative Agent as provided in this Section 2.9(c) and compliance with the other provisions set forth in Section 11.11.

- (d) Each Lender will note on its internal records the amount of each Loan made by it and each payment in respect thereof and the outstanding principal amount of Loans evidenced thereby. Failure to make such notation shall not affect the Borrowers' obligations in respect of such Loans.
- Conversions. The Borrowers shall have the option to convert on any Business Day the principal amount of the Loans made pursuant to one or more Borrowings from one Type of Loan into a Borrowing of another Type of Loan; provided, however, that (i) Loans of a particular Tranche may not be so converted to another Type unless the aggregate principal amount of such Loans to be so converted equals \$1,000,000 or an integral multiple of \$500,000 in excess thereof, (ii) except as provided in Sections 2.12, 2.15 and 2.18, no conversion of all or any portion of any Eurodollar Loan into a Base Rate Loan may be effected on any day other than the last day of an Interest Period applicable to such Eurodollar Loan, (iii) no partial conversion of Eurodollar Loans shall reduce the outstanding principal amount of such Eurodollar Loans made pursuant to a single Borrowing to be less than \$1,000,000, (iv) Base Rate Loans may only be converted into Eurodollar Loans if no Default or Event of Default has occurred and is continuing on the date of conversion or would result from such conversion, and (v) no conversion pursuant to this Section 2.10 shall result in a greater number of Interest Periods than is permitted under Section 2.7 hereof. Each such conversion shall be effected by the Borrower by giving the Administrative Agent at its Notice Office prior to 11:00 a.m. (New York City time) at least three (3) Business Days' prior notice substantially in the form attached hereto as Exhibit G (each a "Notice of Conversion or Continuation") specifying (w) the principal amount of the

41

Loans to be so converted, (x) the Borrowing or Borrowings pursuant to which such Loans were made, (y) the Type of Loan from which such amount is being converted and the Type of Loan into which such amount will be converted, and (z) if such amount is to be converted into Eurodollar Loans, the Interest Period to be initially applicable thereto or, if such amount is to be continued as a Eurodollar Loan, the portion thereof to be continued and the Interest Period applicable thereto. The Administrative Agent shall give each Lender prompt notice of any such proposed conversion affecting any of its Loans.

#### 2.11 <u>Interest</u>.

- (a) The Borrowers agree to pay interest in respect of the unpaid principal amount of each Base Rate Loan at a rate *per annum* which shall be equal to the sum of (x) the Base Rate in effect from time to time and (y) the Applicable Margin.
- (b) The Borrowers agree to pay interest in respect of the unpaid principal amount of each Eurodollar Loan at a rate *per annum* which shall, during each Interest Period applicable thereto, be equal to the sum of (x) the Adjusted Eurodollar Rate in effect for such Interest Period and (y) the Applicable Margin.
- (c) From and after the occurrence of an Event of Default, all Obligations payable by the Borrowers hereunder or under any other Finance Document (other than any Obligations pursuant to a Required Hedging Agreement) shall bear interest at a rate (the "<u>Default Rate</u>") equal to (x) in the case of Loans, the sum of (i) the interest rate otherwise applicable hereunder and (ii) 2.00% and (y) in the case of any other Obligations, the sum of (i) the Base Rate in effect from time to time, (ii) the Applicable Margin and (iii) 2.00%, in each case with such interest to be payable on demand.
- (d) Accrued (and theretofore unpaid) interest shall be payable (i) in respect of each Base Rate Loan, quarterly in arrears on each Quarterly Date, (ii) in respect of Eurodollar Loans, on the last day of each Interest Period applicable thereto; provided, however, that if any Interest Period for a Eurodollar Loan exceeds three months, interest on such Eurodollar Loan shall be payable every three months after the beginning of such Interest Period, and (iii) in respect of each Loan, on any repayment or prepayment (on the amount repaid or prepaid), conversion (on the amount converted), at maturity (whether by acceleration or otherwise) and, after such maturity, on demand. Notwithstanding the foregoing, interest payable in accordance with Section 2.11(c) shall be payable as provided therein.
- (e) On each Interest Determination Date in respect of any Eurodollar Loan, the Administrative Agent shall determine the Adjusted Eurodollar Rate for the applicable Interest Period to be applicable to such Eurodollar Loans or to any portion thereof and shall promptly notify the Borrower and the Lenders thereof. Each such determination shall, absent manifest error, be final and conclusive and binding on all parties hereto.
- 2.12 <u>Interest Periods.</u> At the time it gives any Notice of Borrowing or Notice of Conversion or Continuation in respect of the making or continuation of, or conversion into, any Eurodollar Loan (in the case of the initial Interest Period applicable thereto) or on the third Business Day prior to the expiration of any Interest Period applicable to any Eurodollar Loan (in

42

the case of any subsequent Interest Period), the applicable Borrower shall have the right to elect, by giving the Administrative Agent written notice thereof, the interest period (each, an "Interest Period") applicable to such Eurodollar Loans, which Interest Period shall, at the option of such Borrower, be a one, two, three or six month period; *provided*, *however*, that:

- (i) all Eurodollar Loans comprising the same Borrowing shall have the same Interest Period;
- (ii) the initial Interest Period for any Eurodollar Loan shall commence on the date of Borrowing of such Loan (including the date of conversion thereof from a Loan of a different Type) and each Interest Period occurring thereafter in respect of such Eurodollar Loan shall commence on the last day of the immediately preceding Interest Period;

- (iii) if any Interest Period begins on a day for which there is no numerically corresponding day in the month at the end of such Interest Period, such Interest Period shall end on the last Business Day of such month;
- (iv) if any Interest Period would otherwise end on a day that is not a Business Day, such Interest Period shall be extended to the next succeeding Business Day unless the result of such extension would be to carry such Interest Period into another month, in which event such Interest Period shall end on the immediately preceding Business Day;
- (v) no Interest Period shall extend beyond the Tranche A Final Maturity Date or Tranche B Final Maturity Date, as applicable;
- (vi) no Interest Period with respect to a Term Loan may be elected that would extend beyond any applicable Term Loan Principal Payment Date unless the aggregate principal amount of Term Loans of the same Tranche outstanding having Interest Periods which end on or before such date shall be at least equal to or greater than the principal due in respect of such Term Loan on such date; and
  - (vii) no Interest Period in respect of any Loans shall be selected at any time when a Default or Event of Default exists.

If upon the expiration of any Interest Period, the Borrowers have failed to elect, or are not permitted to elect, a new Interest Period to be applicable to such Eurodollar Loans as provided above, the Borrowers shall be deemed to have elected to convert such Eurodollar Loans into Base Rate Loans effective as of the expiration date of such current Interest Period.

#### 2.13 Termination and Reduction of Commitments.

(a) Unless otherwise previously terminated, the Tranche A Construction Loan Commitments shall terminate on the Tranche A Construction Loan Maturity Date; *provided* that if such date is also the Facility A Conversion Date, the Tranche A Term Loan Commitments shall terminate only upon and after giving effect to the conversion of the Tranche A Construction Loans into Tranche A Term Loans on such date.

43

- (b) Unless otherwise previously terminated, the Tranche B Construction Loan Commitments shall terminate on the Tranche B Construction Loan Maturity Date; *provided* that if such date is also the Facility B Conversion Date, the Tranche B Term Loan Commitments shall terminate only upon and after giving effect to the conversion of the Tranche B Construction Loans into Tranche B Term Loans on such date.
- (c) If the Initial Funding Date for Tranche A Construction Loans does not occur prior to December 31, 2014, then the Tranche A Construction Loan Commitments shall be terminated in their entirety on such date.
- (d) If the Initial Funding Date for Tranche B Constructions Loans does not occur prior to December 31, 2014, then the Tranche B Construction Loan Commitments shall be terminated in their entirety on such date.
- (e) The Borrowers may at any time terminate, or from time to time reduce, any of the Commitments within a Tranche; *provided* that (i) each reduction of such Commitments shall be in an amount that is an integral multiple of \$5,000,000 (or, if less, such remaining Commitments), (ii) in the case of termination or reduction of Construction Loan Commitments, the applicable Borrower shall have demonstrated to the satisfaction of the Required Lenders the ability to pay, by means satisfactory to the Required Lenders, all remaining Facility A Project Costs necessary to cause the Facility A Project Completion Date to occur no later than the Facility B Date Certain and all remaining Facility B Project Costs necessary to cause the Facility B Project Completion Date to occur no later than the Facility B Date Certain, *provided* that, notwithstanding the foregoing, the Borrowers may terminate the Construction Loan Commitments in their entirety at any time if done so in connection the with full repayment of the outstanding Obligations and (iii) any reduction in the Construction Loan Commitments pursuant to this Section 2.13(e) shall automatically result in a corresponding reduction of the Term Loan Commitments of the same Tranche.
- (f) The Borrowers shall notify the Administrative Agent of any election to terminate or reduce the Commitments under clause (c) above at least ten (10) Business Days prior to the effective date of such termination or reduction, specifying such election and the effective date thereof. Promptly following receipt of any such notice from the Borrowers, the Administrative Agent shall advise the applicable Lenders of the contents thereof. Each notice delivered by the Borrowers to the Administrative Agent pursuant to this **Section 2.13** shall be irrevocable.

#### 2.14 Taxes.

(a) Payments Free of Taxes. Any and all payments by or on account of any obligation of a Finance Party under any Finance Document shall be made without deduction or withholding for any Taxes, except as required by applicable law. If any applicable law (as determined in the good faith discretion of an applicable Withholding Agent) requires the deduction or withholding of any Tax from any such payment by a Withholding Agent, then the applicable Withholding Agent shall make such deduction or withholding and shall timely pay the full amount deducted or withheld to the relevant Governmental Authority in accordance with applicable law and, if such Tax is an Indemnified Tax, then the sum payable by the applicable

44

Finance Party shall be increased as necessary so that after such deduction or withholding has been made (including such deductions and withholdings applicable to additional sums payable under this Section) the applicable Recipient receives an amount equal to the sum it would have received had no such deduction or withholding been made.

- (b) <u>Payment of Other Taxes by the Borrowers</u>. The Borrowers shall timely pay to the relevant Governmental Authority in accordance with applicable law any Other Taxes and shall timely reimburse the Administrative Agent for any Other Taxes paid by the Administrative Agent.
- (c) <u>Indemnification by the Finance Parties</u>. The Finance Parties shall jointly and severally indemnify each Recipient, within 10 days after written demand therefor, for the full amount of any Indemnified Taxes (including Indemnified Taxes imposed or asserted on or attributable to amounts

payable under this Section) payable or paid by such Recipient or required to be withheld or deducted from a payment to such Recipient and any reasonable expenses arising therefrom or with respect thereto, whether or not such Indemnified Taxes were correctly or legally imposed or asserted by the relevant Governmental Authority. A certificate as to the amount of such payment or liability delivered to the Borrower by a Lender (with a copy to the Administrative Agent), or by the Administrative Agent on its own behalf or on behalf of a Lender, shall be conclusive absent manifest error.

- (d) <u>Indemnification by the Lenders</u>. Each Lender shall severally indemnify the Administrative Agent, within 10 days after demand therefor, for (i) any Indemnified Taxes attributable to such Lender (but only to the extent that a Finance Party has not already indemnified the Administrative Agent for such Indemnified Taxes and without limiting the obligation of the Finance Parties to do so), (ii) any Taxes attributable to such Lender's failure to comply with the provisions of <u>Section 11.11(f)</u> relating to the maintenance of a Participant Register and (iii) any Excluded Taxes attributable to such Lender, in each case, that are payable or paid by the Administrative Agent in connection with any Finance Document, and any reasonable expenses arising therefrom or with respect thereto, whether or not such Taxes were correctly or legally imposed or asserted by the relevant Governmental Authority. A certificate as to the amount of such payment or liability delivered to any Lender by the Administrative Agent shall be conclusive absent manifest error. Each Lender hereby authorizes the Administrative Agent to set off and apply any and all amounts at any time owing to such Lender under any Finance Document or otherwise payable by the Administrative Agent to the Lender from any other source against any amount due to the Administrative Agent under this paragraph (d).
- (e) <u>Evidence of Payments</u>. As soon as practicable after any payment of Taxes by a Finance Party to a Governmental Authority pursuant to this <u>Section 2.14</u>, the Finance Party shall deliver to the Administrative Agent the original or a certified copy of a receipt issued by such Governmental Authority evidencing such payment, a copy of the return reporting such payment or other evidence of such payment reasonably satisfactory to the Administrative Agent.
- (f) <u>Status of Lenders</u>. (i) Any Lender that is entitled to an exemption from or reduction of withholding Tax with respect to payments made under any Finance Document shall deliver to the applicable Borrower and the Administrative Agent, at the time or times reasonably requested by such Borrower or the Administrative Agent, such properly completed and executed

45

documentation reasonably requested by such Borrower or the Administrative Agent as will permit such payments to be made without withholding or at a reduced rate of withholding. In addition, any Lender, if reasonably requested by a Borrower or the Administrative Agent, shall deliver such other documentation prescribed by applicable law or reasonably requested by such Borrower or the Administrative Agent as will enable such Borrower or the Administrative Agent to determine whether or not such Lender is subject to backup withholding or information reporting requirements. Notwithstanding anything to the contrary in the preceding two sentences, the completion, execution and submission of such documentation (other than such documentation set forth in Section (ii)(A), (ii)(B) and (ii)(D) below) shall not be required if in the Lender's reasonable judgment such completion, execution or submission would subject such Lender to any material unreimbursed cost or expense or would materially prejudice the legal or commercial position of such Lender.

- (ii) Without limiting the generality of the foregoing,
- (A) any Lender that is a U.S. Person shall deliver to the applicable Borrower and the Administrative Agent on or prior to the date on which such Lender becomes a Lender under this Agreement (and from time to time thereafter upon the reasonable request of such Borrower or the Administrative Agent), executed originals of IRS Form W-9 certifying that such Lender is exempt from U.S. federal backup withholding tax;
- (B) any Foreign Lender shall, to the extent it is legally entitled to do so, deliver to the applicable Borrower and the Administrative Agent (in such number of copies as shall be requested by the recipient) on or prior to the date on which such Foreign Lender becomes a Lender under this Agreement (and from time to time thereafter upon the reasonable request of such Borrower or the Administrative Agent), whichever of the following is applicable:
  - (i) in the case of a Foreign Lender claiming the benefits of an income tax treaty to which the United States is a party (x) with respect to payments of interest under any Finance Document, executed originals of IRS Form W-8BEN establishing an exemption from, or reduction of, U.S. federal withholding Tax pursuant to the "interest" article of such tax treaty and (y) with respect to any other applicable payments under any Finance Document, IRS Form W-8BEN establishing an exemption from, or reduction of, U.S. federal withholding Tax pursuant to the "business profits" or "other income" article of such tax treaty;
    - (ii) executed originals of IRS Form W-8ECI;
  - (iii) in the case of a Foreign Lender claiming the benefits of the exemption for portfolio interest under Section 881(c) of the Code, (x) a certificate substantially in the form of **Exhibit I-1** to the effect that such Foreign Lender is not a "bank" within the meaning of Section 881(c)(3) (A) of the Code, a "10 percent shareholder" of the Borrower within the meaning of Section 881(c)(3)(B) of the Code, or a "controlled foreign corporation" described in Section 881(c)(3)(C) of the Code (a "U.S. Tax Compliance Certificate") and (y) executed originals of IRS Form W-8BEN; or

46

- (iv) to the extent a Foreign Lender is not the beneficial owner, executed originals of IRS Form W-8IMY, accompanied by IRS Form W-8ECI, IRS Form W-8BEN, a U.S. Tax Compliance Certificate substantially in the form of **Exhibit I-2** or **Exhibit I-3**, as applicable, IRS Form W-9, and/or other certification documents from each beneficial owner, as applicable; provided that if the Foreign Lender is a partnership and one or more direct or indirect partners of such Foreign Lender are claiming the portfolio interest exemption, such Foreign Lender may provide a U.S. Tax Compliance Certificate substantially in the form of **Exhibit I-4** on behalf of each such direct and indirect partner;
- (C) any Foreign Lender shall, to the extent it is legally entitled to do so, deliver to the Borrower and the Administrative Agent (in such number of copies as shall be requested by the recipient) on or prior to the date on which such Foreign Lender becomes a Lender under this Agreement (and from time to time thereafter upon the reasonable request of the Borrower or the Administrative Agent), executed originals of any other form prescribed by applicable law as a basis for claiming exemption from or a reduction in U.S. federal withholding Tax, duly completed, together with such supplementary documentation as may be prescribed by applicable law to permit the Borrower or the Administrative Agent to determine the withholding or deduction required to be made; and

(D) if a payment made to a Lender under any Finance Document would be subject to U.S. federal withholding Tax imposed by FATCA if such Lender were to fail to comply with the applicable reporting requirements of FATCA (including those contained in Section 1471(b) or 1472(b) of the Code, as applicable), such Lender shall deliver to the applicable Borrower and the Administrative Agent at the time or times prescribed by law and at such time or times reasonably requested by such Borrower or the Administrative Agent such documentation prescribed by applicable law (including as prescribed by Section 1471(b)(3)(C)(i) of the Code) and such additional documentation reasonably requested by such Borrower or the Administrative Agent as may be necessary for such Borrower and the Administrative Agent to comply with their obligations under FATCA and to determine that such Lender has complied with such Lender's obligations under FATCA or to determine the amount to deduct and withhold from such payment. Solely for purposes of this clause (D), "FATCA" shall include any amendments made to FATCA after the date of this Agreement.

Each Lender agrees that if any form or certification it previously delivered expires or becomes obsolete or inaccurate in any respect, it shall update such form or certification or promptly notify the applicable Borrower and the Administrative Agent in writing of its legal inability to do so.

- (g) <u>Survival</u>. Each party's obligations under this <u>Section 2.14</u> shall survive the resignation or replacement of the Administrative Agent or any assignment of rights by, or the replacement of, a Lender, the termination of the Commitments and the repayment, satisfaction or discharge of all obligations under any Finance Document.
  - (h) <u>Defined Term</u>. For purposes of this <u>Section 2.14</u>, the term "applicable law" includes FATCA.

47

#### 2.15 <u>Illegality</u>.

- (a) If any Lender determines that the introduction of any Law, or any change in any Law, or in the interpretation or administration of any Law, makes it unlawful, or that any central bank or other Governmental Authority has asserted that it is unlawful, for any Lender or its Applicable Lending Office to make a Eurodollar Loan and such determination is applied to other borrowers of such Lender or its Affiliates, then, on notice thereof by such Lender to the Borrowers through the Administrative Agent, any obligation of such Lender to make such Loan shall be suspended until such Lender notifies the Administrative Agent and the Borrowers that the circumstances giving rise to such determination no longer exist.
- (b) If any Lender determines that the introduction of any Law, or any change in any Law, or in the interpretation or administration of any Law, makes it unlawful, or that any central bank or other Governmental Authority has asserted that it is unlawful, for any Lender or its Applicable Lending Office to make a Eurodollar Loan and such determination is applied to other borrowers of such Lender or its Affiliates, the Borrowers shall, upon its receipt of notice of such fact and demand from such Lender (with a copy to the Administrative Agent), prepay in full the Eurodollar Loans of such Lender then outstanding, together with interest accrued thereon and amounts required under Section 2.16 or 2.17, if any, either on the last day of the Interest Period in respect of such Eurodollar Loan, if such Lender may lawfully continue to maintain such Eurodollar Loans to such day, or immediately, if such Lender may not lawfully continue to maintain such Eurodollar Loan, then concurrently with such prepayment, the Borrowers shall, notwithstanding any otherwise applicable restriction, prohibition or requirement herein, borrow (and be deemed to have so borrowed) from the affected Lender a Base Rate Loan in the amount of the principal amount of the Eurodollar Loan so prepaid.
- (c) If the obligation of any Lender to make or maintain Eurodollar Loans has been so terminated or suspended, the Borrowers may elect, by giving notice to such Lender through the Administrative Agent, that all Loans which would otherwise be made by such Lender as Eurodollar Loans shall instead be Base Rate Loans.
- (d) Before giving notice to the Administrative Agent under this <u>Section 2.15</u>, the affected Lender shall designate a different Applicable Lending Office with respect to its Eurodollar Loans if such designation will avoid the need for giving such notice or making such demand and will not, in the sole judgment of such Lender, be illegal or otherwise disadvantageous to such Lender.

## 2.16 Increased Costs and Reduction of Return.

- (a) If, due to any Change in Law, there shall be:
  - (i) any increase in the cost to such Lender of agreeing to make or making, funding or maintaining any Loans,
- (ii) subject any Lender to any Taxes (other than (A) Indemnified Taxes, (B) Taxes described in clauses (b) through (d) of the definition of Excluded Taxes and (C) Connection Income Taxes) on its loans, loan principal, letters of credit, commitments, or other obligations, or its deposits, reserves, other liabilities or capital attributable thereto; or

48

(iii) any reduction in the amount of any sum received or receivable by such Lender hereunder (whether of principal, interest or any other amount),

then the Borrowers shall be liable for, and shall from time to time, within thirty (30) days of demand therefor by such Lender (with a copy of such demand to the Administrative Agent), pay to the Administrative Agent for the account of such Lender, additional amounts as are sufficient to compensate such Lender for such increased costs; provided, that the Borrowers shall not be required to compensate any Lender pursuant to this Section for any increased costs incurred or reductions suffered more than 180 days prior to the date that such Lender notifies the Borrowers in writing of the increased costs or reductions and of such Lender's intention to claim compensation thereof; provided, further, that if the circumstance giving rise to such increased costs or reductions is retroactive, then the 180-day period referred to above shall be extended to include the period of retroactive effect thereof.

(b) If any Lender shall have determined that any Change in Law regarding any Capital Adequacy Regulation has or would have the effect of reducing the rate of return on such Lender's capital to a level below that which such lender could have achieved but for such Change in Law (taking into consideration such Lender's or such Person's policies with respect to capital adequacy and such Lender's desired return on capital) and such

determination is applied to other borrowers of such Lender or its Affiliates, then, within thirty (30) days of demand therefor by such Lender to the Borrowers through the Administrative Agent, the Borrowers shall pay to such Lender, from time to time as specified by such Lender, additional amounts sufficient to compensate such Lender for such reduction; provided, that the Borrowers shall not be required to compensate any Lender pursuant to this Section for any increased costs incurred or reductions suffered more than 180 days prior to the date that such Lender notifies the Borrowers in writing of the increased costs or reductions and of such Lender's intention to claim compensation thereof; provided, further, that if the circumstance giving rise to such increased costs or reductions is retroactive, then the 180-day period referred to above shall be extended to include the period of retroactive effect thereof.

- (c) Before giving notice to the Administrative Agent under Section 2.16(a), the affected Lender shall designate a different Applicable Lending Office with respect to its Loans if such designation will avoid the need for giving such notice or making such demand and will not, in the sole judgment of such Lender, be illegal or otherwise disadvantageous to such Lender.
- 2.17 <u>Funding Losses</u>. The Borrower shall reimburse each Lender and hold each Lender harmless from any loss or expense which such Lender may sustain or incur as a consequence of:
  - (a) the failure of either Borrower to make on a timely basis any scheduled payment of principal of any Loan;

49

- (b) the failure of either Borrower to borrow or convert a Loan after such Borrower has given (or is deemed to have given) a Notice of Borrowing or a Notice of Conversion or Continuation;
  - (c) the failure of either Borrower to make any prepayment in accordance with any notice delivered under Section 8.2;
- (d) subject to <u>Section 8.7(d)</u>, the prepayment or repayment (including pursuant to <u>Section 8.1</u>, <u>8.2</u> or <u>8.3</u>) or other payment (including after acceleration thereof) of a Eurodollar Loan on a day that is not the last day of the applicable Interest Period; or
  - (e) the conversion of any Eurodollar Loan to a Base Rate Loan on a day that is not the last day of the applicable Interest Period;

including any such loss or expense arising from the liquidation or reemployment of funds obtained by it to maintain its Loan or from fees payable to terminate the deposits from which such funds were obtained.

- 2.18 <u>Inability to Determine Rates</u>. If the Administrative Agent reasonably determines that for any reason adequate and reasonable means do not exist for determining the Eurodollar Rate for any Interest Period with respect to any Eurodollar Loans, or that the Eurodollar Rate applicable for any Interest Period with respect to a Eurodollar Loan does not adequately and fairly reflect the cost to the Lenders of funding such Eurodollar Loan and such determination is applied to other borrowers of the Administrative Agent and its Affiliates, the Administrative Agent will promptly so notify the Borrowers and each Lender. Thereafter, the obligation of the Lenders to make or continue Eurodollar Loans or to convert any Base Rate Loan into a Eurodollar Loan hereunder shall be suspended until the Administrative Agent revokes such notice in writing. Upon the receipt of such notice, the Borrowers may revoke any Notice of Borrowing or Notice of Conversion or Continuation then submitted by them. If the Borrowers do not revoke any such notice, the Lenders shall make, convert or continue the Loans, as proposed by the Borrowers, in the amount specified in the applicable notice submitted by the Borrowers, but such Loans shall be made, converted or continued as Base Rate Loans instead of Eurodollar Loans.
- 2.19 <u>Survival</u>. The agreements and obligations of the Borrowers in <u>Sections 2.14</u> through <u>2.18</u> shall survive the payment of the Loans, the Notes and all other Obligations (other than Obligations under any Required Hedging Agreement).
- 2.20 <u>Replacement of Lenders.</u> If any Lender (x) is owed increased costs or other amounts under <u>Section 2.14</u>, <u>2.15</u> or <u>2.16</u>, (y) shall have become a Defaulting Lender or (z) fails to consent to a requested amendment, waiver or modification to any Finance Document in which Required Lenders have already consented to such amendment, waiver or modification but the consent of each Lender (or each Lender directly affected thereby, as applicable) is required with respect thereto, the Borrowers shall have the right, if no Default or Event of Default then exists and (in the case of (x) above) such Lender has not changed its Applicable Lending Office with the effect of eliminating such increased cost or other amount, to replace such Lender (the "<u>Replaced Lender</u>") with one or more commercial banks or other financial institutions

50

(collectively, the "Replacement Lender") reasonably acceptable to the Administrative Agent; provided that (a) at the time of any replacement pursuant to this Section 2.20, the Replacement Lender shall enter into one or more Assignment and Acceptances pursuant to Section 11.11 hereof pursuant to which the Replacement Lender shall acquire all of the Commitments and outstanding Loans of the Replaced Lender at par and, in connection therewith, shall pay to the Replaced Lender in respect thereof an amount equal to the sum of (i) an amount equal to the principal of, and all accrued interest on, all outstanding Loans of the Replaced Lender and (ii) an amount equal to all accrued, but theretofore unpaid, Fees owing to the Replaced Lender and (b) all obligations of the Borrowers owing to the Replaced Lender (other than those specifically described in clause (a) above in respect of which the assignment purchase price has been, or is concurrently being, paid) shall be paid in full to such Replaced Lender concurrently with such replacement. Upon the execution of the respective assignment documentation pursuant to clause (a) above and the payment of the amounts referred to in clauses (a) and (b) above, the Replacement Lender shall become a Lender hereunder, and the Replaced Lender shall cease to constitute a Lender hereunder, except with respect to Sections 11.1, 11.2 and 11.18 and any other indemnification or expense reimbursement provisions under this Agreement or any other Finance Document, which shall survive as to such Replaced Lender in accordance with their terms. Nothing in this Section 2.20 shall be deemed to prejudice any other rights that the Borrower may have against any Lender that is a Defaulting Lender.

## 2.21 <u>Defaulting Lenders</u>.

(a) <u>Defaulting Lender Adjustments</u>. Notwithstanding anything to the contrary contained in this Agreement, if any Lender becomes a Defaulting Lender, then, until such time as such Lender is no longer a Defaulting Lender, to the extent permitted by applicable law:

- (i) <u>Waivers and Amendments</u>. Such Defaulting Lender's right to approve or disapprove any amendment, waiver or consent with respect to this Agreement shall be restricted as set forth herein (including in the definition of Required Lenders).
- (ii) <u>Defaulting Lender Waterfall</u>. Any payment of principal, interest, fees or other amounts received by any Agent for the account of such Defaulting Lender (whether voluntary or mandatory, at maturity, pursuant to <u>Section 9</u> or otherwise) or received by any Agent from a Defaulting Lender pursuant to <u>Section 11.14</u> shall be applied at such time or times as may be determined by the Administrative Agent as follows: *first*, to the payment of any amounts owing by such Defaulting Lender to any Agent hereunder; *second*, if requested by the Borrowers (so long as no Default or Event of Default exists), to the funding of any Loan in respect of which such Defaulting Lender has failed to fund its portion thereof as required by this Agreement, as determined by the Administrative Agent; *third*, if so determined by the Administrative Agent and the Borrowers, to be held in a deposit account and released *pro rata* in order to satisfy such Defaulting Lender's potential future funding obligations with respect to Loans under this Agreement; *fourth*, to the payment of any amounts owing to the Lenders as a result of any judgment of a court of competent jurisdiction obtained by any Lender against such Defaulting Lender as a result of such Defaulting Lender's breach of its obligations under this Agreement; *fifth*, so long as no Default or Event of Default exists, to the payment of any amounts owing to the Borrowers as a result of any judgment of a court of competent

jurisdiction obtained by the Borrowers against such Defaulting Lender as a result of such Defaulting Lender's breach of its obligations under this Agreement; and *sixth*, to such Defaulting Lender or as otherwise directed by a court of competent jurisdiction; *provided*, *however*, that if (i) such payment is a payment of the principal amount of any Loans in respect of which such Defaulting Lender has not fully funded its appropriate share, and (ii) such Loans were made at a time when the conditions set forth in **Section 3.3** and, if applicable, **Section 3.2**, were satisfied or waived, such payment shall be applied solely to pay the Loans of all Non-Defaulting Lenders on a *pro rata* basis prior to being applied to the payment of any Loans of such Defaulting Lender until such time as all Loans are held by the Lenders *pro rata*. Any payments, prepayments or other amounts paid or payable to a Defaulting Lender that are applied (or held) to pay amounts owed by a Defaulting Lender pursuant to this **Section 2.21(a)(ii)** shall be deemed paid to and redirected by such Defaulting Lender, and each Lender irrevocably consents hereto

- (iii) <u>Certain Fees</u>. No Defaulting Lender shall be entitled to receive any fees pursuant to <u>Section 8.6(a)</u> for any period during which that Lender is a Defaulting Lender (and the Borrowers shall not be required to pay any such fee that otherwise would have been required to have been paid to that Defaulting Lender).
- (b) <u>Defaulting Lender Cure</u>. If the Borrowers and the Administrative Agent agree in writing that a Lender is no longer a Defaulting Lender, the Administrative Agent will so notify the parties hereto, whereupon as of the effective date specified in such notice and subject to any conditions set forth therein, that Lender will, to the extent applicable, purchase at par that portion of outstanding Loans of the other Lenders or take such other actions as the Administrative Agent may determine to be necessary to cause the Loans to be held *pro rata* by the Lenders, whereupon such Lender will cease to be a Defaulting Lender; *provided*, *however*, that no adjustments will be made retroactively with respect to fees accrued or payments made by or on behalf of the Borrowers while that Lender was a Defaulting Lender; and *provided*, *further*, that except to the extent otherwise expressly agreed by the affected parties, no change hereunder from Defaulting Lender to Lender will constitute a waiver or release of any claim of any party hereunder arising from that Lender's having been a Defaulting Lender.

## 2.22 <u>Interest During Construction.</u>

(a) <u>Deemed Loan Requests</u>. Unless a Borrower shall have otherwise notified Administrative Agent at least five Business Days prior to the relevant date on which an interest payment is due (such dates, "<u>Interest Payment Dates</u>") for loans of the applicable Tranche, such Borrower shall be deemed to have requested that the amount of interest due on each Interest Payment Date occurring prior to applicable Conversion Date be paid pursuant to this <u>Section 2.22</u>. Subject to the conditions set forth in <u>Section 2.22(b)</u>, each Lender agrees to effect payment of the interest due to such Lender on each such Interest Payment Date pursuant to <u>Section 2.22(c)</u>. If a Borrower advises the Administrative Agent that it elects not to have any such interest paid pursuant to this <u>Section 2.22</u>, or if the conditions for any such payment set forth in <u>Section 2.22(b)</u> are not satisfied, then Administrative Agent shall so advise the Lenders (and the Borrowers in the case of a failure to satisfy such conditions) in writing, in which case no Loans of such Tranche shall be made pursuant to this <u>Section 2.22</u> on the relevant Interest Payment Date and such Borrower shall pay the amount of interest due on such Interest Payment Date in cash.

52

- (b) <u>Conditions</u>. The Borrowers shall not be entitled to make any payment of interest pursuant to this <u>Section 2.22</u> unless (i) the conditions precedent set forth in <u>Sections 3.3(b)</u>, (c), (h), and (k) are determined by the Administrative Agent to have been satisfied with respect to the Loans to be funded to effect such payment pursuant to <u>Section 2.22(c)</u>, (ii) if and to the extent such Loans are to be Tranche A Construction Loans, the initial Tranche B Construction Loans have already been funded and made.
- (c) <u>Deemed Funding and Payment</u>. Subject to <u>Sections 2.22(a)</u> and <u>(b)</u>, on each Interest Payment Date occurring before the Conversion Date, each Lender shall book a Loan of the applicable Tranche to each Borrower in an amount equal to the amount of interest due from such Borrower to such Lender on such Interest Payment Date, and shall apply the proceeds of such Loan to payment of the interest owed by such Borrower to such Lender on such Interest Payment Date. Each such Loan shall be a Eurodollar Loan with an Interest Period of one month unless converted pursuant to <u>Section 2.10</u>. Notwithstanding anything to the contrary contained herein, in no event shall any Loans made pursuant to this Section 2.22 shall (i) cause the amount of Tranche A Construction Loans or Tranche B Construction Loans disbursed by any Lender to exceed such Lender's Tranche A Construction Loan Commitment or Tranche B Construction Loan Commitment, as applicable or (ii) cause the amount of Total Funded Construction Loans to exceed the Total Construction Commitment.

## SECTION 3. CONDITIONS PRECEDENT.

3.1 <u>Conditions to Closing</u>. The effectiveness of this Agreement and the other Finance Documents to be executed on the date hereof shall be subject to the conditions precedent that the Administrative Agent shall have received, or shall have waived receipt of, the following, each of which shall be in form and substance reasonably satisfactory to each Lender, and that the other conditions set forth below shall have been satisfied or waived by each Lender:

` _	saction Documents. Each of the following Transaction Documents shall have been executed and delivered by each party Agent shall have received a copy thereof executed by all parties thereto:
(i)	This Agreement;
(ii)	The Security Agreement;
	To the extent requested by any Lender, an original Tranche A Construction Note or Tranche B Construction Note for the extent Notes shall be appropriately completed with the name of the payee, the maximum principal amount thereof and the h shall be the Closing Date) inserted therein;
(iv)	The Pledge Agreement;

The Equity Contribution Agreement;

- (vi) The Sponsor Guaranty;
- (vii) The Warrant
- (viii) The Warrant Agreement;
- (ix) The Registration Rights Agreement; and
- (x) The Closing Date Project Documents and the Consent Agreements related thereto.
- (b) [Reserved]

(v)

- (c) <u>Charter Documents</u>. The Administrative Agent shall have received the following documents, each certified as indicated below:
- (i) a copy of the Charter Documents of each Borrower, the Sponsor each Affiliate Project Participant and Holdings, as in effect on the Closing Date, certified by the Secretary of State or other appropriate officer of the jurisdiction of such Person's formation or incorporation, and a certificate, where available, as to the good standing of and payment of franchise taxes by each Borrower, the Sponsor, each Affiliate Project Participant and Holdings, certified by the Secretary of State or other appropriate officer of the jurisdiction of such Person's formation or incorporation and, in respect of the Borrowers only, each jurisdiction in which each Borrower is qualified as a foreign entity, dated as of a recent date;
- (ii) a certificate of an Authorized Officer of each Borrower, the Sponsor, each Affiliate Project Participant and Holdings, dated the Closing Date, certifying (A) that attached thereto is a true, correct and complete copy of the Charter Documents of such Person, as in effect at all times from the date on which the resolutions referred to in clause (B) below were adopted to and including the date of such certificate, (B) that attached thereto is a true, correct and complete copy of resolutions duly adopted by the board of directors (or other equivalent body) or evidence of all corporate, partnership or limited liability company action, as the case may be, of such Person, authorizing the execution, delivery and performance of the Transaction Documents to which such Person is or is intended to be a party, and that such resolutions have not been modified, rescinded or amended and are in full force and effect, and (C) as to the name, incumbency and specimen signature of each officer of such Person executing the Transaction Documents to which such Person is intended to be a party and each other document to be delivered by such Person from time to time in connection therewith (and the Secured Parties may conclusively rely on such certificate until the Administrative Agent receives a replacement certificate in the form described in this clause (C) from such Person); and
- (iii) a certificate of another Authorized Officer of each Borrower, the Sponsor, each Affiliate Project Participant and Holdings, as applicable, as to the name,

54

incumbency and specimen signature of the Authorized Officer of such Person that signed the certificate referred to in clause (ii) above.

- (d) <u>Filings, Registrations and Recordings</u>. Any document (other than the Mortgages) required to be filed, registered, notarized or recorded in order to create and perfect the Security Interests as first priority Liens (subject only to (i) Permitted Liens and (ii) the priority afforded to such Permitted Liens by operation of Law) shall have been properly filed, registered, notarized or recorded in each office in each jurisdiction in which such filings, registrations, notarizations and recordations are required, and any other action required in the judgment of the Collateral Agent to perfect such Security Interests as such first priority Liens (subject only to (i) Permitted Liens and (ii) the priority afforded to such Permitted Liens by operation of Law) shall have been effected.
- (e) <u>Pledge Agreement</u>. Each Lender and the Administrative Agent shall have received evidence satisfactory to such Person that the pledged securities required to be delivered to the Collateral Agent pursuant to the Pledge Agreement have been delivered, together with such other documents as are necessary to perfect the interests of the Secured Parties in and to the Collateral covered thereby as a first-priority security interest.
  - (f) Officers' Certificates.

- (i) <u>Borrowers' Certificates</u>. The Administrative Agent shall have received a certificate signed by an Authorized Officer of each Borrower, dated the Closing Date, certifying that: (i) the representations and warranties of such Borrower contained in <u>Section 4</u> hereof and the representations and warranties of such Borrower contained in each of the other Transaction Documents to which such Borrower is a party are true and correct in all material respects on and as of such date as if made on and as of such date (or, if expressly stated to have been made solely as of an earlier date, were true and correct in all material respects as of such earlier date), (ii) all Transaction Documents then in existence to which such Borrower is a party are in full force and effect, (iii) no Default or Event of Default has occurred and is continuing and (iv) no default or event of default has occurred and is continuing under any Closing Date Project Document.
- (ii) Sponsor's Certificate. The Administrative Agent shall have received a certificate signed by an Authorized Officer of the Sponsor, dated the Closing Date, certifying that (i) the representations and warranties of the Sponsor set forth in the Transaction Documents to which the Sponsor is a party are true and correct in all material respects on and as of such date as if made on and as of such date (or, if stated to have been made solely as of an earlier date, were true and correct in all material respects as of such earlier date) and (ii) the Sponsor is in compliance with all of its agreements contained in any Transaction Document to which it is a party.
- (iii) <u>Holdings' Certificate</u>. The Administrative Agent shall have received a certificate signed by an Authorized Officer of Holdings, dated the Closing Date, certifying that (i) the representations and warranties of Holdings set forth in the Pledge Agreement are true and correct in all material respects on and as of such date as if made on and as of such date (or, if stated to have been made solely as of any earlier date, were true and correct in all material respects as of such earlier date) and (ii) Holdings is in compliance with all of its agreements contained in the Pledge Agreement.

(iv) Affiliate Project Participant Certificates. The Administrative Agent shall have received a certificate signed by an Authorized Officer of each Affiliate Project Participant, dated the Closing Date, certifying that (i) the representations and warranties of such Affiliate Project Participant set forth in each applicable Affiliate Project Document are true and correct in all material respects on and as of such date as if made on and as of such date (or, if stated to have been made solely as of any earlier date, were true and correct in all material respects as of such earlier date) and (ii) each Affiliate Project Participant is in compliance with all of its agreements contained in each applicable Affiliate Project Document.

#### (g) <u>Financial Information, etc.</u>

- (i) The Administrative Agent shall have received copies of (A) the unaudited consolidated financial statements of the Sponsor for the fiscal quarter ended June 30, 2012, and (B) the audited consolidated financial statements of the Sponsor for the fiscal year ended December 31, 2011. The Administrative Agent shall be deemed to have received any financial statement referred to in this clause (i) upon its filing with the Securities and Exchange Commission.
- (ii) The Administrative Agent shall have received such other financial, business and other information regarding the Closing Date Project Participants (other than the Affiliate Project Participants) as such Person shall have reasonably requested.

## (h) [Reserved]

(i) <u>Legal Opinions</u>. The Administrative Agent, the Collateral Agent and the Lenders shall have received a written legal opinion of Morrison & Foerster LLP, counsel to each Borrower, the Sponsor, Holdings, and each Affiliate Project Participant, which legal opinion shall be dated the Closing Date and addressed to the Administrative Agent, the Collateral Agent and the Lenders and in form, scope and substance satisfactory to each such Person.

## (j) [Reserved]

- (k) <u>Corporate Structure</u>. The corporate governance and capital structure of each Borrower and Holdings shall be satisfactory to the Administrative Agent.
- (l) <u>Material Adverse Effect; Default</u>. No event shall have occurred and no condition shall exist that has had or could reasonably be expected to have a Material Adverse Effect. No Default or Event of Default shall have occurred and be continuing or would result from this Agreement or the other Transaction Agreements becoming effective in accordance with their respective terms.
- (m) <u>Fees</u>. The Borrowers shall have paid, or made arrangements satisfactory to the Administrative Agent to pay, all fees, costs and charges payable by them (including fees of counsel to the Secured Parties) under the Finance Documents, not to exceed \$500,000, on or prior to the Closing Date.

- (n) <u>Patriot Act</u>. The Secured Parties shall have received, sufficiently in advance of the Closing Date, all documentation and other information that may be required by the Secured Parties in order to enable compliance with Anti-Terrorism Laws.
- (o) <u>Preliminary Base Case Projections</u>. Each Lender and the Administrative Agent shall have received the Base Case Projections for each Project, which shall project (a) a Tranche A Debt Service Coverage Ratio and a Tranche B Debt Service Coverage Ratio, as applicable, of not less than 1.25:1.00 as of the end of each fiscal quarter during the period covered thereby and (b) that no more than twenty percent (20%) of the original principal amount of Tranche A Term Loans or Tranche B Term Loans shall be outstanding as of the Tranche A Final Maturity Date and the Tranche B Final Maturity Date, as applicable. The Base Case Projections shall (i) be prepared using assumptions provided in good faith by the applicable Borrower, (ii) be certified as true, complete and correct by an Authorized Officer of such Borrower and (iii) be otherwise in form and substance satisfactory to each Lender.

- (p) Project Documents. (i) The representations and warranties of the Closing Date Project Participants contained in each Closing Date Project Document shall be true and correct in all material respects on and as of such date as if made on and as of such date (or, if stated to have been made solely as of an earlier date, were true and correct in all material respects as of such earlier date), (ii) all Closing Date Project Documents shall be in full force and effect under the terms and conditions set forth in such Closing Date Project Documents, and (iii) no default or event of default shall have occurred and be continuing under any Closing Date Project Document.
- (q) Additional Matters. All corporate and other proceedings, and all documents, instruments and other legal matters in connection with the transactions contemplated by this Agreement and the other Transaction Documents shall be satisfactory in form and substance to each Lender and the Administrative Agent, and each Lender and the Administrative Agent shall have received such other documents, certificates, and instruments relating to this Agreement or any other Transaction Document or the transactions contemplated hereby or thereby as such Person shall have reasonably requested, in each case in form and substance satisfactory to each such Person.

For the purpose of determining compliance with the conditions specified in this <u>Section 3.1</u>, each Lender shall be deemed to have accepted, and to be satisfied with, each document or other matter required under this <u>Section 3.1</u> unless the Administrative Agent shall have received written notice from such Lender prior to the proposed Closing Date specifying its objection thereto.

3.2 <u>Conditions to Initial Funding</u>. The obligation of any Lender to make its initial Construction Loan to a Borrower in respect of a Tranche on the Initial Funding Date relating to such Tranche shall be subject to the conditions precedent that, both immediately prior to the making of such initial Construction Loan and also after giving effect thereto:

57

- Insurance. The Administrative Agent shall have received a binder certified by a nationally recognized insurance broker, with respect to each policy of insurance required to be in effect pursuant to Section 5.9 hereof in respect of such Borrower, and the designation of the Collateral Agent as loss payee thereunder to the extent required by Section 5.9 hereof. In addition, such Borrower shall have delivered to the Administrative Agent written letters, dated no earlier than ten (10) days prior to the Initial Funding Date, from a firm of independent insurance brokers of nationally recognized standing satisfactory to the Administrative Agent, as to such insurance being in full force and effect, subject only to payment of the premium therefor (which payment shall be made on or before the Initial Funding Date). In addition, the Administrative Agent shall have received a report from the Insurance Advisor stating that, in its opinion, all insurance policies required to be maintained (or caused to be maintained) by such Borrower pursuant to Section 5.9 hereof have been obtained and are in full force and effect on the Initial Funding Date, such insurance policies comply in all respects with the requirements of Section 5.9 hereof, and the insurance coverage required by Section 5.9 hereof is comparable in all respects to insurance carried by responsible owners and operators of properties similar to the applicable Project.
- (b) <u>Updated Base Case Projections; Drawdown Schedule; Budget</u>. The Administrative Agent shall have received (i) an updated version of the Base Case Projections for the applicable Project, which shall project (a) a Tranche A Debt Service Coverage Ratio or Tranche B Debt Service Coverage Ratio, as applicable, of not less than 1.25:1.00 as of the end of each fiscal quarter during the period covered thereby and (b) that no more than twenty percent (20%) of the original principal amount of Tranche A Term Loans or Tranche B Term Loans shall be outstanding as of the Tranche A Final Maturity Date and the Tranche B Final Maturity Date, as applicable, (ii) the Construction Budget for the applicable Project, (iii) the Drawdown Schedule for the applicable Project, and (iv) the Borrower's proposed Operating Budget for the first Operating Year of the applicable Project, in each case prepared using assumptions provided in good faith by such Borrower, certified as true, complete and correct by an Authorized Officer of such Borrower and the Project Engineer and otherwise in form and substance reasonably satisfactory to each Lender and the Project Engineer.
- (c) <u>Project Engineer's Report</u>. The Administrative Agent shall have received a report of the Project Engineer as to such matters with respect to the applicable Project as the Administrative Agent shall reasonably request, including the technical and economic feasibility of such Project, the ability of such Project to meet regulatory and contractual requirements, the status of all Necessary Governmental Approvals for such Project, the reasonableness of the Base Case Projections for such Project, the Construction Budget and the Drawdown Schedule for such Project, the operating performance assumptions of such Project, the engineering design of such Project, the commercial operating arrangements for such Project, and the adequacy and appropriateness of the Acceptance Tests of such Project, the applicable Construction Agreements, the applicable LNG Purchase Agreement and such other Project Documents as the Administrative Agent shall request.
- (d) <u>Necessary Governmental Approvals</u>. The Administrative Agent shall have received (i) copies of all Necessary Governmental Approvals (including all Necessary Governmental Approvals required to initiate construction), which shall be final and in full force and effect and for which either the applicable administrative and statutory appeal periods have

58

expired or those applicable administrative and statutory appeal periods greater than ninety (90) days that have not expired and their relevant expiration dates are identified on Schedule 4.6 and, (ii) if requested, copies of all applications made for such Necessary Governmental Approvals and all material correspondence received or sent in respect of such applications; provided that with respect to Necessary Governmental Approvals (other than any Necessary Governmental Approvals required to initiate construction) that are not required by applicable Law to have been obtained on or prior to the Initial Funding Date for such Tranche in the exercise of reasonable diligence (but which are routinely obtainable and can be obtained only at a later stage of construction, only after completion of certain operations testing or only after a period of operations), the Secured Parties shall have received reasonably satisfactory assurances that such Necessary Governmental Approvals will be obtained by the time when needed in connection with the construction or operation of the applicable Project.

## (e) <u>Environmental Matters</u>.

(i) The Administrative Agent shall have received from Environ Corporation, or another environmental consultant reasonably acceptable to the Administrative Agent, an environmental assessment report (with a copy to the Collateral Agent) with respect to the applicable Site (which shall include, at a minimum, a Phase I Environmental Site Assessment and Limited Compliance Review), in form, scope, and substance reasonably acceptable to the Administrative Agent; and

Laws.

(f) <u>Utilities</u>. The Administrative Agent shall have received a certificate of an Authorized Officer of such Borrower (with a concurring certification by the Project Engineer), dated the Initial Funding Date, to the effect that all utility services necessary for the construction and operation of the applicable Project (including potable and raw water supply, storm, electric, telephone and sewage services and facilities) have been committed to such Project (with a true copy of binding agreements (if any) which evidence the same) by appropriate utilities, authorities or other Persons, or are otherwise available to such Borrower in the ordinary course of business, in each case on terms consistent with those reflected in the Construction Budget and the Base Case Projections relating to such Project.

The applicable Project and such Project's design and operation shall be in compliance with all applicable Environmental

(g) The Sites. With respect to the Site relating to the applicable Project:

(ii)

(i) Such Borrower shall have (A) acquired a valid and indefeasible fee simple title to or leasehold interests (or other applicable interests) in the applicable Site, free and clear of all Liens other than Permitted Liens described in Section 3.2(g)(iii)(A)(1) and (B) caused one or more Mortgages in respect of the applicable Site to be duly recorded in accordance with applicable Law, together with such financing statements and any other instruments necessary to grant a mortgage lien on the applicable Site under the laws of any applicable jurisdiction, all of which shall be in form and substance reasonably satisfactory to the Administrative Agent, and such Borrower shall have provided evidence reasonably acceptable to the Administrative Agent of payment by such Borrower of all mortgage recording taxes, fees, charges, costs and expenses required for the recording of such Mortgages and such financing statements and other ancillary instruments.

59

- (ii) Such Borrower shall have obtained, at its sole cost and expense, all such consents, approvals, amendments, supplements, estoppels, tenant subordination agreements or other instruments as necessary to consummate the transactions contemplated hereby, or as shall be deemed reasonably necessary by any Lender in order for such Borrower to grant the Lien contemplated by each Mortgage relating to the applicable Site.
- (iii) Such Borrower shall have obtained, at its sole cost and expense, (A) one or more Lender's or mortgagee's policy of title insurance issued by the Title Insurance Company, in favor of the Collateral Agent, for the benefit of the Secured Parties, that shall (1) insure the validity and priority of the Lien created under the applicable Mortgage as a valid first mortgage Lien on the applicable Site, such policy or policies to be in an aggregate amount not less than \$200,000,000 (subject to a pending disbursement provision as set forth in clause (2) below) (unless approved by the Administrative Agent, in its reasonable discretion), in form and substance, and listing only those exceptions, reasonably satisfactory to the Administrative Agent, (2) contain a pending disbursement provision reasonably satisfactory to the Administrative Agent, but without an exception for unfiled mechanics' and materialmen's lien claims, (3) be accompanied by such reinsurance agreements as may be reasonably requested by the Administrative Agent, (4) contain such endorsements and other additional coverages and modifications as the Administrative Agent shall reasonably require, in the exercise of its sole discretion, and (5) otherwise be in form and substance reasonably satisfactory to the Administrative Agent (collectively, the "Title Insurance") and (B) a survey of recent date of the applicable Site certified to the Collateral Agent by the Title Insurance Company and such Borrower, which survey shall be in form and substance satisfactory to each Lender and the Title Insurance Company.
- (iv) Such Borrower shall have executed and delivered such affidavits, certificates, information (including financial data) and instruments of indemnification (including a so-called "gap" indemnification) as shall be required to induce the Title Insurance Company to issue the Title Insurance and endorsements contemplated above.
- (v) Such Borrower shall have delivered evidence reasonably acceptable to the Administrative Agent of payment by such Borrower of all premiums for the Title Insurance in respect of such Site, and all related search and examination charges, escrow charges and related charges of the Title Insurance Company.
- (vi) The Administrative Agent shall have received a completed "Life-of-Loan" Federal Emergency Management Agency Standard Flood Hazard Determination with respect to such Site (together with a notice about special flood hazard area status and flood disaster assistance duly executed by such Borrower).
- (h) <u>Legal Opinions</u>. The Administrative Agent shall have received such written opinions of counsel to each Finance Party and the Affiliate Project Participants as the

60

Administrative Agent may reasonably request, including a legal opinion by an attorney acceptable to the Administrative Agent and licensed in the state or states in which the applicable Site is located that all Necessary Governmental Approvals with regard to such Site have been obtained, are final and in full force and effect, and for which either the applicable administrative and statutory appeal periods have expired or those applicable administrative and statutory appeal periods greater than ninety (90)\_days that have not expired and their relevant expiration dates are identified on Schedule 4.6. Such legal opinions shall be reasonably acceptable to the Administrative Agent.

- (i) <u>Required Hedging Agreements</u>. Such Borrower shall have entered into the Required Hedging Agreements required as of such Initial Funding Date under <u>Section 5.12</u> and shall have delivered copies of such Required Hedging Agreements to the Administrative Agent.
- (j) New Project Documents. Such Borrower shall have entered into (i) a GE EPC Contract and such other Construction Agreements as are determined by the Project Engineer and the Administrative Agent in their reasonable discretion to be necessary or desirable to complete the design, development, construction, testing and start-up of the applicable Project, (ii) a Gas Supply Agreement and (iii) a Power Purchase Agreement, each of which shall be in form and substance reasonably satisfactory to the Administrative Agent and, in the case of each Gas Supply Agreement and Power Purchase Agreement shall include invoicing and payment mechanics that coordinate with the applicable LNG Purchase Agreement such there is no increase in such Borrower's working capital requirements.

#### (k) Officers' Certificates.

- (i) <u>Borrowers' Certificates</u>. The Administrative Agent shall have received a certificate signed by an Authorized Officer of each Borrower, dated the Initial Funding Date, certifying that: (i) the representations and warranties of each Borrower contained in <u>Section 4</u> hereof and the representations and warranties of each Borrower contained in each of the other Transaction Documents to which such Borrower is a party are true and correct in all material respects on and as of such date as if made on and as of such date (or, if expressly stated to have been made solely as of an earlier date, were true and correct in all material respects as of such earlier date), (ii) all Transaction Documents then in existence to which either Borrower is a party are in full force and effect, (iii) no Default or Event of Default has occurred and is continuing under any Affiliate Project Document and (v) to such Borrower's knowledge, no default or event of default has occurred and is continuing under any other Project Document, (other than any Immaterial Additional Project Document).
- (ii) Sponsor's Certificate. The Administrative Agent shall have received a certificate signed by an Authorized Officer of the Sponsor, dated the Initial Funding Date, certifying that (i) the representations and warranties of the Sponsor set forth in the Finance Documents and the Project Documents to which the Sponsor is a party are true and correct in all material respects on and as of such date as if made on and as of such date (or, if stated to have been made solely as of an earlier date, were true and correct in all material respects as of such earlier date) and (ii) the Sponsor is in compliance with all of its agreements contained in any Transaction Document to which it is a party.

61

- (iii) <u>Holdings' Certificate</u>. The Administrative Agent shall have received a certificate signed by an Authorized Officer of Holdings, dated the Initial Funding Date, certifying that (i) the representations and warranties of Holdings set forth in the Pledge Agreement are true and correct in all material respects on and as of such date as if made on and as of such date (or, if stated to have been made solely as of any earlier date, were true and correct in all material respects as of such earlier date) and (ii) Holdings is in compliance with all of its agreements contained in the Pledge Agreement.
- (iv) Affiliate Project Participant Certificates. The Administrative Agent shall have received a certificate signed by an Authorized Officer of each Affiliate Project Participant, dated the Initial Funding Date, certifying that (i) the representations and warranties of such Affiliate Project Participant set forth in each Affiliate Project Document are true and correct in all material respects on and as of such date as if made on and as of such date (or, if stated to have been made solely as of any earlier date, were true and correct in all material respects as of such earlier date) and (ii) each such Affiliate Project Participant is in compliance with all of its agreements contained in each Affiliate Project Document.
- (l) <u>Equity Letters of Credit</u>. The Administrative Agent shall have received the original Tranche A Equity Letters of Credit or the Tranche B Equity Letters of Credit, as applicable (as such terms are defined in the Equity Contribution Agreement).
- (m) <u>Solvency Certificates</u>. The Administrative Agent shall have received a solvency certificate signed by an Authorized Officer of the Sponsor and each Borrower, substantially in the form of **Exhibit H**.
- (n) <u>Updated Attachments</u>. The Administrative Agent shall have received, at least 10 Business Days prior to the applicable Initial Funding Date, (i) an updated version of <u>Appendix A</u>, in form and substance reasonably acceptable to the Administrative Agent and the Insurance Advisor, (ii) an updated version of <u>Schedule 4.6</u>, in form and substance reasonably acceptable to the Administrative Agent and the Project Engineer and (iii) a schedule for completion of the Work relating to the applicable Facility (a "<u>Project Schedule</u>"), in form and substance reasonably acceptable to the Administrative Agent and the Project Engineer.

For the purpose of determining compliance with the conditions specified in this <u>Section 3.2</u>, each Lender shall be deemed to have accepted, and to be satisfied with, each document or other matter required under this <u>Section 3.2</u> unless the Administrative Agent shall have received written notice from such Lender prior to the proposed Initial Funding Date specifying its objection thereto.

3.3 <u>Initial and Subsequent Construction Loans</u>. The obligation of any Lender to make its initial Construction Loan or any subsequent Construction Loan to a Borrower on any Disbursement Date shall be subject to the conditions precedent that, both immediately prior to

62

the making of such initial Construction Loan and each such subsequent Construction Loan and also after giving effect thereto:

#### (a) Construction Requisitions; Notices of Borrowing.

- (i) Not less than ten (10) Business Days prior to such Disbursement Date, the Administrative Agent shall have received (A) a Construction Requisition executed and delivered by an Authorized Officer of the applicable Borrower in respect of the Disbursement of Construction Loans to be made on such Disbursement Date, and (B) a certificate of the Project Engineer in respect of such proposed Disbursement in the form attached hereto as **Exhibit C-2** (such form being subject to review by the Project Engineer subsequent to the appointment thereof), in each case containing no exceptions or qualifications that are unsatisfactory to the Required Lenders.
- (ii) The Administrative Agent shall have received a Notice of Borrowing pursuant to and in compliance with <u>Section 2.5</u> hereof in respect of the Disbursement of Construction Loans on such Disbursement Date.
- (b) <u>Representations and Warranties</u>. The representations and warranties of each Borrower contained in <u>Section 4</u> hereof and the representations and warranties of each Finance Party, each Affiliate Project Participant and, to the knowledge of the applicable Borrower, each other Project Participant (other than any Lender Affiliate Project Participant or any Immaterial Project Participant) contained in any other Finance Document or Project Document then in effect shall be true and correct in all material respects on and as of such Disbursement Date as if made on and as of such date (or, if expressly stated to have been made solely as of an earlier date, were true and correct in all material respects as of such date).

- (c) No Default. No Default or Event of Default shall have occurred and be continuing or would result from the making of such Construction Loan and no default by any Finance Party, any Affiliate Project Participant or, to the knowledge of the applicable Borrower, any other Project Participant under any of the Transaction Documents (other than any Immaterial Additional Project Document) then in effect shall have occurred and be continuing or would result from the making of such Construction Loan.
- (d) Governmental Approvals, etc. (i) All Necessary Governmental Approvals in respect of the applicable Project that were not required to be obtained by the applicable Borrower or any Project Participant (other than any Immaterial Project Participant) on or prior to the Closing Date but which under applicable Law were required to be obtained prior to such Disbursement Date shall have been duly obtained and shall be final and in full force and effect and either the applicable administrative and statutory appeal periods shall have expired or those applicable administrative and statutory appeal periods greater than ninety (90) days that have not expired and their relevant expiration dates are identified on Schedule 4.6; (ii) in addition to and not in limitation of Section 3.3(d)(i) above and unless waived by the Administrative Agent, the Administrative Agent shall have received a legal opinion by an attorney acceptable to the Administrative Agent and licensed in the state or states in which the applicable Site is located that all permits and approvals required under Environmental Laws with regard to such Site have been obtained, are final and in full force and effect and either all applicable administrative and

statutory appeal periods have expired or those applicable administrative and statutory appeal periods greater than ninety (90) days that have not expired and their relevant expiration dates are identified on Schedule 4.6, (iii) there shall have been no change in any applicable Law, and no issuance of any order, writ, injunction or decree of any Governmental Authority or arbitral tribunal, which, in either such case, could reasonably be expected to have a Material Adverse Effect; and (iv) there shall have been no proposed change in or modification of any applicable Law which could reasonably be expected to be enacted and which if enacted could reasonably be expected to have a Material Adverse Effect.

- (e) <u>Material Adverse Effect</u>. There shall exist no circumstance, event or condition which has had or could reasonably be expected to have a Material Adverse Effect.
- (f) <u>Litigation</u>. No legal or arbitral proceedings or investigations, or any proceedings by or before any Governmental Authority, shall be pending or threatened against either Borrower, or its Properties or rights or, to the knowledge of each Borrower, against any Project Participant or its respective Properties or rights, which, if adversely determined, could reasonably be expected to have a Material Adverse Effect.
- (g) <u>Construction Budget; Drawdown Schedule</u>. Such Construction Loan shall be in accordance with the applicable Construction Budget and the applicable Drawdown Schedule.
- (h) <u>Fees and Expenses</u>. Each Borrower shall have paid or arranged for the payment when due (including, to the extent permitted, arrangement for payment out of such Disbursement) of all fees, expenses and other charges payable by it on or prior to such Disbursement Date under this Agreement or under any other Finance Document.
- (i) Title Continuation Report; Survey. (i) The applicable Borrower shall have delivered to the Administrative Agent a title continuation report and a down date endorsement from the Title Insurance Company to such Disbursement Date, in form and substance satisfactory to the Administrative Agent, setting forth no additional exceptions other than Permitted Liens of the type described in Section 6.1(d) and increasing the amount of the Title Insurance to the aggregate amount of all Disbursements (which, for the avoidance of doubt, shall expressly include, but shall not be limited to, the amount of the initial Construction Loan and such current Disbursement and (ii) in the case of the first Disbursement Date with respect to Construction Loans which is at least thirty (30) days after the foundations of the applicable Site have been completed, such Borrower shall have delivered to the Administrative Agent an updated survey of such Site, certified to the Collateral Agent, for the benefit of the Secured Parties, the Title Insurance Company and such Borrower, showing the Project as built to the date of such survey, which survey shall be in form and substance satisfactory to the Administrative Agent and the Title Insurance Company, and shall disclose no easements, rights of way or encumbrances, other than Permitted Liens of the type described in Section 6.1(d).
- (j) <u>Project Costs</u>. The aggregate outstanding principal amount of all Construction Loans of the applicable Tranche shall not exceed 75% of the aggregate Project Costs theretofore incurred in respect of the applicable Project.

64

- (k) <u>Equity Contribution</u>. The Administrative Agent shall have received satisfactory evidence that Equity Contributions have been deposited in the Tranche A Construction Account or the Tranche B Construction Account, as applicable, in an aggregate amount such that the ratio of the Construction Loans of the applicable Tranche (after giving effect to the proposed disbursement of Construction Loans to be made on the Disbursement Date) to such Equity Contributions does not exceed 75:25; <u>provided</u>, <u>however</u>, that for the purposes of this <u>Section 3.3(k)</u>, Equity Contributions for each Tranche shall include the allocable portion of all Project Costs directly paid by the Sponsor in connection with the Projects prior to the applicable Initial Funding Date (including without limitation any payments made by the Sponsor with respect to any letter of credit or the issuance thereof in connection with the Projects) to the extent the same have been approved by the Administrative Agent in its reasonable discretion.
- (l) Other Documents. The Administrative Agent shall have received such other statements, certificates, documents and approvals as the Administrative Agent may reasonably request.

The acceptance of the proceeds of each Construction Loan shall constitute a certification by the applicable Borrower to the Lenders confirming the satisfaction of the conditions set forth in clauses (a) through (!) of this Section 3.3 upon the making of such Construction Loan.

3.4 <u>Conversion Dates</u>. The occurrence of the Facility A Conversion Date and the Facility B Conversion Date shall be subject to the conditions precedent that the Administrative Agent shall have received, or the Required Lenders shall have waived receipt of, the following, each of which shall be in form and substance reasonably satisfactory to the Required Lenders in their sole discretion, and that the other conditions set forth in this <u>Section 3.4</u> shall have been satisfied or waived by the Required Lenders:

- (a) <u>Term Notes</u>. If a Lender shall request, such Lender shall have received an original Tranche A Term Note or Tranche B Term Note, as applicable, in respect of the Term Loans made or maintained by it, duly completed, executed and delivered by the applicable Borrower, which shall (i) be dated the Facility A Conversion Date or the Facility B Conversion Date, as applicable, (ii) mature on the Tranche A Final Maturity Date or the Tranche B Final Maturity Date, as applicable, and (iii) bear interest as provided in <u>Section 2.11</u> hereof.
- (b) <u>Representations and Warranties</u>. The representations and warranties of each Borrower contained in <u>Section 4</u> hereof and the representations and warranties of each Finance Party, each Affiliate Project Participant and, to the knowledge of the applicable Borrower, each other Project Participant (other than any Lender Affiliate Project Participant or any Immaterial Project Participant) contained in any other Finance Document or Project Document then in effect shall be true and correct in all material respects on and as of such Conversion Date as if made on and as of such date (or, if expressly stated to have been made solely as of an earlier date, were true and correct in all material respects as of such date).
- (c) <u>Insurance</u>. The Administrative Agent shall have received (i) a certified copy of the insurance policies required by <u>Section 5.9</u> hereof or certificates of insurance with respect thereto, together with evidence of the payment of all premiums therefor, (ii) a certificate

of the Insurance Advisor, certifying that insurance complying with <u>Section 5.9</u> hereof, covering the risks referred to therein, has been obtained, and (iii) a letter from a nationally-recognized insurance broker stating that insurance complying with <u>Section 5.9</u> hereof is in full force and effect.

- (d) No Default. No Default or Event of Default shall have occurred and be continuing or would result from the occurrence of such Conversion Date and no default by any Finance Party, any Affiliate Project Participant or, to the knowledge of the applicable Borrower, any other Project Participant under any of the Transaction Documents then in effect shall have occurred and be continuing or would result from the occurrence of such Conversion Date.
- (e) <u>Material Adverse Effect</u>. There shall exist no circumstance, event or condition which has had or could reasonably be expected to have a Material Adverse Effect.
- (f) <u>Governmental Approvals</u>. All Necessary Governmental Approvals required for the construction and operation of the applicable Project shall have been duly obtained and shall be final and in full force and effect and either the applicable administrative and statutory appeal periods shall have expired or those applicable administrative and statutory appeal periods greater than ninety (90) days that have not expired and their relevant expiration dates are identified on <u>Schedule 4.6</u>.
- (g) <u>Financial Covenants; Updated Base Case Projections.</u> After giving pro forma effect to the Borrowing of Term Loans to be made on such date, the Borrower shall be in compliance with <u>Section 7</u> hereof and the Administrative Agent shall have received an updated version of the Base Case Projections for the applicable Project, which shall project (a) a Tranche A Debt Service Coverage Ratio or Tranche B Debt Service Coverage Ratio, as applicable, of not less than 1.25:1.00 as of the end of each fiscal quarter during the period covered thereby and (b) that no more than twenty percent (20%) of the original principal amount Tranche A Term Loans or Tranche B Term Loans shall be outstanding as of the Tranche A Final Maturity Date and the Tranche B Final Maturity Date, as applicable, in each case prepared using assumptions provided in good faith by such Borrower, certified as true, complete and correct by an Authorized Officer of such Borrower and the Project Engineer and otherwise in form and substance reasonably satisfactory to each Lender and the Project Engineer.
- (h) <u>Completion Certificates</u>. The Administrative Agent shall have received (i) an executed Borrower Completion Certificate (the statements contained in which shall be true and correct in all material respects) and (ii) an executed Project Engineer Completion Certificate.
- (i) <u>Project Completion Date</u>. The Facility A Project Completion Date or the Facility B Project Completion Date, as applicable, shall have occurred.
- (j) Officer's Certificates. The Administrative Agent shall have received a executed certificate of an Authorized Officer of the Borrowers, dated such Conversion Date, certifying that (i) the representations and warranties made by each Borrower in Section 4 hereof and the representations and warranties made by each Finance Party, each affiliate Project Participant and, to the knowledge of the applicable Borrower, each other Project Participant (other than any Lender Affiliate Project Participant or any Immaterial Project Participant) in each

66

of the other Finance Documents and the Project Documents then in effect are true and correct in all material respects on and as of such Conversion Date with the same force and effect as if made on and as of such date (or, if stated to have been made solely as of an earlier date, were true and correct in all material respects as of such date), (ii) no Default or Event of Default has occurred and is continuing on such Conversion Date, and (iii) no default by either Borrower, any Affiliate Project Participant or, to the knowledge of the applicable Borrower, any other Project Participant under any of the Transaction Documents then in effect has occurred and is continuing on such Conversion Date.

- (k) <u>Opinions</u>. The Administrative Agent shall have received such supplemental written opinions of counsel to each Finance Party and the Affiliate Project Participants as the Administrative Agent may reasonably request.
- (l) <u>Title Insurance</u>. The Collateral Agent shall have received (i) a final down date endorsement to the Title Insurance, in favor of the Collateral Agent, for the benefit of the Secured Parties, in form, scope and substance satisfactory to the Collateral Agent and each Lender, and showing no exceptions other than those set forth in the Title Insurance, and deleting any pending disbursement provisions that may have been contained in the Title Insurance issued pursuant to <u>Section 3.2(g)(iii)</u> and (ii) an as-built ALTA survey of the applicable Site certified to the Administrative Agent, the Title Insurance Company and the applicable Borrower, which survey shall be in form and substance satisfactory to the Administrative Agent and the Title Insurance Company.
- (m) <u>Debt Service Reserve Account.</u> As of the Conversion Date the amounts on deposit in the applicable Debt Service Reserve Account shall equal or exceed the applicable Required Debt Service Reserve Amount.

- (n) <u>Phase I</u>. The Administrative Agent shall have received a new Phase I Environmental Site Assessment with respect to the applicable Site, in form, scope, and substance reasonably acceptable to the Administrative Agent.
- (o) Other Documents. The Administrative Agent shall have received such other statements, certificates, documents and approvals as the Administrative Agent may reasonably request.

#### SECTION 4. REPRESENTATIONS, WARRANTIES AND AGREEMENTS

In order to induce each of the Lenders to enter into this Agreement and to make the Loans, the Borrowers make the following representations, warranties and agreements, all of which shall survive the execution and delivery of this Agreement and the Notes:

4.1 <u>Organization</u>. Each Borrower is a limited liability company duly incorporated, validly existing and in good standing under the laws of the State of Delaware. The correct legal name of the Tranche A Borrower is "Clean Energy Tranche A LNG Plant, LLC". The correct legal name of the Tranche B Borrower is "Clean Energy Tranche B LNG Plant, LLC". Each Borrower is duly authorized and qualified to do business and is in good standing in each jurisdiction in which it owns or leases Property or in which the conduct of its business requires it so to qualify. Each Borrower has the requisite power and authority to own or lease and operate

67

its Properties, to carry on its business (including with respect to the Projects), to borrow money, to create the Security Interests as contemplated by the Security Documents and to execute, deliver and perform each Transaction Document (including the Notes) to which it is or will be a party.

#### 4.2 <u>Authority and Consents</u>.

- (a) The execution, delivery and performance by each Borrower of each Transaction Document to which it is or will be a party, and the transactions contemplated by the Transaction Documents (i) have been duly authorized by all necessary limited liability company action (including any necessary member or manager action), (ii) will not breach, contravene, violate, conflict with or constitute a default under (A) any of its Charter Documents, (B) any applicable Law, or (C) any contract, loan, agreement, indenture, mortgage, lease or other instrument to which it is a party or by which it or any of its Properties may be bound or affected, including all Governmental Approvals and the Transaction Documents (except in the case of clauses (B) and (C) such as would not reasonably be expected to have a Material Adverse Effect), and (iii) except for the Liens created by the Security Documents, will not result in or require the creation or imposition of any Lien upon or with respect to any of the Properties of either Borrower.
- (b) Each Transaction Document (i) has been duly executed and delivered by each Borrower and (ii) when executed and delivered by each of the other parties thereto will be the legal, valid and binding obligation of each Borrower, enforceable against each such Borrower in accordance with its terms, except as the enforceability thereof may be limited by (A) applicable bankruptcy, insolvency, moratorium or other similar Laws affecting the enforcement of creditors' rights generally and (B) the application of general principles of equity (regardless of whether such enforceability is considered in a proceeding at law or in equity).
- (c) No authorization, consent or approval of, or notice to or filing with, any Governmental Authority or any other Person has been, is or will be required to be obtained or made (i) for the due execution, delivery, recordation, filing or performance by either Borrower of any of the Transaction Documents to which it is a party or any transaction contemplated by the Transaction Documents (other than any Necessary Governmental Approvals that will be obtained by the time needed in connection with the construction or operation of the Projects), (ii) for the grant by either Borrower, or the perfection and maintenance, of the Liens contemplated by the Security Documents then required to have been delivered (including the first priority nature thereof) (subject only to (i) Permitted Liens and (ii) the priority afforded to such Permitted Liens by operation of Law), except, in each case in this clause (ii), those which cannot be made or obtained on the date this representation is made and which are reasonably expected to be able to made or obtained at the appropriate future dates or (iii) for the exercise by the Administrative Agent or any other Secured Party of any of its rights under any Finance Document or any remedies in respect of the Collateral pursuant to the Security Documents, except, in each case in this clause (iii), those which cannot be made or obtained on the date this representation is made and which are reasonably expected to be able to made or obtained at the appropriate future dates, except, in each case in clauses (i), (ii) and (iii), for the authorizations, consents, approvals, notices and filings that have been duly obtained, taken, given or made and are in full force and effect.

68

#### 4.3 <u>Capitalization; Indebtedness; Investments</u>.

- (a) Schedule 4.3 (as may be updated by the Borrowers from time to time by written notice to the Administrative Agent) contains a true, correct and complete list of all of the authorized and outstanding Capital Stock of the Borrowers by class. All of the Capital Stock of the Borrowers has been duly authorized and validly issued and is fully paid and non-assessable. None of such Capital Stock has been issued in violation of any applicable Law. Other than the Finance Documents, neither Borrower is a party or subject to, has outstanding or is bound by, any subscriptions, options, warrants, calls, agreements, preemptive rights, acquisition rights, redemption rights or any other rights or claims of any character that restrict the transfer of, require the issuance of, or otherwise relate to any shares of its Capital Stock. The Capital Stock of each Borrower is owned beneficially and of record by the Persons set forth on Schedule 4.3 hereto (as may be updated by the Borrowers from time to time by written notice to the Administrative Agent). Except for the Liens created by the Pledge Agreement, there is no Lien on any of the Capital Stock of either Borrower, and neither Borrower has been notified of the assignment of all or any part of Holdings' Investments in such Borrower other than the assignment in favor of the Collateral Agent pursuant to the Pledge Agreement.
- (b) As of the Closing Date, (i) neither Borrower has any Indebtedness of any nature, whether due or to become due, absolute, contingent or otherwise, other than Indebtedness arising under the Finance Documents and (ii) neither Borrower holds any Investments other than Investments permitted by **Section 6.4**.

## 4.4 Financial Condition.

- (a) Neither Borrower has any outstanding obligations or liabilities, fixed or contingent, except those arising under the Transaction Documents or permitted under this Agreement. Since December 31, 2011, no event, condition or circumstance has occurred which could reasonably be expected to have a Material Adverse Effect.
  - (b) Each Borrower is Solvent.
- 4.5 <u>Litigation; Labor Disputes</u>. There is no action, suit, other legal proceeding, arbitral proceeding, inquiry or investigation pending or, to the best of each Borrower's knowledge, threatened, by or before any Governmental Authority or in any arbitral or other forum, nor any order, decree or judgment in effect, pending, or, to the best of each Borrower's knowledge, threatened, against or affecting (a) either Borrower or any of its Properties or rights or (b) any Affiliate Project Participant and to each Borrower's knowledge any other Project Participant (other than any Immaterial Project Participant) or any of its Properties or rights, that, in the case of this clause (b), relates to the Projects, to any of the Transaction Documents or to any of the transactions contemplated thereby. There are no ongoing, or, to the knowledge of each Borrower, currently threatened, strikes, slowdowns or work stoppages by the employees of either Borrower, any Contractor, any Approved Gas Supplier, any Approved Power Supplier or the LNG Purchaser.

#### 4.6 <u>Governmental Approvals</u>.

- (a) All Governmental Approvals necessary in connection with (i) the due execution and delivery of, and performance by each Borrower, each Affiliate Project Participant and, to each Borrower's knowledge, each other Project Participant (other than any Immaterial Project Participant) of their respective obligations, and the exercise of their respective rights under, the Transaction Documents, (ii) the legality, validity and binding effect or enforceability thereof, and (iii) on and after the applicable Initial Funding Date, the design, development, ownership, construction, installation, operation and maintenance of each Project as contemplated by the Transaction Documents (collectively, the "Necessary Governmental Approvals"), are set forth on Schedule 4.6 hereto and, except for those set forth in Part B of Schedule 4.6 hereto, each such Necessary Governmental Approval has been duly obtained or made, was validly issued, is in full force and effect, is final and not subject to modification or appeal, is held in the name of the applicable Borrower (except as specifically indicated in such Schedule) and is free from conditions or requirements the compliance with which could reasonably be expected to have a Material Adverse Effect and which the applicable Borrower does not reasonably expect to be able to satisfy. No event has occurred that could reasonably be expected (A) to result in the revocation, termination or adverse modification of any such Necessary Governmental Approval or (B) adversely to affect any rights of the Borrower (or, as applicable, any Project Participant) under any such Necessary Governmental Approval.
- (b) As of the applicable Initial Funding Date, the Governmental Approvals set forth in Part B of Schedule 4.6 hereto are not required for the current stage of installation and construction or operation and maintenance of the Projects and are not customarily obtained until a later stage of installation and construction or after operation of the Projects has commenced. Neither Borrower has any reason to believe that any Necessary Governmental Approvals which are not obtainable or not required to have been obtained by such Borrower as of the date of this Agreement, but which will be required in the future (including those set forth in Part B of Schedule 4.6 hereto), will not be granted in due course prior to the time when needed free from conditions or requirements which such Borrower does not reasonably expect to be able to satisfy or compliance with which could reasonably be expected to have a Material Adverse Effect.
- (c) On and after the applicable Initial Funding Date, the information set forth in each application submitted by or on behalf of each Borrower in connection with each Necessary Governmental Approval and in all correspondence sent by or on behalf of each Borrower in respect of each such application is accurate and complete in all material respects.
- (d) On and after the applicable Initial Funding Date, the applicable Facility, if installed, constructed, owned and operated in accordance with the Plans and Specifications and the Transaction Documents, will conform to and comply in all material respects with all covenants, conditions, restrictions and requirements in all Necessary Governmental Approvals, in the Transaction Documents applicable thereto, and under all zoning, environmental, land use and other Laws applicable thereto.

70

#### 4.7 <u>Margin Stock, Etc.</u>

- (a) Neither Borrower is engaged principally, or as one of its important activities, in the business of extending credit for the purpose, whether immediate, incidental or ultimate, of buying or carrying Margin Stock and no part of the proceeds of any Loan will be used to purchase or carry any Margin Stock.
- (b) Neither the making of any Loan nor the use of the proceeds thereof will violate or be inconsistent with the provisions of Regulation U or Regulation X.
- 4.8 <u>ERISA Compliance</u>. Neither Borrower nor Holdings nor any Subsidiary of either Borrower maintains or contributes to or has ever maintained or contributed to (or has or has ever had an obligation to maintain or contribute to) any Plan subject to Section 412 of the Code or Multiemployer Plan. No ERISA Affiliate of either Borrower maintains or contributes to or has within the prior six years maintained or contributed to or had an obligation to maintain or contribute to any Plan subject to Section 412 of the Code or a Multiemployer Plan, in either case, that could reasonably be expected to result in a Material Adverse Effect.

#### 4.9 <u>Taxes</u>.

(a) Each Borrower has timely filed all Federal and state tax returns and all material local and foreign income tax returns, and all other material tax and informational returns which are or were required to be filed by it, if any, and has paid all material Taxes owed by it (whether or not shown on any tax return) when due, except such Taxes, if any, as are being contested in good faith and by proper proceedings and as to which adequate reserves have been provided in accordance with GAAP. No claim for assessment or collection of taxes has been asserted against either Borrower or its Properties by any Governmental Authority except such taxes as are being contested in good faith and by proper proceedings and as to which adequate reserves have been provided in accordance with GAAP. Each Borrower is an entity that is disregarded as separate from its owner within the meaning of Treasury regulation

section 301.7701-3. Each set of Base Case Projections accurately reflect in all material respects all Taxes that, under present Law, will be due and payable by the applicable Borrower assuming that such Borrower has the income and expenses reflected in such Base Case Projections.

(b) No liability for any Tax (other than customary filing and recording fees) will be incurred by either Borrower as a result of the execution, delivery or performance of this Agreement or any other Finance Document.

#### 4.10 Investment Company Act; Public Utility Holding Company Act

(a) Neither Borrower is an "investment company," or an "affiliated person" of, or "promoter" or "principal underwriter" for, an "investment company," as such terms are defined in the Investment Company Act of 1940, as amended. Neither the making of any Disbursement, nor the application of the proceeds or repayment thereof by either Borrower, nor the consummation of the other transactions contemplated hereby will violate any provisions of the Investment Company Act or any rule, regulation or order of the U.S. Securities and Exchange Commission thereunder.

71

(b) Neither Borrower is a "holding company" as defined in, or subject to regulation under, the Public Utility Holding Company Act of 1935.

#### 4.11 <u>Title; Security Documents</u>.

- (a) On and after the Initial Funding Date in respect of a Tranche, the applicable Borrower owns and has good and marketable title to or leasehold interests in the applicable Facility and the applicable Site, in each case free and clear of all Liens other than Permitted Liens.
- (b) Each Borrower has good and marketable title to all of the Property purported to be owned by it, free and clear of all Liens, other than Permitted Liens, and holds such title and all of such Property in its own name and not in the name of any nominee or other Person. Each Borrower is lawfully possessed of a valid and subsisting leasehold estate in and to all Property which it purports to lease, free and clear of all Liens, other than Permitted Liens, and holds such leaseholds in its own name and not in the name of any nominee or other Person. Neither Borrower has created, or is contractually bound to create, any Lien on or with respect to any of its assets, Properties, rights or revenues, except for Permitted Liens, and, except for this Agreement, neither Borrower is restricted by contract, law or otherwise from creating Liens on any of its Properties.
- (c) All Property owned, leased or otherwise used by each Borrower is located on the Sites or, prior to the applicable Initial Funding Date, at the addresses listed on **Schedule 4.11**.
- (d) The provisions of the Security Documents then required to have been delivered are effective to create, in favor of the Collateral Agent, for the benefit of the Secured Parties, legal, valid and enforceable Liens on or in all of the Collateral intended to be covered thereby, and all necessary recordings and filings have been made in all necessary public offices and all other necessary and appropriate action has been taken so that the Liens created by each Security Document constitute perfected Liens on or in the Collateral intended to be covered thereby, prior and superior to all other Liens, and all necessary consents to the creation, effectiveness, priority and perfection of each such Lien have been obtained. No mortgage or financing statement or other instrument or recordation covering all or any part of the Collateral is on file in any recording office, except such as may have been filed in favor of the Secured Parties or in respect of Permitted Liens.

## 4.12 Environmental Matters.

- (a) Each Borrower has complied and is now complying in all material respects with (i) all Environmental Laws applicable to the Projects and (ii) the requirements of any Governmental Approvals issued under such Environmental Laws with respect to the Projects.
- (b) Each Facility is designed and will be constructed and operated in such a manner that it will comply in all material respects with (i) all applicable Environmental Laws, and (ii) the requirements of any Governmental Approvals issued under such Environmental Laws with respect to such Facility (the representations in this clause (b) only being made on and after the Initial Funding Date relating to each Facility, as applicable).

- (c) To the knowledge of each Borrower, there are no facts, circumstances, conditions or occurrences regarding the Projects that (i) could reasonably be anticipated to form the basis of a material Environmental Claim against either Project, either Borrower or any other Project Participant (other than an Immaterial Project Participant), or (ii) could reasonably be anticipated to cause the Sites to be subject to any material restrictions on its ownership, occupancy, use or transferability under any Environmental Law.
- (d) There are no past, pending, or, to the knowledge of each Borrower, threatened, material Environmental Claims with respect to the Projects against (i) either Borrower or either Project; or (ii) to the knowledge of each Borrower, any other Project Participant (other than any Lender Affiliate Project Participant or any Immaterial Project Participant) or any other Person occupying, using, or conducting operations on or about the Sites.
- (e) To the knowledge of each Borrower, Hazardous Materials have not at any time been generated, used, treated, recycled, stored on, or transported to or from, or Released, deposited or disposed of on all or any portion of the Sites other than in material compliance with all applicable Environmental Laws.
- (f) There are not now and, to the knowledge of each Borrower (after due inquiry), never have been any underground storage tanks located on the Sites, there is no asbestos contained in, forming part of, or contaminating any part of the Projects, and no polychlorinated biphenyls (PCBs) or radioactive materials are used, stored, located at or contaminate any part of the Projects (the representations in this clause (f) only being made on and after the Initial Funding Date relating to each Facility, as applicable).
- (g) To the knowledge of each Borrower, there is no soil or groundwater contamination in, on, or under the Sites that requires remediation pursuant to Environmental Laws, and any soil or groundwater remediation that has previously occurred at the Sites has been determined by the

applicable Governmental Authorities to be completed.

- (h) Copies of all final reports, audits, studies, and material, non-privileged correspondence with Governmental Authorities regarding compliance with Environmental Laws or the presence of Hazardous Materials at the Project and/or the Sites that either Borrower is aware or has in its possession have been delivered to the Administrative Agent.
  - 4.13 <u>Subsidiaries</u>. Neither Borrower has any Subsidiaries or beneficially owns any Capital Stock or other ownership interest of any other Person.
- 4.14 <u>Intellectual Property.</u> Each Borrower owns or has the right to use all material patents, trademarks, permits, service marks, trade names, copyrights, franchises, formulas, intellectual property licenses and other intellectual property rights with respect thereto, and has obtained assignment of all licenses and other rights of whatsoever nature materially necessary for the design, development, construction, operation, maintenance or ownership of the Projects and the operation of its business as currently contemplated without any material conflict with the rights of others. No material product, process, method, substance, part or other material sold or employed or presently contemplated to be sold by or employed by either Borrower in connection with its business infringes or will infringe any patent, trademark, permit, service mark, trade name, copyright, franchise, formula, license or other intellectual property right.

73

4.15 <u>Insurance</u>. On and after the applicable Initial Funding Date, all Insurance required by <u>Section 5.9</u> has been obtained and is in full force and effect.

#### 4.16 <u>Project Documents</u>.

- (a) On and after the Initial Funding Date with respect to a Tranche, the Project Documents constitute all contracts, agreements, side letters, leases, powers of attorney or other instruments or documents that are necessary (other than utility services necessary for the construction and the operation of the Project for its intended purposes that are available at the Project or can reasonably be expected to be so available as and when required upon commercially reasonable terms, provided, that in any event the Borrower shall have entered into a Gas Supply Agreement and a Power Purchase Agreement) for (A) the construction, completion, operation, maintenance and ownership of the applicable Project and (B) the conduct of the business of the applicable Borrower as contemplated by the Transaction Documents. Each Project Document has been duly authorized, executed and delivered by each Borrower, each Affiliate Project Participant and, to the knowledge of the Borrower, each other Project Participant (other than any Lender Affiliate Project Participant and any Immaterial Project Participant) party thereto, is in full force and effect and is binding upon and enforceable against each Borrower, each Affiliate Project Participant and, to the knowledge of each Borrower, each other Project Participant (other than any Lender Affiliate Project Participant and any Immaterial Project Participant) party thereto in accordance with its terms. Each Borrower, each Affiliate Project Participant and, to the knowledge of each Borrower, each other Project Participant (other than any Immaterial Project Documents to which it is a party, and no event has occurred that could reasonably be expected to (1) result in an event of default under, or a material breach of, any Project Document (other than any Immaterial Additional Project Document), (2) result in the revocation, termination or adverse modification of any Project Document (other than any Immaterial Additional Project Document), or (3) adversely affect the rights of either Borrower unde
- (b) All representations and warranties of the Borrowers, the Affiliate Project Participants and, to the knowledge of each Borrower, the other parties thereto (other than any Lender Affiliate Project Participant and any Immaterial Project Participant) contained in the Project Documents (other than any Immaterial Additional Project Document) are true and correct in all material respects (except to the extent that any such representation or warranty is expressed to be made only as of an earlier date, in which case such representation or warranty was true and correct in all material respects on and as of such earlier date).
- (c) All conditions precedent to the obligations of the respective parties under the Project Documents (other than any Immaterial Additional Project Document) have been satisfied or waived, except for such conditions precedent which by their terms cannot be (and are not required to be) met until a later stage in the construction or operation of the applicable Project, and neither Borrower has any reason to believe that any such conditions precedent cannot be satisfied prior to the time when such conditions are required to be met pursuant to the applicable Project Documents (other than any Immaterial Additional Project Document).

74

- (d) Neither Borrower is a party to any agreement or contract other than the Transaction Documents. As of the Closing Date, each of the Project Documents then in effect consists only of the original document (including exhibits and schedules) and the amendments thereto expressly described in the relevant definitions appearing in Section 1.1 hereto, and there are no other agreements, side letters, amendments or waivers or supplements, written or oral, with respect thereto. The Administrative Agent has received a true, correct and complete copy of each Project Document (other than any Immaterial Additional Project Document), including all exhibits, schedules and disclosure letters referred to therein or delivered pursuant thereto, if any. None of the Project Documents has been amended or modified except as permitted under this Agreement.
  - 4.17 No Default. No Default or Event of Default has occurred and is continuing.
- 4.18 <u>Compliance with Laws</u>. Neither Borrower is in violation of any Law, Governmental Approval, order, writ, injunction or decree or its Charter Documents.

## 4.19 <u>Disclosure</u>.

(a) All documents, reports or other information pertaining to the Borrowers, the Project Documents (other than any Immaterial Additional Project Document), or the Project Participants (other than any Lender Affiliate Project Participant and any Immaterial Project Participant) that have been furnished to the Administrative Agent or any Lender by or on behalf of the Borrowers, Holdings, the Affiliate Project Participants or the Sponsor (including (i) any application to any Lender for the extensions of credit provided for in the Finance Documents, (ii) the Finance Documents,

including the exhibits and schedules attached thereto, and (iii) all other information relating to the Borrowers, the Projects, the Project Documents, or the Project Participants provided by the Borrowers, Holdings, the Affiliate Project Participants or the Sponsor to the Administrative Agent or any Lender, but excluding the Base Case Projections, the Construction Budgets and any other forecasts and projections), in light of the circumstances under which they were or are made and taken as a whole, are true and correct in all material respects and do not contain any material misstatement of fact or omit to state a material fact or any fact necessary to make the statements contained herein or therein not materially misleading. There is no fact, event or circumstance known to the Borrowers that has not been disclosed to the Administrative Agent in writing, the existence of which could reasonably be expected to have a Material Adverse Effect.

(b) Each Construction Budget accurately specifies all costs and expenses incurred and the applicable Borrower's best estimate of all costs and expenses anticipated by such Borrower to be incurred on or prior to the applicable Conversion Date to construct and finance the construction of the applicable Facility and to implement the applicable Project in the manner contemplated by the Transaction Documents. Each Construction Budget and each set of Base Case Projections (i) are, as of the Initial Funding Date, based on reasonable assumptions as to all legal and factual matters material to the estimates set forth therein, (ii) are, as of the Initial Funding Date, consistent with the provisions of the Transaction Documents in all material

75

respects, (iii) have been prepared in good faith and with due care, and (iv) fairly represent each Borrower's reasonable expectations as to the matters covered thereby as of the Initial Funding Date. All projections and budgets to be furnished to the Lenders by or on behalf of the Borrower after the Initial Funding Date (A) will be based on reasonable assumptions as to all legal and factual matters material to the estimates set forth therein, (B) will be consistent with the provisions of the Transaction Documents in all material respects, (C) will be prepared in good faith and with due care and (D) will fairly represent each Borrower's reasonable expectations as to the matters covered thereby as of their respective dates.

- 4.20 <u>Utilities, Etc.</u> All utility services, means of transportation, facilities and other materials necessary for the importation, construction, installation, operation and maintenance of each Facility (including gas, electrical, potable and raw water supply, storm, telephone and sewage services and facilities, as necessary) are or will be available to the Projects (in the case of utility services, at the boundaries of each Site) when necessary for the construction, operation, testing and start-up of each Facility and, to the extent necessary or desirable, arrangements have been made on commercially reasonable terms for such services, means of transportation, facilities and other materials, in each case on terms consistent with those reflected in the applicable Construction Budget and the applicable Base Case Projections.
- 4.21 <u>Transactions with Affiliates</u>. As of the Closing Date, other than pursuant to the LNG Purchase Agreements and as otherwise set forth on <u>Schedule 4.21</u> hereto, neither Borrower has engaged or has agreed to engage in any transactions (including any transactions relating to the buying or selling of any Properties or any products of the Projects or involving the receipt of money as payment for goods or services) with any Affiliate of the Borrower.
  - 4.22 <u>Project Completion Date; Project Costs.</u>
- (a) The Tranche A Borrower estimates, in good faith, that the Facility A Project Completion Date will occur no later than the scheduled date for completion in accordance with the Facility A Project Schedule and that the aggregate proceeds of the Tranche A Construction Loans and the Equity Contributions to be received by the Tranche A Borrower will be sufficient to achieve the Facility A Project Completion Date.
- (b) The Tranche B Borrower estimates, in good faith, that the Facility B Project Completion Date will occur no later than the scheduled date for completion in accordance with the Facility B Project Schedule and that the aggregate proceeds of the Tranche B Construction Loans and the Equity Contributions to be received by the Tranche B Borrower will be sufficient to achieve the Facility B Project Completion Date.
- (c) As of the Initial Funding Date, no Change Order has been proposed and no Change Order is being contemplated for proposal in the future by the Borrowers, or, to the best knowledge of the Borrowers, by any Contractor or any other party pursuant to any Construction Agreement.
- 4.23 <u>Single-Purpose Entity</u>. Neither Borrower has engaged in any business (including any speculative trading) other than the Projects. The Tranche A Borrower has offices at 3020 Old Ranch Parkway, Seal Beach, CA 90740 and does not have a place of business at any other location other than Site A (following the acquisition thereof). The Tranche B Borrower has offices at 3020 Old Ranch Parkway, Seal Beach, CA 90740 and does not have a place of business at any other location other than Site B (following the acquisition thereof).

76

4.24 <u>Anti-Terrorism Laws</u>. Neither Borrower nor any of its Affiliates is in violation of any Laws relating to "know your customer," terrorism or money laundering, including Executive Order No. 13224 on Terrorist Financing, effective September 24, 2001, the Patriot Act, the Terrorism Sanctions Regulations (Title 31 Part 595 of the U.S. Code of Federal Regulations), the Terrorism List Governments Sanctions Regulations (Title 31 Part 596 of the U.S. Code of Federal Regulations), the Foreign Terrorist Organizations Sanctions Regulations (Title 31 Part 597 of the U.S. Code of Federal Regulations) and the laws administered by the United States Treasury Department's Office of Foreign Asset Control (each as from time to time in effect) (the "<u>Anti-Terrorism Laws</u>").

## SECTION 5. <u>AFFIRMATIVE COVENANTS</u>

Each Borrower covenants and agrees with each of the Lenders that, so long as any Commitment or any Loan or any other Obligation is outstanding and until payment in full of all Obligations and any other amounts payable by each Borrower under the Finance Documents:

- 5.1 <u>Financial Statements and Other Information</u>. Each Borrower shall deliver or cause to be delivered to the Administrative Agent and each Lender:
- (a) <u>Quarterly Financial Statements</u>. As soon as available and in any event within sixty (60) days after the end of each fiscal quarter of each of Holdings (beginning, in the case of Holdings, with the fiscal quarter in which the first Initial Funding Date occurs) and the Sponsor, as applicable, a copy of the complete unaudited, consolidated (and in the case of Holdings, consolidating) statements of income, retained earnings and cash flow of such Person and the related unaudited, consolidated (and in the case of Holdings, consolidating) balance sheet of such Person as at the end of such fiscal quarter,

setting forth in each case in comparative form the corresponding figures for the corresponding fiscal quarter in the preceding fiscal year, if any, accompanied by a certificate of the chief financial officer of such Person, which certificate shall state that said financial statements fairly present the financial condition and results of operations of such Person in accordance with GAAP, consistently applied, as at the end of, and for, such fiscal quarters (subject to normal year-end audit adjustments and the absence of footnotes).

(b) Annual Financial Statements. As soon as available and in any event within one hundred twenty (120) days after the end of each fiscal year of each of Holdings (beginning, in the case of Holdings, with the fiscal year in which the first Initial Funding Date occurs) and the Sponsor, as applicable, a copy of the complete audited, consolidated (and in the case of Holdings, consolidating) statements of income, retained earnings and cash flow of such Person and the related audited, consolidated (and in the case of Holdings, consolidating) balance sheets of such Person as at the end of such year and any related audit letter, setting forth in each case in comparative form the corresponding figures for the preceding fiscal year, and accompanied by an opinion thereon of KPMG LLP or such other firm of independent certified public accountants of recognized international standing as may be acceptable to the Required Lenders, which opinion shall state that said financial statements fairly present the financial condition and results of operations of such Person as at the end of, and for, such fiscal year in accordance with GAAP and shall not include any going concern or like qualification or exception.

77

(c) Officer's Certificate. At the time the Sponsor furnishes each set of its financial statements pursuant to Section 5.1(a) or (b) above, a certificate of the chief financial officer (or other senior financial officer) of each Borrower to the effect that no Default or Event of Default has occurred and is continuing (or, if any Default or Event of Default has occurred and is continuing, describing the same in reasonable detail and describing what action such Borrower has taken and proposes to take with respect thereto).

## (d) Financial Ratios.

- (i) Following the Facility A Conversion Date, at the time it furnishes each set of financial statements pursuant to **Section 5.1(a)** above, the Tranche A Borrower shall provide a calculation of:
  - (A) the Tranche A Debt Service Coverage Ratio as of the Quarterly Date most recently falling prior to such date, certified by the chief financial officer of the Tranche A Borrower, together with supporting data in reasonable detail;
  - (B) the Tranche A Projected Debt Service Coverage Ratio as of the Quarterly Date most recently falling prior to such date, certified by the chief financial officer of the Tranche A Borrower, together with supporting data in reasonable detail;
  - (C) the Tranche A Leverage Ratio as of the Quarterly Date most recently falling prior to such date, certified by the chief financial officer of the Tranche A Borrower, together with supporting data in reasonable detail.
- (ii) Following the Facility B Conversion Date, at the time it furnishes each set of financial statements pursuant to **Section 5.1(a)** above, the Tranche B Borrower shall provide a calculation of:
  - (A) the Tranche B Debt Service Coverage Ratio as of the Quarterly Date most recently falling prior to such date, certified by the chief financial officer of the Tranche B Borrower, together with supporting data in reasonable detail;
  - (B) the Tranche B Projected Debt Service Coverage Ratio as of the Quarterly Date most recently falling prior to such date, certified by the chief financial officer of the Tranche B Borrower, together with supporting data in reasonable detail;
  - (C) the Tranche B Leverage Ratio as of the Quarterly Date most recently falling prior to such date, certified by the chief financial officer of the Tranche B Borrower, together with supporting data in reasonable detail.

- (e) <u>Defaults</u>. Promptly after any officer or director of either Borrower knows or has a reasonable basis to believe that any Default or Event of Default or any default by any Project Participant (other than an Immaterial Project Participant) or either Borrower under any Transaction Document has occurred, a written notice of such event describing the same in detail satisfactory to the Administrative Agent and, together with such notice, a description of what action the Borrowers or such Project Participant have taken and propose to take with respect thereto.
- (f) Progress Reports. (x) As soon as available and in any event within ten (10) Business Days after the end of each month prior to the applicable Conversion Date, a construction progress report from the Project Engineer with respect to each Project for such month, which report shall correspond to the items and classifications and periods set forth in the applicable Construction Budget and the Facility A Project Schedule or the Facility B Project Schedule, as applicable, and (y) as soon as available and in any event within thirty (30) days after the end of each fiscal year of the Borrowers following the Facility A Project Completion Date or the Facility B Project Completion Date, as applicable, an operating report from the Project Engineer with respect to the applicable Project for such fiscal year, which report shall certify that such Project has been operated and maintained substantially in accordance with Prudent Industry Practices, the Necessary Governmental Approvals and the Transaction Documents (including Section 5.18 hereof).
- Operating Reports. As soon as available and in any event within forty-five (45) days after the end of each fiscal quarter of each Borrower following the Facility A Project Completion Date or the Facility B Project Completion Date, as applicable, an operating report with respect to each Project for such fiscal quarter and for the portion of the Operating Year then ended, which report shall (i) include the applicable Borrower's production and sales volumes as well as other operating metrics as the Administrative Agent may reasonably request, (ii) correspond to the items and classifications and periods set forth in the applicable Operating Budget and shall show all Project Revenues, all expenditures for Operation and Maintenance Expenses and a reasonably detailed accounting of the use of any amounts transferred to the applicable Borrower from the applicable Operating Account and (iii) be certified as complete and correct in all material respects by an Authorized Officer of the applicable Borrower, which certificate shall also state that the Operation and Maintenance Expenses reflected therein complied with the requirements contained in Section 5.17(c) hereof, or, if any such certifications cannot be given, shall state in detail any necessary qualifications to such certifications.

- (h) Notices. Promptly after delivery or receipt thereof, a copy of each material notice, demand, report or other communication given or received by either Borrower (i) pursuant to or relating to any of the Transaction Documents (other than any Immaterial Additional Project Document) (including all requests for material amendments or waivers) or pursuant to or relating to any Necessary Governmental Approval, or (ii) to or from any Governmental Authority relating in any way to the Projects.
- (i) <u>Environmental Reports</u>. (x) Within sixty (60) days after the end of each year, a report summarizing the environmental performance of each Facility over the preceding year, which report shall include narrative summaries of (i) the results of any environmental monitoring or sampling activity that identified Hazardous Materials contamination, (ii) any

environmental remediation or clean up activities that have been conducted and any remaining remediation or clean up activities required by applicable Environmental Laws to address a Hazardous Materials Release required to be reported to a Governmental Authority, including the estimated cost to complete such remediation or clean up activities; (iii) any other incidents required to be reported pursuant to Environmental Laws, and (iv) any noncompliance with Environmental Law identified or alleged by any Governmental Authority; and (y) promptly after delivery or receipt thereof, copies of all final reports, audits, studies, and material, non-privileged correspondence with Governmental Authorities regarding compliance with Environmental Laws or the presence of Hazardous Materials at the Project and/or the Sites that Borrowers have in their possession.

- (j) <u>Capital Budget</u>. By February 28<sup>th</sup> of each fiscal year, a capital budget for each Borrower for such fiscal year.
- (k) <u>LNG Purchase Agreement Communications</u>. Promptly after delivery or receipt thereof, a copy of each notice, report or other communication given or received by either Borrower pursuant to the applicable LNG Purchase Agreement.
- (l) Other Information. From time to time such other information regarding the financial condition, operations, business or prospects of the Borrowers or the Projects or, to the extent obtainable by the Borrowers upon the exercise of their reasonable efforts, any Project Participant (other than any Lender Affiliate Project Participant or any Immaterial Project Participant), as may be requested by the Administrative Agent.
- (m) <u>Publicly Filed Information</u>. Notwithstanding the foregoing provisions of this <u>Section 5.1</u>, the filing of any of the financial statements referred to in <u>Sections 5.1(a)</u> or <u>(b)</u> with the Securities and Exchange Commission shall be deemed to satisfy any delivery requirement thereof.
- 5.2 <u>Other Notices</u>. Each Borrower shall promptly, but in any event no later than seven (7) Business Days after any officer or director obtains knowledge thereof, give to the Administrative Agent notice of:
- (a) Any pending or threatened application or proceeding by or before any Governmental Authority for the purpose of revoking, terminating, withdrawing, suspending, modifying or withholding any Necessary Governmental Approval;
- (b) Any litigation, investigation or proceeding affecting any Finance Party (other than the Sponsor) or the Projects in which the amount involved is \$250,000 or more in which injunctive, declaratory or similar relief is requested;
- (c) Any litigation, investigation or proceeding affecting the Sponsor or any Project Participant which if adversely determined could reasonably be expected to have a Material Adverse Effect;
- (d) The discovery of (i) any Hazardous Materials in, on, or under the Sites, (ii) any unpermitted Release of Hazardous Materials from, to, in, on, or under the Projects, (iii) any noncompliance with Environmental Law, or (iv) any other condition, fact, or circumstance with respect to the Projects that, in the case of (i), (ii), (iii), or (iv) could give rise to (x) liability under Environmental Law in an amount of \$250,000 or more, or (y) an Environmental Claim that, if resolved adversely, could result in liability of \$250,000 or more;

- (e) The initiation of any investigation, study, sampling, testing, remediation, removal, or cleanup activity intended to identify or address (i) an unpermitted Release of Hazardous Material from, to, in, or, or under the Projects; (ii) the presence of Hazardous Material in, on, or under the Projects; or (iii) any other environmental condition with respect to the Projects, where, in the case of either (i), (ii), or (iii) the investigation, study, sampling, testing, remediation, removal, or clean up activity could cost \$250,000 or more;
- (f) Any Environmental Claim threatened or asserted against or affecting (i) the Borrowers, Holdings or the Projects that, if resolved adversely, could result in liability of \$250,000 or more, or (ii) the Sponsor or the LNG Purchaser that, if resolved adversely, could reasonably be expected to have a Material Adverse Effect or, in each case, any material change or development in, or resolution or settlement of any action, investigation, suit, proceeding, audit, demand, or dispute involving such an Environmental Claim;
- (g) Any request by a Project Participant (other than any Immaterial Project Participant) for an arbitration proceeding under any Project Document (other than an Immaterial Additional Project Document);
- (h) Any (i) Taking or (ii) other casualty, damage or loss to any Property of the Borrowers, whether or not insured, through fire, theft, other hazard or casualty, in excess of \$100,000 for any one casualty or loss or \$1,000,000 in the aggregate in any calendar year;
- (i) Any (i) delay for any reason in any milestones for the construction of the Projects, (ii) unscheduled shutdown or reduction in operation of a Facility for a period in excess of seven (7) days, or (iii) substantial labor dispute which could lead to such a shutdown or reduction;
- (j) Any actual, proposed or threatened material cessation or suspension of the Work for any reason by any Contractor for a period in excess of five (5) days;
  - (k) The execution of any Material Additional Project Document, together with a copy of such Material Additional Project Document;

- (l) A violation of ERISA shall have occurred with respect to any Plan subject to Section 412 of the Code or any Multiemployer Plan that a Borrower or any ERISA Affiliate of a Borrower maintains or contributes to or has ever maintained or contributed to (or has or has ever had an obligation to maintain or contribute to) that could reasonably be expected to result, individually or in the aggregate, in a Material Adverse Effect or in a direct liability against a Borrower in excess of \$500,000;
  - (m) Any material change in accounting policies or financial reporting practices;

Я1

- (n) Any event constituting *force majeure* under any of the Project Documents (other than an Immaterial Additional Project Document) or any claim by any Project Participant (other than an Immaterial Project Participant) alleging that a *force majeure* event thereunder has occurred; and
  - (o) Any other event, circumstance, development or condition which could reasonably be expected to have a Material Adverse Effect.

Each notice pursuant to this <u>Section 5.2</u> shall be accompanied by a statement signed by an Authorized Officer of the Borrowers setting forth a description in reasonable detail of the occurrence referred to therein and stating what action the Borrowers propose to take with respect thereto.

- 5.3 <u>Maintenance of Existence; Conduct of Business</u>. Each Borrower shall (a) preserve and maintain its legal existence as a limited liability company under the laws of the State of Delaware and all of its material licenses, rights, privileges and franchises necessary for the maintenance of its corporate existence, and comply, in all material respects, with its Charter Documents, (b) not change its name, (c) engage solely in the business of constructing, owning, operating and maintaining the applicable Project and performing its obligations pursuant to the Transaction Documents to which it is a party, (d) not engage in any speculative trading and (e) not cancel, terminate, permit the cancellation or termination of, amend, modify or change any material terms or conditions of, or grant any material consent, waiver or approval under, or take or fail to take any other action that would impair the value of the interest or impair the rights of either Borrower under, any of its Charter Documents.
- 5.4 <u>Compliance with Laws</u>. Each Borrower shall conduct its business in material compliance with all applicable requirements of Law, including all relevant Governmental Approvals and Environmental Laws, except that each Borrower may, at its expense, contest by appropriate proceedings conducted in good faith the validity or application of any such requirement of Law, so long as (a) none of the Secured Parties or either Borrower could be subject to any criminal liability for failure to comply therewith, (b) all proceedings to enforce such requirement of Law against the Secured Parties, the Borrowers, the Project or any part thereof shall have been, to the extent applicable, duly stayed, and (c) such contest does not involve any risk of the sale, forfeiture or loss of any of the Collateral.
- 5.5 <u>Payment of Taxes, Etc.</u> Each Borrower shall duly pay and discharge before they become delinquent (a) all material Taxes levied or imposed upon its Property, earnings or business, (b) all material utility and other governmental charges incurred in the ownership, operation, maintenance, use, occupancy and upkeep of its business, and (c) all lawful claims (including mechanics liens) and obligations that, if unpaid, might result in the imposition of a Lien upon a material portion of its Property; *provided, however*, that the Borrowers may contest in good faith any such Taxes, charges, claims or obligations and, in such event, may permit the Taxes, charges, claims or obligations to remain unpaid during any period, including appeals, when the Borrowers are in good faith contesting the same, so long as (i) adequate reserves shall have been established in accordance with GAAP with respect to any such Taxes, charges, claims (including mechanics liens) or obligations, accrued interest thereon and potential penalties or other costs relating thereto, or other adequate provision for payment thereof satisfactory to the

82

Administrative Agent shall have been made, (ii) in the case of any mechanics', materialmen's, repairmen's, supplier's or other similar liens filed against the Projects or any part of the Work, such disputed amounts have been bonded for not less than the amount in dispute with a surety acceptable to the Administrative Agent, or other adequate provision for payment thereof which shall be satisfactory to the Administrative Agent shall have been made, (iii) such contest does not involve any risk of the sale, forfeiture or loss of any of the Collateral, and (iv) enforcement of the contested item shall be effectively stayed.

- 5.6 Accounting and Financial Management. Each Borrower shall (a) maintain adequate management information and cost control systems, (b) maintain a system of accounting in which full and correct entries shall be made of all financial transactions and the assets and business of such Borrower in accordance with GAAP, and (c) promptly deliver to the Administrative Agent a copy of any "management letter" or other similar communication received by such Borrower from such Borrower's accountants relating to such Borrower's financial, accounting and other systems, management or accounts. In the event that either Borrower replaces its existing auditors for any reason, such Borrower shall appoint and maintain as auditors another firm of independent public accountants, which firm shall be internationally recognized and approved by the Required Lenders.
  - 5.7 <u>Inspection</u>. Subject to requirements of any Governmental Authority and safety requirements:
- (a) Each Borrower shall permit, at the expense of such Borrower, representatives of the Secured Parties and the Project Engineer, during normal business hours and at such intervals as such Person shall desire, to visit and inspect the Projects to examine, copy and make extracts from its books and records, to inspect its Properties, and to discuss its business and affairs with its officers and engineers. Each Borrower will authorize its auditors (whose fees and expenses shall be for the account of such Borrower) to communicate directly with the officers and designated representatives of each Secured Party and the Project Engineer regarding its accounts and operations at any reasonable time upon prior written notice to such Borrower.
- (b) Each Borrower shall permit the Administrative Agent, the Project Engineer, and any other consultant engaged by the Administrative Agent to review (i) all Plans and Specifications, (ii) any quality control data and performance test data, and (iii) any other data relating to the Projects or to the progress of construction as may be reasonably requested by the Administrative Agent, the Project Engineer or such other consultant. Further, each Borrower shall permit the Administrative Agent, the Project Engineer and any other consultant engaged by the Administrative Agent to monitor, witness and review the Work.

(d) Notwithstanding anything to the contrary herein or in any other Transaction Document, no act or omission of the Administrative Agent, any other Secured Party, the Project Engineer or any other consultant engaged by any Lender (other than any act of gross negligence or willful misconduct) shall in any way (i) affect the obligations of the Borrowers, the Contractor or any other Person under any Transaction Document or any other contract relating to any Construction Agreement, (ii) be deemed to be the acceptance of any defective work performed by the Contractor or any other Person under any Construction Agreement, or (iii) be deemed to be a waiver of any rights against the Contractor or any other Person under any Construction Agreement or otherwise.

Notwithstanding the foregoing, Borrowers shall bear the costs of only two visits, inspections, audits or examinations per Project annually, unless an Event of Default has occurred and is continuing, in which case Borrowers shall bear the cost all such visits, inspections, audits or examinations as Administrative Agent may require in its sole discretion.

5.8 <u>Governmental Approvals</u>. Each Borrower shall (i) from time to time obtain and maintain, and comply with, all Necessary Governmental Approvals then required to have been obtained by such Borrower as shall now or hereafter be required under applicable Laws, (ii) cause the Projects to be duly constructed, completed, operated and maintained in all material respects in accordance with all applicable Laws, and (iii) intervene in and contest any proceeding which seeks, or may reasonably be expected, to rescind, to terminate, to modify or to suspend any Necessary Governmental Approval and, if reasonably requested by the Required Lenders, appeal any such rescission, termination, modification or suspension in the manner and to the full extent permitted by applicable Law; *provided* that the obligations of the Borrowers under this <u>Section 5.8</u> shall not in any way limit or impair the rights or remedies of the Secured Parties under any Finance Document directly or indirectly arising as a result of any such rescission, termination, modification or suspension.

#### 5.9 Insurance.

- Insurance Requirements. Each Borrower shall maintain or cause to be maintained in full force and effect at all times on and after the applicable Initial Funding Date (unless otherwise specified in **Appendix A**) and continuing throughout the term of this Agreement (unless otherwise specified in **Appendix A**) insurance coverages for the applicable Project meeting the requirements set forth in **Appendix A** (such **Appendix A** to be reasonably satisfactory to the Insurance Advisor and the Administrative Agent and prepared by such Persons in consultation with the applicable Borrower on or before the applicable Initial Funding Date considering the facts and circumstances as they then exist) with responsible insurance companies authorized to do business in the State in which such Project is located with (i) an AM Best Insurance Reports rating of "A" or better and a financial size category of "X" or higher or (ii) a Standard & Poor's financial strength rating of "A" or higher, or other companies acceptable to the Administrative Agent, with limits and coverage provisions sufficient to satisfy the requirements set forth in each of the Project Documents, but in no event less than the limits and coverage provisions set forth in **Appendix A**.
  - (b) <u>Endorsements</u>. Policies of general liability insurance required to be maintained shall be endorsed as follows:

84

- (i) to name the applicable Borrower and its officers and employees as named insureds, and to name the Secured Parties as additional insureds;
  - (ii) to provide a severability of interests and cross liability clause; and
- (iii) to provide that the insurance shall be primary and not excess to or contributing with any insurance or self-insurance maintained by the applicable Borrower or the Secured Parties.
- (c) <u>Waiver of Subrogation</u>. Each Borrower hereby waives any and every claim for recovery from the Secured Parties for any and all loss or damage covered by any of the insurance policies to be maintained under this Agreement to the extent that such loss or damage is recovered under any such policy. Inasmuch as the foregoing waiver will preclude the assignment of any such claim to the extent of such recovery, by subrogation (or otherwise), to an insurance company (or other Person), such Borrower shall give written notice of the terms of such waiver to each insurance company which has issued, or which may issue in the future, any such policy of insurance (if such notice is required by the insurance policy) and shall cause each such insurance policy to be properly endorsed by the issuer thereof to, or to otherwise contain one or more provisions that, prevent the invalidation of the insurance coverage provided thereby by reason of such waiver.
- (d) Amendment of Requirements. The Administrative Agent (acting on the instruction of the Required Lenders) may at any time, upon written notice to the Borrowers, amend the requirements (including the amount and scope of insurance coverage) and approved insurance companies described in this Section 5.9 due to (i) new information (Project exposure or insurance products) not known on the Closing Date which in the reasonable judgment of the Administrative Agent (acting on the instruction of the Required Lenders) either renders required coverage materially inadequate (when compared to insurance carried by prudent operators and owners of facilities similar to the Project) or materially reduces the financial ability of the approved insurance companies to pay claims, (ii) changed circumstances after the Closing Date which in the reasonable judgment of the Administrative Agent (acting on the instruction of the Required Lenders) either render required coverage materially inadequate (when compared to insurance carried by prudent operators and owners of facilities similar to the Project) or materially reduce the financial ability of the approved insurance companies to pay claims, or (iii) more appropriate insurance heretofore unavailable or unaffordable becomes available on commercially reasonable terms; provided that in the case of this clause (iii), such changes shall be limited to requiring the Borrowers to obtain such insurance.

# (e) Additional Provisions.

- (ii) <u>Payment of Loss Proceeds</u>. Any builders all risk, delay in startup, marine cargo, marine cargo delay in startup, machinery breakdown, physical damage or business interruption insurance policies shall include a standard lender's loss payable endorsement in favor of the Collateral Agent for the benefit of the Secured Parties and shall name the Collateral Agent (for the benefit of the Secured Parties) as sole loss payee for all losses greater than \$500,000.
- (iii) Loss Adjustment and Settlement. A loss under any builders all risk, delay in startup, marine cargo, marine cargo delayed in startup, machinery breakdown, physical damage or business interruption insurance policies shall be adjusted with the insurance companies, including the filing in a timely manner of appropriate proceedings, by the applicable Borrower, subject to the approval of the Administrative Agent if such loss is in excess of \$500,000. In addition the applicable Borrower may in its reasonable judgment consent to the settlement of any loss, provided that in the event that the amount of the loss exceeds \$500,000 the terms of such settlement are approved by the Administrative Agent (acting on the instructions of the Required Lenders).
- (iv) <u>Policy Cancellation and Change</u>. Subject to the exceptions set forth on <u>Appendix A</u> (if any) and with the exception of any builders all risk policy which shall be non-cancelable except for nonpayment of premiums (10 days) or for non-compliance with an insurance company's recommendations, all policies of insurance required to be maintained pursuant to this <u>Section 5.9</u> shall be endorsed so that if at any time they should be canceled, or coverage be reduced in any manner, such cancellation or reduction shall not be effective as to the Secured Parties for forty-five (45) days, except for non-payment of premium which shall be for ten (10) days, after receipt by the Administrative Agent of written notice from such insurer of such cancellation or reduction.
- (v) <u>Miscellaneous Policy Provisions</u>. Subject to the exceptions set forth on <u>Appendix A</u> (if any), any builders all risk, delay in startup, marine cargo, marine cargo delay in startup, machinery breakdown, physical damage or business interruption insurance policies shall (A) not include any annual or term aggregate limits of liability or clause requiring the payment of an additional premium to reinstate the limits after loss except as regards the insurance applicable to the perils of flood, earth movement, sabotage and terrorism, (B) include the Secured Parties as additional named insureds as their interests may appear, and (C) include a clause requiring the insurer to make final payment on any claim within ninety (90) days after the submission of proof of loss and its acceptance by the insurer.
- (vi) <u>Separation of Interests</u>. Subject to the exceptions set forth on <u>Appendix A</u> (if any), all policies shall insure the interests of the Secured Parties regardless of any breach or violation by a Borrower or any other Person of warranties, declarations or conditions contained in such policies, or any action or inaction of a Borrower or others. This provision may be satisfied through the use of an acceptable multi-insured endorsement.

- (vii) <u>Acceptable Policy Terms and Conditions</u>. Except as otherwise agreed by the Administrative Agent in its sole discretion, all policies of insurance required to be maintained pursuant to this <u>Section 5.9</u> shall be the same in all material respects as the policies furnished to the Insurance Advisor prior to the applicable Initial Funding Date.
- (viii) <u>Waiver of Subrogation</u>. Subject to the exceptions set forth on <u>Appendix A</u> (if any) and with the exception of workers compensation insurance, all policies of insurance required to be maintained by the provisions of this <u>Section 5.9</u> shall (A) provide for waivers by the insurers of subrogation in favor of the Secured Parties and their respective officers and employees and (B) provide that there shall be no recourse against any Secured Party for payment of premiums or other amounts with respect thereto.
- (f) Evidence of Insurance. On or before the applicable Initial Funding Date and on an annual basis prior to each policy anniversary, each Borrower shall furnish the Administrative Agent with (i) certification of all required insurance marked "premium paid" or accompanied by other evidence of payment satisfactory to the Administrative Agent and (ii) a schedule of the insurance policies held by or for the benefit of such Borrower and required to be in force by the provisions of this Section 5.9. Such certification shall be executed by each insurer or by an authorized representative of each insurer where it is not practical for such insurer to execute the certificate itself. Such certification shall identify underwriters, the type of insurance, the insurance limits and the policy term and shall specifically list the special provisions enumerated for such insurance required by this Section 5.9. The schedule of insurance shall include the name of the insurance company, policy number, type of insurance, major limits of liability and expiration date of the insurance policies. Upon request by the Administrative Agent, each Borrower shall promptly furnish the Administrative Agent with copies of all insurance policies, binders and cover notes or other evidence of such insurance relating to the insurance required to be maintained hereunder.
- (g) Reports. Concurrently with the furnishing of the certification referred to in Section 5.9(f) above, each Borrower shall furnish the Administrative Agent with a report of a nationally recognized independent insurance broker, signed by an officer of the broker, stating that in the opinion of such broker, the insurance then carried or to be renewed is in accordance with the terms of this Section 5.9.
- (h) <u>Failure to Maintain Insurance</u>. In the event a Borrower fails, or fails to cause any Contractor, to take out or maintain the full insurance coverage required to be taken out and maintained by such Party by this <u>Section 5.9</u>, the Administrative Agent, upon thirty (30) days' prior notice (unless the aforementioned insurance would lapse within such period, in which event notice should be given as soon as possible) to such Borrower of any such failure, may (but shall not be obligated to) take out the required policies of insurance and pay the premiums on the same. All amounts so advanced therefor by the Administrative Agent shall become an additional Obligation, and the Borrowers shall forthwith pay such amounts to the Administrative Agent, together with interest thereon at the Default Rate from the date so advanced until fully paid.

(i) No <u>Duty of Administrative Agent to Verify or Review</u>. No provision of this <u>Section 5.9</u> or any provision of any other Transaction Document shall impose on any Secured Party any duty or obligation to verify the existence or adequacy of the insurance coverage maintained pursuant to this <u>Section 5.9</u>, nor shall any Secured Party be responsible for any representations or warranties made by or on behalf of the Borrowers to any insurance company or underwriter. Any failure on the part of any Secured Party to pursue or obtain the evidence of insurance required by this Agreement and/or failure of any Secured Party to point out any non-compliance of such evidence of insurance shall not constitute a waiver of any of the insurance requirements in this Agreement.

#### (j) Waivers of Insurance Requirements.

- (i) If at any time the Administrative Agent determines in its reasonable judgment (after consultation with the Insurance Advisor) that any insurance (including the limits or deductibles thereof) required to be maintained by this Section 5.9 is not available on commercially reasonable terms due to prevailing conditions in the commercial insurance market at such time, then upon the written request of the Borrowers together with a written report of the Borrowers' insurance broker or another independent insurance broker of nationally recognized standing in the insurance industry (x) certifying that such insurance is not available on commercially reasonable terms (and, in any case where the required maximum coverage is not reasonably available, certifying as to the maximum amount which is so available), (y) explaining in detail the basis for such broker's conclusions, and (z) containing such other information as the Administrative Agent or the Insurance Advisor may reasonably request, the Administrative Agent may (after consultation with the Insurance Advisor) temporarily waive such requirement; provided, however, that the Administrative Agent may, in its sole judgment, decline to waive any such insurance requirement without the consent of the Required Lenders.
- (ii) At any time after the granting of any temporary waiver pursuant to clause (i) above, but not more than once in any calendar year, the Administrative Agent may request, and the Borrowers shall furnish to the Administrative Agent within fifteen (15) days after such request, an updated insurance report reasonably acceptable to the Administrative Agent and the Insurance Advisor from the Borrowers' independent insurance broker.
- (iii) Any waiver granted pursuant to clause (i) above shall expire, without further action by any party, immediately upon (A) such waived insurance requirement becoming available on commercially reasonable terms, as reasonably determined by the Administrative Agent (in consultation with the Insurance Advisor) or (B) failure of the Borrowers to deliver an updated insurance report pursuant to clause (ii) above.

## 5.10 Events of Loss and Project Document Claims.

(a) If an Event of Loss shall occur with respect to any Collateral, the Borrowers shall (i) diligently pursue all its rights to compensation against any Person with

88

respect to such Event of Loss and (ii) cause the Net Available Amount of any Loss Proceeds to be deposited in the applicable Loss Proceeds Account pursuant to the Collateral Account Agreement. To the extent that any such Loss Proceeds are paid to the Borrowers, such Loss Proceeds shall be held in trust for the Collateral Agent for the benefit of the Secured Parties segregated from other funds of the Borrowers, and the Borrowers shall cause such Loss Proceeds to be deposited into the applicable Loss Proceeds Account as contemplated above as promptly as possible.

- (b) The Administrative Agent shall be entitled to participate in any compromise, adjustment or settlement in connection with any Event of Loss under any policy or policies of insurance or in respect of any proceeding with respect to any Taking of Property of the Borrowers involving in excess of \$500,000.
- (c) If either Borrower shall have a Project Document Claim against any Project Participant, such Borrower shall (i) diligently pursue all its rights to compensation against such Project Participant with respect to such Project Document Claim and (ii) cause the Net Available Amount received in respect of any such Project Document Claim to be deposited in the applicable Collateral Account as required by the Collateral Account Agreement and, in the case of Project Document Claims other than indemnity claims and delay liquidated damages, applied to prepay the Obligations, such prepayment to be made no later than the date which is three (3) days after the receipt of such proceeds. To the extent that any such amounts are paid to a Borrower, such amounts shall be held in trust for the Collateral Agent for the benefit of the Secured Parties segregated from other funds of such Borrower, and such Borrower shall cause such amounts to be deposited in the applicable Collateral Account as contemplated above as promptly as possible.

## 5.11 <u>Application of Loss Proceeds</u>.

- (a) If an Event of Loss shall occur with respect to any Collateral, the applicable Borrower shall cause any Net Available Amount of any Loss Proceeds to be applied to the payment of Obligations within three (3) days of the receipt of such proceeds in accordance with Section 8.3(b) hereof; provided that, with the consent of the Required Lenders, such Borrower shall be permitted to apply such proceeds to the costs of Restoring the portion of the Project that was the subject of such Event of Loss; and provided, further, that, so long as no Default or Event of Default shall then exist, if such proceeds are \$500,000 or less, such Borrower shall be permitted (without the consent of the Required Lenders) to apply such proceeds to the payment of the costs of Restoring the portion of the Project that was the subject of such Event of Loss (and such proceeds shall be delivered to such Borrower in accordance with the following sentence and Section 4.8 of the Collateral Account Agreement). If any Loss Proceeds are to be applied to the payment of Restoration costs pursuant to the preceding sentence, the Net Available Amount of such Loss Proceeds shall be remitted to the applicable Borrower from time to time as the Restoration Work progresses and shall be applied by such Borrower to the payment of the costs of Restoration, such remittance to be made pursuant to the terms of the Collateral Account Agreement and subject to the following conditions:
  - (i) In the opinion of the Project Engineer, the Net Available Amount of the Loss Proceeds (together with all other funds reasonably expected to be available to

such Borrower) is sufficient to Restore the applicable Facility and to pay all Operation and Maintenance Expenses and Tranche A Debt Service and/or Tranche B Debt Service, as applicable, and to meet its other obligations hereunder and under the other Finance Documents both prior to and following completion of the Restoration Work;

- (ii) The Restoration Work shall be supervised by an architect or engineer who is satisfactory to the Administrative Agent and the Project Engineer and, before any Restoration Work is commenced, other than temporary Restoration Work to protect property or persons, to comply with governmental regulations, or to prevent interference with business, the Administrative Agent and the Project Engineer shall have approved the plans and specifications for the Restoration Work, it being understood that after giving effect to the completion of such proposed Restoration Work, the Project shall be at least equal in value and utility to the Project prior to the damage or destruction;
- (iii) Each request for payment shall be made by the delivery, at least five (5) Business Days prior to the proposed payment date, of a Restoration Requisition to the Depositary Agent (with a copy to the Administrative Agent), accompanied by a certificate signed by the supervising architect or engineer and a certificate signed by the Project Engineer, in each case concurring with the statements made in such Restoration Requisition;
- (iv) No Project Document (other than any Immaterial Additional Project Document) or Necessary Governmental Approval in effect immediately prior to the Event of Loss giving rise to such Loss Proceeds shall (A) have been canceled unless replaced in a manner satisfactory to the Administrative Agent (acting on the instructions of the Required Lenders) or (B) contain any still-exercisable right to cancel due to such Event of Loss;
- (v) No Default or Event of Default (other than a Default or Event of Default arising directly from the event as to which such Loss Proceeds have been paid) shall have occurred and be continuing or would result from such remittance;
- (vi) The Property of such Borrower constituting the Restoration Work shall be subject to the Security Interest (whether by amendment of the Security Documents or by entering into new Security Documents or otherwise); and
- (vii) Such Borrower shall have delivered to the Administrative Agent cash-flow projections and other assurances satisfactory to the Required Lenders demonstrating such Borrower's ability to meet its obligations hereunder and under the other Finance Documents at any time both prior to and following completion of such Restoration Work.
- (b) If such Borrower shall not have commenced the Restoration Work within thirty (30) days after the Required Lenders shall have consented to the disbursement of the Loss Proceeds referred to above to pay the costs of Restoring the Project in accordance with this <u>Section 5.11</u> or if at any time after such 30-day period, one or more of the foregoing conditions shall not be satisfied, then, to the extent that such Loss Proceeds shall not otherwise have been

90

disbursed as aforesaid to such Borrower, the remaining amount of such Loss Proceeds shall be applied, on behalf of such Borrower, to the prepayment of the Loans as provided in <u>Section 8.3(b)</u>. Anything to the contrary in this <u>Section 5.11</u> notwithstanding, if an Event of Default shall have occurred and be continuing (other than as a direct result of the Event of Loss which gave rise to such Loss Proceeds), the Administrative Agent may forthwith direct the Depositary Agent to apply the remaining amount of such Loss Proceeds to the prepayment of the Loans as provided in <u>Section 8.3(b)</u>.

- (c) Notwithstanding anything to the contrary which may be contained in the foregoing provisions of this Section 5.11, if a Taking shall occur with respect to any Collateral, the applicable Borrower shall (i) immediately upon discovery or receipt of notice of any occurrence thereof provide written notice to the Administrative Agent, (ii) not, without the written consent of the Required Lenders, compromise or settle any claim with respect to such Taking, and (iii) apply the Net Available Amount of any Loss Proceeds received in respect of such Taking to the prepayment of the Obligations in accordance with Section 8.3(b). The Borrowers consent to the participation to the extent permitted by Law of the Administrative Agent in any proceedings regarding any Taking, and the Borrowers shall from time to time deliver to the Administrative Agent all documents and instruments requested by the Administrative Agent to permit such participation. Nothing in this Section 5.11(c) shall be deemed to impair any rights any Secured Party may have with respect to any such Taking.
- Required Hedging Agreements. Each Borrower shall maintain in full force and effect one or more Interest Rate Hedging Agreements (collectively, the "Required Hedging Agreements") with an Acceptable Swap Counterparty, which shall effectively enable such Borrower to protect itself in a manner satisfactory to the Administrative Agent against the risk of interest rate fluctuations as to a notional principal amount at least equal to 75% of the outstanding principal amount of, (i) with respect to the Tranche A Borrower, the Tranche A Construction Loans from time to time and (ii) with respect to the Tranche B Borrower, the Tranche B Construction Loans from time to time, which Required Hedging Agreements shall be entered into and become effective on or prior to the applicable Initial Funding Date and shall be maintained until the Facility A Conversion Date or the Facility B Conversion Date, as applicable; provided that, in each case, if any Required Hedging Agreement is with an Acceptable Swap Counterparty other than a Secured Swap Counterparty, such Borrower shall not post any cash collateral, letter of credit or other margin thereunder, and the obligations of such Borrower thereunder shall not be secured by the Security Documents or any other Lien on the Property of such Borrower and shall be subordinated to the Obligations in a manner and pursuant to terms and conditions which are in all respects satisfactory to the Required Lenders.
- 5.13 <u>Chief Place of Business; etc.</u> Each Borrower shall (a) maintain its chief place of business at 3020 Old Ranch Parkway, Seal Beach, CA 90740 and maintain the office where it keeps its records concerning the Transaction Documents at such address, (b) not keep any place of business or assets outside of the Sites other than as provided in clause (a) above, (c) not change its name, and (d) not do business under any other name than "Clean Energy Tranche A LNG Plant, LLC", with respect to the Tranche B Borrower, unless, in any such case, such Borrower shall have given the Administrative Agent at least thirty (30) days' prior written notice, and all action requested by the Administrative Agent necessary or advisable in the Administrative Agent's opinion to preserve and perfect the Security Interests with respect to the Collateral shall have been taken.

#### 5.14 <u>Use of Proceeds</u>.

- (a) (i) The Tranche A Borrower shall use the proceeds of the Tranche A Construction Loans in accordance with the provisions of this Agreement and the Collateral Account Agreement to pay Facility A Project Costs and (ii) the Tranche B Borrower shall use the proceeds of the Tranche B Construction Loans in accordance with the provisions of this Agreement and the Collateral Account Agreement to pay Facility B Project Costs, in each case, in accordance with the applicable Construction Budget.
- (b) Neither Borrower shall, without the prior written consent of the Administrative Agent, amend, revise or modify a Construction Budget to increase or decrease or otherwise change the number or type of Construction Budget categories, or request any Loans for the purpose of funding any Project Costs in excess of the amount contained in such Construction Budget for such category of Project Costs or request any Loans in respect of the Contingency in an amount which would result in the Contingency being reduced to less than 3% of the total of such Construction Budget.

#### 5.15 <u>Project Construction; Maintenance.</u>

- (a) The Borrowers shall cause the Projects to be duly constructed and completed in accordance with the Construction Budgets, the Construction Agreements, the Plans and Specifications and Prudent Industry Practice and shall cause the Facility A Project Completion Date to occur on or before the Facility A Date Certain and the Facility B Project Completion Date to occur on or before the Facility B Date Certain.
- (b) The Borrowers shall maintain and preserve the Projects and all of their other material Properties necessary for the proper conduct of their business in good working order and in such condition that each Facility will have the capacity and functional ability to perform, on a continuing basis (ordinary wear and tear excepted), in normal commercial operation, the functions for which it was specifically designed in accordance with the applicable Construction Agreements at substantially the levels contemplated thereby. The Borrowers shall cause each Project to be operated, serviced, maintained and repaired so that the condition and operating efficiency thereof will be maintained and preserved (ordinary wear and tear excepted) in all material respects in accordance and in compliance with (i) Prudent Industry Practices, (ii) such operating standards as shall be required to enforce any material warranty claims against dealers, manufacturers, vendors, contractors and sub-contractors, (iii) the terms and conditions of all insurance policies maintained with respect to such Project at any time, (iv) all requirements of Law and all Governmental Approvals applicable to such Project and (v) the terms of the Project Documents (other than any Immaterial Additional Project Documents).
- (c) The Borrowers shall not alter, remodel, add to, reconstruct, improve or demolish any part of the Projects or any other Collateral, except as contemplated by or in accordance with the Plans and Specifications.

92

(d) The Borrowers shall not, directly or indirectly, make or commit to make any expenditure in respect of the purchase or other acquisition of fixed or capital assets, other than (i) expenditures contemplated by the Construction Budgets or the prevailing Operating Budgets, as appropriate and (ii) expenditures permitted to be made pursuant to <u>Section 5.11</u>.

## 5.16 <u>Performance of Project Documents.</u>

- (a) Each Borrower shall perform and observe all of its covenants and agreements contained in any of the Project Documents (other than any Immaterial Additional Project Documents) to which it is or becomes a party, to the extent that the failure of such Borrower to perform or observe such covenant or agreement would permit any other party to such Project Document (other than such Borrower) to terminate such Project Document.
- (b) Each Borrower shall take all commercially reasonable action to prevent the early termination of any Project Document (other than any Immaterial Additional Project Document), (unless such early termination is the result of the completion of performance thereunder) and shall take any and all action as may be reasonably necessary promptly to enforce its rights and to collect any and all sums due to it under the Project Documents.
- (c) Each Borrower shall instruct all applicable Project Participants to make all payments payable to such Borrower to the Depositary Agent for deposit in the appropriate Collateral Account in accordance with the Collateral Account Agreement.

# 5.17 <u>Operating Plan and Budget</u>.

(a) On or before the Facility A Project Completion Date or the Facility B Project Completion Date, as applicable, the applicable Borrower shall adopt an operating plan and a budget of Operation and Maintenance Expenses in respect of the applicable Facility for the period from such date to the end of the first Operating Year of such Facility, and, no less than thirty (30) days in advance of the beginning of each Operating Year of such Facility thereafter, it shall similarly adopt an operating plan and a budget of Operation and Maintenance Expenses for such Operating Year for such Facility (such operating plan and budget for an Operating Year, an "Operating Budget"). Copies of the proposed Operating Budget for each Operating Year shall be furnished at least thirty (30) days before final adoption thereof to the Administrative Agent, and no Operating Budget shall be adopted without the prior written approval of the Administrative Agent and/or the Project Engineer for a proposed Operating Budget is not obtained prior to the first day of the Operating Year to which such proposed Operating Budget relates, the applicable Borrower may continue to operate the applicable Project in accordance with the Operating Budget then in effect with the budgeted cost of each budget item being increased to the lesser of (i) the amount therefor in the proposed Operating Budget or (ii) 110% of the amount of the budgeted cost of such budget item in the current Operating Budget. Copies of the final Operating Budget so adopted shall be furnished to the Administrative Agent promptly upon the adoption thereof.

amended Operating Budget which is proposed shall be furnished at least ten (10) days before final adoption thereof to the Administrative Agent, and no such amended Operating Budget shall be adopted without the prior written approval of the Administrative Agent and the Project Engineer. Copies of the final amended Operating Budget shall be furnished to the Administrative Agent promptly after adoption thereof.

(c) The amounts provided in the annual Operating Budget for each item and classification of Operation and Maintenance Expenses for each month of each Operating Year shall not exceed the amounts reasonably expected to be necessary therefor, and the applicable Borrower shall not expend (or submit a Transfer Date Certificate which contemplates the expenditure of) any amount for Operation and Maintenance Expenses during any month if such expenditure would cause the aggregate amount of Operation and Maintenance Expenses expended by the Borrower in such Operating Year for such Facility, after taking into account amounts theretofore paid in such Operating Year, to exceed by more than 10% the aggregate amount budgeted to be expended through such month in the applicable Operating Budget for such Facility, unless such expenditure could not reasonably be anticipated and failure to make such expenditure would have created an abnormal risk of personal injury to employees or significant physical damage to the applicable Project and, in any such event, such Borrower shall immediately advise the Administrative Agent of such excess expenditures and, within ten (10) days of the making of any such expenditure, prepare and file with the Administrative Agent an amended Operating Budget to reflect such changes.

## 5.18 <u>Environmental Compliance</u>. Each Borrower shall:

- (a) comply in all material respects and cause all other Persons constructing, occupying or conducting operations at the Projects to comply in all material respects with all Environmental Laws now or hereafter applicable to the Projects;
- (b) obtain, at or prior to the time required by applicable Environmental Laws, all Governmental Approvals required pursuant to applicable Environmental Laws for the construction, operation and maintenance of the Projects, and maintain such Governmental Approvals in full force and effect:
- (c) not generate, use, treat, recycle, store, Release or dispose of, or permit the generation, use, treatment, recycling, storage, Release or disposal of Hazardous Materials on the Sites, or transport or permit the transportation of Hazardous Materials to or from the Projects other than in material compliance with all applicable Environmental Laws;
- (d) conduct and complete any investigation, study, sampling and testing and undertake any cleanup, removal, remedial or other action necessary to remove and clean up all Hazardous Materials at, on, in, under or emanating from the Projects to the extent required by and in accordance with the requirements of all applicable Environmental Laws; and
- (e) provide the Administrative Agent with information and documents concerning any Environmental Claim, Release of Hazardous Material, or any other environmental condition or matter relating to the Projects, including but not limited to any matter or condition relating to compliance with applicable Environmental Laws and Governmental Approvals required by such Environmental Laws, as may be reasonably requested by the Administrative Agent.

94

# 5.19 <u>Security Documents</u>.

- (a) Each Borrower shall take all actions necessary or requested by the Administrative Agent or the Collateral Agent to maintain each Security Document in full force and effect and enforceable in accordance with its terms and to maintain and preserve the Liens created by the Security Documents and the priority thereof, including (i) making filings and recordations, (ii) making payments of fees and other charges, (iii) issuing and, if necessary, filing or recording supplemental documentation, including continuation statements, (iv) discharging all claims or other Liens adversely affecting the rights of any Secured Party in any Collateral, (v) publishing or otherwise delivering notice to third parties, (vi) depositing title documents, and (vii) taking all other actions either necessary or otherwise requested by the Administrative Agent or the Collateral Agent to ensure that all Collateral (including any after-acquired Property of the Borrowers intended to be covered by any Security Document) is subject to a valid and enforceable first-priority Lien (subject only to (i) Permitted Liens and (ii) the priority afforded to such Permitted Liens by operation of Law) in favor of the Collateral Agent for the benefit of the Secured Parties. In furtherance of the foregoing, (A) each Borrower shall ensure that all Property acquired by it shall become subject to the Lien of the Security Documents having the priority contemplated thereby promptly upon the acquisition thereof and (B) except with respect to the Checking Accounts, each Borrower shall not open or maintain any bank account without first taking all such actions as may be necessary or otherwise requested by the Administrative Agent to ensure that such bank account is subject to a valid and enforceable first priority Lien in favor of the Collateral Agent for the benefit of the Secured Parties.
- (b) Each Borrower shall take all action necessary to cause each Additional Project Document to which it is a party and each Site Real Estate Right obtained following the Closing Date, including, without limitation, each Specified Site Real Estate Right upon the procurement of a Consent Agreement from the applicable Specified Facility RE Right Grantor, to be or become subject to the Liens of the Security Documents (whether by amendment to any Security Document, execution of a new Security Document or otherwise) in favor of the Collateral Agent, and with respect to any Material Additional Project Document, shall deliver or cause to be delivered to the Administrative Agent and the Collateral Agent such certificates or other documents with respect thereto as the Administrative Agent or the Collateral Agent may reasonably request. Each Borrower shall cause each party to a Material Additional Project Document (other than such Borrower) and each Specified Facility RE Right Grantor to execute and deliver a Consent Agreement with respect to each such Material Additional Project Document or Specified Site Real Estate Right, as applicable.

## 5.20 Required Debt Service Reserve Amounts.

(a) From and after the Facility A Conversion Date, the Tranche A Borrower shall cause the funds on deposit in the Tranche A Debt Service Reserve Account to at all times equal or exceed the Tranche A Required Debt Service Reserve Amount.

(b) From and after the Facility B Conversion Date, the Tranche B Borrower shall cause the funds on deposit in the Tranche B Debt Service Reserve Account to at all times equal or exceed the Tranche B Required Debt Service Reserve Amount.

#### 5.21 <u>Payment of Project Costs with Project Revenues.</u>

- (a) Any Project Revenues received by the Tranche A Borrower prior to the Facility A Conversion Date shall be deposited into the Tranche A Construction Account and applied, in accordance with the Collateral Account Agreement, to pay Tranche A Project Costs.
- (b) Any Project Revenues received by the Tranche B Borrower prior to the Facility B Conversion Date shall be deposited into the Tranche B Construction Account and applied, in accordance with the Collateral Account Agreement, to pay Tranche B Project Costs.
- 5.22 <u>Anti-Terrorism Laws</u>. Each Borrower shall promptly provide to any Secured Party any information that such Secured Party may require in order to comply with any requirement under any Anti-Terrorism Law.
- 5.23 <u>Further Assurances</u>. Each Borrower shall promptly and duly execute and deliver to the Administrative Agent or the Collateral Agent, as applicable, such documents and assurances and take such further action as the Administrative Agent or the Collateral Agent may from time to time reasonably request in order to carry out more effectively the intent and purpose of the Finance Documents and to establish, protect and perfect the rights and remedies created or intended to be created in favor of the Secured Parties pursuant to the Finance Documents.
- 5.24 <u>Separateness</u>. Each Borrower shall (i) maintain books and records separate from those of any other Person, including the other Borrower, the Sponsor and its other Subsidiaries; (ii) maintain its assets in such a manner that it is not more costly or difficult to segregate, identify or ascertain such assets; (iii) observe all organizational formalities; (iv) hold itself out to creditors and the public as separate and distinct from any other Person, including the other Borrower, the Sponsor and its other Subsidiaries; (v) conduct its business in its name or in its own business name or trade name, and use stationary, invoices and checks separate from those of the other Borrower, the Sponsor and its other Subsidiaries; and (vi) not hold itself out as being available to satisfy the obligations of any other Person, including the other Borrower, the Sponsor and its other Subsidiaries.
- 5.25 <u>Collateral Account Agreement</u>. Each Borrower shall cause the Collateral Account Agreement to be finalized and executed by each party thereto and each of the Collateral Accounts thereunder to be established no later than the date thirty (30) days following the Closing Date.

#### SECTION 6. <u>NEGATIVE COVENANTS</u>

- 6.1 <u>Limitation on Liens</u>. Neither Borrower shall create, incur, assume or suffer to exist any Lien upon any of its Property, whether now owned or hereafter acquired, except:
  - (a) the Security Interests;

- (b) carriers', warehousemen's, mechanics', materialmen's, repairmen's or other like Liens arising in the ordinary course of business which are not overdue for a period of more than thirty (30) days to the extent not required to be paid under Section 5.5 or for amounts being contested in good faith and by appropriate proceedings, so long as (i) such proceedings shall not interfere in any material respect with the use, marketability or disposition of either Project or Site, (ii) the negative outcome of such contest could not reasonably be expected to result in a Material Adverse Effect and (iii) a bond, cash reserves or other security reasonably acceptable to Administrative Agent has been posted or provided in such manner and amount as to assure Administrative Agent that any amounts determined to be due will be promptly paid in full when such contest is determined;
- (c) Liens for any tax, assessment or other governmental charge, either secured by a bond or other security reasonably acceptable to Administrative Agent or not yet delinquent or being contested in good faith and by appropriate proceedings, so long as (i) such proceedings shall not interfere in any material respect with the use, marketability or disposition of either Project or Site, (ii) the negative outcome of such contest could not reasonably be expected to result in a Material Adverse Effect and (iii) a bond, cash reserves or other security reasonably acceptable to Administrative Agent has been posted or provided in such manner and amount as to assure Administrative Agent that any amounts determined to be due will be promptly paid in full when such contest is determined;
- (d) Liens arising out of judgments or awards so long as an appeal or proceeding for review is being prosecuted in good faith and for the payment of which adequate reserves, bonds or other security reasonably acceptable to Administrative Agent have been provided or are fully covered by insurance;
- (e) Liens, deposits or pledges to secure statutory obligations or performance of bids, tenders, contracts (other than for the repayment of borrowed money) or leases, or for purposes of like general nature in the ordinary course of its business, not to exceed \$1,000,000 in the aggregate at any one time, and with any such Lien to be released as promptly as practicable;
- (f) pledges or deposits made in the ordinary course of business in connection with the Project owned by such Borrower to secure obligations under worker's compensation, unemployment insurance and other social security and similar laws and regulations;
  - (g) any zoning, building or similar laws or rights reserved to or vested in any Governmental Authority;
- (h) Liens on cash securing obligations in respect of surety bonds or similar instruments and any other Indebtedness in respect of letters of credit, banker's acceptances or similar arrangements in an aggregate amount not exceeding \$1,000,000 at any one time outstanding; and
- (i) those matters constituting exceptions to title contained in the Title Insurance pursuant to <u>Section 3.2(g)(iii)</u>, and such other easements, rights-of-way and other encumbrances on title to real property that have been expressly approved by the Administrative Agent in writing, in the exercise of its sole discretion.

- 6.2 <u>Indebtedness</u>. Neither Borrower shall create, incur, suffer to exist or otherwise become liable for any Indebtedness except:
  - (a) Indebtedness arising under the Finance Documents;
  - (b) Indebtedness (other than Indebtedness for borrowed money) secured by a Permitted Lien;
- (c) contingent liabilities, to the extent otherwise constituting Indebtedness, including those relating to (i) the acquisition of goods, supplies or merchandise in the normal course of business or normal trade credit, (ii) the endorsement of negotiable instruments received in the normal course of its business, and (iii) contingent liabilities incurred with respect to any Necessary Governmental Approval or Transactions Document; and
- (d) obligations in respect of surety bonds or similar instruments and any other Indebtedness in respect of letters of credit, banker's acceptances or similar arrangements incurred in the ordinary course of business and in an aggregate amount not exceeding \$1,000,000 at any one time outstanding.
- 6.3 <u>Leases</u>. Neither Borrower shall enter into any agreement, or be or become liable as lessee under any agreement, for the lease, hire or use of any real or personal Property, except as provided for in the then applicable Operating Budget.
  - 6.4 <u>Investments; Subsidiaries</u>.
    - (a) Neither Borrower shall make or permit to remain outstanding any Investments, except Permitted Investments.
    - (b) Neither Borrower shall establish, create or acquire any Subsidiary without the prior consent of the Administrative Agent.
- 6.5 <u>Distributions</u>. Neither Borrower shall make any distributions or pay any dividends to the Sponsor, Holdings or any other Person in respect of its Capital Stock or any other ownership interest in such Borrower, whether in cash or other Property, or redeem, purchase or otherwise acquire any interest of Holdings or the Sponsor, or permit Holdings or the Sponsor to withdraw any capital from the Borrowers (all of the foregoing being referred to as "<u>Distributions</u>"), or make any payment of any management or other fees to any Affiliate of the Borrowers, or make any payment in respect of any subordinated Indebtedness; *provided* that the Borrowers may make Tax Distributions pursuant to the Collateral Account Agreement; and *provided further*, (y) amounts on deposit in the Tranche A Distribution Account may be remitted to the Tranche A Borrower on any Quarterly Distribution Date and (z) amounts on deposit in the Tranche B Distribution Account may be remitted to the Tranche B Borrower on any Quarterly Distribution Date, in each case for the purpose of making a Distribution or for any other purpose, so long as each of the following conditions is satisfied:
  - (i) both the Facility A Conversion Date and the Facility B Conversion Date shall have occurred;

- (ii) no Default or Event of Default shall have occurred and be continuing or would result from the making of such Distribution;
- (iii) each Borrower shall have delivered the most recent Debt Service Coverage Statement, Projected Debt Service Coverage Statement and Leverage Ratio Statement by the time required by Section 5.1(d) and such statements shall demonstrate that (A) each of the Tranche A Debt Service Coverage Ratio and the Tranche B Debt Service Coverage Ratio on the most recent Quarterly Date prior to such Quarterly Distribution Date was at least 1.25:1.00; (B) each of the Tranche A Projected Debt Service Coverage Ratio and the Tranche B Projected Debt Service Coverage Ratio on the most recent Quarterly Date prior to such Quarterly Distribution Date was at least 1.25:1.00 and (C) each of the Tranche A Leverage Ratio and the Tranche B Leverage Ratio on the most recent Quarterly Date prior to such Quarterly Distribution Date were not more than 2.50:1.00;
- (iv) no default or event of default shall have occurred and be continuing under any Project Document (other than an Immaterial Additional Project Document), other than commercial disputes arising in the ordinary course of business which involve claims of less than \$250,000;
- (v) funds on deposit in each Debt Service Reserve Account shall equal or exceed the applicable Required Debt Service Reserve Amount; and
- (vi) each of the Tranche A Distribution Account and the Tranche B Distribution Account shall have, as of such Quarterly Distribution Date, at least \$1.00 on deposit therein.
- Transactions with Affiliates. Except as provided in the Finance Documents, and the Affiliate Project Documents entered into by a Borrower prior to the applicable Initial Funding Date, neither Borrower shall directly or indirectly (a) make any Investment in or payment to an Affiliate of the Borrowers, (b) transfer, sell, lease, assign or otherwise dispose of any Property to an Affiliate of the Borrowers, (c) purchase or acquire Property from an Affiliate of the Borrowers or (d) enter into any other transaction or arrangement directly or indirectly with or for the benefit of an Affiliate of the Borrowers, unless such transaction is (i) in the ordinary course of such Borrower's (and such Affiliate's) business, (ii) upon fair and reasonable terms no less favorable to such Borrower than it would obtain in a comparable arm's length transaction with a Person that is not an Affiliate and (iii) otherwise acceptable in form and substance to the Administrative Agent in its sole discretion.
- 6.7 <u>Merger; Sales and Purchases of Assets</u>. Neither Borrower shall merge into or consolidate with any other Person, or liquidate or dissolve itself (or suffer any liquidation or dissolution), or sell, lease, transfer, or otherwise Dispose of any assets or Property other than: (a) sales, transfers and other Dispositions of assets of such Borrower, having a value not to exceed \$100,000 in the aggregate, and determined by such Borrower (in its reasonable opinion) to be obsolete or no longer used by or useful to such Borrower for the operation or maintenance of the Project owned by such Borrower; (b) sales of Permitted Investments prior to the maturity thereof; (c) Distributions or other payments in accordance with **Section 6.5** hereof; and (d)

pursuant to the LNG Purchase Agreements. Neither Borrower shall purchase or acquire any assets other than the purchase of (i) assets reasonably required for the completion of the Project owned by such Borrower in accordance with the applicable Construction Budget, (ii) subject to Section 5.17 hereof, assets in the ordinary course of business reasonably required in connection with the operation of the Project owned by such Borrower and (iii) Permitted Investments.

## 6.8 Amendment of Transaction Documents; Additional Project Documents; Change Orders; etc.

- Neither Borrower shall: (i) agree to or permit the cancellation, suspension or termination of any Transaction Document (other than (B) any Immaterial Additional Project Document, (B) any Project Document upon the completion of performance thereunder; or (C) any Financing Document in connection with the termination of the Commitments and the repayments of the Obligations in full); (ii) sell, assign (other than pursuant to the Security Documents) or otherwise dispose of (by operation of Law or otherwise) any part of its interest in any Transaction Document; (iii) waive or extend materially any cure or grace period for any material default under or breach of any Project Document (other than any Immaterial Additional Project Document) or waive, fail to enforce, forgive or release any material right, interest or entitlement, howsoever arising, under or in respect of any Project Document (other than any Immaterial Additional Project Document); (iv) petition, request or take any other legal or administrative action (including any arbitration proceedings) that seeks, or may be expected, to rescind, terminate or suspend any Project Document (other than any Immaterial Additional Project Document) or amend or modify all or any material part thereof; (v) exercise any right to initiate an arbitration proceeding under any Project Document or take any action with respect to any arbitration proceeding involving any other party to a Project Document; (vi) agree to or permit the assignment of any rights or the delegation of any obligations of any Project Participant (other than any Immaterial Project Participant) under any Project Document (other than any Immaterial Additional Project Document) except as permitted without the consent of the Borrowers by the terms of such Project Document; (vii) amend, supplement, modify or give any consent under any Project Document (other than any Immaterial Additional Project Document) or exercise any material option thereunder if such amendment, supplement, modification or consent could reasonably be expected to cause any material milestones to be materially delayed or would increase the costs to the Borrower associated with such Project Document unless those costs are included in the then applicable budget or paid for with an Equity Contribution (other than Primary Equity Contributions as defined in and required by the Equity Contribution Agreement); (viii) except as may be permitted by Section 11.10, amend, supplement, modify or give any consent under any Finance Document or exercise any material option thereunder; or (ix) enter into any Material Additional Project Document other than the Gas Supply Agreements and the Power Purchase Agreements.
- (b) Notwithstanding the provisions of the foregoing subsection (a), each Borrower may, upon ten (10) Business Days' prior notice to the Project Engineer and the Administrative Agent (or such lesser notice as is acceptable to the Administrative Agent), enter into any Change Order if: (i) such Change Order is reasonable and necessary; (ii) such Change Order does not change the Plans and Specifications; (iii) the Cost of such Change Order does not exceed \$100,000 or cause the aggregate Cost of all Change Orders theretofore made, together with the Cost of such Change Order, to exceed \$200,000; (iv) such Change Order does not result

100

in a material extension of any guaranteed completion or acceptance date under any Construction Agreement; (v) such Change Order does not result in any change to, or amendment of, (A) the Acceptance Tests, (B) the compensation and payment provisions set forth in any Construction Agreement, or (C) the conditions pursuant to which payment of any damages is required to be made, either directly or indirectly; and (vi) such Change Order could not otherwise reasonably be expected to have a Material Adverse Effect.

(c) In the event that any Project Document (other than any Immaterial Additional Project Document) is terminated or canceled prior to its expressed expiration date, the applicable Borrower shall promptly, but no later than the date which is thirty (30) days after such termination or cancellation (subject to the consent of the Required Lenders), enter into an Additional Project Document with a Replacement Project Participant constituting a replacement of the Project Document so terminated or canceled. The obligations of each Borrower under this subsection (c) shall not be in derogation or limitation of the obligations of such Borrower under subsection (a) of this **Section 6.8** or in any way limit or impair the rights or remedies of the Secured Parties hereunder or under any other Finance Document directly or indirectly arising out of the termination or cancellation of any Project Document.

#### 6.9 <u>Completion; Acceptance Tests.</u>

- (a) Neither Borrower shall, without the prior approval of the Required Lenders (after consultation with the Project Engineer), (i) take any action (including any postponement of the Work pursuant to any provision of any Construction Agreement) or fail to take any action which could permit an extension of any guaranteed completion or acceptance date under any Construction Agreement, (ii) accept or confirm that a Project or a stage thereof has been provisionally or finally completed pursuant to any Construction Agreement, (iii) fail to advise any counterparty to a Construction Agreement of any defects, deficiencies or discrepancies of which either Borrower has knowledge, (iv) notify any counterparty to a Construction Agreement that it accepts a punch list, (v) issue, approve or execute any acceptance or completion certificate or otherwise confirm acceptance or completion of a Project or any portion or phase thereof, (vi) waive, defer or reduce any of the requirements of any of the Acceptance Tests, (vii) accept or confirm that a Facility has satisfied any of the Acceptance Tests, or (viii) reject a Facility.
- (b) Neither Borrower shall schedule or agree to the scheduling of any Acceptance Tests without at least fifteen (15) days' prior written notice to the Administrative Agent and the Project Engineer.
- 6.10 <u>ERISA</u>. Neither Borrower nor Holdings nor any Subsidiary of either Borrower shall at any time establish, maintain, contribute to or be required or permitted to contribute to any Plan subject to Section 412 of the Code or any Multiemployer Plan.
- 6.11 <u>Sale and Leaseback Transactions</u>. Neither Borrower will, directly or indirectly, become or remain liable as lessee or as a guarantor or other surety with respect to any lease, whether an operating lease or capital lease, of any Property (whether real, personal or mixed), whether now owned or hereafter acquired, (a) which such Borrower has sold or transferred or is to sell or transfer to any other Person or (b) which such Borrower intends to use for substantially the same purpose as any other property which has been or is to be sold or transferred by such Borrower to any Person.

- 6.12 <u>Certain Agreements</u>. Neither Borrower shall enter into any agreement or undertaking (except for the Finance Documents) restricting, or purporting to restrict, the ability of such Borrower to (a) amend this Agreement or any other Finance Document, (b) sell any of its assets, (c) create Liens, (d) create or incur Indebtedness or (e) make any Distribution.
- 6.13 <u>Transfers of Equity Interests.</u> Neither Borrower shall (i) permit or consent to the transfer (by assignment, sale, pledge, collateral assignment or otherwise) of any Capital Stock of such Borrower or (ii) issue any new Capital Stock other than to Holdings to the extent pledged by Holdings to the Collateral Agent.
- 6.14 <u>Bank Accounts</u>. Except for the Checking Accounts, neither Borrower shall at any time own, open or suffer to exist in its name any deposit account, securities account, commodities account or other bank account except for the Collateral Accounts. Neither Borrower shall permit funds on deposit in the applicable Checking Accounts to exceed \$100,000 in the aggregate for each Borrower.
- 6.15 <u>Fiscal Year and Fiscal Quarter; Accounting Changes</u>. Neither Borrower shall make any change in its fiscal year, which shall be the calendar year. Neither Borrower shall make any change in its fiscal quarter, which shall be the calendar quarter. Neither Borrower shall make any change in accounting policies or reporting policies, except as required by GAAP.

## SECTION 7. FINANCIAL COVENANTS

#### 7.1 <u>Debt Service Coverage Ratios.</u>

- (a) From and after the Facility A Conversion Date, the Tranche A Debt Service Coverage Ratio as of any Quarterly Date (beginning with the Quarterly Date at the end of the first full fiscal quarter following the Facility A Conversion Date) shall be no less than 1.20:1.00.
- (b) From and after the Facility B Conversion Date, the Tranche B Debt Service Coverage Ratio as of any Quarterly Date (beginning with the Quarterly Date at the end of the first full fiscal quarter following the Facility B Conversion Date) shall be no less than 1.20:1.00.

#### SECTION 8. PAYMENT PROVISIONS; FEES

## 8.1 Repayment of Principal.

(a) Each Borrower shall repay in full in cash, on a joint and several basis, the aggregate principal amount of (i) the Tranche A Construction Loans, together with accrued and unpaid interest thereon and all other Tranche A Obligations outstanding, and such Tranche A Construction Loans shall mature, on the Tranche A Construction Loan Maturity Date and (ii) the Tranche B Construction Loans, together with accrued and unpaid interest thereon and all other Tranche B Obligations outstanding, and such Tranche B Construction Loans shall mature, on the

102

Tranche B Construction Loan Maturity Date (in each case, except to the extent that such Construction Loans are converted into Term Loans in accordance with <u>Section 2.3</u> or <u>2.4</u> hereof). The Tranche A Construction Loan Commitments shall automatically be reduced to zero on the Tranche A Construction Loan Maturity Date; the Tranche B Construction Loan Commitments shall automatically be reduced to zero on the Tranche B Construction Loan Maturity Date.

- (b) On each Quarterly Date commencing with the first Quarterly Date following the first full fiscal quarter after the Facility A Conversion Date (each such Quarterly Date, a "<u>Tranche A Term Loan Principal Payment Date</u>"), the Borrowers (jointly and severally) shall repay in cash Tranche A Term Loans in an amount (each such amount, a "<u>Tranche A Term Loan Principal Payment</u>") equal to the product of (i) the aggregate outstanding principal amount of the Tranche A Term Loans as of the Facility A Conversion Date multiplied by (ii) the percentage listed opposite such Quarterly Date in **Annex III**.
- (c) On each Quarterly Date commencing with the first Quarterly Date following the first full fiscal quarter after the Facility B Conversion Date (each such Quarterly Date, a "<u>Tranche B Term Loan Principal Payment Date</u>"), the Borrowers (jointly and severally) shall repay in cash Tranche B Term Loans in an amount (each such amount, a "<u>Tranche B Term Loan Principal Payment</u>") equal to the product of (i) the aggregate outstanding principal amount of the Tranche B Term Loans as of the Facility B Conversion Date multiplied by (ii) the percentage listed opposite such Quarterly Date in <u>Annex III</u>.
- (d) The Borrowers (jointly and severally) shall repay in full in cash the aggregate outstanding principal amount of the Tranche A Term Loans together with all interest accrued thereon and all other outstanding Tranche A Obligations on the Tranche A Final Maturity Date.
- (e) The Borrowers (jointly and severally) shall repay in full in cash the aggregate outstanding principal amount of the Tranche B Term Loans together with all interest accrued thereon and all other outstanding Tranche B Obligations on the Tranche B Final Maturity Date.

# 8.2 <u>Voluntary Prepayments</u>.

- (a) Each Borrower shall have the right to prepay the Construction Loans, without premium or penalty, in whole at any time, together with a permanent cancellation of the Construction Loan Commitments, by giving the Administrative Agent at least five (5) Business Days' prior written notice at the Notice Office of its intent to prepay the Construction Loans. Subject to <u>Section 8.7(d)</u>, prepayments of a Eurodollar Loan may only be made pursuant to this <u>Section 8.2(a)</u> on the last day of an Interest Period applicable thereto, unless such Borrower pays all amounts owing under <u>Section 2.17</u> as a result of prepaying such Eurodollar Loan on a day other than the last day of the Interest Period applicable thereto.
- (b) Each Borrower shall have the right to prepay the Term Loans, without premium or penalty, in whole or in part at any time and from time to time after the applicable Conversion Date on the following terms and conditions: (i) prior to the date on which all Tranche A Obligations have been repaid in full and the Tranche A Term Loan Commitments have been

terminated, the Tranche A Borrower may only prepay Tranche A Term Loans; (ii) prior to the date on which all Tranche B Obligations have been repaid in full and the Tranche B Term Loan Commitments have been terminated, the Tranche B Borrower may only prepay Tranche B Term Loans; (iii) such Borrower shall give the Administrative Agent at least seven (7) Business Days' prior written notice at the Notice Office of its intent to prepay Term Loans, the aggregate principal amount of the prepayment, the Types and Tranche of Term Loans to be prepaid and, in the case of Eurodollar Loans, the specific Eurodollar Loan or Eurodollar Loans pursuant to which made (which notice the Administrative Agent shall promptly transmit to each of the Lenders); (iv) such prepayment of Term Loans shall be in an aggregate principal amount of at least \$5,000,000 (or an integral multiple of \$1,000,000 in excess thereof); (v) Subject to Section 8.7(d) prepayments of a Eurodollar Loan may only be made pursuant to this Section 8.2(b) on the last day of an Interest Period applicable thereto, unless such Borrower pays all amounts owing under Section 2.17 as a result of prepaying such Eurodollar Loan on a day other than the last day of the Interest Period applicable thereto; and (vi) each prepayment of Term Loans pursuant to this Section 8.2(b) shall be applied to reduce the applicable Term Loan Principal Payments in inverse chronological order of their due dates.

- 8.3 <u>Mandatory Prepayments</u>. Each Borrower shall make mandatory prepayments of the Loans as follows:
  - (a) <u>Project Document Claims</u>. Each Borrower shall prepay the outstanding Obligations pursuant to <u>Section 5.10(c)</u> hereof.
  - (b) <u>Loss Proceeds</u>. Each Borrower shall prepay the outstanding Obligations to the extent required pursuant to <u>Section 5.11</u> hereof.
- (c) <u>Dispositions</u>. In the event of any Disposition (other than as permitted by <u>Section 6.7</u> hereof), the Borrowers shall cause the Net Disposition Proceeds to be deposited in the applicable Collateral Account pursuant to the Collateral Account Agreement and applied to prepay the outstanding Obligations, such prepayment to be made no later than the date which is three (3) days after the receipt of such Net Disposition Proceeds.
- (d) <u>Debt and Equity Issuances</u>. In the event of any issuance by either Borrower of any Capital Stock (or the receipt of any other equity funding by such Borrower) following the Conversion Date (other than pursuant to the Equity Contribution Agreement) or the issuance by either Borrower of any Indebtedness (other than Indebtedness permitted pursuant to <u>Section 6.2</u>), the Borrowers shall cause the Net Issuance Proceeds to be deposited in the applicable Collateral Account pursuant to the Collateral Account Agreement and applied to prepay the outstanding Obligations, such prepayment to be made no later than the date which is three (3) days after the receipt of such Net Issuance Proceeds; *provided* that no such payment shall cure or otherwise remedy any Default or Event of Default that may be caused by such Borrower issuing such Capital Stock (or receiving such other proceeds of equity funding) or Indebtedness
- (e) <u>Extraordinary Receipts</u>. In the event either Borrower receives any Extraordinary Receipt, the Borrowers shall cause the proceeds of such Extraordinary Receipt to be deposited in the applicable Collateral Account pursuant to the Collateral Account Agreement and applied to prepay the outstanding Obligations, such prepayment to be made no later than the date which is three (3) days after the receipt of such Extraordinary Receipt.

104

- (f) <u>Failure to Satisfy Distribution Release Conditions.</u> Each Borrower shall prepay the outstanding Obligations to the extent required pursuant to <u>Section 4.7(c)</u> of the Collateral Account Agreement.
- (g) Excess Cash Sweep. Commencing on the first (i) Tranche A Term Loan Principal Payment Date occurring after the Facility A Conversion Date, the Tranche A Borrower shall prepay the outstanding Obligations on each Tranche A Term Loan Principal Payment Date in an amount equal to seventy-five percent (75%) of the Tranche A Excess Cash Flow and (ii) Tranche B Term Loan Principal Payment Date occurring after the Facility B Conversion Date, the Tranche B Borrower shall prepay the outstanding Obligations on each Tranche B Term Loan Principal Payment Date in an amount equal to seventy-five percent (75%) of the Tranche B Excess Cash Flow.
- (h) <u>Conversion Date</u>. (i) The Tranche A Borrower shall prepay on the Facility A Conversion Date the outstanding Obligations to the extent required pursuant to <u>Section 4.1(e)(v)</u> of the Collateral Account Agreement and (ii) the Tranche B Borrower shall prepay on the Facility B Conversion Date the outstanding Obligations to the extent required pursuant to <u>Section 4.1(e)(v)</u> of the Collateral Account Agreement.
  - (i) [Reserved].
- (j) <u>Application</u>. Subject to <u>Section 8.7(d)</u>, all prepayments made pursuant to clauses (a) through (i) above shall be applied in accordance with <u>Section 4.11</u> of the Collateral Account Agreement and <u>Section 8.7(c)</u> hereof.

## 8.4 Method and Place of Payment

- (a) Except as set forth in the following sentence or as otherwise specifically provided herein, all payments under this Agreement or any Note shall be made to the Administrative Agent for the account of the Lender or Lenders entitled thereto not later than 11:00 a.m. (New York City time) on the date when due and shall be made in Dollars in immediately available funds as follows:
- , or pursuant to such other instructions as the Administrative Agent shall designate to the Borrowers in writing. Whenever any payment to be made hereunder or under any Note shall be stated to be due on a day which is not a Business Day, the due date thereof shall be extended to the next succeeding Business Day and, with respect to payments of principal, interest shall be payable at the applicable rate during such extension; *provided* that in the event that the day on which any such payment relating to a Eurodollar Loan is due is not a Business Day but is a day of the month after which no further Business Day occurs in such month, then the due date thereof shall be the next preceding Business Day.
- (b) With respect to any repayment of Obligations pursuant to <u>Section 8.1</u> hereof or any mandatory prepayment of Obligations pursuant to <u>Section 8.3</u> hereof, the applicable Borrower may, subject to <u>Section 8.3(i)</u> hereof, designate the Types of Loans which

are to be repaid or prepaid and, in the case of Eurodollar Loans, the specific Borrowing or Borrowings pursuant to which such Eurodollar Loans were made; provided that (i) Subject to Section 8.7(d), repayments and prepayments of Eurodollar Loans may only be made on the last day of an Interest Period applicable thereto unless all such Eurodollar Loans with Interest Periods ending on or prior to such date of required repayment or prepayment and all Base Rate Loans have been paid in full, and (ii) each repayment or prepayment of Loans made pursuant to a single Borrowing shall be applied pro rata among such Loans. In the absence of a designation by the Borrower as described in the preceding sentence, the Administrative Agent shall, subject to the above, make such designation in its sole discretion.

- All computations of interest and Fees hereunder shall be made on the basis of a 360-day year and the actual number of days elapsed; provided that computations of interest on Base Rate Loans calculated using the Prime Rate hereunder shall be made on the basis of a 365- or 366-day year, as the case may be, and the actual number of days elapsed.
- Net Payments. All payments made by the Borrowers hereunder or under any Note shall be made without setoff, counterclaim or other 8.5 defense.

#### 8.6 Fees.

- Construction Loan Commitment Fee. The Borrowers agree (jointly and severally) to pay to the Administrative Agent, for the (a) account of each Lender that is not a Defaulting Lender, (i) a commitment fee from the Closing Date through the end of the Tranche A Construction Loan Availability Period, computed at a rate equal to 0.50% *per annum* on the daily average Modified Unutilized Tranche A Construction Loan Commitment Amount of such Lender during such period and (ii) a commitment fee from the Closing Date through the end of the Tranche B Construction Loan Availability Period, computed at a rate equal to 0.50% per annum on the daily average Modified Unutilized Tranche B Construction Loan Commitment Amount of such Lender during such period. Such fees shall be due and payable in arrears on (x) each Quarterly Date from the Closing Date through the end of the Tranche A Construction Loan Availability Period and through the end of the Tranche B Construction Loan Availability Period, as applicable and (y) the last day of the Tranche A Construction Loan Availability Period and the Tranche B Construction Loan Availability Period, as applicable.
- <u>Upfront Commitment Fees</u>. As set forth in the Warrant Agreement, Sponsor will issue on the Closing Date (as defined in the Warrant Agreement) to GE Energy Financial Services, Inc., on behalf of the Borrowers, warrants to purchase 500,000 shares of Sponsor common stock as an upfront commitment fee.
- (c) Additional Fees. The Borrowers agree (jointly and severally) to pay to the Lenders and the Agents, for their respective accounts, such other fees as have been agreed to in writing by the Borrowers and such Persons.

#### 8.7 Application of Payments; Sharing

Subject to the provisions of this Section 8.7, the Administrative Agent agrees that promptly after its receipt of each payment from or on behalf of the Borrowers in respect of any Obligations, it shall promptly distribute such payment to the Lenders and the other Secured Parties, as applicable, *pro rata* based upon their respective shares, if any, of the Obligations with respect to which such payment was received.

106

- Each of the Lenders agrees that, if it should receive any amount hereunder (whether by voluntary payment, by realization upon security, by the exercise of the right of setoff or banker's lien, by counterclaim or cross action, by the enforcement of any right under the Transaction Documents, or otherwise), which, in any such case, is in excess of its ratable share of payments on account of the Obligations (other than Obligations under any Required Hedging Agreement) obtained by all Lenders, then such Lender receiving such excess payment shall purchase for cash without recourse or warranty from the other Lenders an interest in the Obligations (other than Obligations under any Required Hedging Agreement) of the Borrowers to such Lenders in such amount as shall result in a proportional participation by all the Lenders in such amount; provided, however, that if all or any portion of such excess amount is thereafter recovered from such Lender, such purchase shall be rescinded and the purchase price restored to the extent of such recovery, but without interest.
- All prepayments of Obligations pursuant to Sections 8.3(a) through (i) shall be applied as follows: (i) in the case of the Tranche A Borrower (A) first to prepay all outstanding Tranche A Obligations until all such Obligations have been paid in full and (B) second to prepay all outstanding Tranche B Obligations until all such Obligations have been paid in full; and (ii) in the case of the Tranche B Borrower (A) first to prepay all outstanding Tranche B Obligations until all such Obligations have been paid in full and (B) second to prepay all outstanding Tranche A Obligations until all such Obligations have been paid in full.
- Notwithstanding the foregoing or anything to the contrary herein, but only in respect of any mandatory prepayment under Section 8.3, the Borrowers shall have the right, by giving five Business Days' notice to Administrative Agent, in lieu of prepaying a Eurodollar Loan on a day other than the last day of an Interest Period for such Eurodollar Loan, to deposit or cause Administrative Agent to deposit into the applicable Debt Service and Related Payments Account an amount equal to the Eurodollar Loans to be prepaid. Such funds shall be held in the applicable Debt Service and Related Payments Account until the expiration of the Interest Period applicable to the Eurodollar Loan to be prepaid at which time such amount shall be used to prepay such Eurodollar Loan and any interest accrued on such amount shall be deposited into the Project Revenues Account. The deposit of amounts into the applicable Debt Service and Related Payments Account shall be deemed to satisfy Borrower's obligation to make the relevant mandatory prepayment on its due date, but shall not, however, constitute a prepayment of Loans and all Loans to be prepaid using the proceeds from the applicable Debt Service and Related Payments Account shall continue to accrue interest at the then applicable interest rate for such Loans until actually prepaid. All amounts in the applicable Debt Service and Related Payments Account shall only be invested in Permitted Investments as directed in the sole discretion of Administrative Agent. Borrower shall bear the expense and risk of any such investment.

107

#### SECTION 9. **EVENTS OF DEFAULT AND REMEDIES**

9 1 Events of Default. The occurrence of any of the following events or circumstances shall constitute an "Event of Default" hereunder:

- (a) Any Finance Party (i) shall fail to pay any principal payable pursuant to this Agreement or any other Finance Document, when the same becomes or shall be declared to be due and payable (whether prior to its stated maturity or otherwise) or (ii) shall fail to pay any interest payable pursuant to any Note or any other amount payable pursuant to this Agreement or any other Finance Document, in each case within 3 days when the same becomes or shall be declared to be due and payable (whether prior to its stated maturity or otherwise); or
- Any Finance Party (other than the Sponsor) shall default in the payment when due of any principal of or interest on any of its other Indebtedness having a principal or stated amount in excess of \$500,000 beyond any period of grace specified therein, or in the payment when due of any amount under any Hedging Agreement having a termination value owed by such Finance Party in excess of \$500,000 beyond any period of grace specified therein; or any event specified in any note, agreement, indenture or other document evidencing or relating to any such Indebtedness or any event specified in any such Hedging Agreement shall occur and continue if the effect of the occurrence and continuance of such event is to cause or to permit the holder or holders of such Indebtedness (or a trustee or agent on behalf of such holder or holders) to cause such Indebtedness to become due, or to be prepaid in full (whether by redemption, purchase, offer to purchase or otherwise), prior to its stated maturity or to have the interest rate thereon reset or, in the case of a Hedging Agreement, to permit the payments owing under such Hedging Agreement to be liquidated as the result of the early termination thereof; or the Sponsor or any Affiliate Project Participant shall default in the payment when due of any principal of or interest on any of its other Indebtedness having a principal or stated amount in excess of \$18,000,000 beyond any period of grace specified therein, or in the payment when due of any amount under any Hedging Agreement having a termination value owed by the Sponsor or an Affiliate Project Participant in excess of \$18,000,000 beyond any period of grace specified therein; or any event specified in any note, agreement, indenture or other document evidencing or relating to any such Indebtedness or any event specified in any Hedging Agreement shall occur and continue if the effect of the occurrence and continuance of such event is to cause or to permit the holder or holders of such Indebtedness (or a trustee or agent on behalf of such holder or holders) to cause such Indebtedness to become due, or to be prepaid in full (whether by redemption, purchase, offer to purchase or otherwise), prior to its stated maturity or to have the interest rate thereon reset or, in the case of a Hedging Agreement, to permit the payments owing under such Hedging Agreement to be liquidated as the result of the early termination thereof; or
- (c) Any representation, warranty or certification made (or deemed made) by or on behalf of any Finance Party or any Affiliate Project Participant in this Agreement, any other Finance Document, or in any notice or other certificate, agreement, document, financial statement or other statement delivered pursuant hereto or thereto, shall prove to have been false or misleading in any material respect when made or deemed made; or

- (d) The Borrower shall fail to comply with any term, covenant or provision set forth in <u>Sections 5.1</u> through <u>5.5</u>, <u>5.9</u>, <u>5.11</u>, <u>5.13</u>, <u>5.14</u>, <u>5.17</u> through <u>5.19</u>, <u>5.25</u> or in Article <u>6</u> or <u>7</u> hereof; or
- (e) The Sponsor or Holdings shall fail to comply with any term, covenant or provision set forth in the Equity Contribution Agreement or any other Finance Document to which it is a party beyond any period of grace specified therein; or
- (f) Any Finance Party or any Affiliate Project Participant shall fail to comply with or perform any other agreement or covenant (other than those referred to in the foregoing clauses (a), (d) or (e)) contained in this Agreement or in any other Finance Document and, in the case of any such agreement or covenant, such failure shall continue unremedied for thirty (30) days; or
- (g) Any Finance Party or any Project Participant (other than any Lender Affiliate Project Participant or any Immaterial Project Participant) shall admit in writing its inability to, or be generally unable to, pay its debts as such debts become due; or
- (h) Any Finance Party or any Project Participant (other than any Lender Affiliate Project Participant or any Immaterial Project Participant) shall (i) apply for or consent to the appointment of, or the taking of possession by, a receiver, custodian, trustee or liquidator of itself or of all or a substantial part of its Property, (ii) make a general assignment for the benefit of its creditors, (iii) commence a voluntary case under or file a petition to take advantage of any Bankruptcy Law (as now or hereafter in effect), (iv) fail to controvert in an appropriate manner within sixty (60) days of the filing of, or acquiesce in writing to or file an answer admitting the material allegations of any petition filed against it in an involuntary case under any Bankruptcy Law, (v) take any corporate action for the purpose of effecting any of the foregoing, or (vi) take any action under any other applicable Law which would result in a similar or equivalent outcome as set forth in subclauses (i) through (v) hereof; or
- (i) A proceeding or case shall be commenced, without the application or consent of the Borrower, against any Finance Party or any Project Participant (other than any Lender Affiliate Project Participant or any Immaterial Project Participant) in any court of competent jurisdiction, seeking (i) its liquidation, reorganization, dissolution or winding-up, or the composition or readjustment of its debts, (ii) the appointment of a trustee, receiver, custodian, liquidator or the like of such Person or of all or any substantial part of its Property, or (iii) similar relief in respect of any Finance Party or any Project Participant (other than any Lender Affiliate Project Participant or any Immaterial Project Participant) under any Bankruptcy Law, and such proceeding or case shall continue undismissed, or an order, judgment or decree approving or ordering any of the foregoing shall be entered and continue unstayed and in effect, for a period of sixty (60) or more days; or an order for relief against any Finance Party or any Project Participant (other than any Lender Affiliate Project Participant or any Immaterial Project Participant), shall be entered in an involuntary case under any Bankruptcy Law; or any proceeding or action shall be commenced under any other applicable Law which would result in a similar or equivalent outcome as set forth in subclauses (i) through (iii) hereof; or

109

(j) A final judgment or judgments for the payment of money (exclusive of any amounts fully covered by insurance (less any applicable deductible) and as to which the insurer has not disclaimed responsibility to cover such judgment) in excess of, (A) \$250,000 in the case of any one judgment, or \$500,000 in the aggregate, shall be rendered against the Borrower or (B) \$5,000,000 in the case of any one judgment or \$18,000,000 in the aggregate, shall be rendered against the Sponsor or the LNG Purchaser, in either case, by one or more courts, administrative tribunals or other bodies having jurisdiction against the Borrower, the Sponsor or the LNG Purchaser, as the case may be, and the same shall not be discharged (or provision satisfactory to the Required Lenders shall not be made for such discharge), or a stay of execution thereof shall not be procured, within thirty (30) days from the date of entry thereof and the Borrower, the Sponsor or the LNG Purchaser, as the case may be, shall not, within said period of thirty (30) days, or such longer period during which execution of the same shall have been stayed, appeal therefrom and cause the execution thereof to be stayed during such appeal; or

- (k) Any non-monetary judgment or order shall be rendered against any Finance Party or any Project Participant (other than an Immaterial Project Participant) that has had or could reasonably be expected to have a Material Adverse Effect, and a stay of execution thereof shall not have been obtained within thirty (30) days from the date of entry thereof; or
- (l) Either a Borrower or any ERISA Affiliate of a Borrower shall at any time establish, maintain, contribute to, or be required to contribute to, any Plan subject to Section 412 of the Code or any Multiemployer Plan that could reasonably be expected to result, individually, or in the aggregate, in a Material Adverse Effect or in a direct liability against such Borrower in excess of \$500,000; or
- (m) Any Finance Party or Project Participant (other than any Lender Affiliate Project Participant or an Immaterial Project Participant) shall be dissolved or liquidated (as a matter of law or otherwise); or a proceeding shall be commenced by any Person seeking the dissolution or liquidation of any Finance Party or Project Participant (other than any Lender Affiliate Project Participant or an Immaterial Project Participant) and, if such proceedings were commenced by any Person other than a Finance Party or Project Participant (other than any Lender Affiliate Project Participant or an Immaterial Project Participant), either (i) such proceedings shall not be dismissed without any such dissolution, liquidation or other Material Adverse Effect within thirty (30) days from the date of commencement thereof or (ii) the Required Lenders shall reasonably determine that such proceedings and the effect thereof could reasonably be expected to have a Material Adverse Effect; or
- (n) Any of the Security Documents shall be terminated or shall cease to be in full force and effect, for whatever reason (or any Finance Party shall so assert) or any of the Secured Parties shall cease to have a first priority, perfected Lien on any Collateral (including, without limitation, any Site Real Estate Right), subject only to Permitted Liens, and except, in each case, where the occurrence of any of the foregoing is a result of the gross negligence or willful misconduct of the Collateral Agent; or
- (o) Either Borrower or any Project Participant (other than an Immaterial Project Participant) shall fail to obtain any Necessary Governmental Approval, renew, maintain

or comply with any Necessary Governmental Approval or any such Governmental Approval shall be revoked, terminated, withdrawn, suspended, modified or withheld or shall cease to be in full force and effect or any proceeding is commenced to revoke, terminate, withdraw, suspend, modify or withhold such Government Approval and such proceeding is not terminated within thirty (30) days; unless, in any such case, such failure, revocation, termination, withdrawal, suspension, modification, withholding or failure to be in full force and effect could not reasonably be expected to have a Material Adverse Effect; or

- (i)(A) any Project Document (other than any Immaterial Additional Project Document) shall be terminated (other than by virtue of (p) the scheduled expiration in the ordinary course of such Project Document in accordance with its terms), (B) the Borrower or any Project Participant (other than an Immaterial Project Participant) shall take any action to terminate any Project Document (other than any Immaterial Additional Project Document) to which it is a party or (C) any event under a Project Document (other than any Immaterial Additional Project Document) shall occur and continue if the effect of the occurrence and continuance of such event is to cause or to permit a Project Participant (other than an Immaterial Project Participant) to terminate such Project Document, (ii) any Transaction Document (other than any Immaterial Additional Project Document), or any material provision of any Transaction Document, shall at any time for any reason cease to be valid and binding or in full force and effect or any party thereto (other than a Secured Party) shall so assert in writing, (iii) any Transaction Document (other than any Immaterial Additional Project Document), or any material provision of any Transaction Document (other than any Immaterial Additional Project Document), shall be declared to be null and void, or the validity or enforceability thereof shall be contested by any party thereto (other than a Secured Party) or any Governmental Authority, (iv) the Borrower or any Project Participant (other than an Immaterial Project Participant) shall deny that it has any further liability or obligation under any Transaction Document (other than any Immaterial Project Document), other than in connection with the scheduled expiration in the ordinary course of such Transaction Document in accordance with its terms, or (v) any Project Participant (other than an Immaterial Project Participant) shall default in the observance or performance of any of the material covenants or agreements contained in any Project Document (other than any Immaterial Additional Project Document) and such default shall not be cured within the applicable grace period (if any) contained in such Project Document, or any party to any Project Document (other than an Immaterial Project Participant) assigns or transfers all or any part of its rights and obligations in, to or under such Project Document other than to a transferee that has been approved by the Required Lenders; provided that an Event of Default with respect to any Project Document shall be deemed not to have occurred under this clause (p) if such Project Document is replaced by a Replacement Project Document within sixty (60) days of the occurrence of a Default under this clause (p) with respect to such Project Document; or
- (q) The Facility A Project Completion Date shall not have occurred by the Facility A Date Certain or the Facility B Project Completion Date shall not have occurred by the Facility B Date Certain; or
- (r) Either Borrower shall have voluntarily abandoned the construction or the operation of the Facility owned by such Borrower, such abandonment to be evidenced by such Borrower's cessation of such construction or operation for a period of sixty (60) consecutive days for reasons not associated with the occurrence of a force majeure event; or

- (s) There shall have occurred a casualty event resulting in damage to or destruction or loss of all or substantially all of a Facility or the use thereof; or
- (t) Any Environmental Claim shall have been asserted against a Borrower or any Project Participant or any Release, emission, discharge or disposal of any Hazardous Materials shall have occurred in violation of Environmental Laws which, in either case, could reasonably be expected to have a Material Adverse Effect; or
  - (u) A Change of Control shall have occurred; or
  - (v) An Expropriation Event shall have occurred; or

(w) After the Initial Funding Date in respect of either Tranche, the Borrowers shall fail to have cash, unfunded Commitments and unfunded Equity Contributions pursuant to the Equity Contribution Agreement in an aggregate amount sufficient to pay all anticipated Project Costs necessary to cause, (i) in the case of the Tranche A Borrower, the Facility A Project Completion Date to occur on or before the Facility A Date Certain and (ii) in the case of the Tranche B Borrower the Facility B Project Completion Date to occur on or before the Facility B Date Certain, and such failure shall continue unremedied for thirty (30) days;

provided, that, notwithstanding the foregoing, no Event of Default shall occur with respect to a Borrower as a result of an Event of Default specified paragraphs (a), (b), (d), (j), (k), (l) or (o) above having occurred solely with respect to the other Borrower, unless and until such Event of Default shall continue unremedied for (x) an additional ten (10) days in the case of the Events of Default specified in paragraphs (a) and (b) above, (y) an additional thirty (30) days in the case of (A) the Event of Default specified in paragraph (d) above to the extent arising from a breach of a covenant set forth in Section 5.1, 5.2, 5.4, 5.5, 5.9, 5.17 or 5.18, or (B) the Events of Default specified in paragraphs (j), (k) and (l) above or (z) an additional thirty (30) days in the case of paragraph (o) above, which period, in the case of this clause (z), shall be extended up to an additional sixty (60) days so long as (i) the applicable Borrower shall have fully commenced its actions to cure such Event of Default during the initial thirty (30) day period and (ii) during such sixty (60) day extension period, the applicable Borrower shall at all times be diligently pursuing such cure.

#### 9.2 Acceleration

- (a) If an Event of Default specified in <u>paragraph (h)</u> or (i) of <u>Section 9.1</u> hereof shall occur with respect to either Borrower, automatically all Commitments shall immediately terminate and all Loans (with accrued interest thereon) and all other amounts owing under the Finance Documents (other than the Required Hedging Agreements) shall immediately become due and payable, without presentment, demand, protest or other requirements, all of which are hereby expressly waived by the Borrowers.
- (b) If any Event of Default (other than an Event of Default referred to in Section 9.2(a) hereof) shall occur, then the Administrative Agent (acting at the direction of the Required Lenders) may by notice to the Borrowers either (A) declare the Commitments to be terminated, whereupon all Commitments shall immediately terminate and/or (B) declare the Loans, all accrued and unpaid interest thereon and all other amounts owing to the Lenders under the Finance Documents (other than the Required Hedging Agreements) to be due and payable, whereupon the same shall become immediately due and payable, without presentment, demand, protest or other requirements, all of which are hereby expressly waived by the Borrowers.

112

- (c) Except as expressly provided above in this **Section 9.2**, presentment, demand, protest and all other notices and other formalities of any kind are hereby expressly waived by the Borrowers.
  - 9.3 Other Remedies. Upon the occurrence and during the continuation of an Event of Default:
- (a) Either Agent may exercise any or all rights and remedies at law or in equity (in any combination or order that such Agent may elect), including, without prejudice to such Agent's other rights and remedies, any and all rights and remedies available under any of the Finance Documents (other than any Required Hedging Agreement).
- (b) With the prior written consent of the Required Lenders, which consent may be given or withheld in the Required Lenders' sole discretion, any Contractor or any subcontractor thereof may submit an invoice on behalf of the applicable Borrower, and the Lenders may, in their sole discretion, make payments directly to such Person or any other Person. All sums advanced and disbursed pursuant to this paragraph shall be deemed to be Construction Loans of the applicable Tranche disbursed pursuant to the Finance Documents.
- Each Borrower hereby appoints the Administrative Agent and/or the Collateral Agent (acting on the instruction of the Required Lenders) as the attorney-in-fact of such Borrower, with full power of substitution, and in the name of such Borrower, if the Administrative Agent and/or the Collateral Agent elects to do so at any time upon the occurrence and during the continuance of any Event of Default, to (i) make such changes in the Plans and Specifications, employ such engineers and contractors as may be required, and advance and apply such sums as are necessary, including any proceeds of the Loans, for the purpose of completing the construction of the Facilities, (ii) disburse and directly apply the proceeds of any Loan to the satisfaction of any of such Borrower's obligations hereunder or under any other Finance Document, (iii) hold, use, disburse and apply the Loans for payment of any Project Costs, and/or the payment or performance of any obligation of such Borrower under any Project Document, (iv) advance and incur such expenses as the Required Lenders deem necessary for the completion of construction of the Facilities and to preserve the Projects, (v) disburse any portion of any Loan, from time to time, to Persons other than the Borrowers for the purposes specified herein or in any other Transaction Document, (vi) construct, maintain or operate the Facilities, (vii) execute all applications and certificates in the name of such Borrower as may be required for construction and operation of the Facilities, (viii) endorse the name of such Borrower on any checks or drafts representing proceeds of any insurance policies, or other checks or instruments payable to such Borrower with respect to the Projects, (ix) do every act with respect to the Transaction Documents and the construction and operation of the Projects that such Borrower may do, and (x) prosecute or defend any action or proceedings incident to the Projects. The power-of-attorney granted hereby is a power coupled with an interest and is irrevocable. The Administrative Agent and the Collateral Agent shall have no obligation to undertake any of the foregoing actions, and, if it takes any such action it shall have no liability to the Borrowers to continue the same or for the sufficiency or adequacy thereof. At the request of the Administrative Agent or the Collateral Agent, each Borrower shall ratify all actions taken by the Administrative Agent or the Collateral Agent (as applicable) hereunder.

113

(d) Any funds of any Lender or the Administrative Agent (including the proceeds of any Loans) used for any purpose referred to in this <u>Section 9.3</u>, whether or not in excess (without obligating any Lender to fund any Loans in excess of its Commitment) of the relevant Commitments shall (i) be governed hereby, (ii) constitute a part of the Obligations secured by the Security Documents, (iii) bear interest at the Default Rate, and (iv) be payable upon demand by such Lender or the Administrative Agent, as applicable.

- (a) Each Lender hereby irrevocably (subject to Section 10.9 hereof) designates and appoints General Electric Capital Corporation as Administrative Agent and Collateral Agent of such Lender under this Agreement and the other Finance Documents, and each such Lender irrevocably authorizes each of the Administrative Agent and the Collateral Agent, in such capacity, to take such action on its behalf under the provisions of this Agreement and each other Finance Document to which the Administrative Agent or the Collateral Agent (as applicable) is a party and to exercise such powers and perform such duties as are expressly delegated to the Administrative Agent or the Collateral Agent (as applicable) by the terms of this Agreement or any such other Finance Document, together with such other powers as are reasonably incidental thereto.
- (b) Each Lender hereby irrevocably (subject to Section 6.4 of the Collateral Account Agreement and Section 10.9 hereof) consents to the appointment by the Administrative Agent of the Depositary Agent under the Collateral Account Agreement and each other Finance Document to which it is a party, and hereby authorizes such Depositary Agent to take such actions on its behalf under the provisions of the Finance Documents, and to exercise such powers and perform such duties as are expressly delegated to it by the terms thereof, together with such powers as are reasonably incidental thereto.
- (c) Each of the Lenders authorizes and directs, respectively, each Agent to execute, deliver and perform each of the Finance Documents to which such Agent is or is intended to be a party and each Lender agrees to be bound by all of the agreements of such Agent contained in the Finance Documents.
- (d) Notwithstanding any provision to the contrary contained elsewhere in this Agreement or in any other Finance Document, none of the Agents shall have any duties or responsibilities except those expressly set forth herein and in the other Finance Documents to which such Agent is a party, nor shall any of the Agents have or be deemed to have any fiduciary relationship with any Lender, and no implied covenants, functions, responsibilities, duties, obligations or liabilities shall be read into this Agreement or any other Finance Document or otherwise exist against any of the Agents. Without limiting the generality of the foregoing sentence, the use of the terms "Administrative Agent", "Collateral Agent" and "Depositary

Agent" in this Agreement with reference to the Administrative Agent, the Collateral Agent and the Depositary Agent is not intended to connote any fiduciary or other implied (or express) obligations arising under agency doctrine of any applicable Law. Instead, such terms are used merely as a matter of market custom, and are intended to create or reflect only a relationship between independent contracting parties.

- 10.2 <u>Delegation of Duties</u>. Any of the Agents may execute any of its duties under this Agreement or any other Finance Document by or through agents, employees or attorneys-in-fact and shall be entitled to advice of counsel concerning all matters pertaining to such duties. None of the Agents shall be responsible for the negligence or misconduct of any agent or attorney-in-fact that it selects with reasonable care.
- 10.3 <u>Liability of the Agents</u>. None of the Agents or any Agent-Related Persons shall (a) be liable for any action taken or omitted to be taken by any of them under or in connection with this Agreement or any other Transaction Document (other than any Project Document to which it is a party) or the transactions contemplated hereby including, without limitation, any consents, approvals or objections provided by any Agent under the Collateral Account Agreement (except for its own gross negligence or willful misconduct) or (b) be responsible in any manner to any of the Secured Parties or any other Person for any recital, statement, representation or warranty made by a Borrower or any Affiliate of a Borrower, or any officer thereof, contained in this Agreement or in any other Transaction Document, or in any certificate, report, statement or other document referred to or provided for in, or received by any Agent under or in connection with, this Agreement or any other Transaction Document, or for the value of or title to any Collateral, or the validity, effectiveness, genuineness, enforceability or sufficiency of this Agreement or any other Transaction Document, or for any failure of a Borrower or any other party to any Transaction Document (other than itself) to perform its obligations hereunder or thereunder. None of the Agents or any Agent-Related Person shall be under any obligation to any Secured Party to ascertain or to inquire as to the observance or performance of any of the agreements contained in, or conditions of, this Agreement or any other Transaction Document, or to inspect the Properties, books or records of a Borrower or any Affiliate of a Borrower. Notwithstanding any other provision of the Finance Documents, in no event shall the Collateral Agent be required to foreclose on or take possession of the Collateral, if, in the judgment of the Collateral Agent, such action would be in violation of any applicable Law, rule or regulation, or if the Collateral Agent reasonably believes that such action woul
- Reliance by the Agents. Each of the Agents shall be entitled to rely, and shall be fully protected in relying, upon any note, writing, e-mail, resolution, notice, consent, certificate, affidavit, letter, telegram, facsimile, telecopy, telex, teletype, or telephone message, statement, order or other document or conversation believed by it to be genuine and correct and to have been signed, sent or made by the proper Person or Persons, and upon advice and statements of legal counsel (including counsel to a Borrower), independent accountants and other experts selected by any such Agent. Each Agent may deem and treat the payee of any note as the owner thereof for all purposes unless a written notice of assignment, negotiation or transfer thereof shall have been filed with such Agent. Each of the Agents shall be fully justified in failing or refusing to take any action under this Agreement or any other Transaction Document (other than any Project Document to which it or any of its Affiliates is a party) unless such Agent shall first

115

receive such advice or concurrence of the Required Lenders (or, where unanimous consent of the Lenders is expressly required hereunder, all of the Lenders) as such Agent deems appropriate or such Agent shall first be indemnified to its satisfaction by the Lenders against any and all liability and expense which may be incurred by it by reason of taking or continuing to take any such action. Each of the Agents shall in all cases be fully protected by the Lenders in acting, or in refraining from acting, under this Agreement or any other Transaction Document in accordance with a request or consent of the Required Lenders (or, where unanimous consent of the Lenders is expressly required hereunder, all of the Lenders) and such request and any action taken or failure to act pursuant thereto shall be binding upon all of the Secured Parties.

10.5 Notice of Default. No Agent shall not be deemed to have knowledge or notice of the occurrence of any Default or Event of Default, unless such Agent shall have actually received written notice from a Lender or a Borrower referring to this Agreement, describing such Default or Event of Default and stating that such notice is a "Notice of Default." If any Agent actually receives any such notice of the occurrence of a Default or an Event of Default, it shall give notice thereof to each other Agent and the Lenders. Each Agent shall take such action with respect to such Default or Event of Default as may be reasonably requested by the Required Lenders in accordance with this Section 10; provided, however, that unless and until such Agent has received any such request, such Agent may (but shall not be obligated to) take such action, or refrain from taking such action, with respect to such Default or Event of Default as it shall deem advisable or in the best interest of the Lenders.

10.6 Credit Decision. Each Lender expressly acknowledges that none of the Agents or the Agent-Related Persons has made any representation or warranty to it, and that no act by any of the Agents hereafter taken, including any review of the Projects or of the affairs of the Borrowers, shall be deemed to constitute any representation or warranty by any Agent or Agent-Related Person to any Lender. Each Lender represents to the Agents that it has, independently and without reliance upon any Agent or Agent-Related Person and based on such documents and information as it has deemed appropriate, made its own appraisal of and investigation into the business, prospects, operations, Property, financial and other condition and creditworthiness of each Borrower, the Projects, the value of and title to any Collateral, and all applicable bank regulatory Laws relating to the transactions contemplated hereby, and made its own decision to enter into this Agreement and to extend credit to the Borrowers hereunder. Each Lender also represents that it will, independently and without reliance upon any Agent or Agent-Related Person and based on such documents and information as it shall deem appropriate at the time, continue to make its own credit analysis, appraisals and decisions in taking or not taking action under this Agreement and the other Transaction Documents, and to make such investigations as it deems necessary to inform itself as to the business, prospects, operations, Property, financial and other condition and creditworthiness of each Borrower and the Projects. Except for notices, reports and other documents expressly required pursuant to any Finance Document to be furnished to the Lenders by the Agents, the Agents shall not have any duty or responsibility to provide any Lender with any credit or other information concerning the business, prospects, operations, Property, financial and other condition or creditworthiness of the Projects or of any Borrower which may come into the possession of any Agent or any of the Agent

116

## 10.7 <u>Indemnification of Agents</u>.

- (a) Whether or not the transactions contemplated hereby are consummated, the Lenders shall indemnify upon demand each Agent (in its capacity as such) and the Agent-Related Persons (to the extent not reimbursed by or on behalf of the Borrowers and without limiting the obligation of the Borrowers to do so), *pro rata* in accordance with the aggregate principal amount of the Loans held by such Lender, from and against any and all Indemnified Liabilities; <u>provided</u>, <u>however</u>, that no Lender shall be liable for the payment to any Agent or the Agent-Related Persons of any portion of such Indemnified Liabilities to the extent determined in the final, non-appealable judgment of a court of competent jurisdiction to have resulted solely from such Agent's gross negligence or willful misconduct.
- (b) Without limitation of the foregoing, each Lender shall reimburse each Agent (in its capacity as such) upon demand for its ratable share as provided above of any costs or out-of-pocket expenses (including reasonable Attorney Costs) incurred by such Agent in connection with the preparation, execution, delivery, administration, modification, amendment or enforcement (whether through negotiations, legal proceedings or otherwise) of, or legal advice in respect of rights or responsibilities under, this Agreement, any other Transaction Document or any document contemplated by or referred to herein, to the extent that such Agent is not reimbursed for such expenses by or on behalf of the Borrower.
- (c) The undertakings of the Lenders in this <u>Section 10.7</u> shall survive the payment of all Obligations hereunder and the resignation or replacement of any Agent.
- Agents in Individual Capacities. Each of the Agents and their respective Affiliates may make loans to, issue letters of credit for the account of, accept deposits from, acquire equity interests in and generally engage in any kind of banking, trust, financial advisory, underwriting or other business with each Borrower or its respective Affiliates as though such Agent were not an Agent hereunder and without notice to or consent of the Lenders. The Lenders acknowledge that, pursuant to such activities, an Agent or its Affiliates may receive information regarding a Borrower or its Affiliates (including information that may be subject to confidentiality obligations in favor of such Borrower or such Affiliates) and acknowledge that the Agents shall be under no obligation to provide such information to them. Any Agent which is also a Lender hereunder shall have the same rights and powers under this Agreement as any other Lender and may exercise the same as though it were not an Agent, and the terms "Lender" and "Lenders" shall include such Agent in its individual capacity.

# 10.9 Successor Agents.

- (a) The Administrative Agent may, on the instructions of the Required Lenders, remove and replace the Depositary Agent pursuant to the terms and conditions of the Collateral Account Agreement and may, on the instructions of the Required Banks, direct the Depositary Agent according to the terms of this Agreement and the relevant Finance Documents.
- (b) Subject to the appointment and acceptance of a successor as provided below, following the Conversion Date, the Administrative Agent or the Collateral Agent may resign at any time by giving notice thereof to the Lenders and the Borrowers, and the

117

Administrative Agent or the Collateral Agent may be removed at any time following the Conversion Date with or without cause by the Required Lenders. Upon any such resignation or removal, the Required Lenders shall have the right to appoint a successor Administrative Agent or the Collateral Agent. If no successor Administrative Agent or Collateral Agent shall have been appointed by the Required Lenders and shall have accepted such appointment with thirty (30) days after the resigning Administrative Agent's or Collateral Agent's giving of notice of resignation or the giving of any notice of removal of the Administrative Agent or the Collateral Agent, then the Borrowers may appoint a successor to such Administrative Agent or Collateral Agent. If the Collateral Agent shall resign or be removed pursuant to the foregoing provisions, upon the appointment of a successor Collateral Agent hereunder, the former Collateral Agent shall deliver all Collateral then in its possession to the successor Collateral Agent. Upon its appointment as a successor Administrative Agent or Collateral Agent hereunder, such successor Administrative Agent or Collateral Agent shall thereupon succeed to and become vested with all the rights, powers, privileges and duties of such resigning or removed Administrative Agent or Collateral Agent, and such resigning Administrative Agent or Collateral Agent or removed Administrative Agent or Collateral Agent shall be discharged from its duties and obligations hereunder.

(c) After any Agent's resignation or removal, the provisions of this Section 10 and of Sections 11.1 and 11.2 hereof shall inure to its benefit as to any actions taken or omitted to be taken by it while it was an Agent.

- Administrative Agent, the Collateral Agent and the Depositary Agent incurred after the Closing Date in connection with the preparation, issuance, execution, delivery, filing, recording and administration of this Agreement, the other Transaction Documents, and any other documents which may be delivered in connection herewith or therewith, including all reasonable costs and expenses with respect to the Platform, all reasonable engineers', architects', environmental, fuel, insurance and other consultants' fees (including any such fees incurred in connection with the preparation of any report referred to herein and any inspections pursuant hereto), all reasonable Attorney Costs (including the reasonable fees and expenses of any local counsel) incurred by the Administrative Agent, the Collateral Agent or the Depositary Agent with respect to the transactions contemplated by this Agreement, and (b) all costs and expenses incurred by any Secured Party (including Attorney Costs) in connection with (i) any and all amounts which any Secured Party has paid relative to curing any Event of Default resulting from the acts or omissions of the Borrower under this Agreement or any other Transaction Document, (ii) the enforcement or attempted enforcement of, or the investigation or preservation of any rights or remedies under, this Agreement or any other Transaction Document.
  - 11.2 Indemnity. Whether or not the transactions contemplated hereby are consummated:

(a) The Borrowers shall pay, indemnify, and hold each Secured Party and each of their respective Affiliates and each of its and their officers, directors, employees, counsel, agents and attorneys-in-fact (each, an "Indemnified Person") harmless from and against any and all liabilities, obligations, losses, damages, penalties, claims, actions, judgments, suits, costs, charges, expenses or disbursements (including Attorney Costs) of any kind or nature whatsoever which may at any time (including at any time following repayment of the Loans or the termination, resignation or replacement of any Agent or any Lender) be imposed on, incurred by or asserted against any such Person in any way relating to or arising out of this Agreement or any other Transaction Document (other than with respect to any Lender Affiliate Project Participant, any Lender Affiliate Project Document to which it is a party), including the Security Documents and any other document or instrument contemplated by or referred to herein or therein, or the transactions contemplated hereby and thereby, or any action taken or omitted by any such Person under or in connection with any of the foregoing, including with respect to the exercise by any Secured Party of any of its respective rights or remedies under any of the Finance Documents, and any investigation, litigation or proceeding (including any bankruptcy, insolvency, reorganization or other similar proceeding or appellate proceeding) related to this Agreement or any other Transaction Document (other than with respect to any Lender Affiliate Project Participant, any Lender Affiliate Project Document to which it is a party) or the Loans, or the use of the proceeds thereof, whether or not any Indemnified Person is a party thereto (all of the foregoing, collectively, the "Indemnified Liabilities"); provided that the Borrowers shall have no obligation hereunder to any Indemnified Person with respect to Indemnified Liabilities to the extent determined in the final, non-appealable judgment of a court of

#### (b) <u>Environmental Indemnity</u>

(i) Without in any way limiting the generality of the other provisions contained in this Section 11.2, each Borrower agrees to defend, protect, indemnify, save and hold harmless each Indemnified Person, whether as beneficiary of any of the Security Documents, as a mortgagee in possession, or as successor-in-interest to such Borrower by foreclosure deed or deed in lieu of foreclosure, or otherwise, from and against any and all liabilities, obligations, losses, damages (including foreseeable and unforeseeable consequential damages and punitive claims payable to third parties pursuant to a final judgment in a third-party claim), penalties, fees, claims, actions, judgments, suits, costs, disbursements (including Attorney Costs and consultants' fees and disbursements) and expenses (collectively, "Losses") that may at any time be incurred by, imposed on, asserted or awarded against any such Indemnified Person directly or indirectly based on, or arising out of or resulting from, (A) the actual or alleged presence of Hazardous Materials on, in, under or affecting all or any portion of the Sites or the Projects whether or not the same originates or emanates from the Sites or the Projects or any property adjoining or adjacent to the Sites or from properties at which any Hazardous Materials generated, stored or handled at or by the Projects were Released or disposed of, (B) any Environmental Claim against an Indemnified Person relating to the Sites or the Projects, or (C) any liability incurred by an Indemnified Person under Environmental Laws as a result of the exercise of any Secured Party's rights under any of the provisions of the Security Documents (the "Indemnified Matters"), whether any of the Indemnified Matters arise before or after foreclosure of any of the Security Interests or other taking of

119

title to all or any portion of the Collateral by any Secured Party, including (x) the costs of remediation of Hazardous Materials required at any portion of the Sites or the Projects or any property adjoining or adjacent to the Sites pursuant to Environmental Law or as the result of the resolution of an Environmental Claim, (y) additional costs required to take reasonable precautions to protect against the Release of Hazardous Materials on, in, under or affecting the Sites or the Projects, into the air, any body of water, any other public domain or any surrounding areas, and (z) costs incurred to comply, in connection with all or any portion of the Sites or any surrounding areas, with all applicable Environmental Laws with respect to Hazardous Materials, regardless of whether any of the Indemnified Matters or Losses are caused by or result from the negligence or strict liability of any Indemnified Person, except with respect to any such Indemnified Matter to the extent determined in the final, non-appealable judgment of a court of competent jurisdiction to have resulted solely from the gross negligence or willful misconduct of such Indemnified Person.

- (ii) In no event shall any site visit, observation, or testing by any Indemnified Person (or any representative of any such Person) be deemed to be a representation or warranty that Hazardous Materials are or are not present in, on, or under, the Sites or the Projects, or that there has been or shall be compliance with any Environmental Law. Neither the Borrowers nor any other Person is entitled to rely on any site visit, observation, or testing by any Indemnified Person. No Indemnified Person owes any duty of care to protect the Borrowers or any other Person against, or to inform the Borrowers or any other Person of, any Hazardous Materials or any other adverse condition affecting the Sites, the Facilities or the Projects. No Indemnified Person shall be obligated to disclose to the Borrowers or any other Person any report or findings made as a result of, or in connection with, any site visit, observation, or testing by any Indemnified Person.
- (iii) Notwithstanding the foregoing, the provisions of this Section 11.2(b) shall not (i) require the Borrowers to indemnify any Lender Affiliate Project Participant for any Losses arising as a result of any action or inaction of such Lender Affiliate Project Participant under the applicable Lender Affiliate Project Document or (ii) release any Lender Affiliate Project Participant from any of its obligations under any Lender Affiliate Project Document.

- (c) <u>Survival; Defense.</u> The obligations in this <u>Section 11.2</u> shall survive payment of the Loans and all other Obligations. At the election of any Indemnified Person, each Borrower's indemnification obligations under this <u>Section 11.2</u> shall include the obligation to defend such Indemnified Person using legal counsel satisfactory to such Indemnified Person, at the sole cost and expense of the Borrowers. All amounts owing under this <u>Section 11.2</u> shall be paid within five (5) days after demand.
- (d) <u>Contribution</u>. To the extent that any undertaking in the preceding paragraphs of this <u>Section 11.2</u> may be unenforceable because it is violative of any law or public policy, each Borrower will contribute the maximum portion that it is permitted to pay and satisfy under applicable Law to the payment and satisfaction of such undertaking.

- (e) <u>Settlement</u>. So long as the Borrowers are in compliance with their obligations under this <u>Section 11.2</u>, the Borrowers shall not be liable to any Indemnified Person under this <u>Section 11.2</u> for any settlement made by such Indemnified Person without the Borrowers' consent.
- (f) <u>Limitation on Liability</u>. Each Borrower agrees not to assert any claim against any Agent, any Lender or any of their Affiliates, or any of their respective officers, directors, employees, agents, and advisors, on any theory of liability, for special, indirect, consequential or punitive damages arising out of or otherwise relating to this Agreement or any of the other Transaction Documents (other than against a Lender Affiliate Project Party pursuant to a Lender Affiliate Project Document), the actual or proposed use of the proceeds of the Loans or any of the transactions contemplated by the Transaction Documents (other than against a Lender Affiliate Project Party pursuant to a Lender Affiliate Project Document).

# 11.3 Notices.

- (a) All notices, requests, demands, documents, reports, information, financial statements, certificates, materials and other communications (collectively, the "Communications") provided for hereunder shall be in writing including, unless the context expressly otherwise provides, by facsimile or other electronic transmission (including by email and, with respect to Communications by a Borrower to the Lenders, as a result of the same having been posted on the Platform by the Administrative Agent following delivery thereof by a Borrower to the Administrative Agent as provided herein); provided that any matter transmitted by a Borrower by facsimile or other electronic transmission (other than any such transmissions which were posted on the Platform for the benefit of the Lenders) (i) shall be immediately confirmed by a telephone call to the recipient at the number specified on the applicable signature page hereof and (ii) shall be followed promptly by a hard copy original thereof to the Administrative Agent by express courier. All Communications shall be transmitted or delivered to the address, facsimile number or email address specified on the applicable signature page hereof or to such other address, facsimile number or email address as shall be designated by such party in a written notice to the other parties hereto.
- (b) All Communications (i) sent by express courier will be effective upon delivery to or refusal to accept delivery by the addressee and (ii) transmitted by facsimile or other electronic transmission will be effective when sent and confirmation of receipt received by the sender; except that all Communications to any Agent shall not be effective until actually received.
- (c) Each Borrower acknowledges and agrees that any agreement of the Secured Parties to receive certain Communications by telephone, facsimile or other electronic transmission is solely for the convenience and at the request of such Borrower. The Secured Parties shall be entitled to rely on the authority of any Person purporting to be a Person authorized by a Borrower to give such Communications and the Secured Parties shall not have any liability to the Borrowers or other Person on account of any action taken or not taken by any of the Secured Parties in reliance upon such telephonic, facsimile or other electronically transmitted Communications.

- (d) Each Borrower further acknowledges and agrees that any Agent may make any Communications received from the Borrowers available to the Lenders by posting such Communications on a secure electronic transmission system (the "Platform"). The Platform is provided "as is" and "as available." The Agents do not warrant the accuracy or completeness of any Communications or the adequacy of the Platform and expressly disclaim liability for errors or omissions in any Communications. No warranty of any kind, express, implied or statutory, including, without limitation, any warranty of merchantability, fitness for a particular purpose, non-infringement of third party rights or freedom from viruses or other code defects, is made by any Agent in connection with any Communications or the Platform. In no event shall any Agent or any of its officers, directors, employees, counsel, agents, attorneys-in-fact and Affiliates have any liability to the Borrowers, any Lender or any other Person for damages of any kind, including direct or indirect, special, incidental or consequential damages, losses or expenses (whether in tort, contract or otherwise) arising out of a Borrower's or such Agent's transmission of any Communications through the Internet, except to the extent the liability of such Person is found in a final non-appealable judgment by a court of competent jurisdiction to have resulted solely from such Person's gross negligence or willful misconduct.
- 11.4 <u>Benefit of Agreement</u>. This Agreement shall be binding upon and inure to the benefit of and be enforceable by the respective successors and permitted assigns of the parties hereto. The Borrowers may not assign or otherwise transfer any of its rights or obligations under this Agreement or any of the other Finance Documents.
- 11.5 No Waiver; Remedies Cumulative. No failure or delay on the part of any of the Secured Parties or the holder of any Note or Loan in exercising any right, power or privilege hereunder or under any other Finance Document and no course of dealing between any Borrower and any Secured Party or the holder of any Note or Loan shall operate as a waiver thereof, nor shall any single or partial exercise of any right, power or privilege hereunder or under any other Finance Document preclude any other or further exercise thereof or the exercise of any other right, power or privilege hereunder or thereunder. No notice to or demand on a Borrower in any case shall entitle such Borrower to any other or further notice or demand in similar or other circumstances or constitute a waiver of the rights of any Secured Party or the holder of any Note or Loan to take any other or further action in any circumstances without notice or demand. All remedies, either under this Agreement or any other Finance Document or pursuant to any applicable Law or otherwise afforded to any Secured Party, shall be cumulative and not alternative.
- 11.6 <u>No Third Party Beneficiaries</u>. The agreement of each Lender to make extensions of credit to the Borrowers on the terms and conditions set forth in this Agreement and the other Finance Documents is solely for the benefit of the Borrowers, and no other Person (including any other Project Participant, or any contractor, sub-contractor, supplier, worker, carrier, warehouseman, materialman or vendor furnishing supplies, goods or services to or for

- 11.7 <u>Reinstatement.</u> To the extent that any Secured Party receives any payment by or on behalf of a Borrower, which payment or any part thereof is subsequently invalidated, declared to be fraudulent or preferential, set aside or required to be repaid to such Borrower or to its estate, trustee, receiver, custodian or any other party under any Bankruptcy Law or otherwise, then to the extent of the amount so required to be repaid, the obligation or part thereof which has been paid, reduced or satisfied by the amount so repaid shall be reinstated by the amount so repaid and shall be included within the Obligations as of the date such initial payment, reduction or satisfaction occurred.
- 11.8 <u>No Immunity</u>. To the extent that a Borrower may be entitled, in any jurisdiction in which judicial proceedings may at any time be commenced with respect to this Agreement or any other Finance Document, to claim for itself or its revenues, assets or Properties any immunity from suit, the jurisdiction of any court, attachment prior to judgment, attachment in aid of execution of judgment, set-off execution of a judgment or any other legal process, and to the extent that in any such jurisdiction there may be attributed to such Person such an immunity (whether or not claimed), the Borrower hereby irrevocably agrees not to claim and hereby irrevocably waives such immunity to the fullest extent permitted by the law of the applicable jurisdiction.
- 11.9 <u>Counterparts</u>. This Agreement may be executed in any number of counterparts and by the different parties hereto on separate counterparts, each of which when so executed and delivered shall be an original, but all of which shall together constitute one and the same instrument.

# 11.10 Amendment or Waiver.

(a) No provision of this Agreement or any other Finance Document may be amended, supplemented, modified or waived, except by a written instrument signed by the Required Lenders (taking into account the provisions in Section 2.21 hereof) and each Finance Party party thereto, and, to the extent its rights or obligations may be affected thereby, the Agent or Agents party thereto. Notwithstanding the foregoing provisions, no such waiver and no such amendment, supplement or modification shall (i) increase or extend the Commitment of any Lender (it being understood that waivers or modifications of conditions precedent, covenants, Defaults or Events of Default or of a mandatory reduction in the Construction Loan Commitments of all Lenders or the Term Loan Commitments of all Lenders, shall not constitute an increase or extension of any Commitment of any Lender), without the prior written consent of such Lender, (ii) postpone or delay any date fixed by this Agreement or any other Finance Document for any payment of principal, interest, fees or other amounts due to any Lender hereunder or under any other Finance Document, without the prior written consent of such Lender, (iii) reduce the principal of, or the rate of interest (other than the application of Default Interest, which may be waived by the Required Lenders) specified in any Finance Document on, any Loan of any Lender, without the prior written consent of such Lender, (iv) release all or substantially all of the Collateral except as shall be otherwise provided in any Security Document or other Finance Document or consent to the assignment or transfer by either Borrower of any of its respective obligations under this Agreement or any other Finance Document, without the prior written consent of each Lender, (vi) amend, modify or waive any condition precedent contained in Section 3 hereof, without the prior written consent of each Lender, (vi) amend, modify or

123

waive any provision of this Section 11.10 or Section 11.1 or 11.2 hereof, without the prior written consent of each Lender, or (vii) reduce the percentage specified in or otherwise amend the definition of Required Lenders without the prior written consent of each Lender; provided that no provision of this Agreement or any other Finance Document may be amended, supplemented, modified or waived without the prior written consent of each Secured Swap Counterparty or, in the case of a Required Hedging Agreement provided or arranged by GE Capital or an Affiliate of GE Capital, GE Capital, if such amendment, supplement, modification or waiver would alter the ratable treatment of the Obligations owing to any Secured Swap Counterparty arising under Required Hedging Agreements resulting in such Obligations being junior in right of payment to principal on the Loans or would result in such Obligations becoming unsecured (other than releases of Liens affecting all Lenders and otherwise permitted in accordance with the terms hereof) in each case in a a manner adverse to any Secured Swap Counterparty; provided, further, that any reference to "Finance Documents" in this Section 11.10(a) shall be deemed to exclude any Required Hedging Agreement.

(b) Any waiver and any amendment, supplement or modification made or entered into in accordance with Section <u>11.10(a)</u> hereof shall be binding upon the Borrowers and the Secured Parties.

# 11.11 <u>Assignments, Participations, etc.</u>

Prior to the occurrence of the Conversion Date with respect to both Tranche A Loans and Tranche B Loans, so long as no Event of Default has occurred and is continuing, each Lender agrees that it shall not be permitted to assign its interest in the Loans or its other rights and obligations hereunder and under the other Finance Documents. On or after the occurrence of such Conversion Dates (and at any time after the occurrence and during the continuance of an Event of Default), any Lender may, with the written consent of the Administrative Agent and, unless an Event of Default has occurred and is continuing, the Borrowers, in each case not to be unreasonably withheld, at any time assign to one or more Eligible Assignees (each, an "Assignee") (provided that no written consent of the Administrative Agent shall be required in connection with any assignment and delegation by a Lender to (i) an entity that is an Affiliate of such Lender or (ii) another Lender and provided that if any Borrower fails to respond to any such request for written consent within five (5) Business Days, such Borrower shall be deemed to have consented to such assignment) all or any part of any Loan and the other rights and obligations of such Lender hereunder and under the other Finance Documents; provided that (A) (x) each such assignment by a Lender of its Loans, its Notes or its Commitments shall be made in such a manner so that the same portion of its Loans, its Notes and Commitments is assigned to the Assignee, (y) each such assignment by a Lender of its Loans, its Notes or its Commitments of a particular Tranche shall be made in such a manner so that the same portion of its Loans, its Notes and Commitments of each other Tranche is assigned to the Assignee and (z) Construction Loans, Construction Notes, and Construction Loan Commitments of a particular Tranche shall only be assigned contemporaneously with Term Loan Commitments of the same Tranche, (B) in the case of an assignment of any part of a Loan or share of an assignor's commitments to any Assignee, such assignment shall not be for an amount less than \$5,000,000 or, if less, such assignor's entire interest in the Loans and/or the Commitments (provided that, in connection with simultaneous assignments to two or more related Approved Funds, such Approved Funds shall be treated as one Assignee for purposes of determining

compliance with the minimum assignment size referred to above), and (C) the Borrower and the Agents may continue to deal solely and directly with the assigning Lender in connection with the interest so assigned until (1) written notice of such assignment, together with payment instructions, addresses and related information with respect to the Assignee, shall have been given to the Borrowers and the Administrative Agent by such assigning Lender and the Assignee, (2) the assigning Lender or Assignee has paid to the Administrative Agent a processing fee in the amount of \$3,500 (provided that only one processing fee shall be payable in connection with simultaneous assignments to two or more related Approved Funds), and (3) the assigning Lender shall have delivered to the Borrowers and the Administrative Agent an Assignment and Acceptance substantially in the form of Exhibit F hereto (an "Assignment and Acceptance") with respect to such assignment from the assigning Lender.

- (b) From and after the date that the Administrative Agent notifies the assigning Lender and the Borrowers that it has received (and provided its consent with respect to) an executed Assignment and Acceptance and payment of the above-referenced processing fee, (i) the Assignee thereunder shall be a party hereto and, to the extent that rights and obligations hereunder have been assigned to it pursuant to such Assignment and Acceptance, shall have the rights and obligations of a Lender hereunder and under the other Finance Documents, and this Agreement shall be deemed to be amended to the extent, but only to the extent, necessary to effect the addition of the Assignee, and any reference to the assigning Lender hereunder or under the other Finance Documents shall thereafter refer to such Lender and to the Assignee to the extent of their respective interests and (ii) the assigning Lender shall, to the extent that rights and obligations hereunder and under the other Finance Documents have been assigned by it pursuant to such Assignment and Acceptance, relinquish its rights and be released from its obligations under the Finance Documents.
- (c) Within five (5) Business Days after its receipt of notice from the Administrative Agent that (i) it has received an executed Assignment and Acceptance and payment of the processing fee and (ii) the Assignee has required a Note or Notes evidencing the Assignee's assigned Commitments and Loans, the Borrowers shall execute and deliver to the Administrative Agent a new Note or Notes evidencing the Assignee's assigned Commitments and Loans and, if the assigning Lender has retained a portion of its Loans, a replacement Note or Notes reflecting the Commitment and the principal amount of the Loans retained by the assigning Lender (such Note or Notes to be in exchange for, but not in payment of, the Note or Notes held by such Lender).
- (d) Notwithstanding the foregoing provisions of this <u>Section 11.11</u> or any other provision of this Agreement, the Administrative Agent has the right, but not the obligation, to effectuate assignments of Loans and Commitments via an electronic settlement system acceptable to the Administrative Agent as designated in writing from time to time to the Lenders by the Administrative Agent (the "<u>Settlement Service</u>"). At any time when the Administrative Agent elects, in its sole discretion, to implement such Settlement Service, each such assignment shall be effected by the assigning Lender and proposed Assignee pursuant to the procedures then in effect under the Settlement Service, which procedures shall be consistent with the other provisions of this <u>Section 11.11</u>. Each assigning Lender and proposed Eligible Assignee shall comply with the requirements of the Settlement Service in connection with effecting any assignment of Loans and Commitments pursuant to the Settlement Service. With the prior

125

approval of the Administrative Agent, the Administrative Agent's approval of such Eligible Assignee shall be deemed to have been automatically granted with respect to any transfer effected through the Settlement Service. Assignments and assumptions of the Loans and the Commitments shall be effected by the provisions set forth herein until the Administrative Agent notifies the Lenders of the Settlement Service as set forth herein.

- (e) In connection with any assignment of rights and obligations of any Defaulting Lender hereunder, no such assignment shall be effective unless and until, in addition to the other conditions thereto set forth herein, the parties to the assignment shall make such additional payments to the Administrative Agent in an aggregate amount sufficient, upon distribution thereof as appropriate (which may be outright payment, purchases by the assignee of participations or subparticipations, or other compensating actions, including funding, with the consent of the Borrowers and the Administrative Agent, the applicable pro rata share of Loans previously requested but not funded by the Defaulting Lender, to each of which the applicable assignee and assignor hereby irrevocably consent), to (A) pay and satisfy in full all payment liabilities then owed by such Defaulting Lender to the Administrative Agent and each other Lender hereunder (and interest accrued thereon), and (B) acquire (and fund as appropriate) its full pro rata share of all Loans. Notwithstanding the foregoing, in the event that any assignment of rights and obligations of any Defaulting Lender hereunder shall become effective under applicable law without compliance with the provisions of this paragraph, then the assignee of such interest shall be deemed to be a Defaulting Lender for all purposes of this Agreement until such compliance occurs.
- (f) Any Lender may at any time on or after the occurrence of the Conversion Date with respect to both Tranche A Loans and Tranche B Loans, without the consent of, or notice to, the applicable Borrower or the Administrative Agent, sell participations to any Person (other than a natural Person or the Borrowers or any of the Borrowers' Affiliates or Subsidiaries) (each, a "Participant") in all or a portion of such Lender's rights and/or obligations under this Agreement (including all or a portion of its Commitment and/or the Loans owing to it); provided that (i) such Lender's obligations under this Agreement shall remain unchanged, (ii) such Lender shall remain solely responsible to the other parties hereto for the performance of such obligations, and (iii) the Borrower, the Administrative Agent and the Lenders shall continue to deal solely and directly with such Lender in connection with obligations and obligations under this Agreement. For the avoidance of doubt, each Lender shall be responsible for the indemnity under Sections 2.14(d) and 10.7 with respect to any payments made by such Lender to its Participant(s). Any agreement or instrument pursuant to which a Lender sells such a participation shall provide that such Lender shall retain the sole right to enforce this Agreement and to approve any amendment, modification or waiver of any provision of this Agreement; provided that such agreement or instrument may provide that such Lender will not, without the consent of the Participant, agree to any amendment, modification or waiver with respect to Section 11.10 that affects such Participant. The Borrowers agree that each Participant shall be entitled to the benefits of Sections 2.17, and Section 2.14 (subject to the requirements and limitations therein, including the requirements under Section 2.14(f) (it being understood that the documentation required under Section 2.14(f) shall be delivered to the participant (A) agrees to be subject to the provisions of Section 2.20 as if it were an assignee under paragraph

acquired the applicable participation. Each Lender that sells a participation agrees, at the Borrower's request and expense, to use reasonable efforts to cooperate with the Borrower to effectuate the provisions of Section 2.20 with respect to any Participant. To the extent permitted by law, each Participant also shall be entitled to the benefits of Section 11.14 as though it were a Lender; provided that such Participant agrees to be subject to Section 8.7 as though it were a Lender. Each Lender that sells a participation shall, acting solely for this purpose as an agent of the Borrowers, maintain a register on which it enters the name and address of each Participant and the principal amounts (and stated interest) of each Participant's interest in the Loans or other obligations under the Finance Documents (the "Participant Register"); provided that no Lender shall have any obligation to disclose all or any portion of the Participant Register (including the identity of any Participant or any information relating to a Participant's interest in any commitments, loans, letters of credit or its other obligations under any Finance Document) to any Person except to the extent that such disclosure is necessary to establish that such commitment, loan, letter of credit or other obligation is in registered form under Section 5f.103-1(c) of the United States Treasury Regulations. The entries in the Participant Register shall be conclusive absent manifest error, and such Lender shall treat each Person whose name is recorded in the Participant Register as the owner of such participation for all purposes of this Agreement notwithstanding any notice to the contrary. For the avoidance of doubt, the Administrative Agent (in its capacity as Administrative Agent) shall have no responsibility for maintaining a Participant Register.

- (g) Notwithstanding any other provision contained in this Agreement or any other Transaction Document to the contrary, any Lender may assign all or any portion of the Loans held by it as collateral security; <u>provided</u> that any payment in respect of such assigned Loans or Notes made by the Borrowers to or for the account of the assigning or pledging Lender in accordance with the terms of this Agreement shall satisfy the Borrowers' obligations hereunder in respect to such assigned Loans or Notes to the extent of such payment. No such assignment shall release the assigning Lender from its obligations hereunder.
- 11.12 <u>Survival</u>. All indemnities set forth herein, including <u>Sections 11.1</u>, <u>11.2</u>, <u>11.13</u> and <u>11.18</u> hereof, shall survive the execution and delivery of this Agreement and the Notes and the making and repayment of the Loans. In addition, each representation and warranty made or deemed to be made pursuant hereto shall survive the making of such representation and warranty, and no Lender shall be deemed to have waived, by reason of making any extension of credit, any Default or Event of Default which may arise by reason of such representation or warranty proving to have been false or misleading, notwithstanding that such Lender may have had notice or knowledge or reason to believe that such representation or warranty was false or misleading at the time such extension of credit was made.
- 11.13 WAIVER OF JURY TRIAL. EACH OF THE PARTIES HERETO HEREBY KNOWINGLY, VOLUNTARILY AND INTENTIONALLY WAIVES THE RIGHT ANY OF THEM MAY HAVE TO A TRIAL BY JURY IN RESPECT OF ANY LITIGATION BASED ON, OR ARISING OUT OF, UNDER OR IN CONNECTION WITH, THIS AGREEMENT, THE NOTES OR ANY OTHER FINANCE DOCUMENT, OR ANY COURSE OF CONDUCT, COURSE OF DEALING, STATEMENTS (WHETHER VERBAL OR WRITTEN) OR ACTIONS OF ANY PARTY RELATING HERETO OR THERETO. THIS PROVISION IS A MATERIAL INDUCEMENT FOR THE SECURED PARTIES TO ENTER INTO THIS AGREEMENT.

127

- 11.14 Right of Set-off. In addition to any rights now or hereafter granted under applicable Law or otherwise, and not by way of limitation of any such rights, upon the occurrence of an Event of Default, each Lender is hereby authorized at any time or from time to time, without presentment, demand, protest or other notice of any kind to the Borrowers or to any other Person, any such notice being hereby expressly waived, to set off and to appropriate and apply any and all deposits (general or special) and any other Indebtedness at any time held or owing by such Lender (including by branches and agencies of any Lender wherever located), to or for the credit or the account of a Borrower against and on account of the Obligations or liabilities of such Borrower to such Lender under this Agreement or any of the other Finance Documents, including all claims of any nature or description arising out of or connected with this Agreement or any other Finance Document, irrespective of whether such Lender shall have made any demand hereunder and although said Obligations, liabilities or claims, or any of them, shall be contingent or unmatured. Each Secured Party agrees promptly to notify the Borrowers and the Administrative Agent after any such application made by such Secured Party, provided that the failure to give such notice shall not affect the validity of such application. Notwithstanding the foregoing, in the event that any Defaulting Lender shall exercise any right of setoff pursuant to this Section 11.14, (a) all amounts so set off shall be promptly paid over to the Administrative Agent for further application in accordance with the provisions of Section 2.21 and, pending such payment, shall be segregated by such Defaulting Lender from its other funds and deemed held in trust for the benefit of the Administrative Agent and the Lenders, and (b) the Defaulting Lender shall provide promptly to the Administrative Agent a statement describing in reasonable detail the Obligations owing to such Defaulting Lender sa to which
- 11.15 <u>Severability</u>. Any provision hereof which is prohibited or unenforceable in any jurisdiction shall, as to such jurisdiction, be ineffective to the extent of such prohibition or unenforceability without invalidating the remaining provisions hereof and without affecting the validity or enforceability of any provision in any other jurisdiction.
- 11.16 <u>Domicile of Loans</u>. Subject to Sections <u>2.14(e)</u>, <u>2.15(d)</u> and <u>2.16(c)</u>, each Lender may transfer and carry its Loans at, to or for the account of any office, Subsidiary or Affiliate of such Lender.
  - 11.17 [Reserved]
  - 11.18 Governing Law; Submission to Jurisdiction
- (a) THIS AGREEMENT SHALL BE GOVERNED BY, AND CONSTRUED IN ACCORDANCE WITH, THE LAWS OF THE STATE OF NEW YORK.
- (b) Each Borrower hereby submits to the nonexclusive jurisdiction of the United States District Court for the Southern District of New York and of any New York State

128

court sitting in New York County for the purposes of all legal proceedings arising out of or relating to this Agreement, any other Finance Document or the transactions contemplated hereby or thereby. Each Borrower hereby irrevocably waives, to the fullest extent permitted by applicable Law, any objection which it may now or hereafter have to the laying of the venue of any such proceeding brought in such a court and any claim that any such proceeding brought in such a court has been brought in an inconvenient forum. Nothing herein shall affect the right to serve process in any other manner permitted by applicable

Law or any right to bring legal action or proceedings in any other competent jurisdiction, including judicial or non-judicial foreclosure of real property interests which are part of the Collateral. To the extent permitted by applicable Law, each Borrower further irrevocably agrees to the service of process of any of the aforementioned courts in any suit, action or proceeding by the mailing of copies thereof by certified mail, postage prepaid, return receipt requested, to such Borrower at the address referenced in <u>Section 11.3</u> hereof, such service to be effective upon the date indicated on the postal receipt returned from such Borrower.

- 11.19 <u>Patriot Act Notification</u>. The Administrative Agent (for itself and not on behalf of any Lender) and each Lender hereby notifies each Borrower that pursuant to the requirements of the Patriot Act, it is required to obtain, verify, and record certain information and documentation that identifies such Borrower, which information includes the name and address of such Borrower and such other information that will allow the Administrative Agent or such Lender, as applicable, to identify such Borrower in accordance with the Patriot Act.
- 11.20 <u>Complete Agreement</u>. THIS AGREEMENT AND THE OTHER FINANCE DOCUMENTS REPRESENT THE FINAL AND COMPLETE AGREEMENT OF THE PARTIES HERETO, AND ALL PRIOR NEGOTIATIONS, REPRESENTATIONS, UNDERSTANDINGS, WRITINGS AND STATEMENTS OF ANY NATURE ARE HEREBY SUPERSEDED IN THEIR ENTIRETY BY THE TERMS OF THIS AGREEMENT AND THE OTHER FINANCE DOCUMENTS.
- 11.21 <u>No Strict Construction</u>. The parties hereto have participated jointly in the negotiation and drafting of this Agreement. In the event an ambiguity or question of intent or interpretation arises, this Agreement shall be construed as if drafted jointly by the parties hereto and no presumption or burden of proof shall arise favoring or disfavoring any party by virtue of the authorship of any provisions of this Agreement.
- Release of Collateral. Notwithstanding anything to the contrary contained herein or in any other Finance Document, the Administrative Agent and Collateral Agent are hereby irrevocably authorized by each Secured Party (without requirement of notice to or consent of any Secured Party or Affiliate thereof except as expressly required by Section 11.10) to take any action requested by the Borrowers having the effect of releasing any Collateral to the extent necessary or appropriate to permit consummation of any transaction not prohibited by any Finance Document or that has been consented to in accordance with Section 11.10.

# 11.23 Confidentiality.

(a) Each Secured Party agrees to keep confidential all information provided to it by any Finance Party or other Secured Party pursuant to or in connection with any Transaction

129

Document; <u>provided</u> that nothing herein shall prevent any such Person from disclosing any such information (i) to a Secured Party or any Affiliate thereof, (ii) subject to an agreement to comply with the provisions of this <u>Section 11.23</u>, to any actual or prospective Eligible Assignee or Participant, (iii) to its employees, directors, agents, attorneys, accountants and other professional advisors or those of any of its Affiliates (it being understood that the Persons to whom such disclosure is made will be informed of the confidential nature of such information and instructed to keep such information confidential), (iv) upon the request or demand of any Governmental Authority or regulatory authority (including any self-regulatory authority), (v) in response to any order of any court or other Governmental Authority or regulatory authority (including any self-regulatory authority) or as may otherwise be required pursuant to any applicable Law, (vi) if requested or required to do so in connection with any litigation or similar proceeding, (vii) that has been publicly disclosed other than as a result of a breach of this <u>Section 11.23</u>, (viii) to the National Association of Insurance Commissioners or any similar organization or any nationally recognized rating agency that requires access to information about a Secured Party's investment portfolio in connection with ratings issued with respect to such Secured Party, (ix) to any insurance or reinsurance company for the purpose of obtaining insurance in respect of its Loans or Commitments hereunder (provided such persons are informed of the confidential nature of such information), (x) in connection with the exercise of any remedy hereunder or under any other Transaction Document, (xi) solely with respect to the existence of the transactions contemplated in the Finance Documents, the parties thereto and the maximum amounts thereof, to market data collectors and similar service providers to the lending industry, and (xii) solely with respect to the existence of the t

- (b) Each Secured Party acknowledges that information furnished to it pursuant to this Agreement or the other Transaction Documents may include material non-public information concerning the Borrowers and their Affiliates and their respective directors, officers, employees, agents or advisors or their respective securities, and confirms that it has developed compliance procedures regarding the use of material non-public information and that it will handle such material non-public information in accordance with those procedures and applicable law, including Federal and state securities laws.
- (c) Notwithstanding the foregoing clause (a) of this Section, any party to this Agreement (and each Affiliate, director, officer, employee, agent or representative of the foregoing or such Affiliate) may disclose to any and all persons, without limitation of any kind, the tax treatment and tax structure of the transactions contemplated herein and all materials of any kind (including opinions or other tax analyses) that are provided to such party relating to such tax treatment or tax structure. The foregoing language is not intended to waive any confidentiality obligations otherwise applicable under this Agreement except with respect to the information and materials specifically referenced in the preceding sentence. This authorization does not extend to disclosure of any other information, including (i) the identity of participants or potential participants in the transactions contemplated herein (and no party shall disclose any information relating to such tax treatment and tax structure to the extent nondisclosure is reasonably necessary in order to comply with applicable securities laws (it being understood that, for such purpose, the tax treatment of the transactions contemplated by this Agreement is the

130

purported or claimed U.S. Federal income tax treatment of such transactions and the tax structure of such transaction is any fact that may be relevant to understanding the purported or claimed U.S. Federal income tax treatment of such transactions)), (ii) the existence or status of any negotiations, or (iii) any financial, business, legal or personal information of or regarding a party or its affiliates, or of or regarding any participants or potential participants in the transactions contemplated herein (or any of their respective affiliates), in each case to the extent such other information is not related to the tax treatment or tax structure of the transactions contemplated herein.

- (a) Notwithstanding anything in this Agreement or any other Finance Document to the contrary, each Borrower hereby accepts joint and several liability hereunder and under the other Finance Documents in consideration of the financial accommodations to be provided by the Secured Parties under this Agreement and the other Finance Documents, for the mutual benefit, directly and indirectly, of each Borrower and in consideration of the undertakings of the other Borrower to accept joint and several liability for the Obligations. Each Borrower, jointly and severally, hereby irrevocably and unconditionally accepts, not merely as a surety but also as a co-debtor, joint and several liability with the other Borrower, with respect to the payment and performance of all of the Obligations (including, without limitation, any Obligations arising under this Section 11.24), it being the intention of the parties hereto that all of the Obligations shall be the joint and several obligations of each Borrower without preferences or distinction between them. If and to the extent either Borrower shall fail to make any payment with respect to any of the Obligations as and when due or to perform any of the Obligations in accordance with the terms thereof, then in each such event, the other Borrower will make such payment with respect to, or perform, such Obligation. Subject to the terms and conditions hereof, the Obligations of each Borrower under the provisions of this Section 11.24 constitute the absolute and unconditional, full recourse Obligations of each Borrower, enforceable against each such Borrower to the full extent of its properties and assets, irrespective of the validity, regularity or enforceability of this Agreement, the other Finance Documents or any other circumstances whatsoever.
- (b) The provisions of this <u>Section 11.24</u> are made for the benefit of the Secured Parties and their successors and assigns, and may be enforced by them from time to time against either or both of the Borrowers as often as occasion therefor may arise and without requirement on the part of the Secured Parties or such successors or assigns first to marshal any of their claims or to exercise any of their rights against the other Borrower or to exhaust any remedies available to them against the other Borrower or to resort to any other source or means of obtaining payment of any of the Obligations hereunder or to elect any other remedy. The provisions of this <u>Section 11.24</u> shall remain in effect until all of the Obligations shall have been paid in full or otherwise fully satisfied.
- (c) Each Borrower hereby agrees that it will not enforce any of its rights of contribution or subrogation against the other Borrower with respect to any liability incurred by it hereunder or under any of the other Finance Documents, or any payments made by it to the Secured Parties with respect to any of the Obligations or any Collateral, until such time as all of the Obligations have been paid in full in cash. Any claim which either Borrower may have

against the other Borrower with respect to any payments to the Secured Parties hereunder or under any other Finance Documents are hereby expressly made subordinate and junior in right of payment, without limitation as to any increases in the Obligations arising hereunder or thereunder, to the prior payment in full in cash of the Obligations.

11.25 No Fiduciary Duty. Each Agent, each Lender and their respective Affiliates (each, a "Bank Party") may have economic interests that conflict with those of the Borrowers. The Borrowers agree that nothing in the Finance Documents or otherwise will be deemed to create an advisory, fiduciary or agency relationship or fiduciary or other implied duty between the Bank Parties and the Borrowers, their stockholders or Affiliates. The Borrowers acknowledge and agree that (i) the transactions contemplated by the Finance Documents are arm's-length commercial transactions between the Bank Parties, on the one hand, and the Borrowers, on the other hand, (ii) in connection therewith and with the process leading to such transactions, each Bank Party is acting solely as a principal and not the agent or fiduciary of the Borrowers, their management, stockholders, creditors or any other person, (iii) no Bank Party has assumed an advisory or fiduciary responsibility in favor of the Borrowers with respect to the transactions contemplated hereby or in any other Finance Document or the process leading thereto (irrespective of whether any Bank Party or any of its Affiliates has advised or is currently advising the Borrowers on other matters) or any other obligation to the Borrowers except the obligations expressly set forth in the Finance Documents and (iv) the Borrowers have consulted their own legal and financial advisors to the extent it deemed appropriate. The Borrowers further acknowledge and agree that they are responsible for making their own independent judgment with respect to such transactions and the process leading thereto. The Borrowers agree that they will not claim that any Bank Party has rendered advisory services of any nature or respect, or owes a fiduciary or similar duty to the Borrowers, their stockholder or Affiliates, in connection with such transactions or the process leading thereto.

132

IN WITNESS WHEREOF, the parties hereto have caused their duly authorized officers to execute and deliver this Agreement as of the date first above written.

Notice Address:

Address: Clean Energy Tranche A LNG Plant, LLC

c/o Clean Energy Fuels Corp. 3020 Old Ranch Parkway Seal Beach, CA 90740

Attention: Nate Jensen and Rick Wheeler

Telephone No.: (562) 493-7239 Telecopier No.: (562)493-4956 Email: njensen@cleanenergyfuels.com

Notice Address:

Address: Clean Energy Tranche B LNG Plant, LLC

c/o Clean Energy Fuels Corp. 3020 Old Ranch Parkway Seal Beach, CA 90740

Attention: Nate Jensen and Rick Wheeler

Telephone No.: (562) 493-7239 Telecopier No.: (562)493-4956 CLEAN ENERGY TRANCHE A LNG PLANT, LLC, as Borrower

By: /s/ Andrew J. Littlefair

Name: Andrew J. Littlefair

Title: President & Chief Executive Officer

CLEAN ENERGY TRANCHE B LNG PLANT, LLC, as Borrower

By: /s/ Andrew J. Littlefair

Name: Andrew J. Littlefair

### Signature Page to Credit Agreement

By:

Notice Address:

Address:

**GENERAL ELECTRIC CAPITAL CORPORATION**, as Administrative Agent and Collateral Agent

General Electric Capital Corporation

c/o GE Energy Financial Services

800 Long Ridge Road

Stamford, CT 06927

Attention: Dipika Chanana Telephone No.: 203-316-7521

Telecopier No.: 513-770-2270

Email: EFS.AgencyServices@ge.com With copies to: Stephen Lenihan

Tel: 203-961-2273 Fax: 203-357-4897

Fax: 203-357-4897 stephen.lenihan@ge.com

GENERAL ELECTRIC CAPITAL CORPORATION, as Lender

Notice Address:

Address:

Attention:

General Electric Capital Corporation c/o GE Energy Financial Services

800 Long Ridge Road Stamford, CT 06927 Dipika Chanana

Telephone No.: 203-316-7521 Telecopier No.: 513-770-2270

Email: EFS. Agency Services@ge.com

With copies to: Stephen Lenihan

Tel: 203-961-2273 Fax: 203-357-4897 stephen.lenihan@ge.com By: /s/ Tyson Yates

Name: Tyson Yates

/s/ Tyson Yates

Name: Tyson Yates

Title: Authorized Signatory

Title: Authorized Signatory

Signature Page to Credit Agreement

Exhibit A

Credit Agreement

[FORM OF NOTICE OF BORROWING]

# **NOTICE OF BORROWING**

[Date](1)

.](2)

General Electric Capital Corporation, as Administrative Agent for the Lenders party to the Credit Agreement referred to below c/o GE Energy Financial Services 800 Long Ridge Road Stamford, CT 06927 Attn: Dipika Chanana

Ladies and Gentlemen:

The undersigned, [ ], refers to the Credit Agreement dated as of November 7, 2012 (as amended, restated, supplemented or otherwise modified from time to time, the "<u>Credit Agreement</u>," the terms defined therein being used herein as therein defined) among the undersigned, you, as Administrative Agent, and the other parties thereto and hereby gives you notice, irrevocably, pursuant to <u>Section 3.3(a)(ii)</u> of the Credit Agreement, that the undersigned hereby requests a Borrowing of Loans under the Credit Agreement, and in that connection set forth below the information relating to such Borrowing (the "<u>Proposed Borrowing</u>") as required by <u>Section 2.5</u> of the Credit Agreement:

- (i) The Business Day of the Proposed Borrowing is [ ], 20[ ].
- (ii) The aggregate principal amount of the Proposed Borrowing is \$ [ ].
- (iii) The Proposed Borrowing is to consist of [Base Rate Loans] [Eurodollar Loans].
- (iv) [The initial Interest Period for the Proposed Borrowing is [months] and such Interest Period shall expire

The undersigned hereby certifies that the following statements are true on the date hereof and will be true on the date of the Proposed Borrowing:

- (A) the date of the Proposed Borrowing is a Business Day; and
- (B) each of the conditions precedent contained in Section [3.2 and] [3.3] of the Credit Agreement has been fully satisfied.

(1) At least three (3) Business Days prior to the date of the Proposed Borrowing in the case of Eurodollar Loans and at least one (1) Business Day prior to the date of the Proposed Borrowing in the case of Base Rate Loans.

(2) To be included for a Proposed Borrowing of Eurodollar Loans.

Very 1	truly yours,			
[	]			
Ву:	Name: Title:			
[	]			
By:	Name: Title:			

Exhibit B-1 to Credit Agreement

[FORM OF CONSTRUCTION NOTE]

# TRANCHE [A][B] CONSTRUCTION NOTE

U.S. \$[ ] New York, New York [Date]

Each of Clean Energy Tranche A LNG Plant, LLC, a Delaware limited liability company, and Clean Energy Tranche B LNG Plant, LLC, a Delaware limited liability company (collectively, the "Borrowers"), FOR VALUE RECEIVED, hereby jointly and severally promise, subject to [Section 2.3(a)][Section 2.4(a)] of the Credit Agreement referred to below, to pay on the Tranche [A][B] Construction Loan Maturity Date (as defined in the Credit Agreement) to General Electric Capital Corporation (the "Lender"), for the account of its Applicable Lending Office (as defined in the Credit Agreement), in accordance with the Credit Agreement referred to below, the principal sum of [ ]DOLLARS, or, if less, so much thereof as shall constitute outstanding Tranche [A][B] Construction Loans (as defined in the Credit Agreement) which have been lent by the Lender to the Borrowers pursuant to the Credit Agreement referred to below, in lawful money of the United States of America (in freely transferable U.S. dollars and in immediately available funds).

The Borrowers promise also to pay interest on the unpaid principal amount hereof in like money from the date hereof until paid in full at the rate per annum which shall be determined in accordance with the provisions of the Credit Agreement, said interest to be payable at the times and at the place provided for in the Credit Agreement.

The Lender is hereby authorized by the Borrowers to endorse on the schedule attached to this Tranche [A][B] Construction Note (or any continuation thereof) the amount of, and the duration of each Interest Period (if applicable) for, each Tranche [A][B] Construction Loan made by the Lender to the Borrowers under the Credit Agreement, the date such Tranche [A][B] Construction Loan is made and the amount of each payment or prepayment of principal of such Tranche [A][B] Construction Loan received by the Lender; <u>provided</u> that any failure by the Lender to make any such endorsement or any error therein shall not affect the obligations of the Borrowers hereunder or under the Credit Agreement in respect of such Tranche [A][B] Construction Loan.

This Tranche [A][B] Construction Note is one of the Notes referred to in the Credit Agreement, dated as of November 7, 2012 among the Borrowers, General Electric Capital Corporation, as Administrative Agent, General Electric Capital Corporation, as Collateral Agent, the Lenders and other persons from time to time party thereto (as amended, restated, supplemented or otherwise modified from time to time, the "Credit Agreement") and is entitled to the benefits thereof. This Tranche [A][B] Construction Note is secured by the Security

Documents (as defined in the Credit Agreement). This Tranche [A][B] Construction Note is subject to repayment and prepayment, in whole or in part, as specified in the Credit Agreement, and Loans made under the Credit Agreement may be converted from one Type (as defined in the Credit Agreement) into another Type to the extent provided in the Credit Agreement.

In case an Event of Default (as defined in the Credit Agreement) shall occur and be continuing, the principal of and accrued interest on this Tranche [A][B] Construction Note may be declared to be due and payable in the manner and with the effect provided in the Credit Agreement.

The Borrowers hereby waive presentment, demand, protest or notice of any kind in connection with this Tranche [A][B] Construction Note.

THIS TRANCHE [A][B] CONSTRUCTION NOTE SHALL BE CONSTRUED IN ACCORDANCE WITH, AND BE GOVERNED BY, THE LAWS OF THE STATE OF NEW YORK.

[	]				
By:	Name:				
	Title:				
[	]				
By:					
	Name: Title:				
2					

Exhibit B-2 to Credit Agreement

[FORM OF TERM NOTE]

# [TRANCHE [A][B] TERM NOTE

U.S. \$[ ] New York, New York
[Date]

Each of Clean Energy Tranche A LNG Plant, LLC, a Delaware limited liability company, and Clean Energy Tranche B LNG Plant, LLC, a Delaware limited liability company (collectively, the "Borrowers"), FOR VALUE RECEIVED, hereby jointly and severally promise to pay on the Tranche [A][B] Final Maturity Date (as defined in the Credit Agreement referred to below) to General Electric Capital Corporation (the "Lender"), for the account of its Applicable Lending Office (as defined in the Credit Agreement), in accordance with the Credit Agreement, the principal sum of [ ] DOLLARS, or so much thereof as shall constitute outstanding Tranche [A][B] Term Loans (as defined in the Credit Agreement) which have been lent by the Lender to the Borrowers pursuant to the Credit Agreement referred to below, in lawful money of the United States of America (in freely transferable U.S. dollars and in immediately available funds).

The Borrowers promise also to pay interest on the unpaid principal amount hereof in like money from the date hereof until paid in full at the rate per annum which shall be determined in accordance with the provisions of the Credit Agreement, said interest to be payable at the times and at the place provided for in the Credit Agreement.

The Lender is hereby authorized by the Borrowers to endorse on the schedule attached to this Tranche [A][B] Term Note (or any continuation thereof) the amount of, and the duration of each Interest Period (if applicable) for, the Tranche [A][B] Term Loan made by the Lender to the Borrowers under the Credit Agreement, the date such Tranche [A][B] Term Loan is made and the amount of each payment or prepayment of principal of such Tranche [A][B] Term Loan received by the Lender; <u>provided</u> that any failure by the Lender to make any such endorsement or any error therein shall not affect the obligations of the Borrowers hereunder or under the Credit Agreement in respect of such Tranche [A][B] Term Loan.

This Tranche [A][B] Term Note is one of the Notes referred to in the Credit Agreement, dated as of November 7, 2012 among the Borrowers, General Electric Capital Corporation, as Administrative Agent and Collateral Agent, the Lenders and other persons from time to time party thereto (as amended, restated, supplemented or otherwise modified from time to time, the "Credit Agreement") and is entitled to the benefits thereof. This Tranche [A] [B] Term Note is secured by the Security Documents (as defined in the Credit Agreement). This Tranche [A][B] Term Note is subject to repayment and prepayment, in whole or in part, as specified in the Credit Agreement, and Loans made under the Credit Agreement may be converted from one Type (as defined in the Credit Agreement) into another Type to the extent provided in the Credit Agreement.

In case an Event of Default (as defined in the Credit Agreement) shall occur and be continuing, the principal of and accrued interest on this Tranche [A][B] Term Note may be declared to be due and payable in the manner and with the effect provided in the Credit Agreement.

> Exhibit C-1 to Credit Agreement

[FORM OF CONSTRUCTION REQUISITION]

# CONSTRUCTION REQUISITION(3)

NO.

[Date]

General Electric Capital Corporation, as Administrative Agent c/o GE Energy Financial Services
800 Long Ridge Road
Stamford, CT 06927
Attn: Dipika Chanana

Re: Credit Agreement, dated as of November 7, 2012 (as amended, restated, supplemented or otherwise modified and in effect from time to time, the "Credit Agreement"), among Clean Energy Tranche A LNG Plant, LLC and Clean Energy Tranche B LNG Plant, LLC (collectively, the "Borrowers"), General Electric Capital Corporation, as Administrative Agent, General Electric Capital Corporation, as Collateral Agent, the Lenders and other persons from time to time party thereto; and Collateral Account Agreement, dated as of [ ], 2012 (as amended, restated, supplemented or otherwise modified and in effect from time to time, the "Collateral Account Agreement"), among the Borrowers, the Administrative Agent, the Collateral Agent and [ ], as Depositary Agent and securities intermediary.

# Ladies and Gentlemen:

This Construction Requisition (this "Construction Requisition") is delivered to you pursuant to Section 3.3(a)(i) of the Credit Agreement and covers the payment of Project Costs which are due and payable or will become due and payable during the thirty-day period beginning on the date set forth in paragraph 1 below, or if this Construction Requisition is being delivered exclusively in connection with the application of Project Revenues to the payment of Facility [A][B] Project Costs, the earliest date set forth in Schedule I hereto (the "Construction Period"). Each capitalized term used herein and not otherwise defined herein shall have the definition assigned to that term in the Credit Agreement. With respect to this Construction Requisition, Tranche [A][B] Borrower hereby certifies as follows:

1.	Tranche [A][B] Borrow	ver intends to deliver to the A	dministrative Agent a Notice of Borrowing in r	espect of Tranche [A][B] Construction
	Loans in the aggregate	principal amount of \$[	]. The Disbursement Date with respect to sucl	n Tranche [A][B] Construction Loans
	shall be [ ].			

(3) Subject to review by the Project Engineer.

2. Set forth on Schedule I attached hereto is the following information: (a) the name of each Person to whom any payment is to be made from the amounts described in paragraph 1 above, (b) an accurate description of the work performed, services rendered, materials, equipment or supplies delivered or such other purpose for which each such payment was or is to be made, (c) a reasonable estimate of the aggregate amount of each such payment, (d) the proposed (or actual) date of each such payment and (e) the payment or wire transfer instructions for each such payment. The amounts to be withdrawn from the Tranche [A][B] Construction Account pursuant to this Construction Requisition will be used to pay Facility [A][B] Project Costs which have been paid, or are due and payable, or will become due and payable during the Construction Period.

- Attached hereto as Attachment I are copies of all invoices, payment applications and other written information with respect to each item set forth on Schedule I and, in the case of any amount set forth on Schedule I payable to the Contractors, a true and complete copy of the applicable progress report, if any, and payment request submitted to Tranche [A][B] Borrower by such party. The proposed application of amounts to be withdrawn from the Tranche [A][B] Construction Account pursuant to this Construction Requisition complies with the applicable requirements contained in the Credit Agreement (including Section 5.21) and the Collateral Account Agreement.
   The Facility [A][B] Project Costs for which payment is requested under this Construction Requisition have not been the basis for any prior Construction Requisition submitted by Tranche [A][B] Borrower and approved for disbursement by the Project Engineer. Furthermore, all amounts previously drawn from the Tranche [A][B] Construction Account have been applied to pay Facility [A][B] Project Costs listed on the applicable Construction Requisition with respect to which such amounts were drawn
   Attached hereto as Attachment II are copies of partial lien waivers executed by each of the Contractors, covering all work, labor and materials done, performed or furnished for or to Tranche [A][B] Borrower as of the date hereof.
   All Change Orders entered into by Tranche [A][B] Borrower which required the approval of any Lender have been so approved in accordance with the Credit Agreement.
- 7. The work under the Construction Agreements in respect of which payment is requested pursuant to this Construction Requisition has been performed in accordance with each of the Construction Agreements, the Facility [A][B] Plans and Specifications and Prudent Industry Practices.
- 8. Attached hereto as **Attachment III** is an update to the applicable Construction Budget setting forth (i) the actual Facility [A][B] Project Costs incurred to date, (ii) the estimated Project Costs in order to achieve the Facility [A][B] Project Completion Date and (iii) the remaining amount of the Contingency.

- Tranche [A][B] Borrower is not aware of any event that has occurred (a) that has had or could reasonably be expected to have a Material Adverse Effect or (b) the occurrence of which is likely to cause the Facility [A][B] Project Completion Date to occur after the Facility [A][B] Date Certain. Tranche [A][B] Borrower estimates, after reasonable investigation, that the Facility [A][B] Project Completion Date will occur on or prior to [ ].
- 10. All conditions set forth in <u>Section [3.2 and] 3.3</u> of the Credit Agreement to the making of the Construction Loans contemplated by this Construction Requisition have been (or, on or prior to the Disbursement Date with respect to such Construction Loans, will be) satisfied.

Very	truly yours,			
[	]			
By:				
	Name: Title:			
2				
3				

Schedule I

Exhibit C-1

# PAYMENT INFORMATION

Amount of Payee Purpose Payment Date of Payment Instructions

Attachment I

Exhibit C-1

# [LIEN WAIVERS]

Attachment III to Exhibit C-1

# [<u>UPDATE TO THE CONSTRUCTION BUDGET</u>]

Exhibit C-2 to Credit Agreement

[FORM OF PROJECT ENGINEER'S CERTIFICATE]

# PROJECT ENGINEER'S CERTIFICATE (DRAWDOWNS)(4)

[Date]

Re: Construction Requisition No.

To:

General Electric Capital Corporation, as Administrative Agent c/o GE Energy Financial Services 800 Long Ridge Road Stamford, CT 06927 Attn: Dipika Chanana

Re: [ ] and [ ]

[ ], acting as the "<u>Project Engineer</u>" under the Credit Agreement defined below, hereby submits this certificate in connection with the Construction Requisition attached as <u>Exhibit A</u> hereto in relation to the (i) proposed Disbursement of Tranche [A][B] Construction Loans and/or (ii) proposed application of Project Revenues, pursuant to the Credit Agreement.

Except as otherwise defined herein, capitalized terms used herein shall have the meanings assigned to such terms in the Credit Agreement, dated as of November 7, 2012 among Clean Energy Tranche A LNG Plant, LLC, Clean Energy Tranche B LNG Plant, LLC, General Electric Capital Corporation, as Administrative Agent, General Electric Capital Corporation, as Collateral Agent, the Lenders and other persons from time to time party thereto (as in effect on the date hereof, the "Credit Agreement").

The Project Engineer has discussed all matters believed pertinent to this Certificate with Tranche [A][B] Borrower, the GE Contractor and any other third party deemed appropriate, and has made such inspections, site visits, reviews, examinations and investigations as the Project Engineer believed were reasonably necessary to establish the accuracy of this certificate. On the basis of the foregoing and on the understanding and belief that the Project Engineer has been provided true, correct and complete information from such other parties as to the matters covered by this certificate, the Project Engineer hereby certifies, in its professional opinion, as of the date hereof, that:

- 1. The individual executing this certificate is a duly authorized representative of the Project Engineer, authorized to execute and deliver this certificate on behalf of the Project Engineer.
- (4) Subject to review by the Project Engineer.
- 2. The Project Engineer has performed its review of the Construction
  Requisition referenced above (the "Requisition") in a professional manner using sound project management and supervisory principles and procedures and in accordance with the standards of care practiced by leading consulting engineers in performing similar tasks on like projects. The Project Engineer represents that it has the required skills and capacity to perform its services in the foregoing manner.
- 3. The Project Engineer has received all information it has requested relating to each Construction Agreement and any other Transaction Document and has no reason to believe that any of the information is untrue, incorrect or incomplete.
- 4. With respect to the Requisition, the Project Engineer has no reason to believe, except as may be noted below, that any statement made by Tranche [A][B] Borrower in the Requisition is not true.

	[A][B] Plans and Specifications, the quality of the Work completed to date is in o reason to believe that the Project is being built in violation of any applicable ance of the relevant Work, subject to the following:
6. With respect to the amount requested in the Requisition pert of the Work performed under each Construction Agreement, (a) the Contractor Disbursement of Tranche [A][B] Construction Loans pursuant to the terms of been satisfactorily completed.	
Element Not Completed	Value
	2
below opposite each such expenditure. Such payments, when added to other so the aggregate amount of such payments provided for in the Construction Budg	
Expenditure	Line Item Percentage %
	%
8. The Project Engineer believes, except as may be noted belo Authorities required to be obtained or made to date for the construction and op obtained or made.	ow, that all Governmental Approvals and all filings with Governmental peration of the Project as contemplated by the Project Documents have been
9. It is the professional opinion of the undersigned that the Faoccur on or prior to , 20[ ].	acility [A][B] Project Completion Date, as defined in the Credit Agreement, will
	3
IN WITNESS WHEREOF, the undersigned has executed this certificate on the	e date first written above.
	[ ]
	By: Name:
	Title:
	4
	Exhibit A to Exhibit C-2 to
	Credit Agreement

CONSTRUCTION REQUISITION

[attached]

[FORM OF BORROWERS (	COMPLETION	CERTIFICATE]
----------------------	------------	--------------

[	
	-
[	

# **COMPLETION CERTIFICATE(5)**

[INSERT CONVERSION DATE]

c/o GE Energy Finar 800 Long Ridge Roa Stamford, CT 06927 Attn: Dipika Chanan	d .
[ [Address] Attention: [	], as Depositary Agent
connection with the <b>Agreement</b> ") among	cate is being delivered by the undersigned, [ ], a Delaware limited liability company (" <u>Tranche [A][B] Borrower</u> "), in Credit Agreement dated as of November 7, 2012 (as amended, supplemented or modified and in effect from time to time, the " <u>Credit</u> 3 the Borrowers, General Electric Capital Corporation, as Administrative Agent, General Electric Capital Corporation, as Collateral and other persons time to time party thereto.

Each capitalized term used herein and not otherwise defined has the meaning assigned thereto in the Credit Agreement.

After due inquiry and to induce the Secured Parties to rely hereon and to take action in reliance hereon, I do hereby certify that I am an Authorized Officer of the Tranche [A][B] Borrower, and in such capacity do hereby further certify that:

1. Facility [A][B] has been started up and operated.

General Electric Capital Corporation, as Administrative Agent and Collateral Agent

- (5) Subject to review by the Project Engineer.
- 2. The total production capacity of Facility [A][B] is at least 250,000 gallons of LNG per day (taking into account reasonable measurement tolerances), as self-determined using data generally accepted in the natural gas liquefaction industry.
- 3. All portions of the Work (except for painting and punch list items the total cost of which to complete does not exceed \$500,000) have been completed in accordance in all material respects with the Construction Agreements and in compliance with all applicable Laws and Governmental Approvals, and all clearing, landscaping, lighting and paving of Site [A][B] and all ancillary construction, upgrades and improvements necessary for the operation of Facility [A][B] as contemplated by the Transaction Documents have been completed.
- 4. Attached hereto as **Exhibit A** are certified copies of the insurance policies required by **Section 5.9** of the Credit Agreement, or certificates of insurance with respect thereto together with evidence of the payment of all premiums therefor.
- 5. Attached hereto as **Exhibit B** is a certificate of the Insurance Advisor, certifying that insurance complying with **Section 5.9** of the Credit Agreement, covering the risks referred to therein, has been obtained and is in full force and effect.
- 6. Attached hereto as **Exhibit C** is a letter from a nationally-recognized insurance broker stating that insurance complying with **Section 5.9** of the Credit Agreement is in full force and effect.
- 7. No Default or Event of Default has occurred and is continuing or would result from the occurrence of the Facility [A][B] Conversion Date, and no default by any Finance Party, any Affiliate Project Participant or, to the knowledge of the Tranche [A][B] Borrower, any other Project Participant under any of the Transaction Documents then in effect has occurred and is continuing or would result from the occurrence of the Facility [A][B] Conversion Date.
- 8. There exists no circumstance, event or condition which has had or could reasonably be expected to have a Material Adverse Effect.
- 9. All Necessary Governmental Approvals required for the construction and operation of Facility [A][B] have been duly obtained and are final and in full force and effect and either the applicable administrative and statutory appeal periods shall have expired or those applicable administrative and statutory appeal periods greater than ninety (90) days that have not expired and their relevant expiration dates are identified on **Schedule 4.6** of the Credit Agreement.
- 10. The representations and warranties of the Borrowers contained in <u>Section 4</u> of the Credit Agreement and the representations and warranties of each Finance Party, each Affiliate Project Participant and, to the knowledge of the Tranche [A][B] Borrower, each other Project Participant (other than any Lender Affiliate Project Participant or any Immaterial

Project Participant) contained in any other Finance Document or Project Document then in effect are true and correct in all material respects on and as of the date hereof as if made on and as of such date (or, if expressly stated to have been made solely as of an earlier date, were true and correct as of such

11.	Attached hereto as Exhibit D are (i) a final down date endorsement to the Title Insurance, in favor of the Collateral Agent for the benefit of the Secured
	Parties, showing no exceptions other than those set forth in the Title Insurance (as previously endorsed pursuant to Section 3.3(i) of the Credit
	Agreement) and (ii) the final "as-built" ALTA survey of the Site certified to the Collateral Agent, the Title Insurance Company and the Borrowers.

3

IN WITNESS WHEREOF,	he undersigned	has executed this	certificate on tl	he date firs	st written above.

[ By: Name:

4

]

Title:

Exhibit D-2 Credit Agreement

[FORM OF PROJECT ENGINEER COMPLETION CERTIFICATE]

# PROJECT ENGINEER COMPLETION CERTIFICATE(6)

[INSERT CONVERSION DATE]

To:

General Electric Capital Corporation, as Administrative Agent and Collateral Agent c/o GE Energy Financial Services 800 Long Ridge Road Stamford, CT 06927 Attn: Dipika Chanana ], as Depositary Agent [Address]

Attention: [ Re: [] and []

], acting as the "Project Engineer" under the Credit Agreement defined below, hereby submits this Certificate in connection with the Facility [A][B] Project Completion Date, as defined in the Credit Agreement (as defined below).

Except as otherwise defined herein, capitalized terms used herein shall have the meanings assigned to such terms in the Credit Agreement, dated as of November 7, 2012, among Clean Energy Tranche A LNG Plant, LLC, Clean Energy Tranche B LNG Plant, LLC, General Electric Capital Corporation, as Administrative Agent, General Electric Capital Corporation, as Collateral Agent, the Lenders and other persons from time to time party thereto (as in effect on the date hereof, the "Credit Agreement").

The Project Engineer has discussed all matters believed pertinent to this certificate with the Borrowers, the GE Contractor and any other third party deemed appropriate, and has made

(6) Subject to review by the Project Engineer.

]

such inspections, site visits, reviews, examinations and investigations as the Project Engineer believed were reasonably necessary to establish the accuracy of this certificate. On the basis of the foregoing and on the understanding and belief that the Project Engineer has been provided true, correct and complete information from such other parties as to the matters covered by this certificate, the Project Engineer hereby certifies, in its professional opinion, as of the date hereof, that:

The individual executing this certificate is a duly authorized representative of the Project Engineer, authorized to execute and deliver this certificate on behalf of the Project Engineer.

- 2. The Project Engineer has performed its review of Facility [A][B] in a professional manner using sound project management and supervisory principles and procedures and in accordance with the standards of care practiced by leading consulting engineers in performing similar tasks on like projects. The Project Engineer represents that it has the required skills and capacity to perform its services in the foregoing manner.
- 3. The Project Engineer has received all information it has requested relating to the Construction Agreements and any other Transaction Document and has no reason to believe that any of the information is untrue, incorrect or incomplete.
- 4. The total production capacity of Facility [A][B] is at least 250,000 gallons of LNG per day (taking into account reasonable measurement tolerances), as verified by the Project Engineer using data generally accepted in the natural gas liquefaction industry.
- 5. All portions of the Work (except for painting and punch list items the total cost of which to complete does not exceed \$500,000) have been completed in accordance in all material respects with the Construction Agreements and in compliance in all material respects with all applicable Laws and Governmental Approvals, and all clearing, landscaping, lighting and paving of Site [A][B] and all ancillary construction, upgrades and improvements necessary for the operation of Facility [A][B] as contemplated by the Transaction Documents have been completed.
  - 6. The Acceptance Tests have been successfully completed to the satisfaction of the Project Engineer.

[

]

IN WITNESS WHEREOF, the undersigned has executed this certificate on the date first written above.

By:
Name:
Title:

<u>Exhibit E</u> <u>to</u> Credi<u>t Agreement</u>

[Date]

[FORM OF CONSENT AGREEMENT]

# ACKNOWLEDGMENT AND CONSENT AGREEMENT

[ ] (the "Contracting Party.") hereby acknowledges to Clean Energy Tranche A LNG Plant, LLC ("Tranche A Borrower"), Clean Energy Tranche B LNG Plant, LLC ("Tranche B Borrower" and together with Tranche A Borrower, the "Borrowers") and General Electric Capital Corporation, as collateral agent (in such capacity, the "Collateral Agent") that the Borrowers and the Collateral Agent have advised Contracting Party that the Borrowers and the Collateral Agent have entered into the Security Agreement, dated as of November 7, 2012 (as from time to time amended, restated, supplemented or otherwise modified, the "Security Agreement") for the benefit of the Secured Parties (as such term is defined in the Security Agreement), and the Contracting Party hereby agrees as follows:

- - 2. The Contracting Party represents and warrants as follows:
  - (a) Each of this agreement (this "Consent") and the Assigned Agreement has been duly authorized, executed and delivered by the Contracting Party, is in full force and effect and is a legal, valid and binding obligation of the Contracting Party enforceable against the Contracting Party in accordance with its terms, except as the enforceability thereof may be limited by bankruptcy, insolvency, reorganization, debt adjustment, moratorium or other similar laws affecting creditors' rights generally. [Except as described in paragraph 1 above,] there are no amendments, modifications or supplements (whether by waiver, consent or otherwise) to the Assigned Agreement, either oral or written.
  - (b) The Borrowers have complied with all conditions precedent required to be complied with by or on behalf of Tranche [A] [B] Borrower on or prior to the date hereof pursuant to the Assigned Agreement.
  - (c) The Contracting Party is a [ ] duly organized, validly existing and in good standing under the laws of the jurisdiction of its [incorporation] [formation]. The Contracting Party has the [corporate] power to carry on its business as currently being conducted and as proposed to be conducted by it. The Contracting Party has the [corporate] power and authority to execute and deliver this Consent and the Assigned Agreement and to perform its obligations under each thereof.

- (d) The execution and delivery of this Consent and the Assigned Agreement by the Contracting Party did not, and the fulfillment and compliance with the respective provisions hereof and thereof by the Contracting Party do not and will not, conflict with or result in a breach of the terms, conditions or provisions of, or constitute a default under, or result in the creation or imposition of (or the obligation to create or impose) any lien, security interest, charge or encumbrance upon any of the properties or assets of the Contracting Party pursuant to the provisions of, or result in any violation of, the articles, by-laws or any other charter document of the Contracting Party, or any applicable law, statute, rule or regulation, or any agreement, instrument, order, judgment or decree, to which the Contracting Party is subject.
- (e) No consent or approval of or other action by or any notice to or filing with, any court or administrative or governmental body (except those previously obtained) was required in connection with the execution and delivery of the Assigned Agreement, or is required in connection with the execution and delivery of this Consent or the performance by the Contracting Party of its obligations hereunder or thereunder. The Contracting Party has obtained all permits, licenses, approvals, consents and exemptions with respect to the performance of its obligations under this Consent and the Assigned Agreement required by applicable laws, statutes, rules and regulations in effect as of the date hereof.
- (f) There are no proceedings pending or, to the best of the Contracting Party's knowledge, threatened against or affecting the Contracting Party in any court or by or before any governmental authority, arbitration board or tribunal that may result in a material adverse effect upon the property, business, prospects, profits or condition (financial or otherwise) of the Contracting Party, or the ability of the Contracting Party to perform its obligations under this Consent and the Assigned Agreement, and the Contracting Party is not in default with respect to any order of any court, governmental authority, arbitration board or tribunal.
- (g) The Contracting Party affirms that it has no notice of any assignment relative to the right, title and interest of Tranche [A] [B] Borrower in, to and under the Assigned Agreement other than the pledge and assignment referred to in paragraph 1.
- (h) After giving effect to the pledge and assignment referred to in paragraph 1, and after giving effect to the consent to such pledge and assignment by the Contracting Party, there exists no event or condition (a "<u>Termination Event</u>") that would, either immediately or with the passage of time or giving of notice, or both, entitle either the Contracting Party or Tranche [A] [B] Borrower to terminate or suspend its obligations under the Assigned Agreement and there are no claims or rights of set-off pending by any party to the Assigned Agreement. All amounts due under the Assigned Agreement as of the date hereof have been paid in full.
- (i) All representations, warranties and other statements made by the Contracting Party in the Assigned Agreement were true and correct in all material respects as of the date when made and (unless expressly stated to have been made solely as of an earlier date) are true and correct in all material respects as of the date of this Consent.
- 3. (a) From and after the date hereof and unless and until the Contracting Party shall have received written notice from the Collateral Agent that the lien of the Security Agreement has been released in full, the Collateral Agent shall have the right, but not the obligation, to pay all sums due under the Assigned Agreement by Tranche [A] [B] Borrower and to perform any other act, duty or obligation required of Tranche [A] [B] Borrower thereunder at any time; <u>provided</u>, that no such payment or performance shall be construed as an assumption by the Collateral Agent or any Secured Party of any covenants, agreements or obligations of Tranche [A] [B] Borrower under or in respect of the Assigned Agreement.
- (b) The Contracting Party agrees that it will not terminate or suspend its obligations under the Assigned Agreement without first giving the Collateral Agent notice and opportunity to cure as provided below.
- (c) If a Termination Event shall occur, and the Contracting Party shall desire to terminate or suspend its obligations under the Assigned Agreement, the Contracting Party first shall give written notice to the Collateral Agent of such Termination Event (which notice shall not be given pursuant to this paragraph until the date upon which the Contracting Party is entitled to terminate or suspend its obligations under the Assigned Agreement). If the Collateral Agent elects to exercise its right to cure as herein provided, it shall, within thirty (30) days after the receipt by it of the notice from the Contracting Party referred to in the preceding sentence, deliver to the Contracting Party a written notice stating that it has elected to exercise such right to cure, together with a written statement of the Collateral Agent that it will promptly commence to cure the Termination Event(s) described in the Contracting Party's notice, and that it will, during the cure period, diligently attempt in good faith to complete the curing of such Termination Event(s).
- (d) The Collateral Agent shall have ninety (90) days after the delivery of the notice by the Collateral Agent referred to in paragraph 3(c) in which to cure the Termination Event(s) specified in such notice; <u>provided</u> that in the case of termination due to monetary defaults, such period to cure the Termination Event(s) shall be thirty (30) days. In the event that any such Termination Event (except monetary defaults) shall be incurable by the Collateral Agent within such ninety (90) day period, the Contracting Party shall not exercise any remedies under the Assigned Agreement if the Collateral Agent shall, within such ninety (90) day period, initiate action to cure such Termination Event and diligently attempt to complete the curing thereof. Any curing of or attempt to cure any Termination Event shall not be construed as an assumption by the Collateral Agent or the other Secured Parties of any covenants, agreements or obligations of Tranche [A] [B] Borrower under or in respect of the Assigned Agreement.
- (e) If, before the Collateral Agent shall have cured any Termination Event pursuant to paragraph 3(d), Tranche [A] [B] Borrower shall have cured such Termination Event, the Contracting Party promptly shall provide the Collateral Agent with notice of such cure and the discontinuance of such Termination Event.
- 4. (a) Notwithstanding any provision in the Assigned Agreement to the contrary, in the event of the rejection of the Assigned Agreement by a receiver of Tranche [A] [B] Borrower or otherwise pursuant to bankruptcy or insolvency proceedings, at the Collateral Agent's request, the Contracting Party will enter into a new agreement with the Collateral Agent's nominee, for the remainder of the originally scheduled term of the Assigned Agreement, effective as of the date of such rejection, with the same covenants, agreements, terms, provisions and limitations as are contained in the Assigned Agreement.
- (b) If the Collateral Agent or its nominee is prohibited by any process or injunction issued by any court having jurisdiction of any bankruptcy or insolvency proceeding involving Tranche [A] [B] Borrower, from continuing the Assigned Agreement in place of Tranche [A] [B] Borrower or

from otherwise exercising any of its rights or remedies hereunder or under the Security Agreement in respect of the Assigned Agreement, then the times specified herein for the exercise by the Collateral Agent of any right or benefit

granted to it hereunder (including without limitation the time period for the exercise of any cure rights granted hereunder) shall be extended for the period of such prohibition; <u>provided</u> that the Collateral Agent is diligently pursuing such rights or remedies (to the extent permitted) in such bankruptcy or insolvency proceeding or otherwise.

- 5. The Contracting Party acknowledges that upon an exercise of rights and remedies, the Collateral Agent may (but shall not be obligated to) assume, or cause any purchaser at any foreclosure sale or any assignee or transferee under any instrument of assignment or transfer in lieu of foreclosure to assume, all of the interests, rights and obligations of Tranche [A] [B] Borrower thereafter arising under the Assigned Agreement. If the interest of Tranche [A] [B] Borrower in the Assigned Agreement shall be assumed, sold or transferred as hereinbefore provided, the assuming party shall agree in writing to be bound by and to assume the terms and conditions of the Assigned Agreement and any and all obligations to the Contracting Party arising or accruing thereunder from and after the date of such assumption, and the Contracting Party shall continue to perform its obligations under the Assigned Agreement in favor of the assuming party as if such party had thereafter been named as Tranche [A] [B] Borrower under the Assigned Agreement; provided that if the Collateral Agent (or any entity acting on behalf of the Collateral Agent or any of the other Secured Parties) assumes the Assigned Agreement as provided above, it shall not be personally liable for the performance of the obligations thereunder except to the extent of all of its right, title and interest in and to the Project (as defined in the Security Agreement). Notwithstanding any such assumption or disposition by the Collateral Agent, a purchaser, an assignee or a transferee, Tranche [A] [B] Borrower shall not be released or discharged from and shall remain liable for any and all of its obligations to the Contracting Party arising or accruing under the Assigned Agreement prior to such assumption.
- 7. The Contracting Party shall deliver to the Collateral Agent, concurrently with the delivery thereof to Tranche [A] [B] Borrower, a copy of each notice, request or demand given by the Contracting Party pursuant to the Assigned Agreement.
- 8. No amendment or modification of, or waiver by or consent of Tranche [A] [B] Borrower in respect of, any provision of the Assigned Agreement shall be effective unless the same shall be in writing and the Collateral Agent shall have given its prior written consent thereto.
- 9. Notice to any party hereto shall be in writing and shall be deemed to be delivered on the earlier of: (a) the date of personal delivery, (b) if deposited in a United States Postal Service depository, postage prepaid, registered or certified mail, return receipt requested, or sent by express courier, in each case addressed to such party at the address indicated below (or at such other address as such party may have theretofore specified by written notice delivered in accordance herewith), upon delivery or refusal to accept delivery, or (c) if transmitted by facsimile, the date when sent and facsimile confirmation is received; provided that any facsimile communication shall be followed promptly by a hard copy original thereof by express courier:

The	Col	lateral	Agent:
-----	-----	---------	--------

The Borrowers:

The Contracting Party:

- 10. The Contracting Party agrees to execute such acknowledgments or other similar instruments as the Collateral Agent shall reasonably request in connection with the transactions provided for in this Consent.
- 11. This Consent shall be binding upon and shall inure to the benefit of the successors and assigns of the Contracting Party, and shall inure to the benefit of the Collateral Agent, the Secured Parties and their respective successors, transferees and assigns.
- 12. This Consent may be executed in one or more counterparts with the same effect as if the signatures thereto and hereto were upon the same instrument.
- 13. THIS CONSENT SHALL BE GOVERNED BY, AND CONSTRUED IN ACCORDANCE WITH, THE LAWS OF THE STATE OF NEW YORK.
- 14. In the event of any conflict between the terms, conditions and provisions of this Consent and those of any other agreement, document or instrument (including, without limitation, the Assigned Agreement), the terms, conditions and provisions of this Consent shall prevail.

IN WITNESS WHEREOF, the parties hereto have caused their duly authorized officers to execute and deliver this Consent as of the date first written above.

By: Name: Title:	
<b>GENERAI</b> as Collatera	L ELECTRIC CAPITAL CORPORATION, al Agent
By: Name: Title:	
Acknowled	ged and Agreed:
[	]
By: Name: Title:	
]	]
By: Name: Title:	

[CONTRACTING PARTY]

<u>Exhibit F</u> <u>to</u> <u>Credit Agreement</u>

[FORM OF ASSIGNMENT AND ACCEPTANCE]

# ASSIGNMENT AND ACCEPTANCE

[Date]

Reference is made to the Credit Agreement described in Item 2 of <u>Annex I</u> annexed hereto (as such Credit Agreement may hereafter be amended, supplemented or otherwise modified from time to time, the "<u>Credit Agreement</u>"). Unless otherwise defined herein, terms defined in the Credit Agreement are used herein as therein defined.

[Name of Assignor] (the "Assignor") and [Name of Assignee] (the "Assignee") hereby agree as follows:

- 1. The Assignor hereby sells and assigns to the Assignee without recourse and without representation or warranty (other than as expressly provided herein), and the Assignee hereby purchases and assumes from the Assignor, a portion of the Assignor's rights and obligations under the Credit Agreement as of the Settlement Date (as hereinafter defined) as specified in Item 5 of **Annex I** (the "**Assigned Share**") including, without limitation, all rights and obligations with respect to the Assigned Share of the outstanding Loans. After giving effect to such sale and assignment, the amount of the outstanding Loans owing to the Assignee will be as set forth in Item 5 of **Annex I**.
- 2. The Assignor (i) represents and warrants that it is the legal and beneficial owner of the interest being assigned by it hereunder and that such interest is free and clear of any adverse claim; (ii) makes no representation or warranty and assumes no responsibility with respect to any statements, warranties or representations made in or in connection with the Credit Agreement or the other Finance Documents or the execution, legality, validity, enforceability, genuineness, sufficiency or value of the Credit Agreement or the other Finance Documents or any other instrument or document furnished pursuant thereto; and (iii) makes no representation or warranty and assumes no responsibility with respect to the financial condition or prospects of the Borrowers or the performance or observance by the Borrowers of any of their obligations under the Credit Agreement or the other Finance Documents or any other instrument or document furnished pursuant thereto.
- 3. The Assignee (i) confirms that it has received a copy of the Credit Agreement and the other Finance Documents, and such other documents and information as it has deemed appropriate to make its own credit analysis and decision to enter into this Assignment and Acceptance; (ii) agrees that it will, independently and without reliance upon the Administrative Agent, the Assignor or any other Lender and based on such documents and information as it shall deem appropriate at the time, continue to make its own credit decisions in taking or not taking action under the Credit Agreement; (iii) appoints and authorizes the Agents to take such action each as an agent on its behalf and to exercise such powers under the Credit Agreement

and the other Finance Documents as are delegated to the Agents by the terms thereof, together with such powers as are reasonably incidental thereto; (iv) agrees that it will perform in accordance with their terms all of the obligations which by the terms of the Credit Agreement are required to be performed by it as a Lender; and (v) confirms that it is an Eligible Assignee as defined in the Credit Agreement.

- 4. Following the execution of this Assignment and Acceptance by the Assignor and the Assignee, an executed original hereof will be delivered to the Administrative Agent. The effective date of this Assignment and Acceptance shall be [Date] (the "Settlement Date").
- 5. Upon the delivery of a fully executed original hereof (accompanied by any documentation required under Section 2.14(f) of the Credit Agreement) to the Administrative Agent, as of the Settlement Date, (i) the Administrative Agent shall record the Assignment and Acceptance in the Register pursuant to Section 2.9(c) of the Credit Agreement, the Assignee shall be a party to the Credit Agreement and, to the extent provided in this Assignment and Acceptance, have the rights and obligations of a Lender thereunder and under the other Finance Documents and (ii) the Assignor shall, to the extent provided in this Assignment and Acceptance, relinquish its rights and be released from its obligations under the Credit Agreement and the other Finance Documents.
- 6. On the Settlement Date, the Assignee shall pay to the Assignor an amount specified by the Assignor in writing which represents the Assigned Share of the principal amount of the Loans made pursuant to the Credit Agreement which are outstanding on the Settlement Date, and which are being assigned hereunder. The Assignor and the Assignee shall make all appropriate adjustments in payments under the Credit Agreement for periods prior to the Settlement Date directly between themselves on the Settlement Date. It is agreed that the Assignee shall be entitled to all interest on the Assigned Share of the outstanding Loans at the rates specified in the Credit Agreement which accrues from and after the Settlement Date, such interest to be paid by the Administrative Agent directly to the Assignee. It is further agreed that all payments of principal made on the Assigned Share of the outstanding Loans which occur from and after the Settlement Date will be paid directly by the Administrative Agent to the Assignee.
- 7. The Assignee agrees that it will furnish to the Administrative Agent and the Borrowers the documentation required under Section 2.14(f) of the Credit Agreement.
- 8. THIS ASSIGNMENT AND ACCEPTANCE SHALL BE GOVERNED BY, AND CONSTRUED IN ACCORDANCE WITH, THE LAWS OF THE STATE OF NEW YORK.
  - 9. The addresses of the Assignee for notice and payment purposes are set forth in Items 3 and 4, respectively, of **Annex I** hereto.

2

IN WITNESS WHEREOF, the parties hereto have caused this Assignment and Acceptance to be executed by their duly authorized officers as of the date first written above.

# [NAME OF ASSIGNOR] By: Name: Title: Name: Title: [NAME OF ASSIGNEE] By: Name: Title: Name: Title:

The undersigned, as Administrative Agent, hereby consents to this Assignment and Acceptance

**GENERAL ELECTRIC CAPITAL CORPORATION**, as Administrative Agent

By:				
	Name:			
	Title			

	undersigned Borrowers hereby consent to this Assignment and ptance
CLE.	AN ENERGY TRANCHE A LNG PLANT, LCC
By:	
	Name: Title:
CLE	AN ENERGY TRANCHE B LNG PLANT, LLC
By:	
J	Name: Title:(7)
	's he incorted to the output required under Section 11 11(a) of the Credit Agreement
(/) 1	To be inserted to the extent required under Section 11.11(a) of the Credit Agreement.
	4
	Annex I
	to Exhibit F
	ANNEX I
	<u>TO</u>
	ASSIGNMENT AND ACCEPTANCE
1.	Borrowers: Clean Energy Tranche A LNG Plant, LLC and Clean Energy Tranche B LNG Plant, LLC
2.	Name and date of Credit Agreement and other documents or agreements evidencing or securing the Obligations: (1) Credit Agreement, dated as of November 7, 2012 among [ ], [ ], General Electric Capital Corporation, as Administrative Agent and Collateral Agent, the Lenders and other persons from time to time party thereto; and (2) the Security Documents described in the Credit Agreement.
3.	Notice Address (for Assignee):
4.	Assignee's Payment Instruction:
5.	Assigned Share (as of Settlement Date):

Exhibit G

(<u>)</u>% (<u>)</u>%

)%

)%

)%

\$

\$

Credit Agreement

[FORM OF NOTICE OF CONVERSION OR CONTINUATION]

# NOTICE OF CONVERSION OR CONTINUATION

[Date](8)

(a)

(b)

(c)

(d)

(e)

Aggregate principal amount of outstanding Loans:

Tranche A Construction Loan Commitment:

Tranche B Construction Loan Commitment:

Tranche A Term Loan Commitment:

Tranche B Term Loan Commitment:

800 Long Ridge Road Stamford, CT 06927 Attn: Dipika Chanana				
Ladies and Gentlemen:				
The undersigned, [	] refers to the Credit Agreement dated a	as of November 7, 20	012	
therein defined) among (i Administrative Agent and pursuant to <u>Section 2.10</u> oset forth below, and in that	) the undersigned and Clean Energy Tra Collateral Agent and (iii) the Lenders a of the Credit Agreement, that the unders	nche [A][B] LNG Pland other persons from higned hereby request nation relating to sucl	redit Agreement," the terms defined therein being used her ant, LLC, as borrowers, (ii) General Electric Capital Corpora time to time party thereto, and hereby gives you notice, in the toto convert or continue certain Loans under the Credit Agree the conversion or continuation (the "Proposed")	ation, as revocably,
(i) followin	the Business Day of the Proposed Cong the date of this Notice of Continuation		n is , [Insert date that is no sooner than the third Busines	ss Day
(ii) ;	the Loan (or portion thereof) to be Co.	ntinued/Converted in	the Proposed Continuation/Conversion was originally borro	wed on
(iii) Rate Loan].	each Loan (or portion thereof) to be C	Continued/Converted	in the Proposed Continuation/Conversion is a [Eurodollar L	oan] [Base
[The Interest Per	iod applicable to such Loan is	.](9);		
(iv) Loan];(1)	the Type of Loans (or portion thereof)	resulting from the Pi	roposed Continuation/Conversion is a [Base Rate Loan] [Eu	rodollar
[(v) (10)]; and (vi) Continuation/Con	the aggregate principal amount of Loa		alting from the Proposed Continuation/Conversion shall be of) proposed to be Continued/Converted in this Proposed	
The undersigned Continuation/Conversion:	•	nents are true on the	date hereof and will be true on the date of the Proposed	
(A) the date	e of the Proposed Continuation/Convers	ion is a Business Day	y; and	
(B) [no Def	ault or Event of Default has occurred ar	nd is continuing, or w	vould result from such Proposed Continuation/Conversion.](	11)
		Very tr	uly yours,	
		]	1	
		Ву:	Name: Title:	
			THE.	
(10) Applicable only to E (11) Applicable only to c	Curodollar Loans. onversion from Base Rate Loans to Eur	odollar Loans.		
		2		
				Exhibit

as Administrative Agent for the Lenders party to the Credit Agreement referred to below

c/o GE Energy Financial Services

[

<u>to</u>

Credit Agreement

# [ ] [ ]

# **SOLVENCY CERTIFICATE**

Pursuant to Section 3.2(m) of the Credit Agreement dated as of November 7, 2012 (the "Credit Agreement"; all capitalized terms defined therein being used
herein as therein defined), by and among (i) Clean Energy Tranche A LNG Plant, LLC, a Delaware limited liability company ("Tranche A Borrower") and
Clean Energy Tranche B LNG Plant, LLC, a Delaware limited liability company (" <u>Tranche B Borrower</u> " and, together with [ ], the
"Borrowers"), (ii) the financial institutions from time to time party thereto as Lenders, (iii) each Secured Swap Counterparty and (iv) General Electric
Capital Corporation, as Administrative Agent and Collateral Agent, the undersigned Authorized Officer of each Borrower and the Sponsor hereby certifies as
of the date hereof that such Borrower is, and immediately after giving effect to the incurrence of all Indebtedness and Obligations being incurred in
connection with the Credit Agreement will be, Solvent.

connection with the Credit Agreement will be, Solvent.	ng effect to ti	ie incurrence of an indeptedness and Obligations being incurred in
IN WITNESS WHEREOF, I have hereunto set my hand this [	] day of [ ],	[2012].
	C	CLEAN ENERGY TRANCHE A LNG PLANT, LLC
	В	y: Name:
		Title:
	C	LEAN ENERGY TRANCHE B LNG PLANT, LLC
	В	y: Name:
		Title:
	C	LEAN ENERGY LNG HOLDINGS, LLC
	В	y: Name:
		Title:

Exhibit I-1 to Credit Agreement

# FORM OF U.S. TAX COMPLIANCE CERTIFICATES:

# FORM OF U.S. TAX COMPLIANCE CERTIFICATE

(For Foreign Lenders That Are Not Partnerships For U.S. Federal Income Tax Purposes)

Reference is hereby made to the Credit Agreement dated as of November 7, 2012 (as amended, supplemented or otherwise modified from time to time, the "Credit Agreement"), among Clean Energy Tranche A LNG Plant, LLC and Clean Energy Tranche B LNG Plant, LLC, as Borrowers, the Lenders from time to time party thereto, and General Electric Capital Corporation, as Administrative Agent, General Electric Capital Corporation as Collateral Agent, and each other party from time to time party thereto.

Pursuant to the provisions of <u>Section 2.14(f)(ii)(B)</u> of the Credit Agreement, the undersigned hereby certifies that (i) it is the sole record and beneficial owner of the Loan(s) (as well as any Note(s) evidencing such Loan(s)) in respect of which it is providing this certificate, (ii) it is not a bank within the meaning of Section 881(c)(3)(A) of the Code, (iii) it is not a ten percent shareholder of either Borrower within the meaning of Section 871(h)(3) (B) of the Code and (iv) it is not a controlled foreign corporation related to either Borrower as described in Section 881(c)(3)(C) of the Code.

The undersigned has furnished the Administrative Agent and the Borrowers with a certificate of its non-U.S. Person status on IRS Form W-8BEN. By executing this certificate, the undersigned agrees that (1) if the information provided on this certificate changes, the undersigned shall promptly so inform the Borrowers and the Administrative Agent, and (2) the undersigned shall have at all times furnished the Borrowers and the Administrative Agent with a properly completed and currently effective certificate in either the calendar year in which each payment is to be made to the undersigned, or in either of the two calendar years preceding such payments.

Unless otherwise defined herein, terms defined in the Credit Agreement and used herein shall have the meanings given to them in the Credit Agreement.

[NAME OF LENDER]

Ву:			
Name:			
Title:			
Date:	, 20[]		
atc.	, 20[ ]		

Exhibit I-2 to Credit Agreement

# FORM OF U.S. TAX COMPLIANCE CERTIFICATE

(For Foreign Participants That Are Not Partnerships For U.S. Federal Income Tax Purposes)

Reference is hereby made to the Credit Agreement dated as of November 7, 2012 (as amended, supplemented or otherwise modified from time to time, the "Credit Agreement"), among Clean Energy Tranche A LNG Plant, LLC and Clean Energy Tranche B LNG Plant, LLC, as Borrowers, the Lenders from time to time party thereto, and General Electric Capital Corporation, as Administrative Agent, General Electric Capital Corporation as Collateral Agent, and each other party from time to time party thereto.

Pursuant to the provisions of Section 2.14(f)(ii)(B) of the Credit Agreement, the undersigned hereby certifies that (i) it is the sole record and beneficial owner of the participation in respect of which it is providing this certificate, (ii) it is not a bank within the meaning of Section 881(c)(3)(A) of the Code, (iii) it is not a ten percent shareholder of either Borrower within the meaning of Section 871(h)(3)(B) of the Code, and (iv) it is not a controlled foreign corporation related to either Borrower as described in Section 881(c)(3)(C) of the Code.

The undersigned has furnished its participating Lender with a certificate of its non-U.S. Person status on IRS Form W-8BEN. By executing this certificate, the undersigned agrees that (1) if the information provided on this certificate changes, the undersigned shall promptly so inform such Lender in writing, and (2) the undersigned shall have at all times furnished such Lender with a properly completed and currently effective certificate in either the calendar year in which each payment is to be made to the undersigned, or in either of the two calendar years preceding such payments.

Unless otherwise defined herein, terms defined in the Credit Agreement and used herein shall have the meanings given to them in the Credit Agreement.

rigice	men.		
[NAM	IE OF LENI	DER]	
By:			
	Name:		
	Title:		
Date:		, 20[]	

Exhibit I-3 to Credit Agreement

# FORM OF U.S. TAX COMPLIANCE CERTIFICATE

(For Foreign Participants That Are Partnerships For U.S. Federal Income Tax Purposes)

Reference is hereby made to the Credit Agreement dated as of November 7, 2012 (as amended, supplemented or otherwise modified from time to time, the "<u>Credit Agreement</u>"), among Clean Energy Tranche A LNG Plant, LLC and Clean Energy Tranche B LNG Plant, LLC, as Borrowers, the Lenders from time to time party thereto, and General Electric Capital Corporation, as Administrative Agent, General Electric Capital Corporation as Collateral Agent, and each other party from time to time party thereto.

Pursuant to the provisions of Section 2.14(f)(ii)(B) of the Credit Agreement, the undersigned hereby certifies that (i) it is the sole record owner of the participation in respect of which it is providing this certificate, (ii) its direct or indirect partners/members are the sole beneficial owners of such participation, (iii) with respect such participation, neither the undersigned nor any of its direct or indirect partners/members is a bank extending credit pursuant to a loan agreement entered into in the ordinary course of its trade or business within the meaning of Section 881(c)(3)(A) of the Code, (iv) none of its direct or indirect partners/members is a ten percent shareholder of either Borrower within the meaning of Section 871(h)(3)(B) of the Code and (v) none of its direct or indirect partners/members is a controlled foreign corporation related to either Borrower as described in Section 881(c)(3)(C) of the Code.

The undersigned has furnished its participating Lender with IRS Form W-8IMY accompanied by one of the following forms from each of its partners/members that is claiming the portfolio interest exemption: (i) an IRS Form W-8BEN or (ii) an IRS Form W-8IMY accompanied by an IRS Form W-8BEN from each of such partner's/member's beneficial owners that is claiming the portfolio interest exemption. By executing this certificate, the

undersigned agrees that (1) if the information provided on this certificate changes, the undersigned shall promptly so inform such Lender and (2) the undersigned shall have at all times furnished such Lender with a properly completed and currently effective certificate in either the calendar year in which each payment is to be made to the undersigned, or in either of the two calendar years preceding such payments. Unless otherwise defined herein, terms defined in the Credit Agreement and used herein shall have the meanings given to them in the Credit Agreement. [NAME OF PARTICIPANT] Bv: Name: Title: , 20[] Date: Exhibit I-4 Credit Agreement FORM OF U.S. TAX COMPLIANCE CERTIFICATE (For Foreign Lenders That Are Partnerships For U.S. Federal Income Tax Purposes) Reference is hereby made to the Credit Agreement dated as of November 7, 2012 (as amended, supplemented or otherwise modified from time to time, the "Credit Agreement"), among Clean Energy Tranche A LNG Plant, LLC and Clean Energy Tranche B LNG Plant, LLC, as Borrowers, the Lenders from time to time party thereto, and General Electric Capital Corporation, as Administrative Agent, General Electric Capital Corporation as Collateral Agent, and each other party from time to time party thereto. Pursuant to the provisions of Section 2.14(f)(ii)(B) of the Credit Agreement, the undersigned hereby certifies that (i) it is the sole record owner of the Loan(s) (as well as any Note(s) evidencing such Loan(s)) in respect of which it is providing this certificate, (ii) its direct or indirect partners/members are the sole beneficial owners of such Loan(s) (as well as any Note(s) evidencing such Loan(s)), (iii) with respect to the extension of credit pursuant to this Credit Agreement or any other Loan Document, neither the undersigned nor any of its direct or indirect partners/members is a bank extending credit pursuant to a loan agreement entered into in the ordinary course of its trade or business within the meaning of Section 881(c)(3)(A) of the Code, (iv) none of its direct or indirect partners/members is a ten percent shareholder of either Borrower within the meaning of Section 871(h)(3)(B) of the Code and (v) none of its direct or indirect partners/members is a controlled foreign corporation related to either Borrower as described in Section 881(c)(3)(C) of the Code. The undersigned has furnished the Administrative Agent and the Borrowers with IRS Form W-8IMY accompanied by one of the following forms from each of its partners/members that is claiming the portfolio interest exemption: (i) an IRS Form W-8BEN or (ii) an IRS Form W-8IMY accompanied by an IRS Form W-8BEN from each of such partner's/member's beneficial owners that is claiming the portfolio interest exemption. By executing this certificate, the undersigned agrees that (1) if the information provided on this certificate changes, the undersigned shall promptly so inform the

Borrowers and the Administrative Agent, and (2) the undersigned shall have at all times furnished the Borrowers and the Administrative Agent with a properly completed and currently effective certificate in either the calendar year in which each payment is to be made to the undersigned, or in either of the

Unless otherwise defined herein, terms defined in the Credit Agreement and used herein shall have the meanings given to them in the Credit

two calendar years preceding such payments. Agreement. [NAME OF LENDER] By: Name: Title: Date: , 20[]

Credit Agreement

# **CONSTRUCTION LOAN COMMITMENTS**

Lender	er		Tranche A Construction Loan Commitment		Tranche B Construction Loan Commitment	
	General Electric Capital Corporation	S	\$	100,000,000	\$	100,000,000

Annex II to

Credit Agreement

# APPLICABLE LENDING OFFICES

 $\frac{\text{Annex III}}{\text{to}}$   $\frac{\text{Credit Agreement}}{\text{Credit Agreement}}$ 

# **TERM LOAN AMORTIZATION**

Quarterly Dates following the	
applicable	Percentage
Conversion Date	Amortization 0.45%
2	0.45%
3	0.46%
4	0.46%
5	0.46%
6	0.46%
7	0.46%
8	3.20%
9	3.20%
10	3.20%
11	3.20%
12	3.20%
13	3.20%
14	3.20%
15	3.20%
16	3.20%
17	3.20%
18	3.20%
19	3.20%
20	3.20%
21	3.20%
22	3.20%
23	3.20%
24	3.20%
25	3.20%
26	3.20%
27	3.20%
28	3.20%
29	3.20%
30	3.20%
31	3.20%
Maturity Date	00.000/
Payment	20.00%

# **EQUITY CONTRIBUTION AGREEMENT**

by and among

CLEAN ENERGY TRANCHE A LNG PLANT, LLC and CLEAN ENERGY TRANCHE B LNG PLANT, LLC, as Borrowers

CLEAN ENERGY FUELS CORP., as Equity Investor

GENERAL ELECTRIC CAPITAL CORPORATION, as Administrative Agent

and

GENERAL ELECTRIC CAPITAL CORPORATION, as Collateral Agent

dated as of November 7, 2012

# **EQUITY CONTRIBUTION AGREEMENT**

This EQUITY CONTRIBUTION AGREEMENT (this "Agreement"), dated as of November 7, 2012, by and among CLEAN ENERGY TRANCHE A LNG PLANT, LLC, a limited liability company organized under the laws of the State of Delaware (the "Tranche A Borrower"), (ii) CLEAN ENERGY TRANCHE B LNG PLANT, LLC, a limited liability company organized under the laws of the State of Delaware (the "Tranche B Borrower" and, together with the Tranche B Borrower, collectively, the "Borrowers"), CLEAN ENERGY FUELS CORP., a Delaware corporation (the "Equity Investor"), GENERAL ELECTRIC CAPITAL CORPORATION, as administrative agent (in such capacity, together with its successors and permitted assigns in such capacity, the "Administrative Agent") for the Lenders (as defined in the Credit Agreement referred to below), and GENERAL ELECTRIC CAPITAL CORPORATION, as collateral agent (in such capacity, together with its successors and permitted assigns in such capacity, the "Collateral Agent") for the Secured Parties (as defined in the Credit Agreement referred to below).

### WITNESSETH:

WHEREAS, the Borrowers intend to undertake the development, construction and ownership of two (2) micro-LNG facilities, each with a capacity of approximately 250,000 gallons per day (the "**Projects**");

WHEREAS, the Borrowers, the various financial institutions from time to time party thereto, as Lenders, General Electric Capital Corporation, as administrative agent, and the Collateral Agent have entered into a Credit Agreement dated as of November 7, 2012 (as the same may be amended, amended and restated, supplemented or otherwise modified from time to time, the "<u>Credit Agreement</u>"), providing for the making of Loans to the Borrowers as contemplated therein; and

WHEREAS, the Equity Investor is an indirect owner of all of the equity of each Borrower;

WHEREAS, the Equity Investor will receive substantial direct and indirect benefits through the development of the Projects, and such development is dependent upon the Equity Contributions and Contingent Equity Contributions provided for herein;

WHEREAS, the execution and delivery by the Equity Investor and the Borrowers of this Agreement is a condition precedent to each Lender's obligation to make the Loans;

NOW, THEREFORE, in consideration of the premises contained herein and other benefits to the Equity Investor and the Borrowers, the receipt and sufficiency of which are hereby acknowledged, each of the parties hereto hereby agrees as follows:

# Section 1. <u>Definitions and Interpretation</u>.

- 1.1 <u>Definitions</u>. All capitalized terms used herein (including in the preamble and recitals hereto) and not otherwise defined herein shall have the meanings ascribed thereto in the Credit Agreement. The following terms shall have the following respective meanings:
- "Acceptable Issuing Bank" shall mean a bank or other financial institution reasonably acceptable to the Collateral Agent (i) whose long-term senior unsecured debt is rated at least A- by Standard & Poor's and A3 by Moody's and (ii) that has a capital, surplus and undivided profits of at least \$1,000,000,000.
  - "Administrative Agent" shall have the meaning set forth in the preamble.
  - "**Agreement**" shall have the meaning set forth in the preamble.
- "Approved Sponsor Expenditure" shall mean all Project Costs paid by the Sponsor prior to the applicable Initial Funding Date that have been approved by the Administrative Agent pursuant to Section 3.3(k) of the Credit Agreement.
  - "Borrowers" shall have the meaning set forth in the preamble.

- "CapEx Equity Contribution" shall have the meaning set forth in Section 2.2(c).
- "Collateral Agent" shall have the meaning set forth in the preamble.
- "Commercial Operations Date" shall have the meaning set forth in the applicable LNG Purchase Agreement.
- "Contingent Equity Contributions" shall mean the CapEx Equity Contributions, the LNG Purchase Agreement Equity Contributions and the WC Equity Contributions.
  - "Credit Agreement" shall have the meaning set forth in the recitals.
  - "Defaulted Payment" shall have the meaning set forth in Section 7.1.
  - "Discharge of Obligations" shall have the meaning set forth in Section 4.3.
  - "ECA Collateral" shall have the meaning set forth in Section 5.1.
  - "Equity Commitment" shall mean, as of any date of determination, \$66,666,666.67, minus the Funded Equity Amount.
  - "Equity Investor" shall have the meaning set forth in the preamble.
- "Equity Letters of Credit" shall mean, collectively, the Tranche A Equity Letters of Credit and the Tranche B Equity Letters of Credit and any letters of credit issued in substitution thereof.

"Funded Equity Amount" shall mean, as of any date of determination, the sum of (i) the aggregate amount of Approved Sponsor Expenditures made prior to such date of determination, (ii) the aggregate amount of Primary Equity Contributions made during the period from the Closing Date to such date of determination and (iii) the aggregate amount drawn under the Equity Letters of Credit prior to such date of determination.

- "Insolvency Proceeding" shall have the meaning set forth in Section 6.1.
- "LNG Purchase Agreement Equity Contribution" shall have the meaning set forth in Section 2.2(a).
- "Monthly Liquefaction Charge" shall have the meaning set forth in the applicable LNG Purchase Agreement.
- "Primary Equity Contribution" shall have the meaning set forth in Section 2.1(a).
- "Projects" shall have the meaning set forth in the recitals.
- "Retained Interest" shall have the meaning set forth in Section 7.3.
- "Support Period" shall mean (i) as to the Tranche A Borrower, the period between the Closing Date and the Facility A Conversion Date and (ii) as to the Tranche B Borrower, the period between the Closing Date and the Facility B Conversion Date.
  - "Take or Pay Payment Obligation" shall have the meaning set forth in the applicable LNG Purchase Agreement.
  - "Tranche A Borrower" shall have the meaning set forth in the preamble.
  - "Tranche A Equity Letters of Credit" shall have the meaning set forth in Section 9(a).
  - "Tranche B Borrower" shall have the meaning set forth in the preamble.
  - "Tranche B Equity Letters of Credit" shall have the meaning set forth in Section 9(b).
  - "WC Equity Contribution" shall have the meaning set forth in Section 2.2(b).
- 1.2 <u>Interpretation</u>. <u>Sections 1.2</u> and <u>1.3</u> of the Credit Agreement shall apply to this Agreement as if set forth in this Agreement, *mutatis mutandis*.
- 1.3 No Strict Construction. The parties hereto have participated jointly in the negotiation and drafting of this Agreement. If an ambiguity or question of intent or interpretation arises, this Agreement shall be construed as if drafted jointly by the parties hereto and no presumption or burden of proof shall arise favoring or disfavoring any party by virtue of the authorship of any provisions of this Agreement.

- (a) The Equity Investor hereby agrees to make during the Support Period capital contributions to the Borrowers in immediately available funds when, as and to the extent expressly required pursuant to <u>Section 2.1(b)</u> in an aggregate amount not to exceed the Equity Commitment as determined on the Closing Date (each such capital contribution, a "<u>Primary Equity Contribution</u>").
- (b) Subject to the limitations contained in Section 2.1(a), during the Support Period and in accordance with Section 3.3(k) of the Credit Agreement, the Equity Investor shall make, or shall cause to be made, at any time when a Borrower requests that the Lenders make a Construction Loan pursuant to a Notice of Borrowing and at least two (2) Business Days prior to each Disbursement Date, Primary Equity Contributions to such Borrower in an amount such that the ratio of (i) the aggregate amount of all Construction Loans made to such Borrower (including the Construction Loans to be advanced on such Disbursement Date) to (ii) the Funded Equity Amount (including the Primary Equity Contribution then being made) does not exceed 75:25. The Equity Investor and the Borrowers hereby acknowledge and agree that the making of such Primary Equity Contributions shall be a condition precedent to the Lenders' obligations to make each requested Construction Loan pursuant to the Credit Agreement.
- (c) At any time that an Event of Default has occurred and is continuing, the Equity Investor shall, upon the written request of the Collateral Agent (at the direction of the Required Lenders), make Primary Equity Contributions in an amount equal to the unfunded Equity Commitment existing at such time on the first Business Day following such written request, such Primary Equity Contributions to be allocated between the Borrowers as directed by the Collateral Agent. Such Primary Equity Contributions shall be deposited directly into the applicable Construction Account and shall be applied in accordance with the terms of the Collateral Account Agreement.
- 2.2 <u>Contingent Equity Contributions</u>. In addition to the Primary Equity Contributions described in <u>Section 2.1</u> above, the Equity Investor hereby agrees to make additional capital contributions to each Borrower in immediately available funds as follows:
- (a) (i) On the first Business Day of each month, an amount equal to any reduction in the Monthly Liquefaction Charge set forth in the invoice for such Monthly Liquefaction Charge pursuant to Section 10.1(b)(ii) of the applicable LNG Purchase Agreement and (ii) on the Business Day following the date that any Take or Pay Payment Obligation is not paid by the LNG Purchaser when due in accordance with the applicable LNG Purchase Agreement, in an amount equal to such unpaid Take or Pay Payment Obligation (each such capital contribution, an "LNG Purchase Agreement Equity Contribution shall be deposited directly into the applicable Project Revenues Collection Account and shall be applied in accordance with the terms of the Collateral Account Agreement.

- (b) On or prior to the date that such Borrower is expected to be unable to make a payment under any Project Document due to a mismatch in the timing of payments expected to be received under other Project Documents, in an amount equal to the amount required to be paid (each such capital contribution, a "WC Equity Contribution"). Each WC Equity Contribution shall be deposited directly into the applicable Operating Account and shall be applied in accordance with the terms of the Collateral Account Agreement.
- (c) On or prior to the date that any payment is required to be made for the acquisition, purchase, construction or completion of any capital expenditure required to be made by such Borrower to a Project in order to comply with any Law (other than any such capital expenditure provided for as a Project Cost in the Construction Budget for such Project or any payment for which an insurance or warranty provider has assumed responsibility), in an amount equal to such required payment (each such capital contribution, a "CapEx Equity Contribution"). Each CapEx Equity Contribution shall be made by direct payment to the Person to whom such payment is due. The Equity Investor shall promptly provide a copy of each payment requisition or invoice with respect to such capital expenditure, together with evidence of making each such payment, to the Administrative Agent.

# Section 3. <u>Specific Provisions.</u>

- 3.1 Reinstatement. This Agreement and the obligations of the Equity Investor hereunder shall automatically be reinstated if and to the extent that for any reason any Primary Equity Contribution or Contingent Equity Contribution made pursuant to this Agreement is rescinded or otherwise restored to the Equity Investor, whether as a result of any Insolvency Proceeding or otherwise with respect to the Borrowers or any other Person or as a result of any settlement or compromise with any Person (including the Equity Investor) in respect of such payment, and the Equity Investor shall pay the Collateral Agent on demand all of its reasonable out-of-pocket costs and expenses (including reasonable fees of outside counsel) incurred by the Collateral Agent in connection with such rescission or restoration.
- 3.2 <u>Specific Performance</u>. To the extent it may do so under applicable legal requirements, the Collateral Agent may demand specific performance of this Agreement. The Equity Investor hereby irrevocably waives, to the extent the Equity Investor may do so under applicable legal requirements, any defense based on the adequacy of a remedy at law that may be asserted as a bar to the remedy of specific performance in any action brought against the Equity Investor for specific performance of this Agreement by the Collateral Agent or any successor or assign thereof or for its benefit by a receiver, custodian or trustee appointed for the Borrowers or in respect of all or a substantial part of their assets, under the Bankruptcy Laws of any jurisdiction to which the Borrowers or their assets are subject.

# Section 4. <u>Unconditionality; Waivers; Subrogation</u>.

4.1 <u>Obligations Unconditional</u>. The liability of the Equity Investor hereunder shall be irrevocable, absolute and unconditional irrespective of, and the Equity Investor hereby irrevocably waives any defenses (other than a defense of payment in full) it may now have or hereafter acquire in any way relating to, any or all of the following:

- (a) any lack of validity or enforceability of any Finance Document or any other agreement or instrument relating thereto;
- (b) any change in the time, manner or place of payment of, or in any other term of, all or any of the obligations of the Equity Investor or any other Finance Party or any other amendment or waiver of or any consent to departure from any Finance Document, including any increase in the Obligations resulting from the extension of additional credit to the Equity Investor, any other Finance Party or otherwise; provided that the amount of the Equity Commitment shall not be increased unless consented to in accordance with **Section 10.2**;

- (c) any taking, exchange, release or non-perfection of any Collateral or any other collateral, or any taking, release or amendment or waiver of, or consent to departure from, any guaranty, for all or any of the Obligations;
- (d) any manner of application of Collateral or any other collateral, or proceeds thereof, to all or any of the Obligations, or any manner of sale or other disposition of any Collateral or any other collateral for all or any of the Obligations or any other Property of the Equity Investor or any other Finance Party;
  - (e) any change, restructuring or termination of the corporate structure or existence of the Equity Investor or any other Finance Party;
- (f) any failure of any Secured Party to disclose to the Equity Investor or any other Finance Party any information relating to the business, condition (financial or otherwise), operations, performance, Properties or prospects of the Equity Investor or any other Finance Party now or hereafter known to such Secured Party (the Equity Investor waiving any duty on the part of the Secured Parties to disclose such information);
- (g) the failure of any other Person to execute or deliver this Agreement, any guaranty or other agreement or the release or reduction of liability of the Equity Investor or any guarantor or surety with respect to the obligations hereunder; or
- (h) any other circumstance (including any statute of limitations), whether or not similar to any of the foregoing, or any existence of or reliance on any representation by any Secured Party that might otherwise constitute a defense available to, or a discharge of, the Equity Investor, any other Finance Party or any guarantor or surety other than payment in full of the Obligations.

# 4.2 <u>Waiver and Acknowledgments</u>.

(a) The Equity Investor hereby unconditionally and irrevocably waives promptness, diligence, notice of acceptance, presentment, demand for performance, notice of nonperformance, default, acceleration, protest or dishonor and any other notice with respect to its obligations hereunder and this Agreement except as expressly provided herein and any requirement that any Secured Party protect, secure, perfect or insure any Lien or any Property subject thereto or exhaust any right or take any action against the Equity Investor, any other Finance Party or any other Person or any Collateral.

6

- (b) The Equity Investor hereby unconditionally and irrevocably waives any right to revoke this Agreement and acknowledges that its obligations hereunder are continuing in nature and apply to all Obligations, whether existing now or in the future.
- (c) The Equity Investor hereby unconditionally and irrevocably waives (i) any defense arising by reason of any claim or defense (other than the defense of payment in full) based upon an election of remedies by any Secured Party that in any manner impairs, reduces, releases or otherwise adversely affects the subrogation, reimbursement, exoneration, contribution or indemnification rights of the Equity Investor or other rights of the Equity Investor to proceed against any of the other Finance Parties, any guarantor or any other Person or any Collateral and (ii) any defense (other than the defense of payment in full) based on any right of set-off or counterclaim against or in respect of the obligations of the Equity Investor hereunder.
  - (d) [Intentionally Omitted].
- (e) The Equity Investor hereby unconditionally and irrevocably waives any duty on the part of any Secured Party to disclose to the Equity Investor any matter, fact or thing relating to the business, condition (financial or otherwise), operations, performance, Properties or prospects of any other Finance Party now or hereafter known by such Secured Party.
- (f) The Equity Investor acknowledges that it will receive substantial direct and indirect benefits from the financing arrangements contemplated by the Finance Documents, and that the waivers set forth in <u>Section 4.1</u> and this <u>Section 4.2</u> are knowingly made in contemplation of such benefits.
- 4.3 <u>Subrogation</u>. Notwithstanding any payment or payments made by the Equity Investor under <u>Section 2</u> hereof, prior to the payment in full in cash of all the Obligations (other than contingent indemnity obligations in respect of which no claim has been made) and the expiration or termination of the Commitments of the Lenders to make Loans under the Credit Agreement (the "<u>Discharge of Obligations</u>"), the Equity Investor waives any claim, right or remedy which it may now have or hereafter acquire against the Borrowers that arises hereunder and/or from the performance by the Equity Investor of its obligations hereunder, whether or not such claim, right or remedy arises in equity, under contract, by statute, under common law or otherwise. Any amount paid by the Borrowers to the Equity Investor in violation of the immediately preceding sentence prior to the date of Discharge of Obligations shall be held in trust for the benefit of the Collateral Agent (on behalf of the Secured Parties) and shall promptly thereafter be paid to the Collateral Agent for application in accordance with the Finance Documents.

# Section 5. <u>Grant of Security Interest.</u>

5.1 <u>Grant of Security Interest.</u> As collateral security for the payment in full when due (whether at stated maturity, by acceleration or otherwise) of the Obligations, each Borrower hereby grants to the Collateral Agent, for the benefit of the Secured Parties, a security interest in and continuing Lien on all of such Borrower's right, title and interest in, to and under this Agreement and all "Proceeds" (as defined in the UCC) of this Agreement (collectively, but excluding any such "Proceeds" that are distributed or otherwise transferred by such Borrower in

5.2 <u>Remedies.</u> Each of the Borrowers and the Equity Investor agrees that the Collateral Agent, acting for the benefit of the Secured Parties, shall be entitled, following the occurrence and during the continuance of an Event of Default, to foreclose on and exercise any and all rights of each Borrower with respect to such Borrower's interest in the ECA Collateral in accordance with the terms hereof and the other Security Documents (in its own name or in the name of such Borrower), and each of the Borrowers and the Equity Investor shall comply with any such exercise.

# Section 6. <u>Bankruptcy; Setoff.</u>

- Bankruptcy Code Waiver by the Equity Investor. The Equity Investor hereby irrevocably waives, to the extent it may do so under 6.1 applicable legal requirements, any protection to which it may be entitled under Sections 365(c)(1), 365(c)(2) and 365(e)(2) of the Bankruptcy Code or equivalent provisions of the laws or regulations of any other jurisdiction with respect to any proceedings, or any successor provision of law or similar import, in the event of any proceeding, voluntary or involuntary, involving the bankruptcy, reorganization, insolvency, receivership, liquidation or arrangement (an "Insolvency Proceeding") with respect to any other Finance Party. Specifically, in the event that the trustee (or similar official) in an Insolvency Proceeding with respect to any other Finance Party or the debtor-in-possession takes any action (including the institution of any action, suit or other proceeding for the purpose of enforcing the rights of such Finance Party under this Agreement), the Equity Investor shall not assert any defense, claim or counterclaim denying liability hereunder on the basis that this Agreement is an executory contract or a "financial accommodation" that cannot be assumed, assigned or enforced or on any other theory directly or indirectly based on Section 365(c)(1), 365(c)(2) or 365(e)(2) of the Bankruptcy Code, or equivalent provisions of the laws or regulations of any other jurisdiction with respect to any proceedings or any successor provision of law of similar import. If an Insolvency Proceeding with respect to any other Finance Party shall occur, the Equity Investor agrees, after the occurrence of such Insolvency Proceeding, to reconfirm in writing, to the extent permitted by applicable legal requirements, its pre-petition waiver of any protection to which it may be entitled under Sections 365(c)(1), 365(c)(2) and 365(e)(2) of the Bankruptcy Code or equivalent provisions of the laws or regulations of any other jurisdiction with respect to proceedings and, to give effect to such waiver, the Equity Investor consents, to the extent permitted by applicable legal requirements, to the assumption and enforcement of each provision of this Agreement by the debtor-in-possession or the applicable Finance Party's trustee in bankruptcy.
- 6.2 <u>Insolvency Proceedings</u>. None of the obligations of the Equity Investor under this Agreement shall be altered, limited or affected by any Insolvency Proceeding relating to any Finance Party, or by any defense which any Finance Party may have by reason of any order, decree or decision of any court or administrative body resulting from any such proceeding.

8

6.3 <u>Set-Off.</u> In addition to any rights now or hereafter granted under applicable legal requirements or otherwise, and not by way of limitation of any such rights, upon the failure of the Equity Investor to make any Primary Equity Contribution or Contingent Equity Contribution as and when required hereunder, the Collateral Agent is hereby authorized at any time or from time to time, without presentment, demand, protest or other notice of any kind to the Equity Investor or to any other person, any such notice being hereby expressly waived, to set-off and to appropriate and apply any and all deposits (general or special) at any time held by the Collateral Agent or any other Secured Party (including by branches and agencies of the Collateral Agent and each Secured Party wherever located) to or for the credit or the account of the Equity Investor, against and on account of the obligations of the Equity Investor under this Agreement, irrespective of whether or not the Collateral Agent or any other Secured Party shall have made any demand hereunder.

# Section 7. <u>Purchase of Participating Interests.</u>

- Required Purchase of Participating Interests. If by reason of an Insolvency Proceeding with respect to any Finance Party, or any act of a Governmental Authority, (a) any Primary Equity Contribution due hereunder has not been deposited in the applicable Construction Account or transferred to the Collateral Agent, as the case may be, within five (5) Business Days after the date on which such amount is payable hereunder, or (b) any Primary Equity Contribution theretofore deposited or transferred pursuant to **Section 2.1** is rescinded or otherwise restored to the Equity Investor and five (5) Business Days have elapsed after the date that such Primary Equity Contribution was rescinded or otherwise restored (such Primary Equity Contribution, whether required but not made as provided in clause (a) or made and returned as provided in clause (b), being herein called the "**Defaulted Payment**"), the Equity Investor shall, without any further notice or demand by the Collateral Agent, purchase an undivided participating interest in each of the Loans and Commitments then outstanding, as provided in the following sentence, in an aggregate principal amount equal to the amount of the Defaulted Payment. The Equity Investor's purchase of an undivided participating interest in such Loans and Commitments based on the respective outstanding amounts thereof. The Equity Investor shall effect its purchase of undivided participating interests in such Loans and Commitments pursuant to this **Section 7.1** by paying to the Collateral Agent, for the account of the holders of such Loans and Commitments, in immediately available funds in Dollars, the amount of the Defaulted Payment.
- 7.2 <u>Effect of Purchase of Participating Interests</u>. The Equity Investor's purchase of an undivided participating interest in the Loans and Commitments in the full amount (if any) required pursuant to <u>Section 7.1</u> following a Defaulted Payment in respect of Primary Equity Contributions, shall satisfy the Equity Investor's obligation pursuant to <u>Section 2.1</u> to make Primary Equity Contributions to the extent of the amount of such Loans and Commitments so purchased by the Equity Investor.
- 7.3 <u>Subordinate Nature of Participating Interest</u>. The Equity Investor hereby agrees that its participating interest in the Loans and Commitments purchased by it pursuant to <u>Section 7.1</u> shall be subordinate in all respects to the interest in such Loans and Commitments retained by the holders (other than the Equity Investor) thereof (the "<u>Retained Interest</u>"), so that all payments received or collected on account of such Loans and Commitments and applied to the

c

payment or termination thereof, whether received or collected through repayment of such Loans and Commitments by any other Finance Party or through right of set-off with respect thereto or realization upon any collateral security therefor or otherwise, shall first be applied to the payment of the principal, interest, fees and other amounts then due (whether at its stated maturity, by acceleration or otherwise) on the Retained Interest until such principal, interest, fees and other amounts are paid in full, before any such payments are applied on account of the Equity Investor's participating interest in such Loans and Commitments.

7.4 <u>Rights of Lenders.</u> Notwithstanding the purchase and ownership by the Equity Investor of participating interests in the Loans and Commitments, and notwithstanding the rights of participants under the Credit Agreement and the other Finance Documents, each Lender shall have the right,

in its sole discretion in each instance and without any notice to the Equity Investor, (a) to agree to the modification or waiver of any of the terms of any of the Finance Documents or any other agreement or instrument relating thereto (but not to reduce any amount payable in respect of the portion of the Loans and Commitments subject to participations purchased pursuant to Section 7.1), (b) to consent to any action or failure to act by any Finance Party or any other Person party to a Finance Document, and (c) to exercise or refrain from exercising any rights or remedies which any Lender may have under the Finance Documents or any other agreement or instrument relating thereto, including the right at any time to declare, or refrain from declaring, the Obligations due and payable upon the occurrence of any Event of Default, and to rescind and annul any such declaration, and to foreclose and sell or exercise any other remedy, or refrain from foreclosing and selling or exercising any other remedy, with respect to any collateral securing the Obligations. No Lender shall be liable to the Equity Investor for any error in judgment or for any action taken or omitted to be taken by it while the Equity Investor holds a participating interest in the Loans and Commitments, except for its gross negligence or willful misconduct. No Lender shall have any duty or responsibility to provide the Equity Investor with any credit or other information concerning the affairs, financial condition or business of any other Finance Party or any other party to a Finance Document which may come into their possession or the possession of any of their respective Affiliates, or to notify the Equity Investor of any Default or Event of Default, or any default by any other Finance Party or any other Person under any of the Finance Documents.

- 7.5 <u>No Voting Rights</u>. Without limiting the generality of the provisions of <u>Section 7.4</u>, in determining whether the required consent of the Lenders (or any portion thereof) has been obtained for all purposes under the Finance Documents, the participating interests in the Loans and Commitments purchased by the Equity Investor pursuant to <u>Section 7.1</u> shall not be deemed to be outstanding.
- 7.6 <u>Obligations Unconditional</u>. The obligations of the Equity Investor under this <u>Section 7</u> to purchase participating interests in or purchase and take an assignment of the Loans and Commitments, as the case may be, are absolute and unconditional and shall not be affected by the occurrence of any Default or Event of Default or any other circumstance, including any circumstance of the nature described in <u>Sections 4.1</u> or <u>4.2</u>.
- Section 8. Representations and Warranties of the Equity Investor. The Equity Investor hereby represents and warrants, as of the Closing Date, to the Administrative Agent and the Collateral Agent, which representations and warranties shall survive the execution and delivery of this Agreement and the consummation of the transactions contemplated hereby:

10

- (a) The Equity Investor (i) is a corporation duly organized, validly existing and in good standing under the laws of the State of Delaware, (ii) is duly qualified to do business as a foreign entity and in good standing under the laws of each jurisdiction in which it owns or leases Property or in which the conduct of its business requires it so to qualify, except the failure of which that could not reasonably be expected to result in a Material Adverse Effect, and (c) has all requisite power and authority to own or lease and operate its Properties, to carry on its business as contemplated by this Agreement and to execute, deliver and perform each Transaction Document to which it is or will be a party.
- (b) The execution, delivery and performance by the Equity Investor of each Transaction Document to which it is a party, and the transactions contemplated thereby (i) have been duly authorized by all necessary corporate action (including any necessary shareholder action), (ii) will not breach, contravene, violate, conflict with or constitute a default under (A) any of the Equity Investor's Charter Documents, (B) any material applicable Law, or (C) any material contract, loan agreement, indenture, mortgage, lease or other instrument to which the Equity Investor is a party or by which it or any of its Properties may be bound or affected, including all Governmental Approvals and the Transaction Documents, and (iii) will not result in or require the creation or imposition of any Lien upon or with respect to any of the Properties of the Equity Investor.
- (c) Each Transaction Document to which the Equity Investor is a party has been duly executed and delivered by the Equity Investor, is the legal, valid and binding obligation of the Equity Investor and enforceable against the Equity Investor in accordance with its terms, except as such enforcement may be limited by (i) the effect of bankruptcy, insolvency, reorganization, receivership, conservatorship, arrangement, moratorium or other laws affecting or relating to the rights of creditors generally, or (ii) the rules governing the availability of specific performance, injunctive relief or other equitable remedies and general principles of equity, regardless of whether considered in a proceeding in equity or at law.
- (d) No Governmental Approval has been, is or will be required to be obtained or made (i) in connection with the due execution, delivery or performance by the Equity Investor of its obligations, and the exercise of its rights under, any of the Transaction Documents to which it is a party or any transaction contemplated thereby or (ii) to ensure the legality, validity and binding effect or enforceability of any such Transaction Document, except for the Governmental Approvals that have been duly obtained or made, were validly issued, and are in full force and effect, are final, not subject to any pending appeal, intervention or similar proceeding, are held in the name of the Equity Investor and are free from conditions or requirements the compliance with which the Equity Investor does not reasonably expect to be able to satisfy. No event has occurred that could reasonably be expected to result in the revocation, termination or adverse modification of any such Governmental Approval or adversely affect any rights of the Equity Investor under any such Governmental Approval.
- (e) The Equity Investor is in compliance with its Charter Documents and is in compliance in all material respects with all Laws and Governmental Approvals applicable to it.

- (f) The Equity Investor has filed all federal, state and material foreign income tax returns which are required to be filed by it, and the Equity Investor has paid all taxes as shown on said returns and on all assessments received by it to the extent that such taxes have become due, except such taxes, if any, as are being contested in good faith and by proper proceedings as to which adequate reserves have been provided in accordance with GAAP.
  - (g) The Equity Investor is Solvent.
- (h) There is no action, suit, other legal proceeding, arbitral proceeding, inquiry or investigation pending or, to the best of the Equity Investor's knowledge, threatened, by or before any Governmental Authority or in any arbitral or other forum, nor any order, decree or judgment in effect, pending, or, to the best of the Equity Investor's knowledge, threatened, against or affecting the Equity Investor or any of its Properties or rights, that relates to the Projects, to any of the Transaction Documents or to any of the transactions contemplated thereby.
  - Section 9. Equity Letters of Credit. The Equity Investor hereby agrees as follows:

- (a) On the Initial Funding Date of the Tranche A Loans the Equity Investor shall deliver to the Collateral Agent one or more letters of credit (the "Tranche A Equity Letters of Credit") naming the Collateral Agent as beneficiary (which letters of credit shall be in form and substance reasonably acceptable to the Collateral Agent and shall each be issued by an Acceptable Issuing Bank) in an aggregate face amount equal to the lesser of (i) the excess of (A) 33% of the Tranche A Construction Loan Commitments over (B) the portion of the Funded Equity Amount allocable to the Tranche A Borrower (including any Primary Equity Contributions being made on such Initial Funding Date) and (ii) the excess of (A) 33% of the Total Construction Commitment over (B) the Funded Equity Amount (including any Primary Equity Contributions being made on such Initial Funding Date).
- (b) On the Initial Funding Date of the Tranche B Loans the Equity Investor shall deliver to the Collateral Agent one or more letters of credit (the "Tranche B Equity Letters of Credit") naming the Collateral Agent as beneficiary (which letters of credit shall be in form and substance reasonably acceptable to the Collateral Agent and shall each be issued by an Acceptable Issuing Bank) in an aggregate face amount equal to the lesser of (i) the excess of (A) 33% of the Tranche B Construction Loan Commitments over (B) the portion of the Funded Equity Amount allocable to the Tranche B Borrower (including any Primary Equity Contributions being made on such Initial Funding Date) and (ii) the excess of (A) 33% of the Total Construction Commitment over (B) the Funded Equity Amount (including any Primary Equity Contributions being made on such Initial Funding Date).
- (c) From the first Initial Funding Date and until the earliest of (A) the later of the Facility A Conversion Date and the Facility B Conversion Date, (B) the date falling thirty (30) days after the later of the Facility A Date Certain and the Facility B Date Certain and (C) the date on which the Equity Commitment equals \$0.00, the Equity Investor shall maintain one or more Equity Letters of Credit (which may include the Tranche A Equity Letters of Credit and the Tranche B Equity Letters of Credit) with an aggregate face amount equal the full amount of the then-effective unfunded Equity Commitment; provided that if the expiration date of an Equity

Letter of Credit will occur before the earliest of the dates described in clauses (A) through (C) above, the Equity Investor must, at least thirty (30) days prior to such expiration date (the "Expiration Date Limit"), provide at its own expense (y) evidence, in form and substance reasonably satisfactory to the Collateral Agent, that such expiration date has been extended for a period of time equal to no less than six (6) months or (z) one or more replacement Equity Letters of Credit, in each case on terms substantively identical to such expiring Equity Letter of Credit, with a face amount of the then-effective unfunded Equity Commitment and an expiration date no less than six (6) months thereafter; and provided, further, that, for the avoidance of doubt, the Equity Investor may. from time to time after the first Initial Funding Date, (i) reduce the face amount of the Equity Letters of Credit or (ii) provide replacement Equity Letters of Credit, so long as, in each case, the aggregate face amount of the Equity Letters of Credit remains sufficient to satisfy the requirements of this Section 4(c). If such evidence of extension or replacement does not occur prior to any Expiration Date Limit, then the Collateral Agent shall have the right, in its sole discretion, to draw upon such Equity Letter of Credit at any time thereafter.

- (d) The Equity Investor agrees that if at any time the issuer of any Equity Letter of Credit shall cease to be an Acceptable Issuing Bank, then, within ten (10) days of such occurrence, the Equity Investor shall replace any Equity Letters of Credit issued by such issuer with Equity Letters of Credit issued by an Acceptable Issuing Bank. If such replacement does not occur within such fifteen (15) day period, then the Collateral Agent shall have the right, in its sole discretion, to draw upon such Equity Letters of Credit at any time thereafter.
- (e) The Equity Investor agrees that the Collateral Agent may proceed directly to draw under the Equity Letters of Credit in respect of any of the obligations of the Equity Investor under Section 2.1 hereof as and when they become due (including obligations that are due as a result of the acceleration thereof) without first proceeding against the Equity Investor. Any amounts paid to the Collateral Agent under an Equity Letter of Credit shall be immediately credited against and shall discharge the obligations of the Equity Investor hereunder in respect of which such payment was made.

# Section 10. Miscellaneous.

- 10.1 <u>No Waiver; Cumulative Remedies</u>. No failure or delay on the part of any Person in exercising any right, power or remedy hereunder shall operate as a waiver thereof, nor shall any single or partial exercise of any such right, power or remedy preclude any other or further exercise thereof or the exercise of any other right, power or remedy hereunder. The remedies prescribed in this Agreement or by Law shall be cumulative and not in limitation of or substitution for any other remedies available hereunder or otherwise, all of which shall be available to the parties hereto and their respective successors and permitted assigns.
- Amendments. None of the terms and conditions of this Agreement may be amended, supplemented, modified or waived, nor may any consent under or with respect to such terms and conditions be granted, unless (a) each of the parties hereto agrees thereto in writing, and (b) such amendment supplement, modification, waiver or consent is executed in accordance with the provisions of **Section 11.10** of the Credit Agreement.

- Notices. Except as otherwise expressly provided herein, all notices, requests and demands to or upon the respective parties hereto to be effective shall be in writing (including by telecopy), and shall be deemed to have been duly given or made when delivered by hand, or upon actual receipt if deposited in the United States mail, postage prepaid, or, in the case of telecopy notice, when confirmation of successful transmission is received, or, in the case of a nationally recognized overnight courier service, charges prepaid, one (1) Business Day after delivery to such courier service, addressed, in the case of each party hereto, at its address specified opposite its name on **Schedule 10.3** hereto, or to such other address as may be designated by any party in a written notice to the other party hereto.
- 10.4 <u>Successors and Assigns</u>. This Agreement shall be binding upon and inure to the benefit of the parties hereto, all future holders of the Obligations and their respective successors and assigns. The Administrative Agent and the Collateral Agent may assign or transfer their obligations and rights hereunder only to their respective permitted successors under the Credit Agreement. The Equity Investor may not assign or transfer any of its rights or obligations hereunder without the prior written consent of the Collateral Agent (acting at the direction of the Required Lenders).
- 10.5 <u>No Waiver; Remedies Cumulative</u>. No failure or delay on the part of the Collateral Agent or any other Secured Party in exercising any right, power or privilege hereunder and no course of dealing between or among the Equity Investor and the Collateral Agent or any other Secured Party shall operate as a waiver thereof, nor shall any single or partial exercise of any right, power or privilege hereunder preclude any other or further exercise thereof or

the exercise of any other right, power or privilege hereunder or thereunder. The rights and remedies herein expressly provided are cumulative and not exclusive of any rights or remedies which the Collateral Agent or any other Secured Party would otherwise have. No notice to or demand on the Equity Investor in any case shall entitle the Equity Investor to any other or further notice or demand in similar or other circumstances or constitute a waiver of the rights of the Collateral Agent or any other Secured Party to any other or further action in any circumstances without notice or demand.

# 10.6 GOVERNING LAW; SUBMISSION TO JURISDICTION; WAIVER OF JURY TRIAL.

- (a) THIS AGREEMENT AND THE RIGHTS AND OBLIGATIONS OF THE PARTIES HEREUNDER SHALL BE CONSTRUED IN ACCORDANCE WITH AND BE GOVERNED BY THE LAWS OF THE STATE OF NEW YORK.
- (b) ANY LEGAL ACTION OR PROCEEDING AGAINST THE BORROWERS OR THE EQUITY INVESTOR WITH RESPECT TO THIS AGREEMENT AND ANY ACTION FOR ENFORCEMENT OF ANY JUDGMENT IN RESPECT THEREOF MAY BE BROUGHT IN THE COURTS OF THE STATE OF NEW YORK IN THE COUNTY OF NEW YORK OR OF THE UNITED STATES OF AMERICA FOR THE SOUTHERN DISTRICT OF NEW YORK, AND, BY EXECUTION AND DELIVERY OF THIS AGREEMENT, THE EQUITY INVESTOR AND EACH BORROWER HEREBY ACCEPTS FOR ITSELF AND IN RESPECT OF ITS PROPERTY, GENERALLY AND UNCONDITIONALLY, THE NON-EXCLUSIVE JURISDICTION OF THE AFORESAID

14

COURTS AND APPELLATE COURTS FROM ANY APPEAL THEREOF. THE EQUITY INVESTOR AND EACH BORROWER IRREVOCABLY CONSENTS TO THE SERVICE OF PROCESS OUT OF ANY OF THE AFOREMENTIONED COURTS IN ANY SUCH ACTION OR PROCEEDING BY THE MAILING OF COPIES THEREOF BY REGISTERED OR CERTIFIED MAIL, POSTAGE PREPAID, TO THE EQUITY INVESTOR OR SUCH BORROWER, AS APPLICABLE, AT THE ADDRESS PROVIDED FOR IT IN **SECTION 10.3**. THE EQUITY INVESTOR AND EACH BORROWER HEREBY IRREVOCABLY WAIVES ANY OBJECTION WHICH IT MAY NOW OR HEREAFTER HAVE TO THE LAYING OF VENUE OF ANY OF THE AFORESAID ACTIONS OR PROCEEDINGS ARISING OUT OF OR IN CONNECTION WITH THIS AGREEMENT BROUGHT IN THE COURTS REFERRED TO ABOVE AND HEREBY FURTHER IRREVOCABLY WAIVES AND AGREES, TO THE EXTENT PERMITTED BY APPLICABLE LAW, NOT TO PLEAD OR CLAIM IN ANY SUCH COURT THAT ANY SUCH ACTION OR PROCEEDING BROUGHT IN ANY SUCH COURT HAS BEEN BROUGHT IN AN INCONVENIENT FORUM. NOTHING HEREIN SHALL AFFECT THE RIGHT OF ANY PARTY HERETO TO SERVE PROCESS IN ANY OTHER MANNER PERMITTED BY LAW OR TO COMMENCE LEGAL PROCEEDINGS OR OTHERWISE PROCEED IN ANY OTHER JURISDICTION.

- (c) EACH OF THE PARTIES HERETO HEREBY IRREVOCABLY WAIVES ALL RIGHT OF TRIAL BY JURY IN ANY ACTION, PROCEEDING OR COUNTERCLAIM ARISING OUT OF OR IN CONNECTION WITH THIS AGREEMENT OR ANY MATTER ARISING HEREUNDER.
- 10.7 <u>Headings Descriptive</u>. The headings of the several Sections and subsections of this Agreement are inserted for convenience only and shall not in any way affect the meaning or construction of any provision of this Agreement.
- 10.8 <u>No Third Party Beneficiaries</u>. The agreements of the parties hereto are solely for the benefit of the Equity Investor, the Borrowers, the Collateral Agent, the Administrative Agent, the Depositary Agent and the Secured Parties and their respective successors and assigns and no Person (other than the parties hereto and such Secured Parties) shall have any rights hereunder.
- 10.9 <u>Survival</u>. All agreements, statements, representations and warranties made by the Equity Investor herein shall be considered to have been relied upon by the Secured Parties and shall survive the execution and delivery of this Agreement and the other Finance Documents regardless of any investigation made by the Secured Parties or on their behalf until the Obligations (other than contingent indemnity obligations in respect of which no claim has been made) shall have been paid in full in cash.
- 10.10 <u>Severability</u>. In case any provision in or obligation under this Agreement shall be invalid, illegal or unenforceable in any jurisdiction, the validity, legality and enforceability of the remaining provisions or obligations, or of such provision or obligation in any other jurisdiction, shall not in any way be affected or impaired thereby.
  - 10.11 <u>Independent Obligations</u>. The Equity Investor's obligations under this Agreement are independent of those of the Borrowers.

15

- 10.12 <u>Counterparts</u>. This Agreement may be executed in any number of counterparts and by the different parties hereto on separate counterparts, each of which when so executed and delivered shall be an original, but all of which shall together constitute one and the same instrument.
- 10.13 <u>Entire Agreement</u>. This Agreement, together with any other agreement or instrument executed in connection herewith, is intended by the parties hereto as a final expression of their agreement as to the matters covered hereby and is intended as a complete and exclusive statement of the terms and conditions thereof.
- 10.14 <u>Termination</u>. Notwithstanding anything contained herein to the contrary, this Agreement and the obligations of the Borrowers and the Equity Investor hereunder shall terminate upon the payment in full in cash of all Obligations (other than contingent indemnity obligations in respect of which no claim has been made) and the termination of all Commitments under the Credit Agreement.

[SIGNATURE PAGES FOLLOW]

IN WITNESS WHEREOF, the parties hereto have caused their duly authorized officers to execute and deliver this Agreement as of the date first above written.

> CLEAN ENERGY FUELS CORP., as Equity Investor

/s/ Andrew J. Littlefair Bv:

Name: Andrew J. Littlefair

Title: President & Chief Executive Officer

CLEAN ENERGY TRANCHE A LNG PLANT, LLC, as Tranche A Borrower

/s/ Andrew J. Littlefair By:

Name: Andrew J. Littlefair

Title: President & Chief Executive Officer

CLEAN ENERGY TRANCHE B LNG PLANT, LLC,

as Tranche B Borrower

By: /s/ Andrew J. Littlefair

Name: Andrew J. Littlefair

Title: President & Chief Executive Officer

GENERAL ELECTRIC CAPITAL CORPORATION, as Administrative Agent and Collateral Agent

By: /s/ Tyson Yates Name: Tyson Yates

Title: **Authorized Signatory** 

Signature Page to Equity Contribution Agreement

Schedule 10.3 to **Equity Contribution Agreement** 

# **ADDRESSES**

If to Equity Investor: Clean Energy Fuels Corp.

c/o Clean Energy Fuels Corp. 3020 Old Ranch Parkway Seal Beach, CA 90740

Attn: Nate Jensen and Rick Wheeler Telephone: (562) 493-7239 Facsimile: (562) 493-4956

If to Tranche A Borrower: Clean Energy Tranche A LNG Plant, LLC

c/o Clean Energy Fuels Corp. 3020 Old Ranch Parkway Seal Beach, CA 90740

Attn: Nate Jensen and Rick Wheeler Telephone: (562) 493-7239 Facsimile: (562) 493-4956

Clean Energy Tranche B LNG Plant, LLC

c/o Clean Energy Fuels Corp. 3020 Old Ranch Parkway Seal Beach, CA 90740

Attn: Nate Jensen and Rick Wheeler Telephone: (562) 493-7239

Facsimile: (562) 493-4956

General Electric Capital Corporation

c/o GE Energy Financial Services

800 Long Ridge Road Stamford, CT 06927 Attn: Dipika Chanana

If to Tranche B Borrower:

If to the Administrative Agent or the Collateral Agent:

Tel: 203-316-7521 Fax: 513-770-2270

Email: EFS. Agency Services@ge.com

With copies to: Attn: Stephen Lenihan Tel: 203-961-2273 Fax: 203-357-4897

Email: stephen.lenihan@ge.com

## WARRANT AGREEMENT

This **WARRANT AGREEMENT** (including all Exhibits hereto, this "*Agreement*") is dated as of November 7, 2012 between Clean Energy Fuels Corp., a Delaware corporation (the "*Issuer*") and GE Energy Financial Services, Inc., a Delaware corporation (together with its successors and permitted assigns "*GE*").

## **RECITALS**

WHEREAS, Issuer indirectly owns all of the issued and outstanding equity interests in Clean Energy Tranche A LNG Plant, LLC and Clean Energy Tranche B LNG Plant, LLC (the "Borrowers");

WHEREAS, in order to induce GE to provide the credit facilities to the Borrowers pursuant to the Credit Agreement dated as of November 7, 2012 by and among the Borrowers, Clean Energy LNG Holdings, LLC ("*Holdings*"), GE and the other parties thereto (the "*Credit Agreement*"), Issuer has agreed to issue the Warrants (as defined herein) and provide certain other rights to GE as set forth herein; and

WHEREAS, this Agreement is being executed and delivered in connection with the closing under the Credit Agreement;

NOW, THEREFORE, in consideration of the foregoing recitals, and the mutual agreements among the parties, and of other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, and intending to be legally bound hereby, the parties hereto hereby agree as follows:

## 1. **DEFINITIONS**.

- 1.1 <u>Certain Defined Terms.</u> As used in this Agreement, the following terms have the respective meanings set forth below.
- "Additional Shares" means all Shares issued by the Issuer after the Closing Date, other than Permitted Shares.
- "Affiliate" means, in relation to any Person, any other Person that directly or indirectly Controls, is Controlled by or is under common Control with such Person. For purposes of this definition, "Control" means the ability to direct or cause the direction of the management or policies of another Person, whether by means of the ownership of stock or other securities with the right to vote, by contract or otherwise.
  - "Board" means the Board of Directors of the Issuer.
  - "Borrower" has the meaning set forth in the recitals of this Agreement.
- "*Business Day*" means any day that is not a Saturday, Sunday or other day when commercial banks are required or permitted by law to be closed in New York, New York.
  - "Closing" has the meaning set forth in Section 2.2.

1

- "Closing Date" means the date of the Closing.
- "Common Stock" means the common stock of the Issuer, par value \$0.0001 per share, and any other class of securities into which such securities may hereafter be reclassified or changed into.
  - "Convertible Securities" means any security convertible or exchangeable into Shares.
- "Current Market Price" means, in respect of any Shares on any date herein specified, the average of the daily market prices for the ten (10) consecutive Business Days before such date. The daily market price for each such Business Day shall be (i) the last sale price on such date on the Trading Exchange on which the Shares are then listed or admitted to trading, (ii) if no sale takes place on such day on any Trading Exchange, the average of the last reported closing bid and asked prices on such day as officially quoted on any Trading Exchange or (iii) if the Shares are not then listed or admitted to trading on any Trading Exchange, the average of the last reported closing bid and asked prices on such day in the over-the-counter market, as furnished by the National Association of Securities Dealers Automatic Quotation System or the National Quotation Bureau, Inc.
  - "Exchange Act" means the U.S. Securities Exchange Act of 1934, as amended.
  - "GE" has the meaning set forth in the introductory paragraph of this Agreement.
- "Holder" means any Person in whose name any Warrant is registered on the books of the Issuer maintained for such purpose and its successors and assigns.
  - "Holdings" has the meaning set forth in the introductory paragraph of this Agreement.
  - "Issuer" has the meaning set forth in the introductory paragraph of this Agreement.
- "Majority Holders" means the holders of Warrants exercisable for a majority of the aggregate number of Shares then purchasable upon exercise of all Warrants, whether or not then exercisable.
- "*Material Adverse Effect*" means a material adverse effect on or material adverse developments with respect to the business operations, properties, assets, condition (financial or otherwise) or prospects of the Issuer.

- "NASD" means the National Association of Securities Dealers, Inc., or any successor corporation thereto.
- "OFAC" means the Office of Foreign Assets Control of the U.S. Treasury Department.
- "Option" means rights, options or warrants to subscribe for, purchase or otherwise acquire Shares, Convertible Securities or other equity interests in the Issuer.
- "Permitted Shares" means (i) Warrant Shares and (ii) Shares issued or issuable on conversion or exercise of Convertible Securities or Options outstanding on the date hereof.

2

- "*Person*" means any individual, corporation, general partnership, limited partnership, limited liability partnership, joint venture, association, joint-stock company, trust, limited liability company, unincorporated organization or government or any agency or political subdivision thereof.
  - "Registration Rights Agreement" means the registration rights agreement in the form attached hereto as Exhibit B.
  - "SEC" means the U.S. Securities and Exchange Commission.
- "SEC Documents" means all reports, schedules, forms, statements and other documents required to be filed by the Issuer with the SEC pursuant to the reporting requirements of the Securities Act or the Exchange Act and all exhibits included therein and financial statements, notes and schedules thereto and documents incorporated by reference therein.
  - "Securities Act" means the U.S. Securities Act of 1933, as amended.
  - "Shares" means shares of Common Stock of the Issuer.
- "Subsidiary" means any corporation, association, trust, limited liability company, partnership, joint venture or other business association or entity (i) at least 50% of the outstanding voting securities of which are at the time owned or controlled, directly or indirectly, by the Issuer; or (ii) with respect to which the Issuer possesses, directly or indirectly, the power to direct or cause the direction of the affairs or management of such Person.
- "*Trading Market*" means the following markets or exchanges on which the Common Stock is listed or quoted for trading on the date in question: the American Stock Exchange, the Nasdaq Capital Market, the Nasdaq Global Market, the Nasdaq Global Select Market or the New York Stock Exchange.
- "*Transfer*" means any disposition of any Warrant or Warrant Shares or of any interest in either thereof, which would constitute a sale thereof within the meaning of the Securities Act.
- "Warrant Documents" means this Agreement, the Warrants, the Registration Rights Agreement and all other documents and instruments delivered hereunder or thereunder.
  - "Warrant Shares" means the Shares issuable or issued under the Warrants upon the exercise thereof.
- "*Warrants*" means the warrant in the form attached hereto as <u>Exhibit A</u>, exercisable for an aggregate of up to 5,000,000 Shares, and all warrants issued upon permitted Transfer, division or combination of, or in substitution for, any thereof; <u>provided</u> that all Warrants shall at all times be identical as to terms and conditions, except as to the number of Shares for which they may be exercised.
  - 1.2 Rules of Construction.
- (a) Unless the context requires otherwise, any reference in this Agreement to any Warrant Document means such Warrant Document, respectively, and all schedules, exhibits and attachments thereto, as amended, supplemented or otherwise modified and in effect from time to time. Unless otherwise stated, any reference in this Agreement to any Person shall include its permitted successors and assigns.

3

- (b) Defined terms in this Agreement shall include in the singular number the plural and in the plural number the singular.
- (c) The words "hereof", "herein" and "hereunder" and words of similar import when used in this Agreement shall, unless otherwise expressly specified, refer to this Agreement as a whole and not to any particular provision of this Agreement and all references to Sections, subsections, Exhibits and Schedules shall be references to Sections, subsections, Exhibits and Schedules of this Agreement unless otherwise expressly specified.
- (d) Unless otherwise expressly specified, any agreement, contract, or document defined or referred to herein means such agreement, contract or document in the form (including all amendments relating thereto) delivered to GE on the Closing Date as the same may thereafter be amended, restated, supplemented, or otherwise modified from time to time in accordance with the terms of this Agreement.
  - (e) The words "include," "includes" and "including" are not limiting and shall mean "including, without limitation".

## 2. WARRANTS; CLOSING; DELIVERIES.

2.1 <u>Warrants</u>. The Issuer shall issue a Warrant to GE upon execution and delivery of the Credit Agreement. As of the Closing, GE shall have been granted a Warrant to purchase 5,000,000 Shares (subject to adjustment as provided therein) for \$0.01 per Share in the form attached hereto as <u>Exhibit A</u>. Such Warrant shall be validly issued and fully paid on the Closing Date.

- 2.2 <u>Closing</u>. The issuance of the Warrant contemplated hereby shall be held or take place concurrently with and at the same location as the closing of the Credit Agreement. Such closing is hereinafter referred to as the "*Closing*."
  - 2.3 <u>Deliveries at Closing</u>. Subject to the terms of this Agreement, at the Closing:
  - (a) The Issuer shall duly execute and deliver to GE a Warrant in the form attached hereto as Exhibit A; and
  - (b) The Issuer and GE shall duly execute and deliver counterparts to the Registration Rights Agreement in the form attached hereto as <u>Exhibit</u>

## 3. REPRESENTATIONS AND WARRANTIES OF THE ISSUER.

<u>B</u>.

In order to induce GE to enter into this Agreement, the Issuer hereby represents and warrants to GE that:

3.1 <u>Organization and Qualification</u>. The Issuer is an entity duly incorporated and validly existing in good standing under the laws of the jurisdiction in which it is incorporated and

4

has the requisite power and authority to own its properties and to carry on its business as now being conducted. The Issuer is duly qualified as a foreign corporation to do business and is in good standing in every jurisdiction in which its ownership of property or the nature of the business conducted by it makes such qualification necessary, except to the extent that the failure to be so qualified or be in good standing would not reasonably be expected to have a Material Adverse Effect.

- 3.2 <u>Authorization; Enforcement; Validity.</u> The Issuer has the requisite corporate power and authority to enter into and perform its obligations under the Warrant Documents and to issue the Warrant Shares in accordance with the terms of the Warrant Documents. The execution and delivery of the Warrant Documents by the Issuer and the consummation by the Issuer of the transactions contemplated by the Warrant Documents have been duly authorized by all required corporate action, and no further filing, consent or authorization is required by the Issuer, the Board or the Issuer's stockholders. The Warrant Documents have been duly executed and delivered by the Issuer, and constitute the legal, valid and binding obligations of the Issuer, enforceable against the Issuer in accordance with their respective terms, except as such enforceability may be limited by general principles of equity or applicable bankruptcy, insolvency, reorganization, moratorium, liquidation or similar laws relating to, or affecting generally, the enforcement of applicable creditors' rights and remedies.
- 3.3 <u>Issuance of Securities; No Registration Required</u>. The Warrant Shares have been duly reserved for issuance, are duly authorized and, upon issuance in accordance with the terms hereof, will be validly issued and free from all preemptive or similar rights, taxes, liens and charges with respect to the issue thereof and will be fully paid and nonassessable with the holders being entitled to all rights accorded to a holder of Common Stock. Assuming the accuracy of each of the representations and warranties of GE set forth in <u>Section 4</u>, the offer and issuance by the Issuer of the Warrants and the Warrant Shares are exempt from registration under the Securities Act and will be issued in compliance with all applicable federal and state securities laws subject to any filings which may be required to be made by the Issuer with the SEC, state securities administrators or NASDAQ, subsequent to Closing.
- 3.4 No Conflicts. The execution, delivery and performance of the Warrant Documents by the Issuer and the consummation by the Issuer of the transactions contemplated hereby and thereby will not (a) result in a violation of any certificate of incorporation, certificate of formation, certificate of designations or other constituent documents of the Issuer, any capital stock of the Issuer or bylaws of the Issuer; (b) conflict with, or constitute a default (or an event which with notice or lapse of time or both would become a default) under, or give to others any rights of termination, amendment, acceleration or cancellation of, any agreement, indenture or instrument to which the Issuer is a party; or (c) result in a violation of any applicable law, rule, regulation, order, judgment or decree, including without limitation, federal and state securities laws and regulations and the rules and regulations of NASDAQ or any other national securities exchange applicable to the Issuer or by which any property or asset of the Issuer is bound or affected, except in the cases of clauses (b) and (c) such as would not reasonably be expected to have a Material Adverse Effect.
- 3.5 <u>Consents.</u> Assuming the accuracy of the representations and warranties of GE set forth in <u>Section 4</u>, the Issuer is not required to obtain any consent, authorization or order of, or make any filing or registration with, any court, governmental agency or any regulatory or self-

5

regulatory agency or any other Person in order for it to execute, deliver or perform any of its obligations under or contemplated by the Warrant Documents, in each case in accordance with the terms thereof, other than (a) any consent, authorization or order that has been obtained as of the date hereof; (b) any filing or registration that has been made as of the date hereof; or (c) any filings which may be required to be made by the Issuer with the SEC, state securities administrators or NASDAQ, subsequent to the Closing. The Issuer is not in violation of the listing requirements of NASDAQ.

- 3.6 No General Solicitation. Neither the Issuer nor, to the Issuer's knowledge, any Person acting on its behalf, has engaged in any form of general solicitation or general advertising (within the meaning of Regulation D) in connection with the offer or sale of the Warrants and the Warrant Shares. The Issuer will be responsible for the payment of any placement agent's fees, financial advisory fees, or brokers' commissions (other than for Persons engaged by GE) relating to or arising out of the transactions contemplated hereby.
- 3.7 No Integrated Offering. None of the Issuer, or any Person acting on its behalf has, directly or indirectly, made any offers or sales of any security or solicited any offers to buy any security, under circumstances that would require registration of the issuance of the Warrants and the Warrant Shares under the Securities Act, whether through integration with prior offerings or otherwise, or cause this offering of the Warrants and the Warrant Shares to require approval of stockholders of the Issuer for purposes of any applicable stockholder approval provisions, including, without limitation, under the rules and regulations of NASDAQ.

SEC Documents; Financial Statements. During the two (2) years prior to the date hereof, the Issuer has timely filed all SEC Documents. As of their respective filing dates, the SEC Documents complied in all material respects with the requirements of the Securities Act or the Exchange Act, as applicable, and the rules and regulations of the SEC promulgated thereunder applicable to the SEC Documents, and none of the SEC Documents, at the time they were filed with the SEC (solely to the extent any information contained in such SEC Documents has not been amended, modified, supplemented, corrected, rescinded or otherwise withdrawn in subsequent material filed by the Issuer with the SEC), contained any untrue statement of a material fact or omitted to state a material fact required to be stated therein or necessary in order to make the statements therein, in the light of the circumstances under which they were made, not misleading. As of their respective filing dates, the financial statements of the Issuer included in the SEC Documents complied as to form in all material respects with applicable accounting requirements and the published rules and regulations of the SEC with respect thereto. Such financial statements have been prepared in accordance with generally accepted accounting principles, consistently applied, during the periods involved (except (a) as may be otherwise indicated in such financial statements or the notes thereto; or (b) in the case of unaudited interim statements, to the extent they may exclude footnotes or may be condensed or summary statements) and fairly present in all material respects the financial position of the Issuer as of the dates thereof and the results of its operations and cash flows for the periods then ended (subject, in the case of unaudited statements, to normal year-end audit adjustments and to the extent that such unaudited statements exclude footnotes). To the Issuer's knowledge, no other information provided by the Issuer to GE in connection with the transactions contemplated by the Warrant Documents which is not included in the SEC Documents contains any untrue statement of a material fact or omits to state any material fact necessary in order to make the statements therein, in the light of the circumstances under which they are or were made not misleading; provided,

6

however, the Issuer makes no representations or warranties regarding any projections, forecasts or other forward looking statements. KPMG LLP, which has reported on the consolidated financial statements and schedules contained in the SEC Documents, are registered independent public accountants as required by the Securities Act and the rules and regulations promulgated thereunder and by the rules of the Public Accounting Oversight Board.

- 3.9 <u>Absence of Certain Changes</u>. Since December 31, 2011, other than as disclosed in the SEC Documents, (a) there has been no Material Adverse Effect; (b) the Issuer has not paid or declared any dividends or other distributions with respect to its capital stock; and (c) there has not been any change in the capital stock of the Issuer other than grants, exercises, terminations, or forfeitures of securities under the Issuer's equity or compensation plans.
- 3.10 <u>Foreign Corrupt Practices</u>. Neither the Issuer nor, to the Issuer's knowledge, any director, officer, agent, employee or other Person acting on behalf of the Issuer has, in the course of its actions for, or on behalf of, the Issuer: (a) used any corporate funds for any unlawful contribution, gift, entertainment or other unlawful expenses relating to political activity; (b) made any direct or indirect unlawful payment to any foreign or domestic government official or employee from corporate funds; (c) violated or is in violation of any provision of the U.S. Foreign Corrupt Practices Act of 1977, as amended; or (d) made any unlawful bribe, rebate, payoff, influence payment, kickback or other unlawful payment to any foreign or domestic government official or employee.
- 3.11 OFAC. Neither the Issuer nor, to the Issuer's knowledge, any director, officer, agent, employee, Affiliate or Person acting on behalf of the Issuer is currently subject to any U.S. sanctions administered by OFAC; and the Issuer will not directly or indirectly use the proceeds of this offering, or lend, contribute or otherwise make available such proceeds to any Subsidiary, joint venture partner or other Person or entity, for the purpose of financing the activities of any Person currently subject to any U.S. sanctions administered by OFAC.
- 3.12 <u>Equity Capitalization</u>. As of September 30, 2012, the authorized capital stock of the Issuer consisted of 149,000,000 shares of Common Stock and 1,000,000 shares of preferred stock. As of September 30, 2012, (a) 87,232,094 shares of Common Stock were issued and outstanding; and (b) 14,144,330 shares of Common Stock were reserved for issuance pursuant to outstanding Options or Convertible Securities (including 12,013,648 shares of Common Stock reserved for future grant under equity incentive plans). As of September 30, 2012, zero shares of preferred stock were issued and outstanding. All of such outstanding shares have been, or upon issuance will be, validly issued and are fully paid and nonassessable. Except as set forth above in this <u>Section 3.12</u>, or as disclosed in the SEC Documents:
- (a) none of the Issuer's capital stock is subject to preemptive rights or any other similar rights or any liens or encumbrances suffered or permitted by the Issuer;
- (b) there are no outstanding Options, Convertible Securities, rights to subscribe to, calls or commitments of any character whatsoever relating to any capital stock of the Issuer, or contracts, commitments, understandings or arrangements by which the Issuer is or may become bound to issue additional capital stock of the Issuer;

7

- (c) there are no agreements or arrangements under which the Issuer is obligated to register the sale of any of their securities under the Securities Act;
- (d) there are no outstanding securities or instruments of the Issuer which contain any redemption or similar provisions, and there are no contracts, commitments, understandings or arrangements by which the Issuer is or may become bound to redeem a security of the Issuer;
- (e) there are no securities or instruments containing anti-dilution or similar provisions that will be triggered by the issuance of the Warrants; and
- (f) the Issuer does not have any stock appreciation rights or "phantom stock" plans or agreements or any similar plan or agreement. The Issuer has furnished or made available to GE upon GE's request, true, correct and complete copies of the certificate of incorporation and bylaws of the Issuer and the terms of all Options and Convertible Securities and the material rights of the holders thereof in respect thereto.
- 3.13 <u>Absence of Litigation</u>. Except as disclosed in the SEC Documents, there is no action, suit, proceeding, inquiry or investigation before or by NASDAQ, any court, public board, government agency, self-regulatory organization or body pending or, to the knowledge of the Issuer, threatened against or affecting the Issuer that would reasonably be expected to have a Material Adverse Effect.

- Internal Accounting and Disclosure Controls. The Issuer maintains a system of internal accounting controls sufficient to provide reasonable assurance that (a) transactions are executed in accordance with management's general or specific authorizations; (b) transactions are recorded as necessary to permit preparation of financial statements in conformity with generally accepted accounting principles and to maintain asset and liability accountability; (c) access to assets or incurrence of liabilities is permitted only in accordance with management's general or specific authorization; and (d) the recorded accountability for assets and liabilities is compared with the existing assets and liabilities at reasonable intervals and appropriate action is taken with respect to any difference. The Issuer maintains disclosure controls and procedures (as such term is defined in Rule 13a-15 under the Exchange Act) that are effective at the reasonable assurance level to ensure that information required to be disclosed by the Issuer in the reports that it files or submits under the Exchange Act is recorded, processed, summarized and reported, within the time periods specified in the rules and forms of the SEC, including, without limitation, controls and procedures designed to ensure that information required to be disclosed by the Issuer in the reports that it files or submits under the Exchange Act is accumulated and communicated to the Issuer's management, including its principal executive officer or officers and its principal financial officer or officers, as appropriate, to allow timely decisions regarding required disclosure.
- 3.15 Off Balance Sheet Arrangements. There is no transaction, arrangement, or other relationship between the Issuer and an unconsolidated or other off balance sheet entity that is required to be disclosed by the Issuer in the SEC Documents or its financial statements and related footnotes according to GAAP and is not so disclosed.
- 3.16 <u>Investment Company Status</u>. The Issuer is not, and upon issuance of the Warrants will not be, an "investment company," a company controlled by an "investment company" or an "affiliated person" of, or "promoter" or "principal underwriter" for, an "investment company" as such terms are defined in the Investment Company Act of 1940, as amended.

8

3.17 <u>Form S-3 Eligibility</u>. The Issuer is eligible to register the Warrant Shares for resale by the Holders using Form S-3 promulgated under the Securities Act.

## 4. REPRESENTATIONS AND WARRANTIES OF GE.

In order to induce the Issuer to issue the Warrants to GE, GE hereby represents and warrants to the Issuer:

- 4.1 <u>Organization and Standing</u>. GE is a corporation duly incorporated or formed, validly existing and in good standing under the laws of the jurisdiction of its incorporation or formation.
- 4.2 <u>Authorization; Power.</u> GE has all requisite legal power to execute, deliver and perform this Agreement. GE has taken all necessary action for the authorization, execution, delivery and performance of this Agreement and the consummation of the transactions contemplated thereby. This Agreement constitutes a legal, valid and binding obligation of GE, which is enforceable against GE in accordance with its terms, except as such enforceability may be limited by general principles of equity or applicable bankruptcy, insolvency, reorganization, moratorium, liquidation or similar laws relating to, or affecting generally, the enforcement of applicable creditors' rights and remedies.
- 4.3 <u>Investment Representations</u>. GE understands that the Warrants and the Warrant Shares to be issued upon an exercise of each Warrant are restricted securities which are not registered under the Securities Act. GE also understands that the Warrants are being offered and sold pursuant to an exemption from registration contained in the Securities Act based in part upon GE's representations contained in this Agreement. Without limiting the force and effect of the representations and warranties of the Issuer, GE hereby represents and warrants as follows:
- (a) GE is acquiring the Warrants for its own account for investment purposes only, and does not intend to distribute the Warrants or Warrant Shares. GE has not solicited offers for, or offered or sold, and will not solicit offers for, or offer or sell, the Warrants or the Warrant Shares by means of any form of general solicitation or general advertising within the meaning of Rule 502(a) of Regulation D or in any manner involving a public offering within the meaning of Section 4(2) of the Securities Act (other than pursuant to the Registration Rights Agreement).
- (b) GE represents that it: (i) is an accredited investor within the meaning of Regulation D promulgated under the Securities Act; and (ii) acknowledges and agrees that, to the extent it desires to resell the Warrants or the Warrant Shares, it will solicit offers for the Warrants or the Warrant Shares from, and will offer the Warrants or the Warrant Shares only (A) pursuant to an exemption from registration under the Securities Act, or (B) pursuant to an effective registration statement under the Securities Act.
- (c) GE has: (i) received and read all information that GE has requested regarding the Issuer's business, management and financial affairs including the SEC Documents; (ii) had the opportunity to discuss such matters with directors, officers and management of the Issuer; (iii) had the opportunity to review the Issuer's operations and facilities; and (iv) had the opportunity to ask questions of and receive answers from the Issuer and its management regarding the terms and conditions of this investment.

q

## 5. COVENANTS.

- 5.1 <u>Listing</u>. The Issuer shall promptly secure the listing of all of the Warrant Shares upon each national securities exchange and automated quotation system, if any, upon which shares of Common Stock are then listed (subject to official notice of issuance) and shall maintain, so long as any other shares of Common Stock shall be so listed, such listing of all Warrant Shares from time to time issuable under the terms of the Warrant Documents. The Issuer shall pay all fees and expenses in connection with satisfying its obligations under this <u>Section 5.1</u>.
- 5.2 <u>Certain Fees and Expenses</u>. The Issuer shall pay all transfer agent fees, stamp or transfer taxes and other taxes (not including income or similar taxes) and duties levied in connection with the sale and issuance of the Warrant Shares.
- 5.3 <u>Public Statements</u>. The Issuer and GE agree not to issue or cause the publication of any press release or other announcement with respect to the transactions contemplated by this Agreement without the prior consent of the other party, which consent shall not be unreasonably withheld or delayed,

unless such party determines, after consultation with counsel, that it is required by an applicable statute, ordinance, rule or regulation or by any listing agreement with or the listing rules of a national securities exchange or trading market to issue or cause the publication of any press release or other announcement with respect to the transactions contemplated by this Agreement, in which event such party shall endeavor, on a basis reasonable under the circumstances, to provide a meaningful opportunity to the other parties to review and comment upon such press release or other announcement and shall give due consideration to all reasonable additions, deletions or changes suggested thereto.

- 5.4 <u>Restrictions on Transfer of Warrant Shares</u>. Warrant Shares issued upon exercise of a Warrant shall not be Transferred to any Person (other than an Affiliate of the Transferor) for 32 days following such issuance.
  - 5.5 Right of First Refusal on Sale of Borrower.
- (a) If the Issuer, Holdings or any Borrower receives a bona fide offer from a third party (a "Third Party Offer") to purchase all or part of Holdings or the Borrowers, in one or more transactions or a series of transactions, whether by merger, consolidation, combination, amalgamation, transfer, spinoff, sale or other direct or indirect disposition of all or any part of the equity in Holdings or the Borrowers or all or substantially all of the Borrowers' assets (the "Borrower Sale"), and the Issuer, Holdings or such Borrower, as applicable (the "ROFR Seller"), desires, and is permitted, to effect such proposed sale, the Issuer shall deliver or cause to be delivered to GE written notice of such Third Party Offer (the "ROFR Notice" and the date it is delivered, the "ROFR Notice Date") to GE no less than 45 days prior to the date of the proposed Borrower Sale. The ROFR Notice shall set forth the name of the Third Party (including, if such information is not publicly available, information about the identity of the Third Party), the assets or equity to be to be sold and the price offered therefor (the "ROFR Offer Price"), all details of the payment terms and all other terms and conditions of the proposed Borrower Sale.

10

- (b) GE shall have the right to purchase all, but not less than all, of the equity or assets proposed to be sold in the Borrower Sale. Within 30 days after the ROFR Notice Date (such 30th day, the "ROFR Expiration Date"), GE may deliver a written notice to the ROFR Seller and the Issuer of GE's election to purchase such equity or assets, which notice of election shall constitute an irrevocable commitment to consummate such Borrower Sale. The Issuer shall thereafter set a reasonable place and time for the closing of the Borrower Sale, which shall be not less than 60 days nor more than 180 days after the ROFR Notice Date (subject to extension to the extent necessary to pursue any required regulatory or shareholder approvals, including to allow for the expiration or termination of all waiting periods under the HSR Act) unless otherwise agreed by all of the parties to such transaction (such period, the "ROFR Closing Period").
- (c) The purchase price and terms and conditions for the Borrower Sale shall be the ROFR Offer Price and the terms and conditions set forth in the applicable Third Party Offer; provided that the Issuer and the ROFR Seller shall make, at a minimum, customary representations and warranties to GE concerning (i) such ROFR Seller's valid title to and ownership of equity or assets subject to the Borrower Sale, free and clear of all liens, claims and encumbrances (excluding those arising under applicable securities laws), (ii) such ROFR Seller's authority, power and right to enter into and consummate the Borrower Sale, (iii) the absence of any violation, default or acceleration of any agreement to which such ROFR Seller is subject or by which its assets are bound as a result of the agreement to consummate or the consummation the Borrower Sale, and (iv) the absence of, or compliance with, any governmental or third party consents, approvals, filings or notifications required to be obtained or made by Issuer or such ROFR Seller in connection with Borrower Sale. The ROFR Seller also agrees to execute and deliver such instruments and documents and take such actions, including obtaining all applicable approvals and consents and making all applicable notifications and filings, as GE may reasonably request in order to effectively implement the Borrower Sale hereunder.
- (d) Notwithstanding the foregoing, if GE (i) shall not have elected to consummate the Borrower Sale on or prior to the ROFR Expiration Date or (ii) fails to consummate the closing of the Borrower Sale within the ROFR Closing Period and the ROFR Seller has fully complied with the provisions of this Section 5.5, then GE shall not have the right to participate in the Borrower Sale and the ROFR Seller may consummate such Borrower Sale within 120 days after the expiration of the ROFR Closing Period, subject to the other applicable provisions of this Agreement. Any such Borrower Sale shall not be at less than the price or upon terms and conditions materially more favorable, individually or in the aggregate, to the third party purchaser than those specified in the Third Party Offer. If the Borrower Sale is not so consummated within such 120-day period, the ROFR Seller may not sell any of the equity or assets subject to such Borrower Sale without again complying in full with the provisions of this Section 5.5.
- (e) Notwithstanding the foregoing provisions of this <u>Section 5.5</u>, the provisions of this <u>Section 5.5</u> shall not apply to any Borrower Sale to which the provisions of <u>Section 5.6</u> below apply.

11

- (f) This <u>Section 5.5</u> shall expire upon the first to occur of (i) the tenth (10<sup>th</sup>) anniversary of the first Initial Funding Date under the Credit Agreement or (ii) full repayment of all outstanding obligations of the Borrowers under the Credit Agreement.
- Agreement, if the Issuer proposes to Transfer or contribute or cause to be Transferred or contributed, directly or indirectly, the Issuer's equity interests in Holdings or the Borrowers or the Borrowers' assets to, or otherwise restructure or reorganize Holdings or the Borrowers as, an entity with a view towards creating a master limited partnership or similar publicly traded entity (an "MLP"), GE shall have the right to invest in (i) the general partner or other sponsor entity of the MLP or (ii) in the securities in the MLP received by such general partner or sponsor entity, in each case, on the same terms as the Issuer (with (i) and (ii) collectively the "MLP Securities") up to a maximum of the greater of (a) 35% of such MLP Securities, or (b) the amount of MLP Securities offered to any one third-party other than the Issuer or its Affiliates. In connection with any co-investment pursuant to this Section 5.6, GE agrees to execute and deliver to the Issuer any documents reasonably requested by the Issuer to be executed by GE, including, without limitation, a membership interest purchase agreement and a limited liability company agreement (or similar documents, as applicable). Notwithstanding the foregoing, if GE (i) shall not have elected to purchase MLP Securities within 30 days of its receipt of notice (the "MLP Notice") that the Issuer proposes to Transfer or contribute or cause to be Transferred or contributed, directly or indirectly, the Issuer's equity interests in Holdings or the Borrowers' assets to, or otherwise restructure or reorganize Holdings or the Borrowers as, an MLP or (ii) fails to consummate the closing of its purchase of MLP Securities by the later of (a) 60 days from the date on which GE receives the MLP Notice or (b) the date on which the Issuer completes the Transfer or contribution of the Issuer's equity interests in Holdings or the Borrowers or the Borrowers' assets to an MLP, then GE shall not have the right to purchase MLP Securities.

## 6. MISCELLANEOUS.

(a) Notices. Any notices, consents, waivers or other communications required or permitted to be given under the terms of this Agreement must be in writing and shall be deemed to have been delivered: (i) upon receipt, when delivered personally; (ii) upon receipt, when sent by facsimile (provided confirmation of transmission is mechanically or electronically generated and kept on file by the sending party); or (iii) one Business Day after deposit with an overnight courier service, in each case properly addressed to the party to receive the same. The addresses and facsimile numbers for such communications shall be:

if to the Company:

Clean Energy Fuels Corp. 3020 Old Ranch Parkway

Suite 400

Seal Beach, California 90740
Telephone: (562) 493-7239
Facsimile: (562) 493-4956
Attention: J. Nathan Jensen

12

with a copy (for informational purposes only) to:

Morrision & Foerster LLP 12531 High Bluff Drive

Suite 100

San Diego, California 92130
Telephone: (858) 720-5198
Facsimile: (858) 523-2810
Attention: Steven G. Rowles

if to GE:

GE Energy Financial Services, Inc. 333 Clay Street, Suite 4550 Houston, Texas 77002

Telephone: (713) 951-2339 Facsimile: (713) 951-2319 Attention: Brandy Copley

with a copy (for informational purposes only) to:

Vinson & Elkins LLP
666 Fifth Avenue, 26<sup>th</sup> Floor
New York, New York 10103
Telephone: (212) 237-0132
Facsimile: (917) 849-5350
Attention: Robert Seber

- 6.1 <u>Severability</u>. If any one or more of the provisions contained in any of the Warrant Documents is determined to be invalid, illegal or unenforceable in any respect in any jurisdiction, the validity, legality and enforceability of such provision or provisions shall not in any way be affected or impaired thereby in any other jurisdiction, nor shall the validity, legality and enforceability of the remaining provisions contained in the Warrant Documents in any way be affected or impaired thereby.
- 6.2 <u>Construction and Venue</u>. All questions concerning the construction, validity, enforcement and interpretation of this Agreement shall be governed by the internal laws of the State of Delaware, without giving effect to any choice of law or conflict of law provision or rule (whether of the State of Delaware or any other jurisdictions) that would cause the application of the laws of any jurisdictions other than the State of Delaware. Each party hereby irrevocably submits to the exclusive jurisdiction of the Delaware Court of Chancery or, if no such state court has proper jurisdiction, the United States District Court for the District of Delaware, for the adjudication of any dispute hereunder or in connection herewith or with any transaction contemplated hereby or discussed herein, and hereby irrevocably waives, and agrees not to assert in any suit, action or proceeding, any claim that it is not personally subject to the jurisdiction of any such court, that such suit, action or proceeding is brought in an inconvenient forum or that the venue of such suit, action or proceeding is improper. Nothing contained herein shall be deemed to limit in any way any right to serve process in any manner permitted by law. EACH

13

PARTY HEREBY IRREVOCABLY WAIVES ANY RIGHT IT MAY HAVE, AND AGREES NOT TO REQUEST, A JURY TRIAL FOR THE ADJUDICATION OF ANY DISPUTE HEREUNDER OR IN CONNECTION WITH OR ARISING OUT OF THIS AGREEMENT OR ANY TRANSACTION CONTEMPLATED HEREBY.

- 6.3 <u>No Waiver</u>. No payment of proceeds under the Warrant Documents shall constitute a waiver of any of the Issuer's representations, warranties, conditions or covenants under the Warrant Documents.
- 6.4 <u>Entire Agreement</u>. This Agreement and the other Warrant Documents supersede all other prior oral or written agreements between GE, the Issuer, their Affiliates and Persons acting on their behalf with respect to the matters discussed herein and therein and constitute the entire understanding of the

parties with respect to the matters covered herein or therein, and except as specifically set forth herein or therein, neither the Issuer nor GE makes any representation, warranty, covenant or undertaking with respect to such matters.

- 6.5 <u>Headings</u>. The headings of this Agreement are for convenience, are not part of and shall not affect the interpretation of this Agreement.
- 6.6 <u>Counterparts</u>. This Agreement may be executed in two or more counterparts, and it shall not be necessary that the signatures of all parties hereto be contained on any one counterpart hereof. The counterpart shall be deemed an original, but all counterparts together shall constitute one and the same instrument. The parties agree that a telecopy, facsimile or other electronic transmission of this Agreement signed by the parties shall constitute an agreement in accordance with the terms hereof as if all of the parties had executed an original of this Agreement.
- 6.7 <u>Successors and Assigns.</u> This Agreement shall be binding upon and inure to the benefit of the parties and their successors and permitted assigns.
- 6.8 <u>No Third Party Beneficiaries</u> This Agreement is intended for the benefit of the parties hereto and their respective permitted successors and assigns, and is not for the benefit of, nor may any provision hereof be enforced by, any other Person.
- 6.9 <u>Further Assurances</u>. Each party shall do and perform, or cause to be done and performed, all such further acts and things, and shall execute and deliver all such other agreements, certificates, instruments and documents, as any other party may reasonably request in order to carry out the intent and accomplish the purposes of this Agreement and the consummation of the transactions contemplated hereby.
- 6.10 <u>Amendments and Waivers</u>. Except as otherwise set forth herein, any term of this Agreement may be amended, terminated or waived only with the written consent of the Issuer and the Majority Holders.

[Signature Pages Follow]

14

IN WITNESS WHEREOF, each party hereto has caused this Agreement to be duly executed by its authorized officer and dated effective as of the Closing Date.

#### CLEAN ENERGY FUELS CORP.

By: /s/ Andrew J. Littlefair

Name: Andrew J. Littlefair

Title: President & Chief Executive Officer

## GE ENERGY FINANCIAL SERVICES, INC.

By: /s/ Tyson Yates
Name: Tyson Yates
Tyson Yates

Title: Managing Director

SIGNATURE PAGE TO WARRANT AGREEMENT

# Exhibit A Form of Warrant

[see attached]

WARRANT AGREEMENT

## Exhibit B Form of Registration Rights Agreement

[see attached]

WARRANT AGREEMENT

Exhibit A to Warrant Agreement THIS WARRANT AND THE SECURITIES ISSUABLE UPON THE EXERCISE HEREOF HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933, AS AMENDED, OR REGISTERED OR QUALIFIED UNDER THE SECURITIES LAWS OF ANY STATE OR OTHER JURISDICTION AND MAY NOT BE SOLD, PLEDGED, TRANSFERRED OR OTHERWISE ASSIGNED IN VIOLATION OF SUCH ACT AND LAWS OR THE PROVISIONS OF THIS WARRANT.

## FORM OF WARRANT

## TO PURCHASE SHARES OF CLEAN ENERGY FUELS CORP.

THIS CERTIFIES THAT, pursuant to the Warrant Agreement dated November [ ], 2012 (the "Warrant Agreement") by and between GE Energy Financial Services, Inc. (the "Holder") and Clean Energy Fuels Corp. (the "Issuer"), the Holder, or its successor or permitted assign, is entitled, at any time during the Exercise Period (as defined herein), to purchase from the Issuer, an aggregate of up to 5,000,000 shares of Common Stock, par value \$0.0001 per share, of the Issuer (the "Shares") (subject to adjustment and limitation as provided herein), in whole or in part, at a purchase price of \$0.01 per Share (the "Exercise Price"), all on the terms and conditions and pursuant to the provisions hereinafter set forth. This Warrant is issued in consideration of the execution of the Credit Agreement.

**DEFINITIONS.** All capitalized terms defined in the Warrant Agreement and not otherwise defined herein shall have the meanings ascribed to such terms in the Warrant Agreement.

#### 2. EXERCISE OF WARRANT.

- 2.1 <u>General</u>. At any time and from time to time after the Closing Date and until 5:00 p.m., New York time, on the tenth (10<sup>th</sup>) anniversary of the Closing Date or such earlier date as provided in <u>Section 4.2</u> below (the "*Exercise Period*"), the Holder may exercise this Warrant, on any Business Day, for all or any part of such number of Shares, at the stated Exercise Price, equal to the sum of:
  - (a) 10% of the aggregate number of Shares purchasable hereunder as an upfront commitment fee, *plus* (b) or (c)

(b)

- (i) after the first Tranche A Loan (as defined in the Credit Agreement) is made, 25% of the aggregate number of Shares purchasable hereunder,
- (ii) after the first Tranche B Loan (as defined in the Credit Agreement) is made, 25% of the aggregate number of Shares purchasable hereunder.
- (iii) after Tranche A Loans in aggregate principal amount of at least \$15,000,000 have been made, 20% of the aggregate number of Shares purchasable hereunder, and

A-1

- (iv) after Tranche B Loans in aggregate principal amount of at least \$15,000,000 have been made, 20% of the aggregate number of Shares purchasable hereunder, *or* (in the alternative)
- (c) if no Borrowings (as defined in the Credit Agreement) are made prior to December 31, 2014, 10% of the aggregate number of Shares purchasable hereunder.

Upon the occurrence of any of the events described in the foregoing clauses (b) and (c), the Issuer shall deliver written notice thereof to Holder.

<u>Cash Exercise</u>. The Holder may exercise this Warrant, in whole or in part and at any time and from time to time during the Exercise Period, by delivering to the Issuer at the Issuer's principal offices at the address set forth in the Warrant Agreement: (i) a written notice of the Holder's election to exercise this Warrant specifying the number of Shares to be purchased, (ii) payment of the Warrant Price and (iii) this Warrant. "Warrant Price" shall mean an amount equal to (i) the number of Shares being purchased upon any exercise of the Warrant pursuant to this Section 2, multiplied by (ii) the Exercise Price as adjusted pursuant to the terms of the Warrant as of the date of such exercise. Such notice shall be substantially in the form of the subscription form appearing at the end of this Warrant as Annex A, duly executed by the Holder or its agent or attorney. Upon receipt thereof, the Issuer shall, as promptly as practicable, and in any event within five (5) Business Days thereafter, execute or cause to be executed and deliver or cause to be delivered to the Holder a certificate or certificates reflecting the Holder's ownership of the aggregate number of Shares issuable upon such exercise, together with cash in lieu of any fraction of a Share, as hereinafter provided in Section 2.5. The Share certificate or certificates so delivered shall be in such denomination or denominations as the Holder shall request in the notice and shall be registered in the name of the Holder or, subject to any restrictions on Transfer, such other name as shall be designated in the notice. This Warrant shall be deemed to have been exercised and such Warrant Shares shall be deemed to have been issued, and the Holder or any other Person so designated to be named therein shall be deemed to have become a holder of record of such Shares for all purposes, as of the date of delivery of the Warrant Shares by the Issuer. If this Warrant has been exercised in part, the Issuer shall, at the time of delivery of the certificate or certificates representing Warrant Shares, deliver to the Holder a new Warrant evidencing the rights of the Holder to purchase the unpurchased Shares called for by this Warrant, which new Warrant shall in all other respects be identical with this Warrant, or, at the request of the Holder, appropriate notation may be made on this Warrant and the same returned to the Holder. Payment of the Warrant Price and all taxes required to be paid by the Holder, if any, pursuant to Section 2.4, shall be paid by the Holder prior to delivery of the Warrant Shares by the Issuer and shall be made at the option of the Holder by certified or official bank check or by wire transfer of immediately available funds.

## 2.3 <u>Cashless Exercise</u>.

(a) In lieu of the payment of the Warrant Price, the Holder shall have the right (but not the obligation), to require the Issuer to convert this Warrant, in whole or in part, into Shares as provided for in this <u>Section 2.3</u> (the "*Conversion Right*"). Upon exercise of the Conversion Right, the Issuer shall deliver to the Holder (without payment by the Holder of any of the

Warrant Price) that number of Warrant Shares (the "Conversion Shares") equal to the quotient obtained by dividing (x) the value of this Warrant (or portion thereof as to which the Conversion Right is being exercised if the Conversion Right is being exercised in part) at the time the Conversion Right is exercised (determined by subtracting the aggregate Warrant Price of the Warrant Shares as to which the Conversion Right is being exercised in effect immediately prior to the exercise of the Conversion Right from the aggregate Current Market Price of the Warrant Shares as to which the Conversion Right is being exercised immediately prior to the exercise of the Conversion Right) by (y) the Current Market Price of one (1) Warrant Share immediately prior to the exercise of the Conversion Right.

- (b) The Conversion Rights provided under this Section 2.3 may be exercised in whole or in part and at any time and from time to time during the Exercise Period. In order to exercise the Conversion Right, the Holder shall surrender to the Issuer, at its offices, this Warrant, and the Notice of Conversion in the form attached hereto as Annex B duly executed. The presentation and surrender shall be deemed a waiver of the Holder's obligation to pay all or any portion of the aggregate purchase price payable for the Warrant Shares as to which such Conversion Right is being exercised. This Warrant (or so much thereof as shall have been surrendered for conversion) shall be deemed to have been converted immediately prior to the close of business on the day of surrender of such Warrant for conversion in accordance with the foregoing provisions. If this Warrant has been exercised in part, the Issuer shall, at the time of delivery of the acknowledgement reflecting the ownership of the Warrant Shares, deliver to the Holder a new Warrant evidencing the rights of the Holder to purchase the unpurchased Shares called for by this Warrant, which new Warrant shall in all other respects be identical with this Warrant, or, at the request of the Holder, appropriate notation may be made on this Warrant and the same returned to the Holder.
- 2.4 Payment of Taxes. When the Warrant Price is paid to the Issuer, all such Warrant Shares shall be validly issued, fully paid and nonassessable and without any preemptive rights. The Issuer shall pay all expenses in connection with, and all taxes and other governmental charges that may be imposed with respect to, the issue or delivery thereof, unless such tax or charge is imposed by law upon the Holder, in which case, the Holder shall pay such taxes or charges. The Issuer shall not be required to pay any tax or other charge imposed in connection with any Transfer involved in the issue or delivery of any certificate for Shares issuable upon exercise of this Warrant in any name other than that of the Holder, and in such case, the Issuer shall not be required to register such Shares in any name other than the Holder until such tax or other charge has been paid or it has been established to the reasonable satisfaction of the Issuer that no such tax or other charge is due.
- 2.5 <u>Fractional Shares</u>. The Issuer shall not be required to issue a fractional Share upon the exercise of this Warrant. As to any fraction of a Share which the Holder would otherwise be entitled to purchase upon such exercise, the Issuer shall pay a cash adjustment in respect of such final fraction (calculated on an aggregate basis for all Warrants exercised) in an amount equal to the same fraction of the Current Market Price per Share of a Warrant Share on the date of exercise.

A-3

Limitation on Conversion. Notwithstanding anything herein to the contrary, the Issuer shall not be obligated to affect any conversion of all or any portion of this Warrant, and the Holder shall not have the right to convert all or any portion of this Warrant, to the extent that, after giving effect to an attempted conversion, the Holder (together with its Affiliates and its permitted assignees and any other Person whose beneficial ownership of Shares would be aggregated with the Holder and its permitted assignees for purposes of Section 13(d) of the Exchange Act and the applicable rules and regulations of the Securities and Exchange Commission, including any "group" of which Holder and its permitted assignees is a member) would have acquired a number of Shares as a result of one or more conversions or otherwise in excess of 19.99% of the number of Shares outstanding immediately prior to the Closing Date. The Holder and each permitted assignee shall supply all information necessary to ensure compliance with this Section 2.6, and the Issuer shall be entitled to rely on representations made to it by the Holder and its permitted assignees regarding the ownership limitation set forth in this Section 2.6.

## 3. TRANSFER, DIVISION AND COMBINATION.

- 3.1 Transfer. Subject to compliance with all applicable securities laws, this Warrant may be Transferred to an Affiliate of the Holder at any time and from time to time, in whole or, in the case of a partial Transfer, in a Transfer of not less than 1,000,000 Shares subject to this Warrant. Such Transfer shall be registered on the books of the Issuer to be maintained for such purpose, upon surrender of this Warrant at the principal office of the Issuer at the address set forth in the Warrant Agreement, together with a written assignment of this Warrant duly executed by the Holder or its agent or attorney. Upon such surrender, the Issuer shall execute and deliver a new Warrant or Warrants in the name of the transferee or transferees and in the denomination specified in such instrument of assignment, and shall issue to the assignor a new Warrant evidencing the portion of this Warrant not so assigned, and this Warrant shall promptly be cancelled. Such Warrant, if properly assigned in compliance with any restrictions on Transfer and properly registered on the books of the Issuer, may be exercised by a new Holder for the purchase of Shares without having a new Warrant issued.
- 3.2 <u>Maintenance of Books</u>. The Issuer agrees to maintain, at its aforesaid office or agency, a ledger recording the Holder(s) of this Warrant and any subsequent Transfer of this Warrant in compliance with <u>Section 3.1</u>.
- **4. ADJUSTMENTS.** The number of Shares for which this Warrant is exercisable shall be subject to adjustment from time to time as set forth in this <u>Section 4</u>. The Issuer shall give the Holder notice of any event described below, which requires an adjustment pursuant to this <u>Section 4</u> at the time of such event.
  - 4.1 <u>Distributions, Subdivisions and Combinations</u>. If, at any time, the Issuer:
- (a) takes a record of holders of its Shares for the purpose of entitling them to receive a distribution payable in, or other distribution of, Additional Shares,
  - (b) subdivides its outstanding Shares into a larger number of Shares, or

(c) combines its outstanding Shares into a smaller number of Shares,

then the number of Shares for which this Warrant is exercisable immediately after the occurrence of any such event shall be adjusted to equal the number of Shares that a record holder of the same number of Shares for which this Warrant is exercisable immediately prior to the occurrence of such event would own or be entitled to receive after the happening of such event. If the Issuer subdivides its outstanding Shares into a larger number of Shares, the Exercise Price shall be proportionately decreased. If the Issuer combines its outstanding Shares into a lesser number of Shares, the Exercise Price shall be proportionately increased.

- Reorganization, Consolidation, Merger and Other Changes. In case of any capital reorganization or change in the Common Stock (other than as a result of a subdivision, combination, or stock dividend provided for in Section 4.1), or a spinoff, consolidation or merger of the Issuer with or into another entity, or the sale of all or substantially all of its assets to another entity shall be effected in such a way that holders of Common Stock shall be entitled to receive stock, securities or assets with respect to or in exchange for Common Stock, then, as a condition of such reorganization, change, spinoff, consolidation, merger or sale (each a "Change"), lawful provision shall be made, and duly executed documents evidencing the same from the Issuer or its successor shall be delivered to the Holder, so that the Holder shall have the right at any time prior to the Expiration Date to purchase, at a total price equal to that payable upon the exercise of this Warrant, the kind and amount of shares of stock and other securities and property receivable in connection with such Change by a holder of the same number of Shares as were purchasable by the Holder immediately prior to such Change. In any such case appropriate provisions shall be made with respect to the rights and interest of the Holder so that the provisions hereof shall thereafter be applicable with respect to any shares of stock or other securities and property deliverable upon exercise hereof, and appropriate adjustments shall be made to the purchase price per share payable hereunder, provided the aggregate purchase price shall remain the same; provided, however, that notwithstanding the foregoing, if as a result of such Change, holders of the Issuer's Common Stock shall receive consideration other than solely in shares of stock or other securities in exchange for their Common Stock, the Issuer may, at its option, fulfill its obligations hereunder by causing a notice to be mailed to Holder at least 10 days prior to the record date for the Change, which notice shall (i) describe the nature of the proposed Change and any material terms applicable thereto, (ii) indicate the record date applicable to the Change and (iii) set forth the first and last date on which the Holder may exercise this Warrant. The Holder shall have the opportunity to exercise this Warrant in full before the applicable record date for the Change, and thereby receive consideration with respect to any such Change, on the same basis as other previously outstanding shares of Common Stock of the Issuer. If the notice specified above is provided to the Holder in accordance with this Section 4.2, this Warrant to the extent not exercised before the consummation of the Change shall be cancelled and become null and void on the effective date of the Change.
  - 4.3 Adjustment upon Dilutive Issuance.
- (a) If the Issuer at any time, and from time to time, during the Exercise Period issues or sells any Shares, other than Exempted Securities, for a price per share less than a price equal to eighty percent (80%) of the Market Price on the day of such issue or sale (the foregoing a

A-5

"*Dilutive Issuance*"), then immediately after such Dilutive Issuance the number of Shares then purchasable hereunder shall be increased by an amount determined in accordance with the following formula:

$$S2 = [((A - B) * 0.93 * C) * (S1 ÷ D)] ÷ A$$

For purposes of the foregoing formula, the following definitions shall apply:

"S2" means the number of additional Shares purchasable hereunder immediately after the Dilutive Issuance;

"\$1" means the number of Shares purchasable hereunder immediately prior to the Dilutive Issuance;

"A" means the Market Price on the day of the Dilutive Issuance;

"B" means the per share price for which Shares were issued in the Dilutive Issuance;

"C" means the number of Shares issued pursuant to the Dilutive Issuance; and

- **"D"** means the number of Shares deemed outstanding on a fully diluted basis immediately prior to the Dilutive Issuance (treating for this purpose as outstanding all Shares issuable upon exercise of Options outstanding immediately prior to such Dilutive Issuance or upon conversion or exchange of Convertible Securities outstanding (assuming exercise of any outstanding Options therefor) immediately prior to such Dilutive Issuance).
- (b) Notwithstanding any other provision of this Warrant, (i) no adjustment in the number of Shares purchasable hereunder shall be made pursuant to this Section 4.3 if the Market Price on the date of the Dilutive Issuance is greater than or equal to the Market Price on the Closing Date, and (ii) no adjustment shall be made in respect of any Shares issued under this Warrant prior to the date of the Dilutive Issuance.
  - (c) For purposes of this <u>Section 4.3</u>, the following terms shall have the following meanings:
- (i) "Exempted Securities" means (1) Shares issued to (or issuable upon exercise of Options issued to) employees or directors of, or consultants or advisors to, the Issuer or any of its subsidiaries pursuant to a plan, agreement or arrangement approved by the Board of Directors of the Issuer; (2) Shares issued upon the exercise or exchange of or conversion of any Options or Convertible Securities outstanding on the Closing Date and (3) Shares issued pursuant to (or issuable upon the exercise or exchange or conversion of any Options or Convertible Securities issued pursuant to) mergers, consolidations, acquisitions or any other transactions with Persons with whom the Issuer has business relationships, provided such issuances are for other than primarily equity financing purposes.

- (ii) "*Market Price*" shall mean, in respect of any Shares on any date herein specified, the last reported closing price on the Trading Market on which the Shares are listed.
- 4.4 <u>Timing of Issuance of Additional Shares Upon Adjustments</u>. In any case in which the provisions of this <u>Section 4</u> shall require that an adjustment shall become effective immediately after a record date for an event, the Issuer, after such record date and before the occurrence of such event, may defer until the occurrence of such event issuing to the Holder the Additional Shares or other property issuable or deliverable upon exercise by reason of the adjustment required by such event over and above the Shares issuable or deliverable upon such exercise before giving effect to such adjustment; provided, however, that the Issuer shall, upon request of the Holder, deliver to the Holder a due bill or other appropriate instrument evidencing the Holder's right to receive such Additional Shares or other property upon the occurrence of the event requiring such adjustment.
- 4.5 When Adjustment Not Required. If the Issuer takes a record of holders of its Shares for the purpose of entitling them to receive a distribution or subscription or purchase rights and, thereafter and before the distribution to holders thereof, legally abandons its plan to pay or deliver such distribution, subscription or purchase rights, then thereafter no adjustment shall be required by reason of the taking of such record and any such adjustment previously made in respect thereof shall be rescinded and annulled.
- 4.6 Notice of Adjustments. Whenever the number of Shares for which this Warrant is exercisable, or whenever the price at which such Shares may be purchased upon exercise of the Warrants, is adjusted pursuant to this Section 4, the Issuer shall prepare a certificate to be executed by its chief financial officer, if any, or its principal financial officer(s) in case there is no chief financial officer, setting forth, in reasonable detail, the event requiring the adjustment and the method by which such adjustment was calculated, specifying the number of Shares for which this Warrant is exercisable and describing the number and kind of any other Shares or Other Property for which this Warrant is exercisable, and any change in the purchase price or prices thereof, after giving effect to such adjustment or change. The Issuer shall promptly cause a signed copy of such certificate to be delivered to the Holder. The Issuer shall keep at its office or agency copies of all such certificates and cause the same to be available for inspection at said office during normal business hours by the Holder.
- **5. NO IMPAIRMENT.** The Issuer shall not by any action, including, without limitation, through any amendment to its articles of incorporation, through any reorganization, transfer of assets, consolidation, merger, dissolution, issue or sale of securities or any other voluntary action, avoid or seek to avoid the observance or performance of any of the terms of this Warrant, but shall at all times in good faith assist in carrying out all such actions as may be reasonably necessary or appropriate to protect the rights of the Holder against impairment. Without limiting the generality of the foregoing, the Issuer shall (a) take all such action as may be reasonably necessary or appropriate in order that the Issuer may validly and legally issue fully-paid and nonassessable Shares upon the exercise of this Warrant, and (b) use its reasonable best efforts to obtain all such authorizations, exemptions or consents from any public regulatory body having jurisdiction thereof as may be necessary to enable the Issuer to perform its obligations under this Warrant.

A-7

- **6. RESERVATION AND AUTHORIZATION OF SHARES.** From and after the Closing Date, the Issuer shall at all times reserve and keep available for issue upon the exercise of Warrants such number of its authorized but unissued Shares as shall be sufficient to permit the exercise in full of all outstanding Warrants. All Shares, when issued upon exercise of this Warrant and payment therefor in accordance with the terms of this Warrant, shall be duly and validly issued and fully paid and nonassessable, and not subject to preemptive rights.
- **TAKING OF RECORD; SHARES AND WARRANT TRANSFER BOOKS.** In the case of all distributions by the Issuer to holders of its Shares with respect to which any provision of Section 4 refers to the taking of a record of such holders, the Issuer shall in each such case take such a record as of the close of business on a Business Day. The Issuer shall not at any time, except upon dissolution, liquidation or winding up of the Issuer, close its transfer books so as to result in preventing or delaying the exercise or transfer of any Warrant.
- **8. RESTRICTIVE LEGEND.** This Warrant and any Warrant issued upon transfer or partial exercise of this Warrant shall be imprinted with the following legend, in addition to any legend required under applicable state securities laws:

"THIS WARRANT AND THE SECURITIES ISSUABLE UPON THE EXERCISE HEREOF HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933, AS AMENDED, OR REGISTERED OR QUALIFIED UNDER THE SECURITIES LAWS OF ANY STATE OR OTHER JURISDICTION AND MAY NOT BE SOLD, PLEDGED, TRANSFERRED OR OTHERWISE ASSIGNED IN VIOLATION OF SUCH ACT AND LAWS OR THE PROVISIONS OF THIS WARRANT AND THE WARRANT AGREEMENT."

Each Share certificate representing Warrant Shares shall bear the following legend:

"THE SECURITIES REPRESENTED BY THIS CERTIFICATE HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933, AS AMENDED, OR REGISTERED OR QUALIFIED UNDER THE SECURITIES LAWS OF ANY STATE OR OTHER JURISDICTION AND MAY NOT BE SOLD, PLEDGED, TRANSFERRED OR OTHERWISE ASSIGNED IN VIOLATION OF SUCH ACT AND LAWS OR THE PROVISIONS OF THIS WARRANT AND THE WARRANT AGREEMENT."

Upon request of the holder of a Share certificate, the Issuer shall issue to that holder a new certificate free of the foregoing legend, if, with such request, such holder provides the Issuer with an opinion of counsel (including in-house counsel) reasonably acceptable to the Issuer to the effect that the securities evidenced by such certificate may be sold without restriction under Rule 144 (or any other rule permitting resales of securities without restriction) promulgated under the Securities Act.

A-8

**9. SUPPLYING INFORMATION.** The Issuer shall cooperate with each Holder of a Warrant and each holder of Warrant Shares in supplying such information as may be reasonably necessary for such holder to complete and file any information reporting forms presently or hereafter required by the SEC as a condition to the availability of an exemption from the Securities Act for the sale or transfer of any Warrant or Warrant Shares.

- **10. LOSS OR MUTILATION.** Upon receipt by the Issuer of evidence reasonably satisfactory to it of the ownership of and the loss, theft, destruction or mutilation of any Warrant (which evidence shall be, in the case of an institutional investor, notice from such institutional investor of such ownership and such loss, theft, destruction or mutilation), and
- (a) in the case of loss, theft or destruction, of indemnity reasonably satisfactory to it (provided that if the Holder of such Warrant has a minimum net worth of at least \$100,000,000, such Holder's own unsecured agreement of indemnity shall be deemed to be satisfactory), or
  - (b) in the case of mutilation, upon surrender and cancellation thereof,

the Issuer at its own expense shall execute and deliver, in lieu thereof, a new Warrant, dated the date of the original Warrant.

- 11. OFFICE OF THE ISSUER. As long as any Warrant remains outstanding, the Issuer shall maintain an office or agency (which may be the principal executive offices of the Issuer) where the Warrants may be presented for exercise, registration of transfer, division or combination as provided in this Warrant.
- **12. LIMITATION OF LIABILITY.** No provision hereof, in the absence of affirmative action by holder to purchase Shares, and no enumeration of the rights or privileges of holder contained herein, shall give rise to any liability of holder for the purchase price of any Share or as a holder of Shares of the Issuer, whether such liability is asserted by the Issuer or by creditors of Company.

## 13. MISCELLANEOUS.

- 13.1 <u>Nonwaiver and Expenses</u>. If either party fails to comply with any provision of this Warrant, it shall pay to the other party such amounts as shall be sufficient to cover any costs and expenses including, but not limited to, reasonable attorneys' fees, including those of appellate proceedings, incurred by the other party in enforcing any of its rights, powers, or remedies hereunder. No course of dealing or any delay or failure to exercise any right hereunder on the part of a party shall operate as a waiver of such right or otherwise prejudice its rights, powers, or remedies.
- 13.2 <u>Non-Survival</u>. The parties hereby agree that all the provisions of this Warrant shall terminate and be of no further force or effect on the exercise in full of this Warrant.
- 13.3 <u>Miscellaneous Provisions in Warrant Agreement</u>. The provisions set forth in Section 6 of the Warrant Agreement shall apply to this Warrant, *mutatis mutandis*, and are hereby incorporated by reference herein.

A-9			
[Signature page follows.] A-10			
IN WITNESS WHEREOF, each party hereto has caused this Warrant to be duly executed by its authorized officer and dated effective as of the Closing Date.			
	CLEAN ENERGY FUELS CORP.		
	By: Name: Title:		
	GE ENERGY FINANCIAL SERVICES, INC.		
	By: Name: Title:		
A-11			

## ANNEX A SUBSCRIPTION FORM

(To be executed only upon exercise of the attached Warrant)

The undersigned registered Holder of this Warrant irrevocably exercises this Warrant for purchase of Shares of Clean Energy Fuels Corp., Delaware corporation, [and herewith makes payment therefore][pursuant to the cashless exercise provisions set forth in Section 2.3 of the Warrant, with the calculation for such cashless exercise attached to this Subscription Form], all at the price and on the terms and conditions specified in this Warrant and requests that certificates for the Shares hereby purchased (and any securities or property issuable upon such exercise) to be issued in the name of the undersigned and delivered to the undersigned as follows:

Name	Address

If the certificates representing the Shares being purchased pursuant hereto are to be registered in a name or names other than the name of the holder of this Warrant, all transfer taxes payable upon such transfer shall be paid by the undersigned at the time of delivering the notice of exercise and such request.

Solely with respect to the Warrant Shares being purchased pursuant to this Subscription Form, the representations and warranties of the Holder contained in Section 15.1 of the Warrant are repeated at and as of the time of delivery hereof and are true and correct in all material respects at and as of the time of delivery hereof, except to the extent such representations and warranties are expressly limited to an earlier date or the Issuer has expressly consented in writing to the contrary.

The undersigned acknowledges that each certificate for Warrant Shares issued upon exercise of the Warrant shall bear a legend to the effect that such Warrant Shares may not be transferred except upon compliance with the provisions of the Securities Act and applicable state securities laws, and each certificate for Warrant Shares transferred shall bear such a legend unless, in the opinion of counsel for the Issuer, such legend is not required.

If the number of Shares shall not be all the Warrant Shares purchasable under this Warrant, a new Warrant of like tenor is to be issued in the name of and delivered to the undersigned for the remaining balance of the Shares purchasable thereunder.

(Name of Registered Owner)			(Street Address	)	
(Signature of R	egistered Owner)		(City)	(State)	(Zip Code)
		A-1	2		
NOTICE:		otion must correspond with the nament or any change whatsoever.  A-1		oon the face of the attached	d Warrant in every particular,
		ANNE NOTICE OF C			
		(To be executed only upon conve	ersion of the attacl	hed Warrant)	
exercise provise executed a certain lieu of any fr	ions of Section 2.3 of the Warran	at, in respect of Shares und the undersigned's ownership of the ities or other property issuable u	derlying this Warr e aggregate numb	ant, and requests that er of Shares issuable upor	issuable pursuant to the cashless execute or cause to be a such exercise, together with cash be delivered to the undersigned
	Name			Address	
Warrant Shares	ndersigned acknowledges that ea may not be transferred except up Varrant Shares transferred shall a	pon compliance with the provision	ons of the Securiti	es Act and applicable state	
contained in Se		eated at and as of the time of del	ivery hereof and a	are true and correct in all r	
	number of Shares shall not be all undersigned for the remaining b			new Warrant of like tenor	r is to be issued in the name of and
(Name of Regis	stered Owner)		(Street Address	)	

NOTICE: The signature on this subscription must correspond with the names as written upon the face of the within Warrant in every particular, without alteration or enlargement or any change whatsoever.

(City)

(State)

(Zip Code)

(Signature of Registered Owner)

## Exhibit B to Warrant Agreement

## FORM OF REGISTRATION RIGHTS AGREEMENT

This **REGISTRATION RIGHTS AGREEMENT** (this "*Agreement*"), dated as of November [ ], 2012, is entered into by and between Clean Energy Fuels Corp., a Delaware corporation (the "*Issuer*") and GE Energy Financial Services, Inc., a Delaware corporation (the "*Holder*").

## **RECITALS**

**WHEREAS**, the Issuer and the Holder have entered into a Warrant Agreement dated as of November [ ], 2012 (the "Warrant Agreement");

**WHEREAS**, the Warrant Agreement provides for the issuance of a warrant (the "*Warrant*") to purchase shares of the Issuer's Common Stock ("*Warrant Shares*") on the terms and subject to the conditions set forth therein;

**WHEREAS,** in order to induce the Holder to enter into the Credit Agreement and the Warrant Agreement, the Issuer has agreed to provide certain rights to the Holder as set forth in this Agreement; and

WHEREAS, this Agreement is being executed and delivered in connection with the closing under the Warrant Agreement.

NOW, THEREFORE, the parties hereby agree as follows:

## 1. REGISTRATION RIGHTS.

(a) Definitions.

Capitalized terms used herein and not otherwise defined herein shall have the respective meanings set forth in the Warrant Agreement. As used in this Agreement, the following terms shall have the following meanings:

"1933 Act" means the Securities Act of 1933, as amended.

"1934 Act" means the Securities Exchange Act of 1934, as amended.

"Base Price" means \$[ ].(1)

"*Business Day*" means any day that is not a Saturday, Sunday or other day when commercial banks are required or permitted by law to be closed in New York, New York.

"Common Stock" means the shares of common stock of the Issuer, par value \$0.0001.

"Effective Date" means the date the Registration Statement has been declared effective by the SEC.

(1) The closing price of the Common Sock the date before the Closing.

"*Effectiveness Deadline*" means the date that is (i) thirty (30) days after the Filing Deadline if the Registration Statement is not subject to review by the SEC, or (ii) ninety (90) days after the Filing Deadline if the Registration Statement is subject to review by the SEC.

"Filing Deadline" means the date that is the earlier of (i) forty five (45) days following the date that the SEC has confirmed in writing to the Issuer that it has completed its current review of the Issuer's filings with the SEC, which review is ongoing as of the date of this Agreement, and (ii) ninety (90) days following the date of this Agreement.

"Liquidated Damages Amount" means an amount equal to 0.25% of the product of the Base Price times the number of Restricted Warrant Shares (as defined in Section 1(d)(i)) then held by the Holder per 30-day period for the first sixty (60) days, with such payment amount increasing by an additional 0.25% of the product of the Base Price times the number of Restricted Warrant Shares then held by the Holder per 30-day period for each subsequent sixty (60) days, up to a maximum of 1.00% of the product of the Base Price times the number of Restricted Warrant Shares then held by the Holder per 30-day period; provided, however, that notwithstanding anything herein to the contrary, in no event shall the Issuer be liable for an aggregate Liquidated Damages Amount of more than four percent (4%) per annum of the product of the Base Price times the number of Restricted Warrant Shares then held by Holder or its assignees. The Liquidated Damages Amount for any period of less than 30-days shall be prorated by multiplying the Liquidated Damages Amount to be paid in a full 30-day period by a fraction, the numerator of which is the number of days for which such liquidated damages are owed, and the denominator of which is 30. The Base Price is subject to appropriate adjustments for any subdivision or combination of Common Stock after the date thereof.

"*Person*" means an individual, a limited liability company, a partnership, a joint venture, a corporation, a trust, an unincorporated organization and a government or any department or agency thereof.

"register," "registered" and "registration" refer to a registration effected by preparing and filing one or more Registration Statements (as defined below) in compliance with the 1933 Act and the declaration or ordering of effectiveness of such Registration Statement(s) by the SEC.

"Registrable Securities" means the Warrant Shares and any shares of capital stock issued or issuable with respect to the Warrant Shares by reason of or in connection with any stock split, stock dividend, recapitalization, exchange or similar event or otherwise, issued to or held by the Holder.

"Registration Statement" means a shelf registration statement, registration statements or Form S-3 of the Issuer filed under the 1933 Act covering the Registrable Securities.

"*Rule 144*" means Rule 144 promulgated under the 1933 Act or any successor rule or other similar rule or regulation of the SEC that may at any time permit the Holder to sell securities of the Issuer to the public without registration.

B-2

"Rule 415" means Rule 415 promulgated under the 1933 Act, as such rule may be amended or interpreted from time to time, or any similar rule or regulation hereunder adopted by the SEC.

"SEC" means the United States Securities and Exchange Commission.

## (b) <u>Mandatory Registration</u>.

- (i) The Issuer shall prepare and, no later than the Filing Deadline, file with the SEC, a Registration Statement on Form S-3 covering the resale of Warrant Shares. If Form S-3 is unavailable for such a registration, the Issuer shall use such other form as is available for such a registration (the Registration Statement filed pursuant to this Section 1(b)(i), the "Resale Registration Statement").
- (ii) The Issuer shall use its commercially reasonable efforts to have such Registration Statement declared effective by the SEC as soon as practicable, but in no event later than the Effectiveness Deadline, and shall use its commercially reasonable efforts to keep such Registration Statement continuously effective under the 1933 Act until the earlier of the date when all Registrable Securities covered by such Registration Statement (A) have been sold, thereunder or pursuant to Rule 144, or (B) may be sold without volume or manner-of-sale restrictions pursuant to Rule 144 and without the requirement for the Issuer to be in compliance with the current public information requirement under Rule 144, as determined by counsel to the Issuer pursuant to a written opinion letter to such effect, addressed and acceptable to the Issuer's transfer agent and the Holder (the "Effectiveness Period").
- (iii) Notwithstanding the registration obligation set forth in Section 1(b), if the SEC informs the Issuer that all of the Registrable Securities cannot, as a result of the application of Rule 415, be registered for resale as a secondary offering on a single registration statement, the Issuer agrees to promptly inform the Holder and use its commercially reasonable efforts to file amendments to the Registration Statement as required by the SEC, covering the maximum number of Registrable Securities permitted to be registered by the SEC, on Form S-3 or such other form available to register for resale the Registrable Securities as a secondary offering; provided, however, that prior to filing such amendment, the Issuer shall be obligated to use diligent efforts to advocate with the SEC for the registration of all of the Registrable Securities in accordance with any interpretive guidance provided by the SEC, including without limitation, Compliance and Disclosure Interpretation 612.09.

## (c) <u>Piggyback Registration</u>.

(i) If (but without any obligation to do so), at any time during the Effectiveness Period, a Registration Statement is not effective with respect to all of the Registrable Securities and the Issuer proposes to register (including for this purpose a registration effected by the Issuer for stockholders other than the Holder) any of its capital stock or other securities under the 1933 Act in connection with a fully underwritten firm commitment public offering of such securities (other than a registration

B-3

on any form that does not include substantially the same information as would be required to be included in a registration statement covering the sale of the Registrable Securities), the Issuer shall, at such time, give the Holder written notice of such registration (a "*Piggyback Registration*") in accordance with Section 2(f). Upon the written request of the Holder given within five (5) Business Days after delivery of such notice by the Issuer, the Issuer shall, subject to the provisions of Section 1(c)(iii), use all commercially reasonable efforts to cause to be registered under the 1933 Act all of the Registrable Securities that the Holder requests to be registered.

- (ii) The Issuer shall have the right to terminate or withdraw any registration initiated by it under this <u>Section 1(c)</u> prior to the effectiveness of such registration whether or not the Holder has elected to include securities in such registration. The expenses of such withdrawn registration shall be borne by the Issuer in accordance with <u>Section 1(j)</u> hereof.
- (iii) The Issuer shall not be required under this <u>Section 1(c)</u> to include any of the Holder's Registrable Securities in such underwriting unless the Holder accepts the terms of the underwriting as reasonably agreed upon between the Issuer and the underwriters selected by the Issuer (or by other Persons entitled to select the underwriters) and enters into an underwriting agreement in customary form with such underwriters, and then only in such quantity as the underwriters determine in their sole discretion will not jeopardize the success of the offering by the Issuer. If the total amount of securities, including Registrable Securities, requested to be included in such offering exceeds the amount of securities that the underwriters determine in their sole discretion is compatible with the success of the offering, then the Issuer shall be required to include in the offering only that number of Registrable Securities that the underwriters determine in their sole discretion will not jeopardize the success of the offering. Any reduction in the number of Registrable Securities will be made pro rata (based on the number of Warrant Shares then held by the Holder) with the other securities to be registered on behalf of third parties in such offering.

## (d) <u>Effect of Failure to File and Obtain Effectiveness of Registration Statement.</u>

(i) If a Registration Statement covering all of the Registrable Securities required to be covered thereby (or such lesser amount as provided in <u>Section 1(b)(iii)</u>) and required to be filed by the Issuer pursuant to <u>Section 1(b)</u> of this Agreement has not been declared effective by the SEC on or before the Effectiveness Deadline (an "*Effectiveness Failure*"), then each Holder will be entitled to a payment, as liquidated damages

B-4

The aggregate Liquidated Damages Amount payable to each Holder shall be paid to each Holder in immediately available funds within 10 Business Days after the end of each applicable 30-day period. Any payments pursuant to this Section 2(b) shall constitute the Holder's exclusive remedy for such events; provided, however, if the Issuer certifies that it is unable to pay aggregate Liquidated Damages Amount in cash or immediately available funds because such payment would result in a breach under any of the Issuer's credit facilities or other indebtedness filed as exhibits to the Issuer's reports filed under the Securities Act or the Exchange Act, then the Issuer may pay the aggregate Liquidated Damages Amount in kind in the form of the issuance of additional shares of Common Stock. Upon any issuance of shares of Common Stock as liquidated damages, the Issuer shall promptly prepare and file an amendment to the Resale Registration Statement prior to its effectiveness to include such shares of Common Stock issued as liquidated damages as additional Registrable Securities. If shares of Common Stock are issued as liquidated damages after the Resale Registration Statement has been declared effective, the Issuer shall have no obligation to prepare and file a posteffective amendment to the Resale Registration Statement to include such shares nor shall the Issuer be obligated in any way to file a new registration statement for such shares; however if the Issuer is a well-known seasoned issuer (as defined in the rules and regulations of the SEC) ("WKSI"), the Issuer shall be obligated to provide the Holder notice and offer to include such shares in any Piggyback Registration. All shares of Common Stock issued as Liquidated Damages Amounts shall be considered in the calculation of any subsequent Liquidated Damages Amounts. The determination of the number of shares of Common Stock to be issued as the aggregate Liquidated Damages Amount shall be equal to the aggregate Liquidated Damages Amount divided by the average of the closing sale price per share for the Common Stock (or if the Common Stock is not listed or traded on a national securities exchange, the average of the last reported bid and ask prices per share) for each of the 10 consecutive trading days ending on the trading day immediately preceding such date of determination.

## (e) Request for Registration and/or Underwriting.

(i) If, at any time during the Effectiveness Period, a Registration Statement is not effective with respect to all of the Registrable Securities, the Issuer shall, at the request of the Holder, participate in an underwritten offering of Registrable Securities by the Holder under a Registration Statement effected pursuant to Section 1(b) hereof, and shall file any supplements and amendments to such Registration Statement as may be required by applicable law or rules of the SEC. If, at any time after the Effectiveness Period, a Registration Statement is not effective with respect to all of the Registrable Securities, and the Issuer receives a written request from the Holder that the Issuer effect a registration on Form S-3 with respect to an underwritten offering of Registrable Securities, the Issuer shall use commercially reasonable efforts to file a Registration Statement covering the Registrable Securities as soon as reasonably practicable after receipt of the request. For purposes of this Agreement, a "Demand" shall refer to the Holder's request, pursuant to this Section 1(e), for the Issuer to (1) participate in an underwritten offering of Registrable Securities or (2) effect a registration on Form S-3 with respect to an underwritten offering of Registrable Securities. In any

B-5

underwritten offering under this <u>Section 1(e)</u>, the investment banker or bankers and manager or managers that will administer the offering will be selected by, and the underwriting arrangements with respect thereto (including the size of the offering) will be approved by the requesting Holder; <u>provided</u>, <u>however</u>, that such investment bankers and managers and underwriting arrangements must be reasonably satisfactory to the Issuer. The Issuer shall not be required to participate in any underwritten offering contemplated hereby unless each participating Holder (A) agrees to sell its Registrable Securities to be included in the underwritten offering in accordance with any approved underwriting arrangements and (B) completes and executes all reasonable questionnaires, powers of attorney, indemnities, underwriting agreements, lock-up letters and other documents required under the terms of such approved underwriting arrangements. Each participating Holder shall be responsible for any underwriting discounts and commissions and fees and expenses of its own counsel. The Issuer shall pay all expenses customarily borne by issuers in an underwritten offering, including, but not limited to, filing fees, the fees and disbursements of its counsel and independent public accountants and any printing expenses incurred in connection with such underwritten offering.

- (ii) The Issuer shall not be required to participate in or effect any Demand pursuant to this <u>Section 1(e)</u>:
- (A) after the Issuer has participated in or effected two (2) Demands (the Holder shall be deemed to have forfeited its right to a Demand if (1) the Holder withdraws its request that the Issuer effect a registration on Form S-3 with respect to an underwritten offering of Registrable Securities and does not, within thirty (30) days of any such withdrawal, pay all of the Issuer's expenses in connection with such registration or (2) an underwritten offering that is the subject of a Demand is terminated subsequent to the marketing thereof);
  - (B) if the Issuer has participated in or effected a Demand within the preceding twelve (12) months;
- (C) during the period starting with the date sixty (60) days prior to the Issuer's good faith estimate of the date of the filing of, and ending on a date one hundred eighty (180) days following the effective date of, an Issuer-initiated registration subject to Section 1(c), provided that the Issuer is actively employing in good faith all reasonable efforts to cause such registration statement to become effective; or
- (D) if the Issuer shall furnish to the requesting Holder(s) a certificate signed by the Issuer's Chief Executive Officer or Chairman of the Board stating that in the good faith judgment of the Board of Directors of the Issuer (the "*Board*"), it would be seriously detrimental to the Issuer and its stockholders for the Issuer to participate in or effect a Demand at such time, in which event the Issuer shall have the right to defer such Demand for a period of not more than one hundred twenty (120) days after receipt of the request of the Holder.

- (f) <u>Related Obligations</u>. Whenever required under this <u>Section 1</u> to effect the registration of any Registrable Securities, except as otherwise expressly provided herein, the Issuer shall:
  - (i) prepare and file with the SEC a Registration Statement with respect to such Registrable Securities and use all commercially reasonable efforts to cause such Registration Statement to become and remain effective;
  - (ii) prepare and file with the SEC such amendments and supplements to such Registration Statement and the prospectus used in connection with such Registration Statement as may be necessary to comply with the provisions of the 1933 Act with respect to the disposition of all securities covered by such Registration Statement;
  - (iii) furnish to the Holder such number of copies of a prospectus, including a preliminary prospectus, in conformity with the requirements of the 1933 Act, and such other documents as it may reasonably request in order to facilitate the disposition of Registrable Securities owned by it;
  - (iv) if required by applicable law, use all commercially reasonable efforts to register and qualify the securities covered by such Registration Statement under such other securities or "blue sky" laws of such jurisdictions as shall be reasonably requested by the Holder, provided that the Issuer shall not be required in connection therewith or as a condition thereto to qualify to do business or to file a general consent to service of process in any such states or jurisdictions;
    - (v) with a view to making available to the Holder the benefits of Rule 144:
      - (A) make and keep public information available, as those terms are understood and defined in Rule 144;
    - (B) file with the SEC in a timely manner all reports and other documents required of the Issuer under the 1933 Act and the 1934 Act so long as the Issuer remains subject to such requirements and the filing of such reports and other documents is required for the applicable provisions of Rule 144; and
    - (C) furnish or otherwise make available, as applicable, to the Holder so long as the Holder owns Registrable Securities, promptly upon request, (1) a written statement by the Issuer, if true, that it has complied with the reporting requirements of Rule 144, the 1933 Act and the 1934 Act, (2) a copy of the most recent annual or quarterly report of the Issuer and such other reports and documents so filed by the Issuer and (3) such other information as may be reasonably requested to permit the Holder to sell such securities without registration pursuant to Rule 144;

B-7

- (D) in the event of any underwritten public offering, enter into and perform its obligations under an underwriting agreement, in usual and customary form, with the managing underwriter of such offering; and
- (E) notify the holder of Registrable Securities covered by such Registration Statement at any time when a prospectus relating thereto is required to be delivered under the 1933 Act of the happening of any event as a result of which the prospectus included in such Registration Statement, as then in effect, includes an untrue statement of a material fact or omits to state a material fact required to be stated therein or necessary to make the statements therein not misleading in the light of the circumstances then existing.

Notwithstanding the provisions of this Section 1, the Issuer shall be entitled to postpone or suspend the filing, effectiveness or use of, or trading under, any Registration Statement during any period when (i) the SEC or the national securities exchange upon which shares of Common Stock are then listed requests that the Issuer amend or supplement the Registration Statement or the prospectus included therein or requests additional information relating thereto, (ii) the SEC or the national securities exchange upon which shares of Common Stock are then listed issues a stop order or similar order suspending the effectiveness or restricting the use of the Registration Statement or initiates proceedings to issue a stop order or similar order, (iii) the Board in good faith determines that the Registration Statement, the prospectus included therein, any amendment or supplement thereto or any document incorporated or deemed to be incorporated therein contains an untrue statement of a material fact or omits to state a material fact required to be stated therein or necessary to make the statements therein not misleading, in light of the circumstances then existing; provided, however, that the Issuer uses commercially reasonable efforts to prepare and file with the SEC such amendments and supplements to the such Registration Statement or amendment as shall be reasonably necessary to cure such untrue statement or omission, or (iv) the Issuer's management or the Board in good faith determines that the failure to so postpone or suspend would require disclosure of material nonpublic information that, if disclosed at such time, would be materially harmful to the interests of the Issuer and its stockholders; provided, further, that such postponement or suspension (A) shall not exceed a period of forty-five (45) days and (B) shall be exercised by the Issuer not more than twice in any twelve (12) month period (for a maximum of ninety (90) days within any such twelve (12) month period) (each, an "Allowable Grace Period").

- (g) <u>Information from the Holder</u>. It shall be a condition precedent to the obligations of the Issuer to take any action pursuant to this <u>Section 1</u> with respect to the Registrable Securities of the Holder that the Holder shall furnish to the Issuer such information regarding itself, the Registrable Securities held by it, and the intended method of disposition of such securities as shall be reasonably required to effect the registration of such Registrable Securities.
  - (h) Indemnification. If any Registrable Securities are included in a Registration Statement under this Agreement:

B-8

of the 1933 Act or the 1934 Act (each, a "Holder Indemnified Person"), against any losses, claims, damages, liabilities, judgments, fines, penalties, charges, costs, reasonable attorneys' fees, amounts paid in settlement or expenses, joint or several, (collectively, "Claims") incurred in investigating, preparing or defending any action, claim, suit, inquiry, proceeding, investigation or appeal taken from the foregoing by or before any court or governmental, administrative or other regulatory agency, body or the SEC, whether pending or threatened, whether or not an indemnified party is or may be a party thereto ("Indemnified Damages"), to which any of them may become subject insofar as such Claims (or actions or proceedings, whether commenced or threatened, in respect thereof) arise out of or are based upon: (A) any untrue statement or alleged untrue statement of a material fact in a Registration Statement or any post-effective amendment thereto or in any filing made in connection with the qualification of the offering under the securities or other "blue sky" laws of any jurisdiction in which Registrable Securities are offered, or the omission or alleged omission to state a material fact required to be stated therein or necessary to make the statements therein not misleading, (B) any untrue statement or alleged untrue statement of a material fact contained in any preliminary prospectus if used prior to the effective date of such Registration Statement, or contained in the final prospectus (as amended or supplemented, if the Issuer files any amendment thereof or supplement thereto with the SEC) or the omission or alleged omission to state therein any material fact necessary to make the statements made therein, in the light of the circumstances under which the statements therein were made, not misleading, or (C) any violation or alleged violation by the Issuer of the 1933 Act, the 1934 Act, any state securities law, or any rule or regulation thereunder relating to the offer or sale of the Registrable Securities pursuant to a Registration Statement (the matters in the foregoing clauses (A) through (C) being, collectively, "Violations"). Subject to Section 1(h)(iii), the Issuer shall reimburse the Holder Indemnified Persons, promptly as such expenses are incurred and are due and payable, for any legal fees or other reasonable expenses incurred by them in connection with investigating or defending any such Claim. Notwithstanding anything to the contrary contained herein, the indemnification agreement contained in this Section 1(h)(i): (A) shall not apply to a Claim by a Holder Indemnified Person arising out of or based upon a Violation which occurs in reliance upon and in conformity with information furnished in writing to the Issuer by such Holder Indemnified Person expressly for use in connection with the preparation of the Registration Statement or any such amendment thereof or supplement thereto, and (B) shall not apply to amounts paid in settlement of any Claim if such settlement is effected without the prior written consent of the Issuer, which consent shall not be unreasonably withheld or delayed. Such indemnity shall remain in full force and effect regardless of any investigation made by or on behalf of the Holder Indemnified Person.

(ii) In connection with any Registration Statement in which Holder is participating, the Holder agrees to indemnify, hold harmless and defend, to the same extent and in the same manner as is set forth in <u>Section 1(h)(i)</u>, the Issuer, each of its directors, officers, employees and agents and each Person, if any, who controls the Issuer within the meaning of the 1933 Act or the 1934 Act (each, an "*Issuer Indemnified Person*"), against any Claim or Indemnified Damages to which any of them may become

B-9

subject, under the 1933 Act, the 1934 Act or otherwise, insofar as such Claim or Indemnified Damages arise out of or are based upon any Violation, in each case to the extent, and only to the extent, that such Violation occurs in reliance upon and in conformity with written information furnished to the Issuer by the Holder expressly for use in connection with such Registration Statement; and, subject to Section 1(h)(iii), the Holder will reimburse any legal or other expenses reasonably incurred by an Issuer Indemnified Person in connection with investigating or defending any such Claim; provided, however, that the indemnity agreement contained in this Section 1(h)(ii) and the agreement with respect to contribution contained in Section 1(i) shall not apply to amounts paid in settlement of any Claim if such settlement is effected without the prior written consent of the Holder, which consent shall not be unreasonably withheld or delayed; provided, further, however, that the Holder shall be liable under this Section 1(h) (ii) for only that amount of a Claim or Indemnified Damages as does not exceed the net proceeds to the Holder as a result of the sale of Registrable Securities pursuant to such Registration Statement. Such indemnity shall remain in full force and effect regardless of any investigation made by or on behalf of such Issuer Indemnified Person.

Promptly after receipt by a Holder Indemnified Person or an Issuer Indemnified Person (each, an "Indemnified Person") under this <u>Section 1(h)</u> of notice of the commencement of any action or proceeding (including any governmental action or proceeding) involving a Claim, such Indemnified Person shall, if a Claim in respect thereof is to be made against any indemnifying party under this Section 1(h), deliver to the indemnifying party a written notice of the commencement thereof, and the indemnifying party shall have the right to participate in, and, to the extent the indemnifying party so desires, jointly with any other indemnifying party similarly noticed, to assume control of the defense thereof with counsel mutually satisfactory to the indemnifying party and the Indemnified Person; provided, however, that an Indemnified Person shall have the right to retain its own counsel with the fees and expenses of not more than one counsel for such Indemnified Person to be paid by the indemnifying party, if, in the reasonable opinion of counsel retained by the indemnifying party, the representation by such counsel of the Indemnified Person and the indemnifying party would be inappropriate due to actual or potential differing interests between such Indemnified Person and any other party represented by such counsel in such proceeding. In the case of a Holder Indemnified Person, legal counsel referred to in the immediately preceding sentence shall be selected by the Holder. The Indemnified Person shall cooperate fully with the indemnifying party in connection with any negotiation or defense of any such action or Claim by the indemnifying party and shall furnish to the indemnifying party all information reasonably available to the Indemnified Person which relates to such action or Claim. The indemnifying party shall keep the Indemnified Person reasonably apprised at all times as to the status of the defense or any settlement negotiations with respect thereto. No indemnifying party shall be liable for any settlement of any action, claim or proceeding effected without its prior written consent; provided, however, that the indemnifying party shall not unreasonably withhold, delay or condition its consent. No indemnifying party shall, without the prior written consent of the Indemnified Person, consent to entry of any judgment or enter into any settlement or

B-10

other compromise which does not include as an unconditional term thereof the giving by the claimant or plaintiff to such Indemnified Person of a release from all liability in respect to such Claim or litigation, and such settlement shall not include any admission as to fault on the part of the Indemnified Person. Following indemnification as provided for hereunder, the indemnifying party shall be subrogated to all rights of the Indemnified Person with respect to all third parties, firms or corporations relating to the matter for which indemnification has been made. The failure to deliver written notice to the indemnifying party within a reasonable time of the commencement of any such action shall not relieve such indemnifying party of any liability to the Indemnified Person under this Section 1(h), except to the extent that the indemnifying party is prejudiced in its ability to defend such action.

(iv) The indemnification required by this <u>Section 1(h)</u> shall be made by periodic payments of the amount thereof during the course of the investigation or defense, promptly following when bills are received or Indemnified Damages are incurred, and in each case submitted

to the indemnifying party for payment subject to and in accordance with this Section 1(h).

The indemnity agreements contained herein shall be in addition to (A) any cause of action or similar right of the Indemnified Person against the indemnifying party or others, and (B) any liabilities the indemnifying party may be subject to pursuant to the law.

- (i) <u>Contribution.</u> To the extent any indemnification by an indemnifying party is prohibited or limited by law, the indemnifying party agrees to make the maximum contribution with respect to any amounts for which it would otherwise be liable under <u>Section 1(h)</u> to the fullest extent permitted by law; <u>provided</u>, <u>however</u>, that: (i) no Person involved in the sale of Registrable Securities who is guilty of fraudulent misrepresentation (within the meaning of Section 11(f) of the 1933 Act) in connection with such sale shall be entitled to contribution from any Person involved in such sale of Registrable Securities who was not guilty of fraudulent misrepresentation; and (ii) contribution by the Holder shall be limited in amount to the excess of the net amount of proceeds received by the Holder from the sale of such Registrable Securities pursuant to such Registration Statement over the amount of any damages that the Holder has otherwise been required to pay by reason of such untrue or alleged untrue statement or omission or alleged omission.
- (j) <u>Expenses of Registration.</u> All expenses (other than (i) underwriting discounts and commissions relating to the Registrable Securities that are being sold by the Holder and (ii) fees of any counsel for the Holder) that are incurred in connection with registrations, filings or qualifications pursuant to <u>Sections 1(b) and 1(c)</u>, including (without limitation) all registration, filing and qualification fees, printers' and accounting fees, fees and disbursements of counsel for the Issuer, shall be borne by the Issuer.

B-11

## 2. MISCELLANEOUS.

- (a) Governing Law; Jurisdiction; Jury Trial. All questions concerning the construction, validity, enforcement and interpretation of this Agreement shall be governed by the internal laws of the State of Delaware, without giving effect to any choice of law or conflict of law provision or rule (whether of the State of Delaware or any other jurisdictions) that would cause the application of the laws of any jurisdictions other than the State of Delaware. Each party hereby irrevocably submits to the exclusive jurisdiction of the Delaware Court of Chancery, or, if no such state court has proper jurisdiction, the United States District Court for the District of Delaware, for the adjudication of any dispute hereunder or in connection herewith or with any transaction contemplated hereby or discussed herein, and hereby irrevocably waives, and agrees not to assert in any suit, action or proceeding, any claim that it is not personally subject to the jurisdiction of any such court, that such suit, action or proceeding is brought in an inconvenient forum or that the venue of such suit, action or proceeding is improper. Each party hereby irrevocably waives personal service of process and consents to process being served in any such suit, action or proceeding by mailing a copy thereof to such party at the address for such notices to it under this Agreement and agrees that such service shall constitute good and sufficient service of process and notice thereof. Nothing contained herein shall be deemed to limit in any way any right to serve process in any manner permitted by law. EACH PARTY HEREBY IRREVOCABLY WAIVES ANY RIGHT IT MAY HAVE, AND AGREES NOT TO REQUEST, A JURY TRIAL FOR THE ADJUDICATION OF ANY DISPUTE HEREUNDER OR IN CONNECTION WITH OR ARISING OUT OF THIS AGREEMENT OR ANY TRANSACTION CONTEMPLATED HEREBY.
- (b) <u>Counterparts.</u> This Agreement may be executed in two or more identical counterparts, all of which shall be considered one and the same agreement and shall become effective when counterparts have been signed by each party and delivered to the other parties; provided that a facsimile signature shall be considered due execution and shall be binding upon the signatory thereto with the same force and effect as if the signature were an original, not a facsimile signature.
- (c) <u>Headings.</u> The headings of this Agreement are for convenience of reference and shall not form part of, or affect the interpretation of, this Agreement.
- (d) <u>Severability.</u> If any provision of this Agreement shall be invalid or unenforceable in any jurisdiction, such invalidity or unenforceability shall not affect the validity or enforceability of the remainder of this Agreement in that jurisdiction or the validity or enforceability of any provision of this Agreement in any other jurisdiction.
- (e) <u>Entire Agreement; Amendments.</u> This Agreement, the Warrant Agreement and the other Warrant Documents supersede all other prior oral or written agreements between the Holder, the Issuer, their affiliates and Persons acting on their behalf with respect to the matters discussed herein, and this Agreement, the Warrant Agreement, the other Warrant Documents and the instruments referenced herein and therein contain the entire understanding of the parties with respect to the matters covered herein and therein and, except as specifically set forth herein or therein, neither the Issuer nor the Holder makes any representation, warranty, covenant or undertaking with respect to such matters. No provision of this Agreement may be amended other than by an instrument in writing signed by the Issuer and the Holder. No provision hereof may be waived other than by an instrument in writing signed by the party against whom enforcement is sought.

B-12

(f) Notices. Any notices, consents, waivers or other communications required or permitted to be given under the terms of this Agreement must be in writing and will be deemed to have been delivered: (i) upon receipt, when delivered personally; (ii) upon receipt, when sent by facsimile (provided confirmation of transmission is mechanically or electronically generated and kept on file by the sending party); or (iii) one Business Day after deposit with an overnight courier service, in each case properly addressed to the party to receive the same. The addresses and facsimile numbers for such communications shall be:

if to the Issuer:

Clean Energy Fuels Corp. 3020 Old Ranch Parkway Suite 400 Seal Beach, California 90740 Telephone: (562) 493-7239 Facsimile: (562) 493-4956 Attention: J. Nathan Jensen

with a copy (for informational purposes only) to:

Morrison & Foerster LLP 12531 High Bluff Drive Suite 100

San Diego, California 92130 Telephone: (858) 720-5198 Facsimile: (858) 523-2810 Attention: Steven G. Rowles

if to the Holder:

GE Energy Financial Services, Inc. 333 Clay Street, Suite 4550 Houston, Texas 77002 Telephone: (713) 951-2339

Facsimile: (713) 951-2319 Attention: Brandy Copley

with a copy (for informational purposes only) to:

Vinson & Elkins LLP 666 Fifth Avenue, 26<sup>th</sup> Floor New York, New York 10103 Telephone: (212) 237-0132 Facsimile: (917) 849-5350

Attention: Robert Seber

or to such other address and/or facsimile number and/or to the attention of such other Person as the recipient party has specified by written notice given to each other party five (5) days prior to

B-13

the effectiveness of such change. Written confirmation of receipt (A) given by the recipient of such notice, consent, waiver or other communication, (B) mechanically or electronically generated by the sender's facsimile machine containing the time, date, recipient facsimile number and an image of the first page of such transmission or (C) provided by an overnight courier service shall be rebuttable evidence of personal service, receipt by facsimile or receipt from an overnight courier service in accordance with clause (i), (ii) or (iii) above, respectively.

- (g) <u>Successors and Assigns.</u> No party may assign this Agreement or any rights or obligations hereunder without the prior written consent of the other parties, except that the Holder may assign its rights hereunder in connection with any transfer of the Warrant Shares. This Agreement shall be binding upon and inure to the benefit of the parties and their successors and permitted assigns.
- (h) <u>No Third Party Beneficiaries.</u> This Agreement is intended for the benefit of the parties hereto and their respective permitted successors and assigns, and is not for the benefit of, nor may any provision hereof be enforced by, any other Person.
- (i) <u>Further Assurances.</u> Each party shall do and perform, or cause to be done and performed, all such further acts and things, and shall execute and deliver all such other agreements, certificates, instruments and documents, as any other party may reasonably request in order to carry out the intent and accomplish the purposes of this Agreement and the consummation of the transactions contemplated hereby.

[Signature page follows.]

B-14

**IN WITNESS WHEREOF,** the Issuer and the Holder have caused this Registration Rights Agreement to be duly executed as of the date first written above.

CLEAN	ENERGY FUELS CO	ORP.	
By:			
Name: Title:			

HOLDER:

**COMPANY:** 

GE ENERGY FINANCIAL SERVICES, INC.

By:		
Name:		
Title:		

SIGNATURE PAGE TO REGISTRATION RIGHTS AGREEMENT

B-15

## REGISTRATION RIGHTS AGREEMENT

This **REGISTRATION RIGHTS AGREEMENT** (this "*Agreement*"), dated as of November 7, 2012, is entered into by and between Clean Energy Fuels Corp., a Delaware corporation (the "*Issuer*") and GE Energy Financial Services, Inc., a Delaware corporation (the "*Holder*").

## RECITALS

WHEREAS, the Issuer and the Holder have entered into a Warrant Agreement dated as of November 7, 2012 (the "Warrant Agreement");

**WHEREAS**, the Warrant Agreement provides for the issuance of a warrant (the "*Warrant*") to purchase shares of the Issuer's Common Stock ("*Warrant Shares*") on the terms and subject to the conditions set forth therein;

**WHEREAS,** in order to induce the Holder to enter into the Credit Agreement and the Warrant Agreement, the Issuer has agreed to provide certain rights to the Holder as set forth in this Agreement; and

WHEREAS, this Agreement is being executed and delivered in connection with the closing under the Warrant Agreement.

**NOW, THEREFORE,** the parties hereby agree as follows:

## 1. REGISTRATION RIGHTS.

(a) <u>Definitions</u>.

Capitalized terms used herein and not otherwise defined herein shall have the respective meanings set forth in the Warrant Agreement. As used in this Agreement, the following terms shall have the following meanings:

"1933 Act" means the Securities Act of 1933, as amended.

"1934 Act" means the Securities Exchange Act of 1934, as amended.

"Base Price" means \$12.01.

"*Business Day*" means any day that is not a Saturday, Sunday or other day when commercial banks are required or permitted by law to be closed in New York, New York.

"Common Stock" means the shares of common stock of the Issuer, par value \$0.0001.

"Effective Date" means the date the Registration Statement has been declared effective by the SEC.

- "*Effectiveness Deadline*" means the date that is (i) thirty (30) days after the Filing Deadline if the Registration Statement is not subject to review by the SEC, or (ii) ninety (90) days after the Filing Deadline if the Registration Statement is subject to review by the SEC.
- "Filing Deadline" means the date that is the earlier of (i) forty five (45) days following the date that the SEC has confirmed in writing to the Issuer that it has completed its current review of the Issuer's filings with the SEC, which review is ongoing as of the date of this Agreement, and (ii) ninety (90) days following the date of this Agreement.
- "Liquidated Damages Amount" means an amount equal to 0.25% of the product of the Base Price times the number of Restricted Warrant Shares (as defined in Section 1(d)(i)) then held by the Holder per 30-day period for the first sixty (60) days, with such payment amount increasing by an additional 0.25% of the product of the Base Price times the number of Restricted Warrant Shares then held by the Holder per 30-day period for each subsequent sixty (60) days, up to a maximum of 1.00% of the product of the Base Price times the number of Restricted Warrant Shares then held by the Holder per 30-day period; provided, however, that notwithstanding anything herein to the contrary, in no event shall the Issuer be liable for an aggregate Liquidated Damages Amount of more than four percent (4%) per annum of the product of the Base Price times the number of Restricted Warrant Shares then held by Holder or its assignees. The Liquidated Damages Amount for any period of less than 30-days shall be prorated by multiplying the Liquidated Damages Amount to be paid in a full 30-day period by a fraction, the numerator of which is the number of days for which such liquidated damages are owed, and the denominator of which is 30. The Base Price is subject to appropriate adjustments for any subdivision or combination of Common Stock after the date thereof.
- "*Person*" means an individual, a limited liability company, a partnership, a joint venture, a corporation, a trust, an unincorporated organization and a government or any department or agency thereof.
- "register," "registered" and "registration" refer to a registration effected by preparing and filing one or more Registration Statements (as defined below) in compliance with the 1933 Act and the declaration or ordering of effectiveness of such Registration Statement(s) by the SEC.
- "Registrable Securities" means the Warrant Shares and any shares of capital stock issued or issuable with respect to the Warrant Shares by reason of or in connection with any stock split, stock dividend, recapitalization, exchange or similar event or otherwise, issued to or held by the Holder.
- "Registration Statement" means a shelf registration statement, registration statements or Form S-3 of the Issuer filed under the 1933 Act covering the Registrable Securities.
- "*Rule 144*" means Rule 144 promulgated under the 1933 Act or any successor rule or other similar rule or regulation of the SEC that may at any time permit the Holder to sell securities of the Issuer to the public without registration.

"*Rule 415*" means Rule 415 promulgated under the 1933 Act, as such rule may be amended or interpreted from time to time, or any similar rule or regulation hereunder adopted by the SEC.

"SEC" means the United States Securities and Exchange Commission.

## (b) <u>Mandatory Registration</u>.

- (i) The Issuer shall prepare and, no later than the Filing Deadline, file with the SEC, a Registration Statement on Form S-3 covering the resale of Warrant Shares. If Form S-3 is unavailable for such a registration, the Issuer shall use such other form as is available for such a registration (the Registration Statement filed pursuant to this Section 1(b)(i), the "Resale Registration Statement").
- (ii) The Issuer shall use its commercially reasonable efforts to have such Registration Statement declared effective by the SEC as soon as practicable, but in no event later than the Effectiveness Deadline, and shall use its commercially reasonable efforts to keep such Registration Statement continuously effective under the 1933 Act until the earlier of the date when all Registrable Securities covered by such Registration Statement (A) have been sold, thereunder or pursuant to Rule 144, or (B) may be sold without volume or manner-of-sale restrictions pursuant to Rule 144 and without the requirement for the Issuer to be in compliance with the current public information requirement under Rule 144, as determined by counsel to the Issuer pursuant to a written opinion letter to such effect, addressed and acceptable to the Issuer's transfer agent and the Holder (the "Effectiveness Period").
- (iii) Notwithstanding the registration obligation set forth in Section 1(b), if the SEC informs the Issuer that all of the Registrable Securities cannot, as a result of the application of Rule 415, be registered for resale as a secondary offering on a single registration statement, the Issuer agrees to promptly inform the Holder and use its commercially reasonable efforts to file amendments to the Registration Statement as required by the SEC, covering the maximum number of Registrable Securities permitted to be registered by the SEC, on Form S-3 or such other form available to register for resale the Registrable Securities as a secondary offering; provided, however, that prior to filing such amendment, the Issuer shall be obligated to use diligent efforts to advocate with the SEC for the registration of all of the Registrable Securities in accordance with any interpretive guidance provided by the SEC, including without limitation, Compliance and Disclosure Interpretation 612.09.

## (c) <u>Piggyback Registration</u>.

(i) If (but without any obligation to do so), at any time during the Effectiveness Period, a Registration Statement is not effective with respect to all of the Registrable Securities and the Issuer proposes to register (including for this purpose a registration effected by the Issuer for stockholders other than the Holder) any of its capital stock or other securities under the 1933 Act in connection with a fully underwritten firm commitment public offering of such securities (other than a registration

3

on any form that does not include substantially the same information as would be required to be included in a registration statement covering the sale of the Registrable Securities), the Issuer shall, at such time, give the Holder written notice of such registration (a "*Piggyback Registration*") in accordance with Section 2(f). Upon the written request of the Holder given within five (5) Business Days after delivery of such notice by the Issuer, the Issuer shall, subject to the provisions of Section 1(c)(iii), use all commercially reasonable efforts to cause to be registered under the 1933 Act all of the Registrable Securities that the Holder requests to be registered.

- (ii) The Issuer shall have the right to terminate or withdraw any registration initiated by it under this <u>Section 1(c)</u> prior to the effectiveness of such registration whether or not the Holder has elected to include securities in such registration. The expenses of such withdrawn registration shall be borne by the Issuer in accordance with <u>Section 1(j)</u> hereof.
- (iii) The Issuer shall not be required under this Section 1(c) to include any of the Holder's Registrable Securities in such underwriting unless the Holder accepts the terms of the underwriting as reasonably agreed upon between the Issuer and the underwriters selected by the Issuer (or by other Persons entitled to select the underwriters) and enters into an underwriting agreement in customary form with such underwriters, and then only in such quantity as the underwriters determine in their sole discretion will not jeopardize the success of the offering by the Issuer. If the total amount of securities, including Registrable Securities, requested to be included in such offering exceeds the amount of securities that the underwriters determine in their sole discretion is compatible with the success of the offering, then the Issuer shall be required to include in the offering only that number of Registrable Securities that the underwriters determine in their sole discretion will not jeopardize the success of the offering. Any reduction in the number of Registrable Securities will be made pro rata (based on the number of Warrant Shares then held by the Holder) with the other securities to be registered on behalf of third parties in such offering.

## (d) Effect of Failure to File and Obtain Effectiveness of Registration Statement.

(i) If a Registration Statement covering all of the Registrable Securities required to be covered thereby (or such lesser amount as provided in Section 1(b)(iii)) and required to be filed by the Issuer pursuant to Section 1(b) of this Agreement has not been declared effective by the SEC on or before the Effectiveness Deadline (an "Effectiveness Failure"), then each Holder will be entitled to a payment, as liquidated damages and not a penalty, of the Liquidated Damages Amount but only with respect to the number of Warrant Shares then held by the Holder and not included in an effective Registration Statement and not able to be resold pursuant to Rule 144 in the manner set forth in Section 1(b)(ii) (B) ("Restricted Warrant Shares"), for a period beginning on the Effectiveness Deadline and lasting until such time as such a Registration Statement is declared effective.

The aggregate Liquidated Damages Amount payable to each Holder shall be paid to each Holder in immediately available funds within 10 Business Days after the end of each applicable 30-day period. Any payments pursuant to this Section 2(b) shall constitute the Holder's exclusive remedy for such events; provided, however, if the Issuer certifies that it is unable to pay aggregate Liquidated Damages Amount in cash or immediately available funds because such payment would result in a breach under any of the Issuer's credit facilities or other indebtedness filed as exhibits to the Issuer's reports filed under the Securities Act or the Exchange Act, then the Issuer may pay the aggregate Liquidated Damages Amount in kind in the form of the issuance of additional shares of Common Stock. Upon any issuance of shares of Common Stock as liquidated damages, the Issuer shall promptly prepare and file an amendment to the Resale Registration Statement prior to its effectiveness to include such shares of Common Stock issued as liquidated damages as additional Registrable Securities. If shares of Common Stock are issued as liquidated damages after the Resale Registration Statement has been declared effective, the Issuer shall have no obligation to prepare and file a posteffective amendment to the Resale Registration Statement to include such shares nor shall the Issuer be obligated in any way to file a new registration statement for such shares; however if the Issuer is a well-known seasoned issuer (as defined in the rules and regulations of the SEC) ("WKSI"), the Issuer shall be obligated to provide the Holder notice and offer to include such shares in any Piggyback Registration. All shares of Common Stock issued as Liquidated Damages Amounts shall be considered in the calculation of any subsequent Liquidated Damages Amounts. The determination of the number of shares of Common Stock to be issued as the aggregate Liquidated Damages Amount shall be equal to the aggregate Liquidated Damages Amount divided by the average of the closing sale price per share for the Common Stock (or if the Common Stock is not listed or traded on a national securities exchange, the average of the last reported bid and ask prices per share) for each of the 10 consecutive trading days ending on the trading day immediately preceding such date of determination.

## (e) Request for Registration and/or Underwriting.

(i) If, at any time during the Effectiveness Period, a Registration Statement is not effective with respect to all of the Registrable Securities, the Issuer shall, at the request of the Holder, participate in an underwritten offering of Registrable Securities by the Holder under a Registration Statement effected pursuant to Section 1(b) hereof, and shall file any supplements and amendments to such Registration Statement as may be required by applicable law or rules of the SEC. If, at any time after the Effectiveness Period, a Registration Statement is not effective with respect to all of the Registrable Securities, and the Issuer receives a written request from the Holder that the Issuer effect a registration on Form S-3 with respect to an underwritten offering of Registrable Securities, the Issuer shall use commercially reasonable efforts to file a Registration Statement covering the Registrable Securities as soon as reasonably practicable after receipt of the request. For purposes of this Agreement, a "Demand" shall refer to the Holder's request, pursuant to this Section 1(e), for the Issuer to (1) participate in an underwritten offering of Registrable Securities. In any

5

underwritten offering under this <u>Section 1(e)</u>, the investment banker or bankers and manager or managers that will administer the offering will be selected by, and the underwriting arrangements with respect thereto (including the size of the offering) will be approved by the requesting Holder; <u>provided</u>, <u>however</u>, that such investment bankers and managers and underwriting arrangements must be reasonably satisfactory to the Issuer. The Issuer shall not be required to participate in any underwritten offering contemplated hereby unless each participating Holder (A) agrees to sell its Registrable Securities to be included in the underwritten offering in accordance with any approved underwriting arrangements and (B) completes and executes all reasonable questionnaires, powers of attorney, indemnities, underwriting agreements, lock-up letters and other documents required under the terms of such approved underwriting arrangements. Each participating Holder shall be responsible for any underwriting discounts and commissions and fees and expenses of its own counsel. The Issuer shall pay all expenses customarily borne by issuers in an underwritten offering, including, but not limited to, filing fees, the fees and disbursements of its counsel and independent public accountants and any printing expenses incurred in connection with such underwritten offering.

- (ii) The Issuer shall not be required to participate in or effect any Demand pursuant to this <u>Section 1(e)</u>:
- (A) after the Issuer has participated in or effected two (2) Demands (the Holder shall be deemed to have forfeited its right to a Demand if (1) the Holder withdraws its request that the Issuer effect a registration on Form S-3 with respect to an underwritten offering of Registrable Securities and does not, within thirty (30) days of any such withdrawal, pay all of the Issuer's expenses in connection with such registration or (2) an underwritten offering that is the subject of a Demand is terminated subsequent to the marketing thereof);
  - (B) if the Issuer has participated in or effected a Demand within the preceding twelve (12) months;
- (C) during the period starting with the date sixty (60) days prior to the Issuer's good faith estimate of the date of the filing of, and ending on a date one hundred eighty (180) days following the effective date of, an Issuer-initiated registration subject to Section 1(c), provided that the Issuer is actively employing in good faith all reasonable efforts to cause such registration statement to become effective; or
- (D) if the Issuer shall furnish to the requesting Holder(s) a certificate signed by the Issuer's Chief Executive Officer or Chairman of the Board stating that in the good faith judgment of the Board of Directors of the Issuer (the "*Board*"), it would be seriously detrimental to the Issuer and its stockholders for the Issuer to participate in or effect a Demand at such time, in which event the Issuer shall have the right to defer such Demand for a period of not more than one hundred twenty (120) days after receipt of the request of the Holder.

6

- (f) <u>Related Obligations</u>. Whenever required under this <u>Section 1</u> to effect the registration of any Registrable Securities, except as otherwise expressly provided herein, the Issuer shall:
  - (i) prepare and file with the SEC a Registration Statement with respect to such Registrable Securities and use all commercially reasonable efforts to cause such Registration Statement to become and remain effective;

- (ii) prepare and file with the SEC such amendments and supplements to such Registration Statement and the prospectus used in connection with such Registration Statement as may be necessary to comply with the provisions of the 1933 Act with respect to the disposition of all securities covered by such Registration Statement;
- (iii) furnish to the Holder such number of copies of a prospectus, including a preliminary prospectus, in conformity with the requirements of the 1933 Act, and such other documents as it may reasonably request in order to facilitate the disposition of Registrable Securities owned by it;
- (iv) if required by applicable law, use all commercially reasonable efforts to register and qualify the securities covered by such Registration Statement under such other securities or "blue sky" laws of such jurisdictions as shall be reasonably requested by the Holder, provided that the Issuer shall not be required in connection therewith or as a condition thereto to qualify to do business or to file a general consent to service of process in any such states or jurisdictions;
  - (v) with a view to making available to the Holder the benefits of Rule 144:
    - (A) make and keep public information available, as those terms are understood and defined in Rule 144;
  - (B) file with the SEC in a timely manner all reports and other documents required of the Issuer under the 1933 Act and the 1934 Act so long as the Issuer remains subject to such requirements and the filing of such reports and other documents is required for the applicable provisions of Rule 144; and
  - (C) furnish or otherwise make available, as applicable, to the Holder so long as the Holder owns Registrable Securities, promptly upon request, (1) a written statement by the Issuer, if true, that it has complied with the reporting requirements of Rule 144, the 1933 Act and the 1934 Act, (2) a copy of the most recent annual or quarterly report of the Issuer and such other reports and documents so filed by the Issuer and (3) such other information as may be reasonably requested to permit the Holder to sell such securities without registration pursuant to Rule 144;

7

- (D) in the event of any underwritten public offering, enter into and perform its obligations under an underwriting agreement, in usual and customary form, with the managing underwriter of such offering; and
- (E) notify the holder of Registrable Securities covered by such Registration Statement at any time when a prospectus relating thereto is required to be delivered under the 1933 Act of the happening of any event as a result of which the prospectus included in such Registration Statement, as then in effect, includes an untrue statement of a material fact or omits to state a material fact required to be stated therein or necessary to make the statements therein not misleading in the light of the circumstances then existing.

Notwithstanding the provisions of this Section 1, the Issuer shall be entitled to postpone or suspend the filing, effectiveness or use of, or trading under, any Registration Statement during any period when (i) the SEC or the national securities exchange upon which shares of Common Stock are then listed requests that the Issuer amend or supplement the Registration Statement or the prospectus included therein or requests additional information relating thereto, (ii) the SEC or the national securities exchange upon which shares of Common Stock are then listed issues a stop order or similar order suspending the effectiveness or restricting the use of the Registration Statement or initiates proceedings to issue a stop order or similar order, (iii) the Board in good faith determines that the Registration Statement, the prospectus included therein, any amendment or supplement thereto or any document incorporated or deemed to be incorporated therein contains an untrue statement of a material fact or omits to state a material fact required to be stated therein or necessary to make the statements therein not misleading, in light of the circumstances then existing; provided, however, that the Issuer uses commercially reasonable efforts to prepare and file with the SEC such amendments and supplements to the such Registration Statement or amendment as shall be reasonably necessary to cure such untrue statement or omission, or (iv) the Issuer's management or the Board in good faith determines that the failure to so postpone or suspend would require disclosure of material nonpublic information that, if disclosed at such time, would be materially harmful to the interests of the Issuer and its stockholders; provided, further, that such postponement or suspension (A) shall not exceed a period of forty-five (45) days and (B) shall be exercised by the Issuer not more than twice in any twelve (12) month period (for a maximum of ninety (90) days within any such twelve (12) month period) (each, an "Allowable Grace Period").

- (g) <u>Information from the Holder</u>. It shall be a condition precedent to the obligations of the Issuer to take any action pursuant to this <u>Section 1</u> with respect to the Registrable Securities of the Holder that the Holder shall furnish to the Issuer such information regarding itself, the Registrable Securities held by it, and the intended method of disposition of such securities as shall be reasonably required to effect the registration of such Registrable Securities.
  - (h) <u>Indemnification</u>. If any Registrable Securities are included in a Registration Statement under this Agreement:
  - (i) To the fullest extent permitted by law, the Issuer will, and hereby does, indemnify, hold harmless and defend the Holder, the directors, officers, members,

8

partners, employees, agents, representatives of, and each Person, if any, who controls the Holder within the meaning of the 1933 Act or the 1934 Act (each, a "Holder Indemnified Person"), against any losses, claims, damages, liabilities, judgments, fines, penalties, charges, costs, reasonable attorneys' fees, amounts paid in settlement or expenses, joint or several, (collectively, "Claims") incurred in investigating, preparing or defending any action, claim, suit, inquiry, proceeding, investigation or appeal taken from the foregoing by or before any court or governmental, administrative or other regulatory agency, body or the SEC, whether pending or threatened, whether or not an indemnified party is or may be a party thereto ("Indemnified Damages"), to which any of them may become subject insofar as such Claims (or actions or proceedings, whether commenced or threatened, in respect thereof) arise out of or are based upon: (A) any untrue statement or alleged untrue statement of a material fact in a Registration Statement or any post-effective amendment thereto or in any filing made in connection with the qualification of the offering under the securities or

other "blue sky" laws of any jurisdiction in which Registrable Securities are offered, or the omission or alleged omission to state a material fact required to be stated therein or necessary to make the statements therein not misleading, (B) any untrue statement or alleged untrue statement of a material fact contained in any preliminary prospectus if used prior to the effective date of such Registration Statement, or contained in the final prospectus (as amended or supplemented, if the Issuer files any amendment thereof or supplement thereto with the SEC) or the omission or alleged omission to state therein any material fact necessary to make the statements made therein, in the light of the circumstances under which the statements therein were made, not misleading, or (C) any violation or alleged violation by the Issuer of the 1933 Act, the 1934 Act, any state securities law, or any rule or regulation thereunder relating to the offer or sale of the Registrable Securities pursuant to a Registration Statement (the matters in the foregoing clauses (A) through (C) being, collectively, "Violations"). Subject to Section 1(h)(iii), the Issuer shall reimburse the Holder Indemnified Persons, promptly as such expenses are incurred and are due and payable, for any legal fees or other reasonable expenses incurred by them in connection with investigating or defending any such Claim. Notwithstanding anything to the contrary contained herein, the indemnification agreement contained in this Section 1(h)(i): (A) shall not apply to a Claim by a Holder Indemnified Person arising out of or based upon a Violation which occurs in reliance upon and in conformity with information furnished in writing to the Issuer by such Holder Indemnified Person expressly for use in connection with the preparation of the Registration Statement or any such amendment thereof or supplement thereto, and (B) shall not apply to amounts paid in settlement of any Claim if such settlement is effected without the prior written consent of the Issuer, which consent shall not be unreasonably withheld or delayed. Such indemnity shall remain in full force and effect regardless of any investigation made by or on behalf of the Holder Indemnified Person.

(ii) In connection with any Registration Statement in which Holder is participating, the Holder agrees to indemnify, hold harmless and defend, to the same extent and in the same manner as is set forth in <u>Section 1(h)(i)</u>, the Issuer, each of its directors, officers, employees and agents and each Person, if any, who controls the Issuer within the meaning of the 1933 Act or the 1934 Act (each, an "*Issuer Indemnified Person*"), against any Claim or Indemnified Damages to which any of them may become

9

subject, under the 1933 Act, the 1934 Act or otherwise, insofar as such Claim or Indemnified Damages arise out of or are based upon any Violation, in each case to the extent, and only to the extent, that such Violation occurs in reliance upon and in conformity with written information furnished to the Issuer by the Holder expressly for use in connection with such Registration Statement; and, subject to Section 1(h)(iii), the Holder will reimburse any legal or other expenses reasonably incurred by an Issuer Indemnified Person in connection with investigating or defending any such Claim; provided, however, that the indemnity agreement contained in this Section 1(h)(ii) and the agreement with respect to contribution contained in Section 1(i) shall not apply to amounts paid in settlement of any Claim if such settlement is effected without the prior written consent of the Holder, which consent shall not be unreasonably withheld or delayed; provided, further, however, that the Holder shall be liable under this Section 1(h) (ii) for only that amount of a Claim or Indemnified Damages as does not exceed the net proceeds to the Holder as a result of the sale of Registrable Securities pursuant to such Registration Statement. Such indemnity shall remain in full force and effect regardless of any investigation made by or on behalf of such Issuer Indemnified Person.

Promptly after receipt by a Holder Indemnified Person or an Issuer Indemnified Person (each, an "Indemnified Person") under this <u>Section 1(h)</u> of notice of the commencement of any action or proceeding (including any governmental action or proceeding) involving a Claim, such Indemnified Person shall, if a Claim in respect thereof is to be made against any indemnifying party under this <u>Section 1(h)</u>, deliver to the indemnifying party a written notice of the commencement thereof, and the indemnifying party shall have the right to participate in, and, to the extent the indemnifying party so desires, jointly with any other indemnifying party similarly noticed, to assume control of the defense thereof with counsel mutually satisfactory to the indemnifying party and the Indemnified Person; provided, however, that an Indemnified Person shall have the right to retain its own counsel with the fees and expenses of not more than one counsel for such Indemnified Person to be paid by the indemnifying party, if, in the reasonable opinion of counsel retained by the indemnifying party, the representation by such counsel of the Indemnified Person and the indemnifying party would be inappropriate due to actual or potential differing interests between such Indemnified Person and any other party represented by such counsel in such proceeding. In the case of a Holder Indemnified Person, legal counsel referred to in the immediately preceding sentence shall be selected by the Holder. The Indemnified Person shall cooperate fully with the indemnifying party in connection with any negotiation or defense of any such action or Claim by the indemnifying party and shall furnish to the indemnifying party all information reasonably available to the Indemnified Person which relates to such action or Claim. The indemnifying party shall keep the Indemnified Person reasonably apprised at all times as to the status of the defense or any settlement negotiations with respect thereto. No indemnifying party shall be liable for any settlement of any action, claim or proceeding effected without its prior written consent; provided, however, that the indemnifying party shall not unreasonably withhold, delay or condition its consent. No indemnifying party shall, without the prior written consent of the Indemnified Person, consent to entry of any judgment or enter into any settlement or

10

other compromise which does not include as an unconditional term thereof the giving by the claimant or plaintiff to such Indemnified Person of a release from all liability in respect to such Claim or litigation, and such settlement shall not include any admission as to fault on the part of the Indemnified Person. Following indemnification as provided for hereunder, the indemnifying party shall be subrogated to all rights of the Indemnified Person with respect to all third parties, firms or corporations relating to the matter for which indemnification has been made. The failure to deliver written notice to the indemnifying party within a reasonable time of the commencement of any such action shall not relieve such indemnifying party of any liability to the Indemnified Person under this Section 1(h), except to the extent that the indemnifying party is prejudiced in its ability to defend such action.

(iv) The indemnification required by this <u>Section 1(h)</u> shall be made by periodic payments of the amount thereof during the course of the investigation or defense, promptly following when bills are received or Indemnified Damages are incurred, and in each case submitted to the indemnifying party for payment subject to and in accordance with this <u>Section 1(h)</u>.

The indemnity agreements contained herein shall be in addition to (A) any cause of action or similar right of the Indemnified Person against the indemnifying party or others, and (B) any liabilities the indemnifying party may be subject to pursuant to the law.

(i) <u>Contribution.</u> To the extent any indemnification by an indemnifying party is prohibited or limited by law, the indemnifying party agrees to make the maximum contribution with respect to any amounts for which it would otherwise be liable under <u>Section 1(h)</u> to the fullest extent

permitted by law; provided, however, that: (i) no Person involved in the sale of Registrable Securities who is guilty of fraudulent misrepresentation (within the meaning of Section 11(f) of the 1933 Act) in connection with such sale shall be entitled to contribution from any Person involved in such sale of Registrable Securities who was not guilty of fraudulent misrepresentation; and (ii) contribution by the Holder shall be limited in amount to the excess of the net amount of proceeds received by the Holder from the sale of such Registrable Securities pursuant to such Registration Statement over the amount of any damages that the Holder has otherwise been required to pay by reason of such untrue or alleged untrue statement or omission or alleged omission.

(j) <u>Expenses of Registration.</u> All expenses (other than (i) underwriting discounts and commissions relating to the Registrable Securities that are being sold by the Holder and (ii) fees of any counsel for the Holder) that are incurred in connection with registrations, filings or qualifications pursuant to <u>Sections 1(b) and 1(c)</u>, including (without limitation) all registration, filing and qualification fees, printers' and accounting fees, fees and disbursements of counsel for the Issuer, shall be borne by the Issuer.

11

## 2. MISCELLANEOUS.

- (a) Governing Law; Jurisdiction; Jury Trial. All questions concerning the construction, validity, enforcement and interpretation of this Agreement shall be governed by the internal laws of the State of Delaware, without giving effect to any choice of law or conflict of law provision or rule (whether of the State of Delaware or any other jurisdictions) that would cause the application of the laws of any jurisdictions other than the State of Delaware. Each party hereby irrevocably submits to the exclusive jurisdiction of the Delaware Court of Chancery, or, if no such state court has proper jurisdiction, the United States District Court for the District of Delaware, for the adjudication of any dispute hereunder or in connection herewith or with any transaction contemplated hereby or discussed herein, and hereby irrevocably waives, and agrees not to assert in any suit, action or proceeding, any claim that it is not personally subject to the jurisdiction of any such court, that such suit, action or proceeding is brought in an inconvenient forum or that the venue of such suit, action or proceeding is improper. Each party hereby irrevocably waives personal service of process and consents to process being served in any such suit, action or proceeding by mailing a copy thereof to such party at the address for such notices to it under this Agreement and agrees that such service shall constitute good and sufficient service of process and notice thereof. Nothing contained herein shall be deemed to limit in any way any right to serve process in any manner permitted by law. EACH PARTY HEREBY IRREVOCABLY WAIVES ANY RIGHT IT MAY HAVE, AND AGREES NOT TO REQUEST, A JURY TRIAL FOR THE ADJUDICATION OF ANY DISPUTE HEREUNDER OR IN CONNECTION WITH OR ARISING OUT OF THIS AGREEMENT OR ANY TRANSACTION CONTEMPLATED HEREBY.
- (b) <u>Counterparts.</u> This Agreement may be executed in two or more identical counterparts, all of which shall be considered one and the same agreement and shall become effective when counterparts have been signed by each party and delivered to the other parties; provided that a facsimile signature shall be considered due execution and shall be binding upon the signatory thereto with the same force and effect as if the signature were an original, not a facsimile signature.
- (c) <u>Headings.</u> The headings of this Agreement are for convenience of reference and shall not form part of, or affect the interpretation of, this Agreement.
- (d) <u>Severability.</u> If any provision of this Agreement shall be invalid or unenforceable in any jurisdiction, such invalidity or unenforceability shall not affect the validity or enforceability of the remainder of this Agreement in that jurisdiction or the validity or enforceability of any provision of this Agreement in any other jurisdiction.
- (e) Entire Agreement; Amendments. This Agreement, the Warrant Agreement and the other Warrant Documents supersede all other prior oral or written agreements between the Holder, the Issuer, their affiliates and Persons acting on their behalf with respect to the matters discussed herein, and this Agreement, the Warrant Agreement, the other Warrant Documents and the instruments referenced herein and therein contain the entire understanding of the parties with respect to the matters covered herein and therein and, except as specifically set forth herein or therein, neither the Issuer nor the Holder makes any representation, warranty, covenant or undertaking with respect to such matters. No provision of this Agreement may be amended other than by an instrument in writing signed by the Issuer and the Holder. No provision hereof may be waived other than by an instrument in writing signed by the party against whom enforcement is sought.

12

(f) Notices. Any notices, consents, waivers or other communications required or permitted to be given under the terms of this Agreement must be in writing and will be deemed to have been delivered: (i) upon receipt, when delivered personally; (ii) upon receipt, when sent by facsimile (provided confirmation of transmission is mechanically or electronically generated and kept on file by the sending party); or (iii) one Business Day after deposit with an overnight courier service, in each case properly addressed to the party to receive the same. The addresses and facsimile numbers for such communications shall be:

if to the Issuer:

Clean Energy Fuels Corp. 3020 Old Ranch Parkway Suite 400 Seal Beach, California 90740 Telephone: (562) 493-7239 Facsimile: (562) 493-4956 Attention: J. Nathan Jensen

with a copy (for informational purposes only) to:

Morrison & Foerster LLP 12531 High Bluff Drive Suite 100 San Diego, California 92130 Telephone: (858) 720-5198 Facsimile: (858) 523-2810 Attention: Steven G. Rowles

#### if to the Holder:

GE Energy Financial Services, Inc. 333 Clay Street, Suite 4550 Houston, Texas 77002 Telephone: (713) 951-2339

Facsimile: (713) 951-2319 Attention: Brandy Copley

with a copy (for informational purposes only) to:

Vinson & Elkins LLP 666 Fifth Avenue, 26<sup>th</sup> Floor New York, New York 10103 Telephone: (212) 237-0132 Facsimile: (917) 849-5350 Attention: Robert Seber

or to such other address and/or facsimile number and/or to the attention of such other Person as the recipient party has specified by written notice given to each other party five (5) days prior to

13

the effectiveness of such change. Written confirmation of receipt (A) given by the recipient of such notice, consent, waiver or other communication, (B) mechanically or electronically generated by the sender's facsimile machine containing the time, date, recipient facsimile number and an image of the first page of such transmission or (C) provided by an overnight courier service shall be rebuttable evidence of personal service, receipt by facsimile or receipt from an overnight courier service in accordance with clause (i), (ii) or (iii) above, respectively.

- (g) <u>Successors and Assigns.</u> No party may assign this Agreement or any rights or obligations hereunder without the prior written consent of the other parties, except that the Holder may assign its rights hereunder in connection with any transfer of the Warrant Shares. This Agreement shall be binding upon and inure to the benefit of the parties and their successors and permitted assigns.
- (h) <u>No Third Party Beneficiaries.</u> This Agreement is intended for the benefit of the parties hereto and their respective permitted successors and assigns, and is not for the benefit of, nor may any provision hereof be enforced by, any other Person.
- (i) <u>Further Assurances.</u> Each party shall do and perform, or cause to be done and performed, all such further acts and things, and shall execute and deliver all such other agreements, certificates, instruments and documents, as any other party may reasonably request in order to carry out the intent and accomplish the purposes of this Agreement and the consummation of the transactions contemplated hereby.

[Signature page follows.]

14

**IN WITNESS WHEREOF,** the Issuer and the Holder have caused this Registration Rights Agreement to be duly executed as of the date first written above.

## COMPANY:

## CLEAN ENERGY FUELS CORP.

By: /s/ Andrew J. Littlefair

Name: Andrew J. Littlefair

Title: President & Chief Executive Officer

## HOLDER:

## GE ENERGY FINANCIAL SERVICES, INC.

By: /s/ Tyson Yates
Name: Tyson Yates
Title: Managing Director





## News Release

## GE and Clean Energy Fuels Partner to Expand 'America's Natural Gas Highway®'

- · Clean Energy to Use GE's MicroLNG Technology at Two New LNG Facilities
- · New MicroLNG Technology Builds on GE Oil & Gas' Global Success in Large LNG Projects

HOUSTON—November 13, 2012—GE (NYSE: GE) and Clean Energy Fuels (Nasdaq: CLNE) today announced a collaboration to expand the infrastructure for natural gas transportation in the United States. The agreement supports Clean Energy's efforts in developing "America's Natural Gas Highway®," a fueling network that will enable trucks to operate on Liquefied Natural Gas (LNG) coast to coast and border to border. Truck fleets often can reduce fuel costs by more than 25 percent and lower emissions with LNG.

As part of the collaboration, Clean Energy Fuels will initially purchase two ecomagination<sup>TM</sup>-qualified MicroLNG plants from GE Oil & Gas. The plug-and-play modular units, which are designed to rapidly liquefy natural gas while minimizing a site's physical footprint, will support fueling stations along critical transportation corridors that run across the U.S. Further underscoring GE's commitment to expanding natural gas transportation infrastructure, GE Energy Financial Services is providing up to \$200 million in financing for the two GE MicroLNG plants.

"GE is proud to be partnering with Clean Energy Fuels to develop natural gas infrastructure in the U.S. Clean Energy is an industry leader in pioneering a new way for America to fuel its vehicles and to further gain energy independence," said GE Chairman and CEO Jeff Immelt. "With an abundance of cleaner, more affordable natural gas here in the U.S., this is an important opportunity for GE to join Clean Energy in changing the way America drives. It's also a critical step in developing a natural gas-for-transportation fuel model that can be easily exported to other countries interested in exactly these kinds of breakthrough projects."

Clean Energy expects to complete approximately 70 LNG stations by the end of 2012, with more planned for next year to serve the movement of goods along major transportation corridors throughout the U.S. While CNG, or compressed natural gas, is primarily used in cars, buses and smaller trucks, the LNG fueling being rolled out at Clean Energy's stations is targeted at long-haul, heavy-duty trucks, which will have the advantage of longer driving ranges while not impacting tractor weight and incremental costs. In 2013, four major manufacturers will introduce the Cummins Westport 12-liter LNG engine, which is the optimum size for long-haul Class 8 trucks.

Clean Energy plans to use a standardized design of the new GE MicroLNG plants to build additional MicroLNG plants. These first two MicroLNG plants will produce up to 250,000 gallons per day. The plant is designed to be expanded up to 1 million gallons per day as adoption and demand increases. The LNG produced by the MicroLNG plants will be used primarily at Pilot-Flying J truck stops that serve truckers across the country. The two GE MicroLNG plants are targeted to begin operation in 2015. The two companies are currently assessing the best locations for these first two LNG plants.

"The agreement announced today with GE is one of the most significant milestones in Clean Energy's history," said Andrew J. Littlefair, president and CEO of Clean Energy Fuels. "As the long-haul trucking industry begins its transition to natural gas, it will be critical to have a reliable supply of LNG. No other

1

company is as uniquely qualified as GE to help address this need due to its vast experience in energy, technology innovations and financing capabilities. GE partnering with Clean Energy on these two facilities will not only help ensure an adequate LNG supply for our stations, but it is another confirmation that the transition to natural gas as a transportation fuel is gaining momentum."

Natural gas is an abundant, reliable and cleaner-burning source of energy for consumers and commercial users. Clean Energy Fuels and GE are promoting the importance of natural gas to the U.S. economy, enabling energy independence and decreasing CO<sub>2</sub> emissions—in this case, by enabling long-haul trucks and fleets to move from diesel to cleaner, more efficient and readily available domestic natural gas.

"GE is committed to natural gas. From extraction to transport to power generation—we continue to develop solutions that infuse new technologies into the value chain and help improve every step of the natural gas development and deployment life cycle," said Dan Heintzelman, president and CEO of GE Oil & Gas. "Our ecomagination-qualified MicroLNG plant was born from the same turbomachinery technology that has made GE a success in large LNG compression such as in the world-scale plants in Qatar and Australia. By taking this technology and reengineering it so that it's modular and highly efficient, we are able to help customers such as Clean Energy deliver this abundant and cleaner fuel source to the market."

GE's MicroLNG plant can liquefy natural gas at any point along a gas distribution network, making it ideal for supporting the fueling of vehicles in remote locations by reducing the impact of long distance fuel transport. This MicroLNG technology is part of GE's expanding technology offerings in the natural gas-for-transportation sector.

The new GE MicroLNG system that will be used by Clean Energy will produce 250,000 gallons of LNG per day, or about 54 million DGEs (diesel gallon equivalents) per year with the built-in capability for further expansion, which is a 67 percent increase over the capacity of the breakthrough MicroLNG plant that GE Oil & Gas first introduced in January of 2012. The new system will help reduce a fleet operator's fuel costs by more than 25 percent compared to diesel fuel. LNG produced with this MicroLNG system can be used to fuel approximately 28,000 heavy trucks, replacing diesel-powered trucks with equivalent fuel economy. This could enable fleet operators to avoid more than 139,000 metric tons of CO<sub>2</sub>e emissions per year, equivalent to the annual greenhouse gas emissions of approximately 27,000 cars using gasoline or 7,000 trucks using diesel on U.S. roads—assuming an average truck travels approximately 14,000 miles per year.

GE also is providing turnkey process/plant construction and consultations on optimal plant location and power partner. The scope of the agreement also includes project installation. It entails not only the liquefaction but also the complete process design from the pre-treatment of the gas to the storage system.

2

## About GE

GE (NYSE: GE) works on things that matter. The best people and the best technologies taking on the toughest challenges. Finding solutions in energy, health and home, transportation and finance. Building, powering, moving and curing the world. Not just imagining. Doing. GE works. For more information, visit the company's website at www.ge.com.

## **About GE Oil & Gas**

GE Oil & Gas is a world leader in advanced technology equipment and services for all segments of the oil and gas industry, from drilling and production, LNG, pipelines and storage to industrial power generation, refining and petrochemicals. GE Oil & Gas also provides pipeline integrity solutions, including inspection and data management, and designs and manufactures wire-line and drilling measurement solutions for the oilfield services segment.

As part of its customer focus and commitment, GE Oil & Gas exploits technological innovation from other GE businesses, such as aviation and healthcare, to continuously improve oil and gas industry performance and productivity. GE Oil & Gas employs more than 35,000 people worldwide and operates in more than 100 countries.

Follow GE Oil & Gas on Twitter @GE\_OilandGas.

## **About Clean Energy Fuels**

Clean Energy (Nasdaq: CLNE) is the largest provider of natural gas fuel for transportation in North America and a global leader in the expanding natural gas vehicle fueling market. We have operations in compressed natural gas (CNG) and liquefied natural gas (LNG) vehicle fueling and construction and operation of natural gas fueling stations. Wholly-owned subsidiaries include BAF Technologies, which provides natural gas vehicle systems and conversions for taxis, vans, pick-up trucks and shuttle buses; IMW Industries, Ltd., which supplies CNG equipment for vehicle fueling and industrial applications worldwide; NorthStar, which supplies LNG and liquefied to compressed natural gas fueling system technologies and equipment, station construction and operations; and Clean Energy Renewable Fuels (CERF), which develops renewable natural gas (RNG), or biomethane, production facilities in the U.S. For more information, visit www.cleanenergyfuels.com

###

"America's Natural Gas Highway" is a trademark of Clean Energy. ecomagination is a trademark of the General Electric Company.

For more information, contact:

Sean Gannon GE Oil & Gas +1 212 587 5059 sean.gannon@ge.com

Donna Mirandola GE Oil & Gas +1 773 318 7483 donna.mirandola@ge.com

Gary Foster Clean Energy Fuels +1 562 774 7056 gfoster@cleanenergyfuels.com

Ken Darling or Kate Culman Masto Public Relations +1 518 786 6488 kenneth.darling@ge.com kate.culman@mastopr.com